Shared Decision Making in Secondary Mental Healthcare in Taiwan

A thesis submitted to The University of Manchester for the degree of Doctor of Philosophy in the Faculty of Biology, Medicine and Health

2019

Chiu-Yi Lin

School of Health Sciences Division of Nursing, Midwifery & Social Work

LIST OF CONTENTS

		Page
	LIST OF CONTENTS	2
	OTHER LISTS	8
	ABSTRACT	10
	DELARATION	12
	COPYRIGHT STATEMENT	12
	ACKNOWLEDGEMENT	13
	THE AUTHOR	14
	INTRODUCTION	15
Chapter 1	Background	17
1.1	Mental healthcare in Taiwan	17
	1.1.1 National Health Insurance (NHI)	17
	1.1.2 The mental healthcare structure in Taiwan	18
	1.1.2.1 Primary mental health services	19
	1.1.2.2 Hospital mental health services	20
	1.1.2.3 Community mental health service	20
	1.1.3 Psychiatric patient classification system	21
	1.1.4 Patients in halfway houses	22
	1.1.5 Taiwan Mental Health Act	23
1.2	Shared decision making	24
	1.2.1 History of shared decision making	24
	1.2.2 Definition of shared decision making	26
	1.2.3 Different interpretations of shared decision making	26
	1.2.4 Shared decision making is recommended in mental	27
	healthcare	
	1.2.4.1 Ethical perspective	28
	1.2.4.2 Clinical evidence	28
	1.2.5 Challenges to implementation of shared decision making	30
	1.2.5.1 Professional paternalistic attitude	30
	1.2.5.2 Concerns about patients' impaired decisional	31
	capacity	
	1.2.5.3 Passive patient preference	32
	1.2.5.4 Time-consuming issue	32
	1.2.6 Decision aids	33
1.3	Rationale for the study	35
1.4	Summary	36
Chapter 2	Methodology	37
2.1	Research paradigms	37
	2.1.1 Positivism versus interpretivism	38
	2.1.1.1 Positivism	38
	2.1.1.2 Interpretivism	39
	2.1.2 Applying the interpretivist paradigms	40
2.2	Research aim and objectives	41
	2.2.1 Aim	41

	2.2.2 Objectives	41
2.3	Qualitative systematic review methodology	41
	2.3.1 Type of systematic reviews	41
	2.3.2 Selecting qualitative systematic review methodology	42
	2.3.2.1 Search tool	43
	2.3.2.2 Data extraction	44
	2.3.2.3 Quality of the review	45
	2.3.2.4 Synthesis/Analysis of qualitative studies	45
2.4	Qualitative research methodology	48
	2.4.1 Qualitative approaches	48
	2.4.1.1 Grounded theory	48
	2.4.1.2 Phenomenology	48
	2.4.1.3 Ethnography	49
	2.4.1.4 Action research	49
	2.4.1.5 Generic qualitative approach	49
	2.4.2 Qualitative data collection methods	50
	2.4.2.1 Selecting an interview approach	51
	2.4.2.2 Semi-structured interviews	52
	2.4.3 Sampling	52
	2.4.3.1 Sample size	54
	2.4.4 Data analysis	55
	2.4.4.1 Narrative analysis	56
	2.4.4.2 Discourse analysis	56
	2.4.4.3 Constant comparative analysis	56
	2.4.4.4 Content analysis	57
	2.4.4.5 Selecting thematic analysis as a study approach	57
2.5	Ensuring quality and trustworthiness	59
	2.5.1 Credibility	59
	2.5.2 Transferability	60
	2.5.3 Dependability	60
	2.5.4 Confirmability	61
2.6	Ethical principles	61
	2.6.1 Autonomy	62
	2.6.2 Beneficence and non-maleficence	62
	2.6.3 Justice	63
2.7	Summary	64
Chapter 3	Systematic reviews of qualitative studies (Study One)	65
3.1	Review methods	65
	3.1.1 Review aims and objectives	65
	3.1.1.1 Review aim	65
	3.1.1.2 Review objectives	65
	3.1.2 Literature search strategy	65
	3.1.3 Selection criteria	67
	3.1.3.1 Sample/settings	68
	3.1.3.2 Phenomena of interest	69

	3.1.3.3 Design, evaluation, and research type	69
	3.1.4 Eligibility screening	69
	3.1.5 Data extraction	70
	3.1.6 Quality appraisal	70
	3.1.7 Synthesis of qualitative studies	70
	3.1.7.1 Step one: Free line-by-line coding of textual	70
	findings from primary studies	
	3.1.7.2 Step two: Organisation of free codes into	71
	descriptive themes	
	3.1.7.3 Step three: Development of analytical themes	72
3.2	Findings	73
	3.2.1 Search results	73
	3.2.2 Characteristics of the included studies	75
	3.2.2.1 Methodological design	75
	3.2.2.2 Characteristics of participants	75
	3.2.3 Quality of included studies	80
	3.2.4 Analytical themes	83
	3.2.4.1 Theme 1: Barriers to shared decision making	89
	3.2.4.1.1 Sub-theme: Health professionals with	89
	paternalistic attitude	
	3.2.4.1.2 Sub-theme: Poor quality of interaction	90
	with health professionals	
	3.4.2.1.3 Sub-theme: Lack of professional	92
	knowledge and information	
	3.2.4.2 Theme 2: Facilitators of shared decision making	94
	3.2.4.2.1 Sub-theme: Supportive attitude to	94
	patient involvement	
	3.2.4.2.2 Sub-theme: Sufficient information	95
	exchange	
	3.2.4.2.3 Sub-theme: Other support resources	96
3.3	Strengths and limitations	99
3.4	Summary	100
Chapter 4	Working methods for patient and health professional	102
	interviews	
4.1	Participant selection	102
	4.1.1 Patient interviews	102
	4.1.2 Health professional interviews	103
4.2	Recruitment	103
	4.2.1 Patients	104
	4.2.2 Health professionals	105
4.3	Data collection - Qualitative interviews	106
	4.3.1 Topic guide	106
	4.3.1.1 Patient interviews	106
	4.3.1.2 Health professional interviews	106
	4.3.2 Demographic data collection	107

4.4	Data analysis	107
	4.4.1 Qualitative data analysis software	107
	4.4.2 Thematic analysis	107
	4.4.2.1 Familiarisation with the data	108
	4.4.2.2 Generating initial codes	108
	4.4.2.3 Developing for themes	109
	4.4.2.4 Reviewing themes	110
	4.4.2.5 Defining and naming themes	111
	4.4.2.6 Producing the report	111
4.5	Rigour	111
	4.5.1 Credibility	111
	4.5.2 Transferability	112
	4.5.3 Dependability	112
	4.5.4 Confirmability	112
	4.5.4.1 Reflexivity	113
	4.5.4.1.1 The author	113
	4.5.4.1.2 Field notes	114
4.6	Ethical considerations	115
	4.6.1 Autonomy	115
	4.6.1.1 Confidentiality	116
	4.6.2 Beneficence and non-maleficence	117
	4.6.2.1 Potential benefits of participation	118
	4.6.2.2 Potential risks of participation	118
	4.6.3 Justice	118
4.7	Summary	119
Chapter 5	Findings from patient interviews (Study Two)	120
5.1	Characteristics of patient participants	120
5.2	Themes developed from the patient interviews	121
	5.2.1 Theme 1: Barriers to shared decision making	122
	5.2.1.1 Sub-theme: Professional status of health	122
	professionals in a submissive culture	
	5.2.1.2 Sub-theme: Negative perception of making decisions	128
	5.2.1.3 Sub-theme: Limited time resources	130
	5.2.2 Theme 2: Facilitators of shared decision making	132
	5.2.2.1 Sub-theme: Sufficient information exchange	133
	5.2.2.2 Sub-theme: Desire to make decisions	134
5.3	Strengths and limitations	137
5.4	Summary	138
Chapter 6	Findings from health professional interviews (Study Three)	140
- 6.1	Characteristics of professional participants	140
6.2	Themes from the health professional interviews	142
	6.2.1 Theme 1: Barriers to shared decision making	142
	6.2.1.1 Sub-theme: Powerful status of health	143
	professionals and families	

	6.2.1.1.1 Professional-patient relationship	143
	6.2.1.1.2 Family-patient relationship	147
	6.2.1.2 Sub-theme: Patients with impaired decisional	149
	ability due to mental illness	
	6.2.1.3 Sub-theme: Health professionals lack	152
	understanding of shared decision making	
	6.2.1.4 Sub-theme: Insufficient time	154
	6.2.2 Theme 2: Facilitators of shared decision making	156
	6.2.2.1 Sub-theme: Awareness of patients' right to	157
	autonomy	
	6.2.2.2 Sub-theme: Understanding of potential benefits	159
	of shared decision making	
6.3	Strengths and limitations	161
6.4	Summary	162
Chapter 7	Synthesis	164
7.1	Synthesis methods	164
	7.1.1 Evidence resources	164
	7.1.2 Synthesis process	164
7.2	Synthesis outcomes	172
	7.2.1 Capability/skill improvement	172
	7.2.1.1 Increasing patient decisional capability	172
	7.2.1.2 Improving professionals' knowledge of shared	175
	decision making	
	7.2.2 Attitude/motivation modification and improvement	176
	7.2.2.1 Modification of health professionals'	176
	paternalistic attitude to patient involvement	
	7.2.2.2 Attitudinal modification regarding use of	178
	involuntary intervention	
	7.2.3 Limited time resources improvement	179
	7.2.3.1 Using limited time efficiently	179
7.3	Summary	181
Chapter 8	Discussion	182
8.1	8.1 Summary of key findings	182
	8.1.1 Qualitative systematic review	182
	8.1.2 Patient interviews	183
	8.1.3 Health professional interviews	184
	8.1.4 Synthesis of the studies	185
8.2	Strengths and limitations	186
	8.2.1 Qualitative systematic review (Study One)	187
	8.2.2 Patient interviews (Study Two)	187
	8.2.3 Health professional interviews (Study Three)	187
_	8.2.4 Synthesis	188
8.3	Current literature	189
8.4	Clinical implications	194
8.5	Further research	193

8.6	Conclusion	195
References		196

Total word count: 603351

OTHER LISTS

List of tables and figures

Table	Content	page
Table 1.1	Psychiatric patient classification systems	21
Table 2.1	Methods proposed for the synthesis of qualitative evidence	
Table 2.2	Strengths and weaknesses of different types of data	
	collection	
Table 3.1	Search concepts and terms	
Table 3.2	Inclusion criteria and exclusion criteria	68
Table 3.3	An example of use of line-by-line coding	71
Table 3.4	Summary of descriptive themes	71
Table 3.5	Two primary analytical themes from the synthesis of	73
	included studies	
Table 3.6	Studies on shared decision making excluded from the	73
	review	
Table 3.7	Summary of the included studies on perspectives of shared	77
	decision making amongst patients	
Table 3.8	Quality appraisal outcomes	81
Table 3.9	Summary of findings of the included studies on	84
	perspectives of shared decision making from patients	
Table 3.10	Overview of analytical themes	88
Table 4.1	Patients-inclusion and exclusion criteria	102
Table 4.2	Health professionals-inclusion and exclusion criteria	103
Table 4.3	Data extraction with initial codes applied	109
Table 5.1	Characteristics of patient participants by reference number	121
Table 5.2	Themes identified from patient interviews	122
Table 6.1	Characteristics of professional participants by reference number	140
Table 6.2	<i>Themes identified from health professional interviews</i> 14	
Table 7.1	Summary of initial synthesis outcomes and possible solutions	166
Table 7.2	Synthesis summary of improvement mechanisms	169
Figure	Content	page
Figure 1	Structure of the PhD	16
Figure 1.1	The healthcare structure in Taiwan	19

Figure 1.1	The healthcare structure in Taiwan
Figure 2.1	The steps of thematic synthesis
Figure 3.1	Flow of the review

Figure 4.1 Initial thematic map for patients

47 74

110

List of appendices

Appendix	Content	page
Appendix 1	Extraction sheet for qualitative systematic review	223
Appendix 2	Topic guide for patients	227
Appendix 3	Topic guide for health professionals	230
Appendix 4	Background of mental health patients	233
Appendix 5	Background of mental health professionals	234
Appendix 6	Initial themes	235
Appendix 7	Ethical approval from the University of Manchester	238
Appendix 8	Ethical approval from the local committee in Taiwan	240
Appendix 9	Consent Form (patient version)	245
Appendix 10	Consent Form (health professional version)	246
Appendix 11	Distress policy	247
Appendix 12	Example of thematic frameworks for the patient interviews	249
	(Theme 1)	
Appendix 13	Example of thematic frameworks for the patient interviews	255
	(Theme 2)	
Appendix 14	Example of thematic frameworks for the health	259
	professional interviews (Theme 1)	
Appendix 15	Example of thematic frameworks for the health	272
	professional interviews (Theme 2)	
Appendix 16	Extraction sheet for synthesis	276

ABSTRACT

University of Manchester

Chiu-Yi Lin

Doctor of Philosophy

Shared decision making in secondary mental healthcare in Taiwan

23rd September 2019

Background

Although shared decision making has been widely advocated in mental health policies in western countries, there is limited evidence of shared decision making in South Asian cultures, particularly Taiwan. Thus, there is a need to explore shared decision making from the perspectives of Taiwanese health professionals and individuals experiencing mental health disorders. Such knowledge is of key importance to facilitate shared decision making in Taiwan's mental healthcare systems.

Aims

To explore/understand the process of decision making, from both health professional and patient perspectives, in secondary mental healthcare in Taiwan.

Methods

To address this aim, three separate but interrelated studies were conducted:

- **Study One:** A systematic review of previously published qualitative studies of patient perspectives of shared decision making.
- Study Two: Semi-structured qualitative interviews to explore patient perspectives of shared decision making in secondary mental healthcare in Taiwan.
- **Study Three:** Semi-structured qualitative interviews to explore health professional perspectives of shared decision making in secondary mental healthcare in Taiwan.

Results

The qualitative systematic review included 13 studies, and revealed a number of key barriers and facilitators to shared decision making. Barriers included: paternalistic attitudes of health professionals, poor quality interaction with health professionals, and

lack of professional knowledge and information. Factors enhancing shared decision making included a supportive attitude to patient involvement, sufficient information exchange, and other support resources.

Qualitative interviews with 20 patients found that they were not involved in the decisionmaking process due to: the professional status of health professionals in a submissive culture; negative perception of making decisions; and health professionals having limited time. However, patients showed a desire to be involved in decision making but required sufficient information exchange to enable them to do this.

Qualitative interviews with 24 health professionals revealed a number of barriers to and facilitators of shared decision making. Factors reducing shared decision making included: the powerful status of health professionals and families; a belief that patients had impaired decisional ability due to their mental illness; a lack of understanding of shared decision making; and insufficient time. Despite this, there was a view from a few health professionals that they understood the potential benefits of shared decision making.

The results of the above three studies were synthesised and revealed three mechanisms through which implementation of shared decision making could potentially be improved: capability/skill improvement; attitude/motivation modification; and sufficient time resource.

Conclusion

This study has provided an insight into implementing shared decision making in mental healthcare directly from patients' and health professionals' perspectives in Taiwan. The findings revealed that shared decision making was not yet understood or implemented in mental healthcare. Significant barriers and facilitators were identified and mechanisms were proposed to address the barriers to shared decision making. The findings of this study provided potential solutions to aid further training of staff and development of national policies on shared decision making.

DECLARATION

No portion of the work referred to in this thesis has been submitted in support of an application for another degree or qualification of this or any other university or other institute of learning.

COPYRIGHT STATEMENT

i. The author of this thesis (including any appendices and/or schedules to this thesis) owns certain copyright or related rights in it (the "Copyright") and s/he has given The University of Manchester certain rights to use such Copyright, including for administrative purposes.

ii. Copies of this thesis, either in full or in extracts and whether in hard or electronic copy, may be made only in accordance with the Copyright, Designs and Patents Act 1988 (as amended) and regulations issued under it or, where appropriate, in accordance with licensing agreements which the University has from time to time. This page must form part of any such copies made.

iii. The ownership of certain Copyright, patents, designs, trademarks and other intellectual property (the "Intellectual Property") and any reproductions of copyright works in the thesis, for example graphs and tables ("Reproductions"), which may be described in this thesis, may not be owned by the author and may be owned by third parties. Such Intellectual Property and Reproductions cannot and must not be made available for use without the prior written permission of the owner(s) of the relevant Intellectual Property and/or Reproductions.

iv. Further information on the conditions under which disclosure, publication and commercialisation of this thesis, the Copyright and any Intellectual Property and/or Reproductions described in it may take place is available in the University IP Policy (see http://documents.manchester.ac.uk/DocuInfo.aspx?DocID=24420), in any relevant Thesis restriction declarations deposited in the University Library, The University Library's regulations (see http://www.library.manchester.ac.uk/about/regulations/) and in The University's policy on Presentation of Theses

ACKNOWLEDGEMENTS

I would like to thank:

My two excellent supervisors, Professor Karina Lovell and Dr Laoise Renwick, for their continued support and unswerving patience in supervising the PhD.

Dr Owen Price for providing time to support me academically and for his patience listening to me.

Mental health research group members, Professor Penny Bee, Dr Helen Brooks, Dr Judith Gellatly, Dr Amy Blakemore, and Dr Rebecca Pedley, who always provided their help when I needed it.

All the patients and health professionals who participated in and assisted with this study, and particularly Yu-Hui Ku, who provided time to support the study agenda in practice.

All the postgraduate students who kindly shared information and listened patiently to me during this journey.

DEDICATIONS

This thesis is dedicated to my husband Tsen Lu (Nathan) and my son Ding-Lin Lu (Duke). It has been a very long journey, studying abroad, and they have loved me unconditionally throughout. In particular, my parents-in-law, who always love me and continually support me in pursuing my dream. Also, my parents who gave birth to me and raised me.

THE AUTHOR

I qualified as registered nurse in Taiwan in 2006. My first role was as a registered nurse at Tri-Service General Hospital (at the top level of health services in Taiwan). I was promoted to supervisor, with responsibility for managing acute wards, overseeing patient care, developing working guidelines such as physical restraint and de-escalation techniques, supporting and teaching student nurses, and handling administration.

In 2011, I completed my master's degree at National Yang-Ming University (NYMU) and worked as a lecturer in two college-level institutions for teaching mental healthcare and the conducting of literature reviews.

INTRODUCTION

Overview of the study

This PhD study consists of three studies and a final synthesis of findings. The three studies comprise a systematic review of qualitative studies of patient perspectives of shared decision making (Study One), semi-structured qualitative interviews with patients in secondary mental healthcare in Taiwan (Study Two), and semi-structured qualitative interviews eliciting health professionals' perspectives of shared decision making in secondary mental healthcare in Taiwan (Study Three); a final synthesis draws the findings together and presents a conclusion and recommendations. The explanation of studies one, two and three are presented in the following chapters.

Aim of the PhD study

• to explore/understand the process of decision making, from both health professional and patient perspectives, in secondary mental healthcare in Taiwan.

Objectives

The objectives are:

- to examine and synthesise patient perspectives of shared decision making in secondary mental healthcare using a systematic review of qualitative studies;
- to explore patient perspectives of shared decision making in secondary mental healthcare in Taiwan using semi-structured qualitative interviews;
- to explore health professional perspectives of shared decision making in secondary mental healthcare in Taiwan using semi-structured qualitative interviews.

Organisation of the thesis

This PhD comprises three separate, but interrelated, studies: a systematic review of qualitative studies (Study One), semi-structured qualitative interviews with patients in secondary mental healthcare in Taiwan (Study Two), and semi-structured qualitative interviews with health professionals in secondary mental healthcare in Taiwan (Study Three) (Figure 1).

Therefore, the thesis comprises eight chapters, of each of which an outline is presented here. Chapter One gives an overview of the topic and background information about the mental health system in Taiwan, and expands on the background by outlining the evidence regarding shared decision making. Chapter Two presents the aims and objectives, and a justification of the methodology, including its research paradigm. Chapter Three provides details of a qualitative systematic review of existing studies on shared decision making in secondary mental healthcare (Study One). Chapter Four presents the working methods which were used for the patient interviews (Study Two) and health professional interviews (Study Three). Chapters Five and Six provide details of the findings from Study Two and Study Three. Chapter Seven outlines the results of the synthesis of findings from the qualitative systematic review (Study One), patient interviews (Study Two), and health professional interviews (Study Three). Chapter Eight provides a discussion of the thesis, and recommendations for further studies.





Chapter One: Background

This chapter will be presented in four main sections. The first section will provide an introduction to the mental health system in Taiwan and the second section will present an overview of shared decision making. The third section will provide the rationale for the study. The final section will present a summary of this chapter.

1.1 Mental healthcare in Taiwan

Taiwan, located in East Asia, is a relatively small island (36,193 km²) with a population of 23 million. With its rapidly growing economy, Taiwan is recognised as a developing country (United Nations, 2017). In Taiwan, the Ministry of Health and Welfare, which is part of the Executive Yuan (the department of the government in Taiwan), is responsible for formulating healthcare policies and managing the delivery of health services. In 1995, the government adopted a national health insurance system to underpin the Taiwanese healthcare system, including mental healthcare.

1.1.1 National Health Insurance (NHI)

Before nationwide health insurance was adopted, there were a range of insurance systems in Taiwan, such as government employee insurance, labour insurance, fishermens' health insurance, and farmers' health insurance. In 1995, the Taiwanese government consolidated all the insurance schemes into the single, nationwide system, the National Health Insurance (NHI) system.

The current healthcare system (including mental health) in Taiwan is based on National Health Insurance. This maintains that every citizen has an equal right to receive healthcare and treatment. It is compulsory for every citizen to register with the NHI to support their treatment and care (National Health Insurance Administration in Taiwan, 2016a). In 2016, it was estimated that 99% of the population was registered in the NHI (National Health Insurance Administration in Taiwan, 2016a).

The NHI is considered a successful health system with sufficient services because of high patient satisfaction, and high quality of care and treatment delivery (Wu et al., 2010). Results from a survey of client satisfaction with the NHI in 2014 indicated that about 80% of clients reported a high degree of satisfaction (Bureau of the National Health Insurance, 2015). Many features of the NHI contribute to high client satisfaction, including good accessibility of all levels of care, comprehensive coverage of different medical disciplines, short waiting times and low cost (Wu et al., 2010).

However, challenges that emerged included the quality of medical encounters, an unstructured referral system, and systemic financial problems (Wu et al., 2010). There

are some barriers to the quality of medical encounters within the NHI system in Taiwan. In Taiwanese culture, people are used to seeking medical help even for minor medical conditions. People in Taiwan have an average of 13.4 patient visits per annum across healthcare services (Shih et al., 2010), compared with 5 for the United Kingdom (2009) (WHO Regional Office for Europe, 2018). Health professionals' (such as general practitioners and other doctors) income is based on the number of patients and level of treatment/care. It is usual for a health professional to see 50 to 100 patients in one day. The heavy workloads and resulting short duration of patient visits are significant features of the system (Chang et al., 2012). It could be difficult for health professionals and patients to explore their perspectives and to achieve sufficient information exchange in medical encounters. Therefore, these factors all negatively influence the quality of medical encounters.

Primary care does not act as an adequate gate-keeping system, even if this is directed by the government, as patients can bypass this system and directly access specialists services – this being the case, the gatekeeping system is not enforced. This means that individuals can select any level of medical services freely (Wu et al., 2010). Such a system is problematic as the majority of the population believe that higher level medical services provide better quality treatment and care, so they tend to use the higher-level services, and therefore these services, such as medical centres, are always over-subscribed.

Furthermore, the availability of resource-intensive services in the NHI leads to high medical expenditure (Cheng, 2015; Wu et al., 2010). This has led the Taiwanese government to focus recently on the overuse and misuse of health resources. For example, the NHI introduced a new system of calculating premiums for insured people and companies to raise more funding from a variety of sources (Lan, 2017). The NHI is starting to be reformed but still requires significant work to bring it into a financially stable position (Cheng, 2015).

1.1.2 The mental healthcare structure in Taiwan

Almost one in four people in Taiwan suffer from common mental health disorders, including depression and anxiety disorders (Fu et al., 2013). The number of people who have a serious mental illness nationally was estimated to be about 203,587 in 2016 (almost 1% of Taiwan's population) (National Health Insurance Administration in Taiwan, 2016b). A 20-year repeated cross-sectional survey found an increasing trend in the prevalence of common mental disorders in Taiwan, which might be influenced by economic issues, such as the unemployment rate (Fu et al., 2013). Although it is difficult to draw any conclusions about the cause of this trend due to the complexity of mental

illness, given the rapidly increasing prevalence of mental health problems, this sector of the population has started to receive more attention in the healthcare system in Taiwan.

The three levels of mental health services are primary mental healthcare, hospital mental healthcare, and community mental healthcare (see Figure 1.1). In Taiwan, primary mental healthcare refers to the general practitioners (GP) system; hospital mental healthcare focuses on the general hospital or specialised hospital systems; community mental healthcare refers to specialised and long-term rehabilitation systems.

Figure 1.1: The health care structure in Taiwan



The treatment and care team in mental healthcare comprises psychiatrists, mental health nurses, mental health nurse practitioners, clinical psychologists, occupational therapists, case managers, and social workers. Generally, mental health professionals are responsible for delivering mental healthcare and discussions with patients about their treatment and care. Each of the parts of mental healthcare is described below.

1.1.2.1 Primary mental health services

Primary care in Taiwan aims to provide comprehensive first contact and generally promote physical and mental health where people live or work. District public health centres and clinics/general practitioners (GPs), which mainly provide primary care, are mainly made up of general practitioners (GPs), community nurses, and pharmacists. The role of the healthcare professionals in primary care include preventing mental health issues and providing a psychiatric assessment. There are 372 district public health centres

(Health Promotion Administration, 2014) and 11,313 clinics (National Health Insurance Administration in Taiwan, 2016c) covering all regions in Taiwan.

1.1.2.2 Hospital mental health services

Hospital care aims to provide more specialised healthcare for people with mental illness. The core aim of the healthcare is to ensure patients have highly effective care and treatment, as well as to relieve symptoms when they experience a mental illness.

This level refers to hospital systems and mainly comprises inpatient mental healthcare. There are three sub-levels of hospital mental health services. These are medical centres (the highest level), regional hospitals, and district hospitals (the lowest level) across the geographical regions of the country. These facilities are provided by both the public and private sectors.

Unlike primary mental healthcare, hospital systems include most kinds of mental health professionals, including psychiatrists, mental health nurses, clinical psychologists, occupational therapists, and social workers.

1.1.2.3 Community mental health service

Community mental health services aim to provide services to people with long-term serious mental illness. Community rehabilitation institutions in Taiwan are defined as follows:

"The institutions provide rehabilitation therapy including occupational skills, attitude to work, social skills and activities of daily living in order to help patients with mental illness to re-integrate into family life and society."

(The Legislative Yuan of the Republic of China, 2007)

In Taiwan, the major institutions included in community mental services consist of community mental health centres, outpatient departments, day centres, day hospitals, rehabilitation services, halfway houses, and other support services. These different institutions include private and public providers.

The institutions provide professional services to assist the patient to rehabilitate and return to family and community life (The Legislative Yuan of the Republic of China, 2007). Within community mental healthcare facilities, there is at least one full-time mental health professional providing mental healthcare. As some community mental health services are attached to hospital systems, some health professionals may be responsible for both types of mental healthcare.

1.1.3 Psychiatric patient classification system

Overall, mental health treatment and care is provided in a variety of settings for people with mental illness. In Taiwan, the type of care people receive is dependent on how they have been classified by the psychiatric patient classification system. The core standard of psychiatric patient classification is designed to improve the efficiency and quality of care for people with mental illness. People with mental health problems are assigned to one of three types of mental healthcare (see Table 1.1).

However, there are still misgivings about the implementation of this system in practice (Hsieh and Shiau, 2006) and it has not been widely adopted by mental health professionals as a result. As mentioned earlier, due to the inadequacy of the referral system and the easy accessibility of mental health services, people are often not allocated the appropriate level of care.

Level	Criteria	Service
First	Psychotic symptoms	Hospital care
	needing acute treatment	
Secondary	Remission;	Hospital care or community
	unstable condition	care
	needing active treatment	
Third	Continuing psychotic symptoms	Hospital care or community
	which influence social life;	care
	ineffective treatment outcome,	
	needing long-term hospitalisation	
Fourth	Stable psychotic symptoms;	Community care
	partial deficits in functioning;	
	potential for rehabilitation	
	no need to be hospitalised all day	
Fifth	Stable psychotic symptoms;	
	chronic conditions;	
	no need to be whole-day hospitalised	
	needing assistance in daily life	
Sixth	Stable psychotic symptoms;	
	chronic condition;	
	no need to be whole-day hospitalised,	
	including the elderly, those with	
	dementia or mental retardation, or	
	homeless people with mental health	
	issues	

(Ministry of Health and Welfare, 1998)

1.1.4 Patients in halfway houses

In Taiwan's health system, halfway houses are one of the community care services. According to the psychiatric patient classification system, patients at third and fourth level are considered to meet the criteria for halfway houses. Ideally, these halfway houses should offer temporary accommodation where patients with mental illness can stay and get support to reach their recovery goals, and learn to live independently (Shu et al., 2001). Although these patients have long-term mental health conditions and need continual medical support, they may have the potential for community rehabilitation. Based on this, psychiatrists from a psychiatric hospital or an outpatient department assess the patients' conditions and then decide if referral to a halfway house is appropriate. Halfway houses set up regulations and ask patients to comply, and provide therapeutic programmes, such as pharmacotherapy, occupational therapy and vocational training (Shu et al., 2001). People residing in halfway houses are regularly followed up by mental health professionals in mental health outpatient departments in hospitals. Shu et al. (2001) stressed that the halfway houses aim to encourage patients with mental illness to take more responsibility for their own lives, rather than be passive recipients of medical care. However, her study also found that patients in halfway houses in Taiwan experienced a low level of autonomy (Shu et al., 2001).

The reality is that patients with mental illness in Taiwan are more likely to transit between inpatient wards and rehabilitative facilities rather than return to the community. Although these patients' conditions are stable, they are more likely to be forced by health professionals and family to stay in halfway houses, and be absent from the decision-making process. A number of factors contribute to this situation.

Firstly, there is widespread stigma in Taiwanese society towards patients with mental illness, particularly those with schizophrenia (Zhuang et al., 2017). In traditional Chinese culture, these patients are commonly considered to be mad, unpredictable and dangerous (Wong and Xuesong, 2011), which limits their inclusion in society.

Secondly, in Taiwanese traditional culture, it is expected that family members should take responsibility for taking care of patients with mental illness, including financial support. It is likely to make more financial sense for family members to pay for care in a halfway house. National Health Insurance, costing somewhere between £100 and £250 per month, is substantially cheaper than the average cost of living in Taiwan. In some

cases, this fee is waived with evidence from a psychiatrist of a diagnosed serious mental illness (National Health Insurance Administration in Taiwan, 2015).

Finally, recent legislation has also been drafted which prevents long admissions to acute inpatient facilities, due to the high cost (The Dispute of Committee in the National Health Insurance, 2008). Patients care is subsidised for 60-90 days, and if the admission extends beyond this period, that patient must pay higher fees and the hospital must justify the need for an admission that is longer than the average. If health professionals do not make a compelling argument for the need for the admission, any subsidies towards patient care from the government are withdrawn. Therefore, health professionals and families tend to find some mental health facilities, such as halfway houses to place these patients.

1.1.5 Taiwan Mental Health Act

One important development in mental healthcare was the evolution of mental health laws in Taiwan. In the early 1980s, two psychiatrists were assigned by the government to conduct a national survey to examine mental health facilities and staff resources in the country (Chuang et al., 1995). They found unpleasant and prison-like environments to be widespread across mental health facilities, and as a result the Taiwan Mental Health Act was established in 1990 (The Legislative Yuan of the Republic of China, 2007). However, the focus was still on reducing danger from patients with mental illness and protecting the public, rather than on patients' human rights (Tang, 1997).

At this stage, despite the efforts of the government and professionals, patients with mental illness were still compulsorily admitted to mental health facilities where there were no mental health professionals. There was a series of criminal cases related to mental health patients and the abuse of mental patients in a religious institution, which accelerated the enactment of the Mental Health Act in Taiwan. One significant abuse scandal (Long Far Temple) emerged concerning patients with mental illness (Wu and Cheng, 2017). The director of Long Far Temple used metal chains to "help" people with mental illness. This Buddhist temple chained patients with stable mental conditions to patients with more serious conditions, in an attempt to transfer healthy behaviour/thought processes from the stable patient to the unstable one. The pictures of these patients shocked the Taiwanese people and the government started to pay more attention to the patients' human rights.

Accordingly, the Taiwan Mental Health Act was amended several times to ensure patients' rights in cases of compulsory admission. In the most recent version, the act aims to protect patients from discrimination or stigma, to help patients and their families with recovery, to ensure that the final decision on the compulsory admission of patients with serious mental health illness is approved by the authorised review committee, to encourage patients to return to and stay in the community, and to promote and prevent mental health issues (Chou, 2015).

The next section considers the evidence base for patient involvement in shared decision making and the rationale for the thesis.

1.2 Shared decision making

Shared decision making is a two-way communication process between patients and healthcare professionals, where both collaborate in making healthcare decisions. (Charavel et al., 2001; Edwards and Elwyn, 2009; Entwistle, 2009). Charles et al. (1999) conceptualised decision making as a dynamic process and identified a typology of actions that come under this heading including information exchange, deliberation and decision making. In theory, both patients and healthcare professionals follow a series of actions, including sharing information, identifying the task of making a decision, understanding the best evidence about risks, and identifying the benefits of different types of therapy (Légaré and Witteman, 2013; Mead and Bower, 2000). Based on this information, both parties deliberate, culminating in patients and healthcare professionals reaching a consensus (Charles et al., 1997; Charles et al., 1999).

This model assumes that both health professionals and patients can contribute relevant information during the decision-making process. Adams and Drake (2006) note that healthcare professionals have the newest evidence-based information on diagnosis, the course of illnesses, and treatment options. Although patients may have limited knowledge of medical issues, they are the experts on their treatment preference, their own values, and their treatment goals (Charles et al., 1997). Taking into account the voices of both of these parties together, can, therefore, make for better decisions for patients (Slade, 2017).

1.2.1 History of shared decision making

It is acknowledged that the relationship between healthcare professionals and patients is one of the most complicated interpersonal relationships (Edwards and Elwyn, 2009). Prior to the development of the biopsychosocial model, healthcare professionals, as biomedical experts, generally adopted a paternalistic attitude toward patients (Adams and Drake, 2006). In the mid-1900s, some clinicians focusing on the doctor-patient relationship started to develop an increasing interest in patients' perspectives of their illnesses and clinicianclient relationships (Balint, 1957). Following this, others directly advocated that patients' experiences of illness and psychosocial factors should be considered during decision making, rather than only professional knowledge (Ellwood, 1988).

The term "shared decision making" was first adopted in 1982 in the USA (President's Commission for the Study of Ethical Problems in Medicine and Behavioral Research, 1982, p.44). The report states that:

"Shared decision making requires that a practitioner seek not only to understand each patient's needs and develop reasonable alternatives to meet those needs but also to present the alternatives in a way that enables patients to choose one they prefer. To participate in this process, patients must engage in a dialogue with the practitioner and make their views on well-being clear."

It is recognised that shared decision making is designed to shift healthcare to a system where patients are well informed and collaborate with their health professionals (Delbanco et al., 2001). One famous phrase, "nothing about me without me", succinctly expresses the essence of shared decision making (Delbanco et al., 2001). Shared decision making has been widely accepted by experts in evidence-based medicine and widely recommended in general medical encounters (Charles et al., 1999; Slade, 2017).

Shared decision making has gradually become the principal mechanism for ensuring patient involvement. In many countries, shared decision making is promoted and integrated into healthcare policy for its potential to improve the health outcome of patients (Coulter et al., 2011; Légaré and Witteman, 2013). For example, in 2011 the Salzburg Global Seminar, which aims to formulate core goals of shared decision making, suggested that policymakers adopt a policy to encourage implementation of shared decision making in healthcare (Salzburg Global Seminar, 2011).

Most recently, an increasing body of theoretical and empirical literature has examined the issue of shared decision making in healthcare. A review in the early 1990s found that only 10 publications per year were indexed with the key word "shared decision making", increasing to 75 in 2003 (Makoul and Clayman, 2006). From 2004 to 2014 this rose to 34,378 (Koster, 2014).

1.2.2 Definition of shared decision making

Shared decision making has been defined as:

"An approach where clinicians and patients share the best available evidence when faced with the task of making decisions, and where patients are supported to consider options, to achieve informed preferences."

(Elwyn et al., 2010)

Although an agreement on a definition of shared decision making has not been reached, some key attributes have been widely accepted. For example, it is recognised as a dynamic process. As part of this process, at least two people, including a patient and a health professional, are involved and act as partners in decision making. Both parties share information and keep well informed about the benefits and risks of the treatment and expected outcomes, and their preferences, values and circumstances. Based on these, both partners try to reach a mutual agreement regarding preferred treatment options (Hoffmann et al., 2014; Légaré and Stacey, 2009; Sheridan et al., 2004).

1.2.3 Different interpretations of shared decision making

Many interpretations of the concept of shared decision making in healthcare have been described in the literature (Makoul and Clayman, 2006; Stacey et al., 2010). Originally, Emanuel and Emanuel (1992) were concerned with the physician-patient relationship in decision making and argued that a mutualistic relationship is the best way to allow both healthcare professionals and patients to take active roles in decision making. According to this model, both patient and physician hold equal power, and responsibility and deliberation are needed on the part of both; patient values are considered, and healthcare professionals give advice based on these values (Emanuel and Emanuel, 1992). After this, Charles et al. (1997) provided a framework of shared decision making which had been described in the previous section, and has frequently been used in shared decision making intervention studies (Makoul and Clayman, 2006).

Nonetheless, there remains much debate among researchers and clinicians about the interpretations of shared decision making. For example, Elwyn et al. (2000) stated that healthcare professionals should take more responsibility for extracting or responding to patients' preferences during shared decision making. Charles et al. (1997) assumed that healthcare professionals and patients should eventually agree on the decisions being made. However, others have suggested that 'agreeing to disagree' about a course of action is also possible in shared decision making (Joseph-Williams et al., 2017).

While there is no common conceptual interpretation of shared decision making, elements of shared decision making have been widely accepted by many researchers. Makoul's systematic review synthesised 161 conceptual models of shared decision making and identified nine characteristics of health professionals' behaviour in shared decision making, including:

- define/explain the problem
- provide treatment options
- discuss pros/cons (benefits/risks/costs)
- understand the patients' values/preferences
- discuss the patient's ability/self-efficacy
- present professional knowledge/recommendations
- check/clarify the patient's understanding
- make or explicitly defer decisions
- arrange follow-up

(Makoul and Clayman, 2006)

Makoul's framework is applicable across many health service settings. However, his model is limited in that it only focuses on the interaction between patients and health professionals, rather than including patients' families and the wider multi-disciplinary team (Slade, 2017).

1.2.4 Shared decision making is recommended in mental healthcare

Shared decision making is emphasised in policies in many Western countries (Coulter et al., 2011; Del Piccolo and Goss, 2012). In traditional mental healthcare, the main goal was described as helping patients "comply with their treatment plan" rather than involving them in deciding it (known as the paternalistic model) (Deegan, 2007). By implementing shared decision making, the traditional power imbalance between health professionals and patients can be reduced (Capers, 2010). Meanwhile, it is evident that involving both health professionals with expertise in clinical knowledge and patients with expertise in personal beliefs and values produces better decisions, which could bring more benefits to patient recovery (Slade, 2017).

1.2.4.1 Ethical perspective

From an ethical perspective, shared decision making is a basic human right (Drake and Deegan, 2009; Nelson et al., 2001). The right to liberty and self-determination should involve full engagement in the decision-making process and the communication of understandable and unbiased information about treatment and care (Coulter and Collins, 2011). The ethical imperative for shared decision making should build on the principles of good practice, respecting patients' right to know (Elwyn et al., 2013a). In this case, shared decision making which safeguards the autonomy of patients by encouraging their involvement in making decisions about their treatment and care is recommended.

1.2.4.2 Clinical evidence

Research evidence has identified improved outcomes from implementing shared decision making in mental healthcare (Hauser et al., 2015; Joosten et al., 2009; Joosten et al., 2008; Malm et al., 2003). Active patient involvement leads to better engagement, better quality decisions, and improved adherence, which are recognised as indicators of better outcomes (Slade, 2017). The evidence supporting this is as follows.

One systematic review included 11 randomized controlled trials across different health settings (two directly focused on mental healthcare) which aimed to examine the effectiveness of intervention to improve shared decision making in practice (Joosten et al., 2008). The review found that five studies (two of which were focused on mental healthcare) reported that shared decision making had a positive effect on outcome measures including treatment adherence, and depression levels were improved by implementing shared decision making (Joosten et al., 2008). However due to methodological limitations of the included studies (e.g. small sample size), only limited conclusions can be drawn.

A later systematic review aimed to examine the effects of shared decision making on patient-related outcomes, and included 22 studies (Hauser et al., 2015). Five of the included studies involved decision making in mental healthcare, one focusing on people with substance-misuse (Joosten et al., 2009), three on patients with schizophrenia (Hamann et al., 2007; Hamann et al., 2011; Malm et al., 2003), and one on patients with depression (Loh et al., 2007). They found that ten of the included studies indicated positive effects of shared decision making, including two which were focused on mental healthcare. For example, in Sweden, a randomised controlled trial with a two-year follow-up, involving 53 patients with schizophrenia, found that shared decision making led to improved social functioning and higher satisfaction with treatment (Malm et al.,

2003). Another follow-up study with 220 substance-dependent patients in the Netherlands showed that shared decision making significantly improved patient-related outcomes, including reduction in primary substance use and addiction severity (Joosten et al., 2009).

A Cochrane review (Duncan et al., 2010) found only two randomised controlled trials. One study conducted in Germany with 405 patients with depression found that the level of patient participation in treatment decision making and patient satisfaction improved without increasing consultation time (Loh et al., 2007). The second randomized controlled trial (also conducted in Germany) involved 107 patients with schizophrenia, with results showing no difference in the health outcome between intervention and control groups (Hamann et al., 2007). A more recent systematic review which explored patients' experience of clinical assessment involved twelve studies (ten quantitative studies, one qualitative, and one mixed methods) of patients with bipolar disorder (Fisher et al., 2016). In this review, the one qualitative study included took place in the UK and involved 28 patients with mental illness (Bilderbeck et al., 2014). The review found that patients desired more active roles in decision making, and that the advantages of implementing shared decision making included improved treatment adherence, increased patient satisfaction, and reduced suicidal ideation. Whilst the reviews findings are informative, the scarcity of qualitative studies raises concerns of the absence of the patient voice. The most recent systematic review aimed to examine a rationale for shared decision making in mental healthcare (James and Quirk, 2017). The results suggested that shared decision making was widely acceptable and was considered to bring benefits to mental healthcare. Although some positive effects of implementing shared decision making were identified, the reviewers concluded that they were not able to draw any firm conclusions about the strength of the evidence.

The existing quantitative studies have produced some valuable evidence about the benefits and effectiveness of shared decision making, but as yet there has not been adequate critical appraisal of the possible qualitative evidence, particularly in mental healthcare. There is a lack of qualitative systematic reviews which directly examine how patients view shared decision making in mental healthcare. Further reviews are needed to understand this field comprehensively.

In the next section, a discussion of the challenges involved in implementing shared decision making will be presented.

1.2.5 Challenges to implementation of shared decision making

Despite these developments in shared decision making, studies show that patients are often not included in decisions about treatment, and shared decision making is not widely implemented in mental healthcare (Bee et al., 2015; Beitinger et al., 2014; Joseph-Williams et al., 2014a; Patel et al., 2008).

Two narrative reviews (Beitinger et al., 2014; Patel et al., 2008) and one systematic review (Bee et al., 2015) formed similar conclusions about shared decision making. A narrative review including 24 studies revealed that patients desired a more active role during decision making, but patient involvement did not occur (Patel et al., 2008). However, it is evident that implementation of shared decision making benefits to patients (Patel et al., 2008). A recent narrative review (Beitinger et al., 2014) which focused on patients with schizophrenia and included 18 articles found that the outcome of shared decision making is positive but implementation of shared decision making in mental health needs further work. In addition, a recent systematic review aimed to examine how patient-involved care planning is operationalised within mental healthcare (Bee et al., 2015). This review included 117 studies and concluded that there is a lack of patient involvement in mental healthcare (Bee et al., 2015).

This lack of patient involvement is illustrated by the 2017 Community Mental Health Survey in England which found only 52 % of the responders fully agreed that they were involved in decision making about the "medication" they received, and 55% fully agreed that they were involved in decision making about "treatment or therapies" to be used as much as they wanted (N=8,944) (Care Quality Commission, 2017). Of the responders who received new medication, only 54% reported that they were definitely given information about the medication in a way they could understand (Care Quality Commission, 2017). The evidence suggest that efforts to implement shared decision making in mental healthcare have met with resistance. Some of the key challenges are summarised below.

1.2.5.1 Professional paternalistic attitude

Professionals' attitudes could be a key to influencing the level of patient involvement in decision making (Charles et al., 1999; Cribb and Entwistle, 2011; Kon, 2010). Patients have also acknowledged that their ability to participate depends on their healthcare professionals (Frosch et al., 2012). To facilitate shared decision making, healthcare professionals need to actively encourage patients to engage in the shared decision making

process (Edwards and Elwyn, 2009; Mead and Bower, 2000). Therefore, the role of health professionals is vital for patient involvement in making decisions.

However, one study which analysed eighty transcripts in Italy found that health professionals only made limited attempts to involve patients with mental illness in decision making (Goss et al., 2008). In the UK, Quirk et al. (2012) analysed audiotapes of 92 outpatient consultations and examined health professionals' attitudes during consultations. The results revealed that health professionals often adopted a paternalistic attitude in the decision-making process.

1.2.5.2 Concerns about patients' impaired decisional capacity

While many studies generally support the positive impact of shared decision making, health professionals are concerned about the decisional abilities of patients with mental illness. Two qualitative studies have explored barriers to shared decision making from psychiatrists' perspectives. Both studies took place in the UK, one involving 21 general adult psychiatrists (Seale et al., 2006) and the other 26 consultant psychiatrists (Shepherd et al., 2014). Both studies showed that healthcare professionals considered mental health patients to have impaired abilities or poor insight into mental illness, leading to unwillingness to involve them in decision making. In such cases, psychiatrists were afraid that patients would make risky decisions which would worsen patient outcomes.

Despite evidence to the contrary, Auerbach (2000) argued that some healthcare professionals distrust the ability of people with mental illness to make competent decisions. It is challenging for many patients with mental health problems to believe that health professionals are willing to share with them the right to make decisions together (Happell et al., 2004). Thus, some decisions about treatment or care are related to a complex risk-benefit trade-off which is not just about treatment outcome but also about patients' personal lives (Drake and Deegan, 2009).

This view has been contested: for example, a study in the United States which aimed to examine decisional capacity for informed consent in schizophrenic research subjects found that patients with schizophrenia had the same decisional ability (after training) as those without schizophrenia (Carpenter et al., 2000). Similar studies have concluded that patients with mental health problems are able to understand treatment information and make rational decisions (Bunn et al., 1997; Carpenter et al., 2000; Hamann et al., 2006; Stroup et al., 2005).

1.2.5.3 Passive patient preference

Another consideration is patient preference in decision making. Patient preferences about being involved in shared decision making vary in healthcare. Some patients may prefer to make informed decisions, but others may want to leave the final decision to their health professionals (Légaré and Witteman, 2013). It is potentially challenging for health professionals to involve patients with low motivation in decision making. Thus, patient preference about participation in decision making exists on a continuum from passive to highly active, but those who are actively involved reported greater satisfaction or better outcomes (Kessler et al., 2001).

It is argued that patients' role preferences should be discussed and accepted, even if patients prefer a passive role (Elwyn et al., 2000). As this debate has emerged, an increasing majority of patients have indicated that they prefer active involvement in deciding their treatment and care (Chewning et al., 2012). A systematic review which explored the involvement preference of patients with substance use disorder concluded that the patients desire a more active role in decision making (Friedrichs et al., 2016).

Patients' role preferences in decision making have received increasing attention in practice (Clarke et al., 2015). Despite the extended discussion between health professionals and patients in general practice, preference discussion is still limited in mental healthcare. For example, in the USA, researchers who analysed 191 mental health visits found that more time was spent on scientific discussion (about pros and cons, nature of decision, uncertainties, goal/context, and understanding) than on discussing preferences (about the patient's role, alternatives, and preferences) in these medical encounters (Fukui et al., 2015). It could be important for making decisions to encourage the patients to express their views about their preferences, and gain an insight into what factors influence these opinions.

1.2.5.4 Time-consuming issue

A common argument proposed by some health professionals is that the process of involving patients in the decision-making process is time-consuming (Ford et al., 2002; Légaré et al., 2008). Health professionals cite the difficulty of looking up evidence about certain conditions during a busy working day and bringing this evidence into consultations with patients (Ford et al., 2002). Given the relatively short consultation times, health professionals have argued that they do not have sufficient time to implement shared decision making (Charles et al., 1999; Coulter, 2010; Towle et al., 2006).

Although health professionals struggle with their busy timetables, a variety of approaches could be employed to improve the process of shared decision making. For example, many health professionals are concerned about providing patients with sufficient information on treatment and care in a short consultation time. Thus, verbal explanations, which were viewed as the time-consuming part, may be optimised in the following ways: audiotapes, videos, booklets, computer programs, and interactive websites could be used to provide the information more efficiently to help patients to understand.

In addition, there is substantial evidence to the contrary: a Cochrane systematic review which aimed to assess the effects of decision aids for people facing treatment or screening decisions yielded inconsistent results regarding consultation length where shared decision making was practised (Stacey et al., 2014). Nine of the included studies related to consultation length of shared decision making found mixed results. Six studies found there was no difference in consultation lengths between a shared decision-making group and a control group. Two studies showed that the shared decision making group took longer than the comparison group. One indicated that the shared decision making group took less time in consultation. In particular, one study in the USA focused on shared decision making in primary care for depression, and found that the consultation time did not differ between those with and without shared decision making. Therefore, it is difficult to conclude whether shared decision making is a time-consuming procedure.

Overall, although implementation of shared decision making in mental healthcare is challenging in some ways and some concerns have been discussed, there is a clear desire for patient involvement in decision making and various benefits have been recognised. Further efforts are needed to enhance active patient involvement in decision making in mental healthcare.

1.2.6 Decision aids

It is clear that, world-wide policy makers welcome the integration of shared decision making into healthcare (Harter et al., 2011). Shared decision making is now part of patient-centred care policies in some western countries, such as the UK (Coulter et al., 2011; Elwyn et al., 2010). While governments have recently promoted shared decision making in healthcare, its implementation still needs further work, as described in the earlier sections, due to various barriers, such as health professionals' attitudinal issues, limited time resources, and concerns about mental conditions (Coulter et al., 2011). Hence, the use of decision-support tools and resources is fundamental to facilitating shared decision making in practice.

The International Patient Decision Aid Standards Collaboration, which is a world-wide group of researchers, health professionals, patients, and carers who are interested in the design and use of decision aids (DAs), defines them as follows:

"Patient decision aids are tools designed to help people participate in decision making about healthcare options, with the goal of promoting deliberation between patients, healthcare providers, and others about those options."

(Volk et al., 2013)

Despite there being no agreement on the definition, some studies have suggested that implementation of shared decision making requires both skill and awareness on the part of health professionals (Makoul and Clayman, 2006) and the involvement of patients (who are experts on their own values and preferences) (Slade, 2017). A study which included 212 participants (researchers, practitioners, patients, and policy makers) produced a quality criteria framework for Decision Aids, and identified a number of key requirements (Elwyn et al., 2006) including:

- systematic development process
- providing information about options
- presenting probabilities
- clarifying and expressing values
- using patient stories
- guiding/coaching
- disclosing conflicts of interest
- providing internet access
- balanced presentation of options
- using plain language
- basing information on up to date evidence
- establishing effectiveness.

A number of decision support tools have been developed to overcome the challenges and facilitate shared decision making in healthcare, but there is no consensus about the definition of decision aids. Decision aids are often employed in the context of shared decision making to support patients to make medical decisions (Coulter and Collins, 2011). To ensure that patients are able to actively engage in decision making, decision aids are utilised to support them with clear and understandable information about their conditions and treatment options (Drug and Therapeutics Bulletin, 2013). By promoting

active patient involvement, decision aids differ from merely educational materials such as passive informed consent materials (O'Connor, 2001).

A variety of forms of decision aids have been used, including a simple one-page sheet, a script of option outcomes, education and counselling about option outcomes, audiotapes, videos, booklets, computer programmes, decision boards, and interactive websites (Coulter and Collins, 2011; Ng et al., 2013). Some are designed to be completed by patients prior to their medical encounters and others are designed to be completed during the consultation (Substance Abuse and Mental Health Service Administraction, 2010).

1.3 Rationale for the study

In general, positive effects of shared decision making have been identified and some efforts have been made to support its implementation in practice. Although most patients seem to desire a more active role in decision making, from the evidence, shared decision making does not take place routinely in mental healthcare. Many barriers or challenges have been identified in the literature.

Furthermore, current research on shared decision making has primarily been conducted in western countries (Coulter et al., 2011; Elwyn et al., 2010), but is less visible in other parts of the world (Capers, 2010). Although some policy makers in Taiwan have started to promote shared decision making in healthcare since 2016, further work is still needed both in studies and in practice (Liao et al., 2017).

In more submissive cultures, shared decision making could face more challenges. In such cultures, people expect a paternalistic model in decision making and assume that health professionals should take on the responsibility of being decision makers (Coulter and Jenkinson, 2005). Similarly, in a typical East Asian family, such as in Taiwan, authoritative family members are more likely play a decision-maker role and patients are more likely to play a silent role during decision making, due to the traditional cultural pressure (Gilbar and Miola, 2015).

The issues for shared decision making in mental healthcare are likely to be compounded and challenging because of societal perceptions of patients with mental illness, and it is implied that the mental health professionals, particularly psychiatrists, should hold the most power and decide for patients with mental illness based on their best interests (Shepherd et al., 2014). Also, in the view of health professionals, patients with mental illness have a lack of insight, which renders them unable to make reasonable decisions (Shepherd et al., 2014). There is still very little evidence about the process of decision making in Asia, and specifically in Taiwan. Therefore, there is a need to explore and address the use and perspectives of shared decision making in mental healthcare and in other cultures. This study starts to address this gap in evidence.

1.4 Summary

This chapter has provided an overview of mental healthcare in Taiwan and shared decision making in current practice, together with explanations of the various benefits but also challenges of implementing shared decision making in mental healthcare. The next chapter will provide a discussion of the methodology used for this thesis.
Chapter Two: Methodology

This chapter will present a discussion of the methodological foundations for this thesis. An overview of research paradigms and a rationale for adopting the approach taken will be provided. Following this, the aims and objectives of the study will be outlined. Specific details of how the author ensured trustworthiness and methodological quality will be given. Finally, details of ethical considerations and obtaining ethical approval for this research will be presented.

2.1 Research paradigms

The term "research paradigm" is defined as an interpretative framework, which is underpinned by a set of beliefs and views about the world and how it should be understood (Guba, 1990; Kuhn, 1970). A research paradigm creates an integrated view of how knowledge is perceived and how researchers can see themselves in relation to this knowledge when formulating and developing a research proposal. Meanwhile, it is not only imperative to consider what methodological strategies will be used to answer a research question, but also important to understand the justification for the methodological strategies (research design) and the researchers' own philosophical assumptions (Guba, 1990).

There are highly abstract principles guiding researchers in their exploration of the universal aspects of human beings (Denzin and Lincoln, 2005). Beliefs regarding ontology, epistemology and methodology are involved in these principles, which guide researchers in acting in and seeing the world (Denzin and Lincoln, 2003b; Weaver and Olson, 2006).

Ontology refers to the kind of being the human individual is, and what the nature of reality is; epistemology is about the relationship between the enquirer and the known, and the methodology sets out to explore how we gain knowledge about or know the world (Weaver and Olson, 2006). In other words, ontology is about understanding what reality is, epistemology is about how people know something, and methodology is how people find it out. Researchers can apply a variety of fundamental strategies to develop new knowledge.

Selecting a suitable choice of research paradigm is based, therefore, on the nature of the research questions which the researcher intends to ask (Denzin and Lincoln, 2005). Many research paradigms are applied for the purposes of health research (Creswell, 2007; Green and Thorogood, 2009; Teddlie and Tashakkori, 2009). The two most common

research paradigms have significantly influenced researchers for decades: positivism and interpretivism, which lead to the different research methodologies (Tuli, 2010). The positivist paradigm is commonly aligned to a quantitative methodology, whilst interpretivism refers to a qualitative methodology (Rahman, 2016). Both qualitative and quantitative approaches are based on philosophical traditions and have different ontological and epistemological assumptions. The section below provides a more detailed overview of each paradigm.

2.1.1 Positivism versus interpretivism

2.1.1.1 Positivism

A positivist view focuses only on factual knowledge which is gained through observation and measurement, and positivists believe that there is a single reality (Crotty, 1998). Positivism holds the ontological principle that an external objective reality exists for any situation or research phenomenon, which is independent of the researchers, and the epistemological principle that this reality is free from any time and context and can be generalised to all human knowledge (Denzin and Lincoln, 2005). The basis of positivism is that researchers can be totally objective and reality exists independent of the observer (Pring, 2000; Scotland, 2012). In other words, when adopting positivism, researchers are independent of the research and the research can be purely objective. Furthermore, positivist researchers believe that this reality can be found through scientific experiments which are quantitative methods, such as nomothetic experiments, laboratory experiments, and quantitative analysis (Denzin and Lincoln, 2005; Neuman, 2011). Thus, a positivist study is purely objective and trustworthy with a rigorously structured research method.

However, beyond these features of this research paradigm, one disadvantage is obvious. The epistemological disadvantages of positivism give rise to reservations about its appropriateness for the field of nursing studies (Clark, 1998). Positivism refers to the idea that researchers need to concentrate on facts rather than on the meaning of human behaviours and human interests (Aliyu et al., 2014; Rahman, 2016). Researchers adopting positivism are limited to using an objective approach to collect data and interpret the meaning of data (Aliyu et al., 2014). For social science, a positivist approach is significantly lacking in its ability to capture the complexity and meaning of human behaviours and social phenomena (Denzin and Lincoln, 1998).

2.1.1.2 Interpretivism

Conversely, interpretivism holds the view that reality is constructed by social phenomena created by humans, and exists in multiple, context-specific situations (Green and Thorogood, 2009; Guba and Lincoln, 1994). Interpretivist studies aim to generate descriptions of and insights into social phenomena and unveil the meanings of social behaviours (Jacox et al., 1999).

The ontological assumption of interpretivism is relativism (Guba and Lincoln, 1994). Reality is subjective and socially constructed by individual interpretation and cannot be separated from the "person" (Mack, 2010). Interpretivists believe that there is no single reality, and therefore those multiple realities need to be interpreted (Guba and Lincoln, 1994). The reality is mediated by individual senses and the world is pointless without consciousness (Scotland, 2012).

The epistemological assumption of interpretivism is one of subjectivism based on real world phenomena, and the belief that knowledge is inductively gained through individual experience. Only if meaning is constructed though the interaction between consciousness and an individual world can it be discovered and transmitted in a social context (Crotty, 1998). Interpretivists believe that even individuals in the same population can create their own meanings for an event and interpret the same event differently. The way to experience a world is to participate in it, which is both to encounter and to mould (Heron and Reason, 1997). Therefore, through the position of individuals who are participating in it, the social world can be understood (Cohen et al., 2007). Also, individual social constructs that can be understood are extracted through interaction between participants and researchers (Guba and Lincoln, 1994).

Some argue that the interpretivist approach is advanced and more scientific for understanding social science (Travers, 2008). The paradigm presents researchers with the chance to understand individual experiences and meanings in some depth, and has been widely recognised as appropriate for use in qualitative studies (Creswell, 2007). Furthermore, researchers in interpretivist studies should treat participants as human rather than objects of natural science, to capture the subjective meaning of social phenomena (Mack, 2010).

Despite the features of interpretivism, some researchers consider that there are some disadvantages to interpretivist studies due to the subjective nature of this approach. Data, therefore, is heavily influenced by individuals' perspectives and values. In these cases, it

could be difficult to generalise the findings from these studies to other situations (Mack, 2010). However, as interpretivist studies do not aim to be generalisable, such studies can provide thick descriptions of data and a high level of validity (Mack, 2010).

2.1.2 Applying the interpretivist paradigms

Overall, from a philosophical perspective, positivists aim to formulate laws based on prediction and generalisation, and propagate the application of empirical methods to answer a research question (Scotland, 2012). Positivists attempt to identify what factors affect the outcomes which researchers are interested in (Creswell, 2007). Interpretivists aim to understand social phenomena from an individual's perspective, probing interaction between people and the cultural and social contexts where they live (Creswell, 2007). Also, the interpretivist paradigm has proved the best way for researchers to understand and interpret participants' perspectives of the phenomena being studied (Guba and Lincoln, 1994), so these statements support the choice of interpretation for health researchers who propose to investigate a phenomenon in a group of patients or particular health settings.

Furthermore, in health studies, human beings are subjects who can help researchers understand their social behaviours. Adopting the interpretivist paradigms assists researchers to observe the phenomena from the inside, rather than objectively observe it from outside (Mack, 2010). Unlike positivism, it is impossible to separate the researchers from the research participants, especially in social sciences, due to the interactions between humans (Mack, 2010). In other words, from interpretivists' viewpoint, the idea is that pure objectivity on human behaviour is difficult. Therefore, the interpretivist paradigm is beneficial to gain insight and understanding of behaviour, in order to explain the meanings of actions from the perspective of participants (Scotland, 2012).

It is still unclear how patients and healthcare professionals engage with each other around decisions for the care and treatment of the patients in the Taiwanese context. This paradigm could significantly help health researchers when they require in-depth and insightful information about a dynamic process. As the aim of this study was to explore and understand patient and health professional perspectives of shared decision making in secondary mental healthcare, and their social and historical constructions of the context in Taiwan, an interpretive paradigm is the most judicious choice to explore patients' and health professionals' perspectives.

2.2 Research aim and objectives

2.2.1 Aim

• to explore/understand the process of decision making, from both health professional and patient perspectives, in secondary mental healthcare in Taiwan.

2.2.2 Objectives

Key objectives included:

- to examine and synthesise patient perspectives of shared decision making in secondary mental healthcare using a systematic review of qualitative studies;
- to explore patient perspectives of shared decision making in secondary mental healthcare in Taiwan using semi-structured qualitative interviews;
- to explore health professional perspectives of shared decision making in secondary mental healthcare in Taiwan using semi-structured qualitative interviews.

A detailed discussion of methodologies for each objective will be presented, including qualitative systematic review methodology and qualitative research methodology.

2.3 Qualitative systematic review methodology

2.3.1 Types of systematic reviews

A literature review is a summary of a subject field which can be used to support the identification of specific review questions (Rowley and Slack, 2004). Although all review types are required be systematic at some level, the extent could vary between different types of reviews depending on the methods used (search, appraisal, synthesis, and analysis) (Booth et al., 2012). There are two main types of literature review: systematic, and traditional narrative reviews (Cook et al., 1997; Greenhalgh, 2010).

A narrative review is a summary which applies informal or subjective methods to collect and interpret evidence (Grant and Booth, 2009). Narrative reviews are used to summarise the evidence and are mainly descriptive, without involving systematic evidence-based criteria (Greenhalgh, 2010). They often include searches and selections based on the reviewer's personal interests and subjective judgments, potentially leading to bias (Cook et al., 1997). The extraction of information from findings is subjective and lacks explicit criteria for inclusion (Green et al., 2006). Using this approach, the data is described rather than interpreted and it is therefore less rigorous (Paré and Kitsiou, 2017). A systematic review is a summary of the literature that uses systematic, reproducible, explicit methods to search, critically appraise, and synthesise literature on a specific topic (Gopalakrishnan and Ganeshkumar, 2013). The main purpose of a systematic review is to identify key scientific contributions to a particular field, or to answer a specific question based on pre-identified criteria (Higgins and Green, 2011). The use of systematic methods helps minimise bias and provide impartial conclusions (Cipriani and Geddes, 2003; Cook et al., 1997; Higgins and Green, 2011). Systematic reviews can help researchers to understand inconsistencies between various studies, and produce more precise recommendations than those from narrative reviews (Cook et al., 1997).

While quantitative systematic reviews are well established through the Cochrane Collaboration's work, qualitative systematic reviews have been developed more recently and are still evolving (Seers, 2015). Most systematic reviews involve a meta-analysis which requires statistical techniques to synthesise the quantitative data for included studies (Petticrew and Roberts, 2005). The Cochrane Collaboration has raised awareness of qualitative systematic reviews in health and now has a Qualitative and Implementation Methods Group (Gulmezoglu et al., 2013).

2.3.2 Selecting qualitative systematic review methodology

Qualitative systematic review methodology was selected to be consistent with the review aims and on the bases of the research questions. These were: to identify qualitative evidence of patient perspectives of shared decision making in secondary mental healthcare, and its barriers and facilitators. This review aims to systematically identify, extract and synthesise evidence about patient perspectives of shared decision making.

A qualitative systematic review systematically searches for studies on a topic, extracts the findings from each study, and synthesises the evidence together to explore a review question (Grant and Booth, 2009; Higgins and Green, 2011; Seers, 2012). Stern et al. (2014) stated that the questions used in qualitative systematic reviews emphasise the perspective of the participants involved in and experiencing the phenomena with which the researchers are concerned.

Noblit and Hare, who are pioneers in the area of qualitative synthesis, describe qualitative reviews as interpretative (Noblit and Hare, 1988). Interpretative approaches aim to interpret the data, and from that interpretation to develop a theory which is helpful for understanding or predicting human behaviour (Seers, 2015). Furthermore, the value of generating qualitative evidence as thick descriptions has been recognised (Denzin, 1989).

Thick descriptions, which are often extracted from qualitative studies, are "deep, dense and detailed accounts of problematic experience". This means that thick descriptions focus on deeper aspects of phenomena, such as feelings and the meaning of human actions (Holloway and Wheeler, 2010). Consistent with interpretivist paradigms, in this review, the author therefore focused on thick descriptions from qualitative evidence to explore the interpretations in depth.

For complex interventions, the need for rigorous evidence must come from multi-method evaluation and high-quality reports which focus on both the effectiveness of interventions and efforts to implement the interventions more widely (Thomson, 2009). Dixon-Woods et al. (2006) argue that qualitative systematic reviews would not be limited to studying the effectiveness of the interventions but would also be able to make unique contributions to in-depth understanding of the intervention, such as identifying particular patterns and contexts. In other words, qualitative systematic reviews could contribute to determining what interventions work or do not work, and to understanding when, how, or why an intervention could be effective. In this way, evidence-based interventions and practices can be reviewed through a broader, culturally rich and contextual lens.

Because of complex and numerous issues related to providing services and support to patients with mental illness, qualitative approaches that explore experiences, perspectives, relationships, and phenomena are required. Meanwhile, detailed portrayals of aspects of shared decision making, including the mechanisms whereby shared decision making functions, in-depth interpretations of how patients view the implementation of shared decision making in mental healthcare, and facilitators of/barriers to shared decision making, could be extracted from thick descriptions in qualitative studies. These features of qualitative systematic reviews could meet the research question of this review: to systematically explore patients' perspectives of shared decision making within the existing literature.

2.3.2.1 Search tool

A search tool is a technique to identify focused research questions and to improve the comprehensiveness of systematic reviews in various ways, including determining inclusion/exclusion criteria, searching studies, extracting data from studies, and reporting findings (Aslam and Emmanuel, 2010; Higgins and Green, 2011). In this review, the SPIDER tool was used to organise the process of the review.

In the search stage, a search tool standardises systematic search strategies and assists researchers to identify possible search concepts and increases rigour in research (Cooke et al., 2012). The majority of search tools emphasise quantitative studies more, such as PICO (Population, Intervention, Comparison and Outcome), which is designed to identify and particularise quantitative enquiries (Richardson et al., 1995). Thus, Cooke et al. (2012) argue that this tool is inappropriate for a qualitative systematic review.

The SPIDER tool, developed by Cooke et al. (2012), is a five-concept tool which supports incorporating the synthesis of qualitative research into systematic reviews; the concepts include sample (S), phenomenon of interest (PI), design (D), evaluation (E), and research type (R) (Cooke et al., 2012). Based on this tool, one research question of this review is, 'What perspectives of shared decision making prevail amongst patients with mental illness in secondary mental healthcare?'

It might be not necessary to use each concept of the tool to the same extent (Higgins and Green, 2011). For example, if researchers applied the concept, "research type (R)" in the SPIDER tool to narrow down the number of included studies, they could miss out some mixed-methods studies, and meaningful qualitative parts of mixed-methods studies could be ignored by searches. In order to include more potentially relevant studies at the search stage, the researcher only used three of the SPIDER tool's search term concepts to identify studies, namely "sample", "phenomenon of interest", and "evaluation".

2.3.2.2 Data extraction

Data extraction is the process of selecting and recording relevant data from the studies included in the review (Heyvaert et al., 2016). The extraction procedure typically involves summarising all relevant data in the included studies which meet the criteria of the review. This relevant data often includes pertinent contextual issues, characteristics of the population, phenomena of interest, research design, methodology, and the outcome (Heyvaert et al., 2016; White and Schmidt, 2005).

While extracting findings from qualitative studies, it might be difficult to ascertain what counts as relevant findings and to identify the key concepts and succinct summaries of findings (Thomas and Harden, 2008). Therefore, extracting all eligible data could avoid omitting potential findings related to review questions (Noyes and Lewin, 2011). Furthermore, Thomas and Harden (2008) recommend taking all of the eligible texts labelled as results or findings as data to be processed in a thematic synthesis. This

approach therefore guided data extraction in order to extract all relevant findings and results from each included article.

2.3.2.3 Quality of the review

Critical appraisal is the process of systematically assessing and judging the quality of research in a specific field (Higgins and Green, 2011). The critical appraisal of the quality of studies is an essential component in evidence-based practice (Clarke and Oxman, 2003; Mhaskar et al., 2009). The level of quality of a qualitative systematic review can determine whether the review can discover new understandings, often helping to clarify "why", and bringing the potential to establish a new theory (Seers, 2015).

Although a range of critical appraisal tools is available for the assessment of the quality of qualitative studies, there is no significant consensus on the best tool (Dixon-Woods et al., 2006). In this review, the consolidated criteria for reporting qualitative research (COREQ) were applied.

The COREQ, developed by Tong et al. (2007), comprises a 32-item checklist. It helps researchers to understand important aspects of the research team, study methods, context of the study, findings, analysis and interpretations (Tong et al., 2007). The COREQ is recognised as a suitable tool for researchers conducting systematic reviews of qualitative studies (Booth et al., 2014).

2.3.2.4 Synthesis/Analysis of qualitative studies

Qualitative synthesis has been defined as "a methodology whereby study findings are systematically interpreted through a series of expert judgements to represent the meaning of the collected work" (Bearman and Dawson, 2013). By involving qualitative evidence, researchers can interpret social phenomena including the mechanisms of acceptance, feasibility, the meaningfulness of an intervention, and related barriers to and facilitators of an intervention (Booth, 2016). Qualitative synthesis also has the potential to identify omissions and gaps in existing research, and can add interpretation and depth to qualitative evidence (Major and Savin-Baden, 2010).

A variety of methods have been developed to use with these approaches, including narrative synthesis, critical interpretive synthesis, grounded theory synthesis, metaethnography, meta-study, and thematic synthesis (Bearman and Dawson, 2013; Dixon-Woods et al., 2007). There are some descriptions of different approaches in Table 2.1 below. It should be noted that the development of qualitative synthesis is still not complete and there are still relatively few examples of some approaches being applied (Tacconelli, 2010).

Synthesis method	Description
Meta-ethnography	A set of techniques for synthesising qualitative studies. It
	involves the selection, comparison and analysis of studies to
	create new interpretations or concepts.
Thematic synthesis	The identification of recurrent or important themes. Findings
	of included studies are summarised under thematic headings.
	Information is tabulated, allowing identification of
	prominent themes and offering structured ways of dealing
	with the data in each theme.
Grounded theory	Originally, grounded theory was developed for use with
	primary studies and describes methods for sampling, data
	collection and data analysis. Recently, its potential
	application to the synthesis of multiple qualitative studies
	has been explored.
Content analysis	A systematic technique for categorising data into themes and
	counting the frequency of the themes.
Case survey	A formal process for coding data from qualitative cases into
	a quantitative form
Qualitative research	A set of techniques for the interpretive integration of
synthesis: qualitative	qualitative findings.
meta-synthesis	
Qualitative research	A set of techniques for the quantitative aggregation of
synthesis: qualitative	qualitative research findings.
meta-summary	
Narrative synthesis	A general framework and specific tools and techniques that
	help to increase transparency and trustworthiness. It can be
	applied to reviews of quantitative or qualitative research, as
	individual tools and techniques can be selected according to
	the type of study design and the type of data included in the
	review.

Table 2.1: Methods proposed for the synthesis of qualitative evidence

(Tacconelli, 2010)

Thematic synthesis was selected for the qualitative systematic review in this study. Originally, the aim of thematic synthesis was to explore these approaches to see "what works" (Ring et al., 2011). Thematic synthesis involves the clear identification of prominent or recurrent themes, and organises and summarises the findings of different studies according to specified thematic headings (Dixon-Woods et al., 2005; Thomas and Harden, 2008). Thematic synthesis is widely applied in qualitative synthesis to explore participants' experiences and views (Garcia et al., 2002; Mcinnes and Chambers, 2008;

Morton et al., 2010). Thomas and Harden (2008) have also detailed each step of thematic synthesis and provide a clear example process for health studies. Accordingly, thematic synthesis was selected for this study.

Thomas and Harden (2008) proposed three steps in thematic synthesis (see Figure 2.1):

- Free line-by-line coding of textual findings from primary studies;
- Organisation of these free codes into descriptive themes;
- Development of analytical themes.

Figure 2.1: The steps of thematic synthesis



For free coding line-by-line, researchers code a text one line at a time to identify some meaningful codes. Each piece of data in turn could be annotated using a set of preexisting codes, or new codes could be added if necessary.

Next, researchers start to review similarities and differences among the codes assigned in the first step to group them into a parallel structure. In some cases, new codes may be applied to capture the meanings of a group of initial codes. Thus, in this step descriptive themes are generated from an inductive analysis of the findings of the included studies.

Finally, based on the above steps, the findings of the included studies are integrated into a complete picture. By using descriptive themes from the above inductive analysis, "going beyond" the content of the original studies and interpreting the relationships between descriptive themes using a hierarchical tree structure leads researchers to answer their review questions. Thus, analytical themes are generated in this step.

The author sought an approach to summarise and analyse evidence from qualitative studies exploring how patients viewed shared decision making in secondary mental healthcare. It was expected that this approach would also help the author to recognise appropriate themes to meet the aims and objectives of the study.

2.4 Qualitative research methodology

An increasing body of literature has emphasised the value of qualitative studies (Denzin and Lincoln, 2005; Flick, 2009). Qualitative approaches are important in helping researchers to determine the nature of a phenomenon and the meaning of an experience (Fitzpatrick and Boulton, 1994; Lobiondo-Wood and Haber, 2013). The data and findings of qualitative studies provide an in-depth understanding of insights into research problems (Creswell, 2007; Denzin and Lincoln, 2003b). Specific cultural and social factors can be identified through qualitative works, and such factors can be helpful to improve quality of care (Pope et al., 2002). For the development of qualitative research, many methodologies are discussed in the literature (Flick, 2009). Based on the interpretivist paradigm, the qualitative methodology for primary studies (patient interviews and health professional interviews) was selected.

2.4.1 Qualitative approaches

Generally, different qualitative approaches share a similar goal: to reach an understanding of a specific phenomenon from the perspective of people experiencing it (Vaismoradi et al., 2013). Therefore, a considerable overlap can exist between different qualitative approaches in terms of methods, process and techniques (Vaismoradi et al., 2013). Many types of theoretical methodologies for qualitative research are employed in studies: these include: grounded theory, phenomenology, ethnography, case study, action research, and a generic qualitative approach. The approaches will be discussed further in the following sections.

2.4.1.1 Grounded theory

The main purpose of grounded theory is to construct a theory that can help to integrate human behaviours, experiences, and actions (Lobiondo-Wood and Haber, 2013). In a grounded theory approach the research questions focus on a problem that is associated with the complexities and processes of human actions and interactions. The core of grounded theory includes using systematically inductive principles of data collection and data analysis to establish theoretical frameworks (Denzin and Lincoln, 2003b).

2.4.1.2 Phenomenology

Phenomenology aims to investigate or inquire into the meanings of experiences as people live with them, and to reveal an insight into the essence of the lived experience (Flick, 2007; Van Manen, 2007). Therefore, phenomenological research is applied to describe, explain and understand the meaning of human experiences.

2.4.1.3 Ethnography

Ethnography is identified as a study approach to understand behaviours, social interactions, and perceptions that happen within groups, teams, organisations, and communities (Reeves et al., 2008). This approach is designed to describe, understand and interpret a specific social circumstance with a complete range of opinions and cultural diversity (Gerrish et al., 2010). The findings of ethnographic studies can help researchers to explore how cultural beliefs, knowledge and values affect the experiences of a specific population in a particular setting (Lobiondo-Wood and Haber, 2013; Patton, 2002).

2.4.1.4 Action research

Action research involves doing research with and for people in the context of its application, and has been recognised as a facilitator and evaluator of change (Gerrish et al., 2010). In the healthcare field, action researchers work closely with health providers to systematically observe the issues of changing practice, and then develop potential solutions to practical problems (Meyer, 2000). Engaging in action research is an empowering experience for researchers (Gerrish et al., 2010; Meyer, 2000). This approach requires that participants recognise the need to change and are also willing to actively take part in the change process (Meyer, 2000).

2.4.1.5 Generic qualitative approach

A generic qualitative approach is described as "an approach to seek to explore and understand a phenomenon, a process, or the perspectives and worldviews of the people involved" (Merriam, 1998). Generic qualitative approaches offer an opportunity for researchers not to be limited by a single established methodology but to utilise the tools that established methodologies provide, and to formulate research designs which fit their epistemological positions and specific research questions (Kahlke, 2014). Researchers who employ a generic qualitative approach need to make their theoretical position explicit (Caelli et al., 2003; Merriam, 1998). Caelli et al. (2003) argue that although a generic qualitative approach is not guided by a definitive or traditional set of philosophic assumptions, as in one of the qualitative methodologies above, a growing trend of generic studies has appeared, particularly in applied disciplines (such as nursing).

Although other approaches above, such as grounded theory, phenomenology and ethnography, help researchers to frame methodological strategies, in some ways they also limit their applicability (Percy et al., 2015). Also, there is a broader public debate around research genres and studies which do not fit with any established methodologies (Caelli et al., 2003). In some cases, the study questions might fit more than one established

methodology (Cooper and Endacott, 2007; Kahlke, 2014) and researchers need to consider a more generic qualitative approach.

To meet the main aims of this study, to explore patient and health professional perspectives of shared decision making in secondary mental healthcare in Taiwan, a generic qualitative approach was selected. A generic qualitative approach using thematic analysis could identify both study participants' beliefs, experiences, preferences, and views about how decision making has been processed or shared. Meanwhile, this study simply seeks to explore the perspectives of those involved in a specific decision-making process, with no requirement to build a theory (grounded theory) or examine cultural diversity (ethnography). Therefore, a generic qualitative approach was applied to gather the information from patients and health professionals.

2.4.2 Qualitative data collection methods

Within qualitative research, three kinds of data collection are used: interviews, observation, and focus groups (Creswell, 2007; Ellis, 2013; Mack et al., 2005).

Observation is a data collection approach in which researchers directly observe participants in natural settings or occurring events (Creswell, 2007). A focus group is a group discussion on a specific topic designed to answer a research question (Creswell, 2007). The discussion is processed, guided and recorded by a researcher. Interviews share many common features with a focus group, but there is only one participant, who is individually interviewed (Mack et al., 2005). Each method has different strengths and weaknesses, as detailed in Table 2.2.

Туре	Strengths	Weaknesses
Observation	• Allows insight into contexts, relationships, behaviours	 Time-consuming Ethical dilemma of consent of participants
Interview	 Elicits in-depth responses, with nuances and contradictions Gets at an interpretive perspective i.e. the connections and relationships a person sees between particular events, phenomena and beliefs Addresses sensitive topics 	 Lack of professional interviewing experience Long time needed for preparation and management Subjective bias of the interviewer
Focus group	 Elicits information on a range of norms and opinions in a short time Group dynamic stimulates conversation and reactions 	 Difficult to motivate and involve all participants Difficult to address sensitive topics

 Table 2.2: Strengths and weaknesses of different types of data collection

(Ellis, 2013; Mack et al., 2005)

2.4.2.1 Selecting an interview approach

In this study, interviews were used to generate data from both patients and health professionals. An interview has been defined as a conversation between two persons which is instituted by interviewers for the specific purpose of gathering information on specific topics, and focuses on research objectives, description or explanation (Cohen et al., 2007). In-depth interviews are integral to an interpretive approach and the decision to interview people about research topics is derived from the epistemological and ontological positions that knowledge about people's subjective reality can only be obtained from the people themselves (Crotty, 1998). Some scholars (Kvale and Brinkmann, 2009; Patton, 2002) have argued that in-depth interviews aim to directly draw out perspectives, feelings, opinions, views and experiences from participants. Similarly, Miller and Classner (2011) suggested that the in-depth interview could clarify the nature of the specific phenomena of interest, such as the culture framing the subjects' experiences and social world.

Furthermore, qualitative interviewing approaches are suitable for investigating sensitive topics (Elam and Fenton, 2003). It is evident that participants may be more comfortable discussing sensitive perspectives or topics of experience in an individual one-to-one interview rather than through other approaches (Ellis, 2013; Mack et al., 2005). Certain research topics in nursing and health have the potential to involve sensitive topics which cause participants distress and discomfort (Enosh and Buchbinder, 2005). Therefore, in this study interviews were used to explore the perspectives of patients and health professionals.

2.4.2.2 Semi-structured interviews

There are various types of interview, including structured, semi-structured and unstructured (Flick, 2009; Kvale and Brinkmann, 2009).

A structured interview is an interview in which both the questions and the answer categories have been fully developed before the interview (Floyd and Fowler, 2004). By contrast, an unstructured interview has no specific set of pre-determined questions and allows participants to explore their own experience and develop their own directions in the interviews (Fontana, 2004). A semi-structured interview asks a series of preset open-ended questions (Ayres, 2012). The topic guide is developed for semi-structured interviews, which is a series of topics to discuss with participants (Fitzpatrick and Boulton, 1994).

The semi-structured interview is known for its flexibility, enabling researchers to follow up new ideas and produce richer data (Al-Busaidi, 2008). Unlike structured interviews, which have been criticised for their inflexibility and leave little room for unanticipated exploration (Murphy et al., 1998), semi-structured interviews allow the researcher to be more open to exploring the participants' point of view (Flick, 2009). Compared with an unstructured interview, a semi-structured interview is an efficient approach which assists participants to focus more on the topics in which the researcher is interested (Kvale and Brinkmann, 2009). Furthermore, the semi-structured interview is commonly used in health-related qualitative studies (Al-Busaidi, 2008). Therefore, this study applied semistructured interviews to collect data from both patients and health professionals.

2.4.3 Sampling

A sample has been identified as "a finite part or subset of participants drawn from the target population" (Martínez-Mesa et al., 2016). Two main techniques of sampling are probability and non-probability.

Probability sampling aims to randomly select the smallest population which can be considered as representative of the whole population or wider society (Marshall, 1996). In other words, every subject has an equal chance of being selected to be a part of the study. Typically, probability sampling techniques are commonly employed for quantitative studies but are rarely appropriate for conducting qualitative research (Marshall, 1996).

However, non-probability sampling is a sampling technique where not all participants are given an equal chance of being included (Etikan et al., 2016). In other word, non-probability sampling is where researchers select participants based on the subjective judgment of the researchers rather than on a random basis. However, non-probability does not aim to meet the criterion of randomness. Instead, subjective methods based on researcher interest are used to select which criteria are used to select the sample (Etikan et al., 2016; Flick, 2009).

Denzin and Lincoln (2003a) state that the aim of sampling in qualitative studies is to understand a specific topic and generate information-rich data. This means that the findings of qualitative studies are intended to be applied to the specific population which has possibly the richest source of information to answer the study questions, rather than the whole population. In fact, non-probability sampling is more typically employed in qualitative enquiries (Creswell, 2007). For these reasons, non-probability sampling was used to explore the topic in this study.

There are many different types of non-probability sampling in qualitative research, including purposive, snowballing, convenience, and theoretical sampling (Creswell, 2007; Flick, 2009; Marshall, 1996). Purposive sampling is a non-probability sampling technique in which the researcher chooses from a large population subjects who are available to give an account of their experiences. Therefore, the goal of purposive sampling is to select participants with specific characteristics which are of interest, which will best enable researchers to answer their research questions (Flick, 2009). Unlike convenience sampling is used where subject characteristics are purposely chosen to achieve population diversity, which may identify essential differences between participant viewpoints (Teddlie and Yu, 2007).

This sampling approach can provide researchers with the justification for making generalisations from the target sample in which the researcher is interested, whether such generalisations are theoretical, analytic and/or logical in nature (Polit and Tatano Beck, 2004). For these reasons, purposive sampling, which is often widely used in social

sciences, was applied in this study for both the patient and the health professional interviews.

2.4.3.1 Sample size

Sample size is the number of participants in a sample for an experiment, a survey, or other study type (Kadam and Bhalerao, 2010). Qualitative researchers are more concerned with the richness and depth of data than the number of participants (Patton, 2002). In a qualitative study, sample size is based on including participants who are capable of providing rich descriptions of particular phenomena (Denzin and Lincoln, 2003a). Therefore, researchers in qualitative studies generally discuss reaching data saturation when deciding the size of sample.

Data saturation has been defined as the point at which additional data do not generate any new codes or themes, i.e. the researcher finds that no new codes occur in the data (Given, 2016; Urquhart, 2013). In this study, data collection continued until the data saturation point was reached, which was when further sampled data did not lead to any new themes emerging (Given, 2016). Sample size in this study therefore reflects the number of participants who were able to provide new and relevant information for describing and interpreting the phenomenon adequately.

Although data saturation is the guiding principle, an estimated sample size for this study was provided. There are no definite rules for sample size in qualitative studies (Patton, 2002) and a wide range of different sample sizes in qualitative research are reported in the literature. For example, Thomson (2011) reviewed 100 qualitative studies related to grounded theory and found the average sample size was 25, but the suggested sample size should be 30 to fully develop all aspects of the given phenomena (Thomson, 2011). Similarly, another review (Mason, 2010) included 560 PhD studies and the result shows that, although there is no consensus on the size of sample for qualitative studies, the common sample sizes were 20 and 30 (median was 28 and mean was 31). Others have stated that, according to the experience of most qualitative researchers, few new themes come out of transcripts, and data saturation is reached, after researchers have interviewed 20 participants (Green and Thorogood, 2009).

In this study, approximately 20-25 eligible patients in halfway houses and approximately 20-25 eligible registered mental health professionals in secondary mental healthcare in Taiwan were to be recruited to participate in semi-structured interviews, to ensure data

was not lost and to achieve richness of information. Nevertheless, the sample size could have decreased or increased depending on when data saturation was achieved.

2.4.4 Data analysis

Qualitative data analysis methods should be planned prior to data collection as part of the research design process and should fit the theoretical position of the selected research methodology (Creswell, 2007). In qualitative studies, the data analysis process generally involves preparing and managing the data, reducing the data into codes and then themes, and finally producing a report in the form of figures, tables, or a discussion (Creswell, 2007). This process is crucial to provide a high-quality qualitative report. Researchers should be aware that the qualitative data analysis process is circular, not linear, and researchers need to interact closely with the data (Maxwell, 2009). Dey (1993) also mentioned that it is better for qualitative researchers to learn data analysis by actually doing it, to foster analytical skills.

Before the discussion on the data analysis approaches, researchers should identify the types of qualitative data sources in their studies. Some researchers might focus on a single data resource but others might involve more than one data resource in the same study (Sutton and Austin, 2015). There are some common ways to generate qualitative data, such as interviews, observations, documents, and audiovisual material (Creswell, 2007). Qualitative data, which is generally non-numerical, comes in a variety of forms, including:

- audio recordings
- video recordings
- written notes, transcripts
- images (e.g. pictures, photos, drawings or diagrams)
- documents (e.g. diaries, newspapers, magazines, or reports)

(Ellis, 2013; Mack et al., 2005)

There are a number of approaches to qualitative data analysis with different analytical methods, and focusing on different aspects of the data (Thorne, 2000). While there are numerous guidelines for qualitative data analysis, applying these guidelines involves judgment and creativity due to the uniqueness of each qualitative study (Patton, 2002). Some common types of data analysis have been used in health studies:

- narrative analysis
- discourse analysis
- constant comparative analysis
- content analysis
- thematic analysis

2.4.4.1 Narrative analysis

Narrative analysis is an approach for understanding how and why people talk about their lives as a story, which is influenced by self-identity and social interaction between the narrator and audience (Earthy and Cronin, 2008). The main narrative themes within the accounts which people provide about their lives can be detected by analytic processes, and this information can explain how people understand and make sense of their lives (Thorne, 2000). Narrative analysis strongly emphasises how people present their accounts of themselves and the stories that people tell provide insights into their lived experiences, the complexity of human lives, cultures, and people's behaviours (Sandelowski, 1994).

2.4.4.2 Discourse analysis

Discourse analysis is an analysis strategy for exploring texts within the sociocultural practice (Wetherell et al., 2001). The nature of discourse analysis is hermeneutic and phenomenological, focusing on the lifeworld and the meaning people make through the application of language (Miles, 2010). It assumes that there are multiple constructed realities, and researchers working within this assumption aim to understand the interplay between social contexts and language.

Typically, this method involves an analytical process including deconstructing and critiquing language use and the social context of language use (Miles, 2010). In discourse analysis, speech is not just a direct representation of human experience, but an explicit linguistic tool constructed by many social or ideological impacts. The focus of discourse analysis is to develop theories in sociolinguistics and cognitive psychology in order to understand what is represented by the variety of ways in which people communicate ideas (Thorne, 2000).

2.4.4.3 Constant comparative analysis

Constant comparative analysis is an approach for analysing data to develop a grounded theory. In the analysis process, newly gathered data are continually compared with previously collected data to develop conceptualisations of the possible relationships between the data (Glaser and Strauss, 1967). This process aims to examine emerging

ideas which might take the research in new and insightful directions. In constant comparison, the properties and dimensions of the categories are generated and patterns are developed that improve the interpretive power of these categories (Holloway and Todres, 2007). Glaser and Strauss (1967) also argue that the comparative analytical method can be suitable for social units of any size to generate theory. This process involves four stages:

- comparing incidents applicable to each category
- integrating categories and their properties
- delimiting the theory
- writing the theory

(Glaser and Strauss, 1967)

2.4.4 Content analysis

Content analysis is an approach for analysing large volumes of textual data and different data sources to corroborate evidence and make replicable and valid interpretations from the data to its context (Elo and Kyngas, 2008). As Elo and Kyngas (2008) also explained, this approach can be employed with either qualitative or quantitative data. The approach aims to offer an insight into specific phenomena in which the researcher is interested. Furthermore, Cavanagh (1997) described content analysis as a flexible method for analysing textual data, and content analysis has been widely applied in health studies in recent years (Hsieh and Shannon, 2005). However, this approach emphasises quantification of data, so the frequency of different categories and themes in a qualitative study is recognised as a proxy for significance (Vaismoradi et al., 2013).

2.4.4.5 Selecting thematic analysis as a study approach

In this study, thematic analysis was employed to explore patient and health professional perspectives of shared decision making in secondary mental healthcare in Taiwan. Thematic analysis is based on the identification of themes from the data, identified by a coding scheme, often from pre-determined themes. The approach is widely used in qualitative analysis, generally treating accounts as a rich resource to uncover the reality or experiences to which they refer. It is evident that this approach is widely used across various epistemologies and qualitative research questions (Braun and Clarke, 2006; Nowell et al., 2017). In a generic qualitative research approach, thematic analysis is recommended (Percy et al., 2015).

Braun and Clarke (2006) assert that thematic analysis should be the foundation for qualitative analysis due to core skills being developed through thematic analysis for

conducting other forms of data analysis. Thematic analysis emphasises ascertaining, examining, and recording themes within data (Braun and Clarke, 2006) and aims to present the key components of each participant's account (Braun and Clarke, 2006; Green and Thorogood, 2009). This approach also helps researchers to recognise typical responses and identify the salient responses of particular groups of participants (Green and Thorogood, 2009). It is used to analyse qualitative data systematically. Braun and Clarke (2006) argue that qualitative studies can produce trustworthy and insightful findings through a rigorous thematic analysis (Braun and Clarke, 2006).

In addition, due to its theoretical freedom, it is highly flexible and can be modified to meet the needs of many studies, and to provide an information-rich and detailed account of the data (Braun and Clarke, 2006; King, 2004). As the use of this approach does not require technological or theoretical knowledge of other qualitative research methodologies, thematic analysis is more usable for researchers who have little experience of conducting qualitative studies or are not familiar with qualitative analysis (Braun and Clarke, 2006).

There are several key steps related to thematic analysis:

- familiarisation with the data
- generating initial codes
- searching for themes
- reviewing themes
- defining and naming themes
- producing the report

(Braun and Clarke, 2006)

Firstly, researchers read the transcribed data several times and make notes of initial ideas during this process (Braun and Clarke, 2006; Green and Thorogood, 2009). Repeated reading can help researchers to grasp the whole picture of the raw data. Braun and Clarke (2006) indicated that taking notes could ensure researchers are well prepared to carry out coding.

Coding is the next phase, and categorising meaningful data into groups facilitates the development of a logical set of themes. Patton (2002) argued that pattern recognition is a key competency in applying thematic analysis. Initial codes are produced as part of this process. A long list of codes will be generated and then the next step is to codify these into categories in order to search for potential themes. Following this, exploration of the

relationship between codes and themes could be facilitated by reviewing themes and generating a thematic map of the analysis (Braun and Clarke, 2006). Coding schemes can be applied to define and name themes. In this part of the analysis, it is essential to clarify the concept of each theme and discuss it with research colleagues (Green and Thorogood, 2009). Producing the report involves a final analysis of selected extracts and aims to produce a scholarly report of the findings (Braun and Clarke, 2006).

2.5 Ensuring quality and trustworthiness

Confirming trustworthiness is a way to ensure that the findings of qualitative research are worthy of attention (Lincoln and Guba, 1985). To achieve rigour and trustworthiness, researchers must be consistent and strict in employing sampling and analysis methods in a logical flow to ensure quality in qualitative studies (Flick, 2007; Patton, 2002).

Lincoln and Guba (1985) proposed four strategies for establishing the quality of qualitative research:

- credibility
- transferability
- dependability
- confirmability

Strategies for these criteria can be employed throughout the study process to enhance the quality and trustworthiness of qualitative research. Some need to be addressed prior to the study, but others are used during the data collection process and after the data are interpreted (Krefting, 1991). Detailed discussion will be presented in the following sections.

2.5.1 Credibility

Credibility refers to how believable each part of a qualitative study is, comparing the actual findings with the original questions for the study. However, the credibility of qualitative studies is often discredited as the finding of qualitative analysis is judgment dependent (Mays and Pope, 2000).

Thus, peer debriefing has proved to be a key technique to achieve credibility in qualitative studies (Patton, 1999). Peer debriefing is a strategy where researchers, during the research process, seek scholarly guidance and support from other professionals, such as supervisors in the case of postgraduate students (Anney, 2014). This strategy requires academic supervisors to engage in discussion and questioning of the study in a consistent and systematic manner, and all these discussions should be recorded in notes that will be useful in data analysis (Barusch et al., 2011). Through this strategy, researchers can also

explore their perspectives, reactions, and analyses as they go through the research process (Barusch et al., 2011). Also, constructive feedback from academic supervisors can help improve the quality of the qualitative study and ensure credibility (Shenton, 2004). Therefore, peer debriefing was used to ensure credibility in this study.

2.5.2 Transferability

Transferability is the generalisability of the findings of the research from a section of the research sample to the general population of individuals in a similar condition (Lincoln and Guba, 1985). While qualitative studies are traditionally not considered to be generalisable, it is still worth drawing attention to representational generalisation to ensure the quality of the research (Brett, 2018). The researcher should provide a thick description of the participants and the research process for the readers to decide if the results are transferable to other settings or contexts (Korstjens and Moser, 2018). Although the researcher cannot know every site to which a reader may wish to transfer the findings, providing thick descriptions helps other readers to judge transferability (Lincoln and Guba, 1985).

Anney (2014) also argued that thick description involves the researcher clarifying every detail of the research processes, from data collection to production of the final report. Therefore, researchers should provide sufficient contextual information about the field (Korstjens and Moser, 2018) and clearly illustrate that the sample is an accurate reflection of the population studied (Krefting, 1991). During data collection, collecting participants' demographic data and comparing it with the information about the target population will help readers judge the potential of its application in their own settings (Krefting, 1991; Shenton, 2004). For this study, thick descriptions of these aspects were provided to support transferability.

2.5.3 Dependability

The notion of dependability refers to consistency in collecting data, which has implications for the stability of a study's findings over time (Lincoln and Guba, 1985). To ensure dependability, researchers should confirm that the process of the research is logical and explicitly documented (Tobin and Begley, 2004). Korstjens and Moser (2018) also argue that researchers should check whether the analysis process is consistent with the defined standards for a research design. Some strategies can be employed to ensure dependability, including an audit trail, dense description of research methods, step-wise replication and peer examination (Krefting, 1991). In this study, an audit trail was employed to ensure dependability.

An audit trail is kept to document the methodological and theoretical decisions made and how the data were gathered, recorded and analysed throughout the study process (Bowen, 2009). In addressing this, researchers should describe their research design and its application clearly, along with details of how they carried out data collection and reflective appraisal during the study (Shenton, 2004).

2.5.4 Confirmability

Seeking confirmability means that researchers strive to ensure the objectivity of the whole study process (Lincoln and Guba, 1985). Steps should be taken to ensure that the neutrality of the findings is based on the experiences and ideas emerging from the information rather than being based on the preferences and characteristics of the researchers (Mays and Pope, 2000). Sufficient explanation of the study needs to be given, enabling the reader to judge the validity of the research. Studies recommended an audit trail, reflexive journal and triangulation to strengthen the confirmability of a qualitative inquiry (Bowen, 2009; Krefting, 1991; Lincoln and Guba, 1985).

In this study, an audit trail and reflexive journal were used to enhance confirmability. An audit trail was kept to ensure dependability, and was then also useful for confirmability. An audit trail helps other reviewers to trace the whole study through the decision making and the rest of the process as described.

In addition, reflexivity refers to how researchers consider the sensitivity of data collection during the study (Mays and Pope, 2000). A reflexive journal has been recognised as reflexive documents which are held by the researchers in order to reflect on, interpret, and plan data collection (Wallendorf and Belk, 1989). Also, it helps to assess the influence of the researchers' background, perceptions and interests throughout the qualitative study process (Krefting, 1991). Anney (2014) suggest that researchers should document all events that occurred in the field, as well as personal reflections in relation to the study. Similarly, Korstjens and Moser (2018) also argue that this process should review researchers' own biases, preferences, preconceptions, and relationships to the respondents, and how these relationships influence participants' responses.

2.6 Ethical principles

Beneficence, non-maleficence, autonomy and justice are issues addressed by the principles of biomedical ethics (Beauchamp and Childress, 2013). In this study, the potential ethical issues were considered and taken into account; the study had to be approved by the University of Manchester and local research committees in Taiwan. Further details of each principle will be presented in the following sections.

2.6.1 Autonomy

The notion of autonomy encompasses showing respect for the subjects' choices, and this often refers to the protection of a participant's ability to make his or her own decisions. One important component associated with respect for autonomy is disclosure. In order to ensure the participant's participation in the study is fully voluntary, the researchers must share the understanding of what has been agreed (Beauchamp and Childress, 2013). The researchers must provide all relevant information about the research and what will happen as a result of participants being involved to help them make informed decisions (Faden and Beauchamp, 1986). This information should include the purpose and method of the study, the risks and benefits of participation, and the right to withdraw from the study at any point without any questions (Beauchamp and Childress, 2013).

In a qualitative research study, informed consent has been recognised to ensure this principle, which means participants are able to gain sufficient information about the study by this means (Greaney et al., 2012). The importance of informed consent has been identified by Orb et al. (2001) to ensure that patient have sufficient information and to provide an opportunity for participants to refuse to take part in the study. Therefore, every participant should give written informed consent to participate in the study (Creswell, 2007; Munhall, 1993).

In particular, informed consent plays an essential role in protecting the autonomy and dignity of patients with mental illness (Carpenter and Conley, 1999). For such patients, all possible ethical principles need to be given careful consideration by those involved in the research. Meanwhile, the requirement of informed consent has been added in law to ensure patients' rights in both practice and research (Doyal, 2002). For these reasons, this study used informed consent to ensure autonomy.

2.6.2 Beneficence and non-maleficence

The principle of beneficence indicates wanting to do what is good for others: the doing of good, the active promotion of good (Beauchamp, 1984). Munyaradzi (2012) argues that the principle is expressive of humanity, altruism, unconditional love, and optionally moral ideals. In the biomedical field, beneficence requires that, to the best of their ability, health professionals try to bring positive benefits for patients, such as good health, or to prevent and eliminate harmful conditions (Munyaradzi, 2012). It is not just the possible benefits for the participants of taking part in the research but also the potential costs and risks that need to be considered.

On the other hand, non-maleficence is the principle of not inflicting evil or causing any harm to others. (Paavilainen et al., 2014). The principle of non-maleficence refers to the

obligation not to inflict harm intentionally (Beauchamp and Childress, 2013). Further examples of non-maleficence include avoiding different forms of harm in medical care, such as medical error, inequitable treatment, or negative influence on participants' benefits (Beauchamp and Childress, 2013). These are associated with either exposing participants to a risk or causing actual harm to participants. This involves avoiding even the risk of harm. Munson (2004) states that if a researcher has purposely or accidentally subjected a participant to needless risk, then he or she has violated this principle.

A risk-benefit assessment can be used to scrutinise the potential risks and benefits of participating in the study and strike a balance between all aspects, including financial, physical, mental and social (Polit and Beck, 2014). Interestingly, the researchers commonly pay attention only to financial and physical risk, but not mental or social risk (Greaney et al., 2012). For example, in qualitative interview studies, the time sacrificed to take part is commonly mentioned as the cost of participating in the research (Beauchamp and Childress, 2013). However, Greaney et al. (2012) recommend that a risk assessment should also consider psychological distress, and any factors which could cause discomfort or uncomfortable feelings to participants. Also, support mechanisms such as a distress management plan and a complaints process should be in place for participants (Polit and Beck, 2014). These potential risks and benefits were discussed in applying ethical considerations.

2.6.3 Justice

The principle of justice is that all participants have equal rights (Gillon, 1994). This is the right to be treated equally and should not be compromised for any kind of reason. All people should be provided with fair opportunities to be involved in studies (Emanuel et al., 2000).

In addition, the implementation of justice aims to recognise the vulnerability of the participants, and their possible contributions to the study (Orb et al., 2001). The same publication asserts that listening to the voices of minority and vulnerable groups and protecting those who are most vulnerable, such as children, prisoners, and the mentally ill, are further approaches to implementing justice. For example, prejudice and discrimination may be a reflection of the stigmatisation of patients with mental illness, who often experience injustice or are treated differently. It is important to provide an opportunity for this sensitive group to have their voices heard.

Therefore, Greaney et al. (2012) argue that researchers should carefully consider their criteria for including and excluding certain groups from a study. Detailed explanation of

why specific groups are excluded or included should be given, and a rationale should be provided for this decision within the research protocol (Greaney et al., 2012). In order to understand patients and professionals' perspectives of shared decision making, these two participant groups were recruited in this study. All details of inclusion and exclusion criteria will be presented in the working methods chapter (see Chapter Four).

2.7 Summary

In summary, this chapter has detailed both the qualitative systematic review and the primary qualitative studies (patient interviews and professional interviews) used in exploring the aims and objectives of this doctoral study. The epistemological, ontological and methodological decisions relevant to the study were made according to all the principles described in depth above. The following chapter will present the working methods and the findings of the qualitative systematic review.

Chapter Three: Systematic review of qualitative studies (Study One)

This chapter will present working methods and findings for a qualitative systematic review which explored patient perspectives of shared decision making in secondary mental healthcare. The rationale for selecting a qualitative systematic review was presented in Chapter Two. This chapter will present in four sections: the first section will provide details of the working methods of the review and the second will describe the review findings. The third section will present the discussion of the strengths and the limitations of this review. The final section will provide a summary of this chapter.

3.1 Review methods

3.1.1 Review aims and objectives

3.1.1.1 Review aim

• to examine and synthesise patient perspectives of shared decision making in secondary mental healthcare.

3.1.1.2 Review objectives

- to explore the decision-making experiences of patients in secondary mental healthcare;
- to examine the facilitators of and barriers to shared decision making from the perspective of patients in secondary mental healthcare.

3.1.2 Literature search strategy

In this review, the SPIDER tool was utilised to organise the review process. As outlined previously, the SPIDER tool is a search tool used to synthesise qualitative research in systematic reviews through five concepts: sample(S), phenomenon of interest (PI), design (D), evaluation (E), and research type (R) (Cooke et al., 2012). In order to include more potentially relevant studies, the author only used three of the SPIDER tool's search term concepts to search studies at this stage, namely "sample", "phenomenon of interest", and "evaluation".

A long-list of potential search terms was developed during a broader search, again conducted by the author, based on the scoping review, reading of individual studies and browsing medical subject headings (MeSH). The search strategies were also built up from a series of test searches and discussions of the results of those searches with the supervisory team. These search terms were revised and modified in subsequent meetings with the supervisory team. Any differences of opinion were discussed with the supervisory team to make final decisions. All search terms are presented in Table 3.1. The search terms for each concept were combined via the Boolean operators AND, OR and NOT.

Group A: Sample		Group B:		Group C:
		Phenomenon of		Evaluation
		Interest	_	
schizophrenia		"shared decision*"		view*
schizoaffective		"shared decision		experience*
psychosis*		making"		opinion*
psychoses*		"information sharing"		attitude*
psychotic*		"mutual decision*"		perspective*
psychiatric*	AND	"participatory decision	AND	feeling*
"severe mental"		making"		belief
"chronic mental"		"client participation"		value
"serious mental"		involvement		
SMI				
bipolar				
depress*				
"delusional				
disorder"				
"mental health				
service*"				
""= When a word or phrase is in quotes the results include only articles with the				
same words in the same order.				
=Truncation symbol; for instance, decision retrieves 'decisions' as well as				
'decision'.				

Table 3.1: Search concepts and terms

This search strategy was applied to eleven databases, namely:

- CINAHL Plus (Cumulative Index to Nursing and Allied Health Literature) (1937present)
- Medline (1946-present)
- PsycINFO (1806-present)
- Embase (1974-present)
- Health Management Information Consortium, HMIC (1979 to March 2016)
- EBM review Cochrane Database of Systematic Reviews, CDSR (2005-present)
- EBM review Database of Abstracts of Reviews of Effects, DARE (accessed via Ovid)
- British Nursing Index (1994-present)
- Applied Social Sciences Index and Abstracts, ASSIA (1987-present)
- International Bibliography of the Social Sciences, IBSS (1951-present)
- Social Services Abstracts (accessed via ProQuest) (1979-present)

Searches were limited to studies published in English because no financial resources were available for translation or for accessing databases in other languages. All databases were searched from inception to September 5, 2016. After initial searches via these databases, key studies were also hand-searched via databases.

3.1.3 Selection criteria

The process of selecting the relevant articles, i.e. defining eligibility criteria, is one of the essential principles of systematic reviews (Petticrew, 2001). Unlike the search stage, which aimed to include all possible existing studies, this stage aimed to filter the studies to include only relevant ones in the synthesis. Therefore, the all concepts of the SPIDER tool (sample/setting, phenomena of interest, design, evaluation, and research type) were used to develop the inclusion/exclusion criteria. These criteria are detailed in Table 3.2.

Concepts	Inclusion criteria	Exclusion criteria
Sample/setting	 Adults (18 and over) in secondary mental healthcare Population mixed At least 75% of the sample in above settings Or if the study results present patient perspectives separately 	 Children or youths (aged under 18) Adults in any acute psychiatric wards of hospitals
Phenomena of interest	 Patient experiences/perspectives of involvement in decision making about their treatment and care Facilitators of or barriers to involvement in decision making from patient views or perspectives 	Experiences/perspectives of only carers and/or professionals of their involvement in shared decision making about patients' treatment and care
Design	 Primary data from published articles or dissertations Qualitative methodology Interview, focus group, observation Data analysed by (as a minimum) thematic analysis 	 Case reports or one participant studies Using questionnaires to collect data Not primary data Grey literature
Evaluation	Views, experiences, opinions, attitudes, perspectives, beliefs or feelings of patients in secondary mental healthcare	Understanding and knowledge about shared decision making
Research type	 Qualitative data exploring patients' experiences and perspectives of shared decision making about their treatment and care Only English articles 	Quantitative studiesSystematic reviewMeta-analysis

Table 3.2: Inclusion criteria and exclusion criteria

3.1.3.1 Sample/settings

Serious mental illness (SMI) was defined as schizophrenia, schizoaffective disorder, affective disorder (including bipolar disorder and depression), and other psychoses. This is based on how serious mental illness is defined in the Quality and Outcomes Framework

in the UK (Health and Social Care Information Centre, 2015) and prevalence surveys (Kessler et al., 2001; Steadman et al., 2009). In some studies, populations labelled as "serious/severe mental illness" were also included in this review where no specific diagnosis was provided.

Services offering secondary mental healthcare included: community mental health centres, outpatient clinics, outpatient department, clubhouses, day centres, day hospitals, rehabilitation services, group homes, halfway houses, and other support services. Studies solely focused on patients in acute wards of hospitals were excluded. It is likely that there would be differences between the experience of patients with serious mental illness in acute inpatient wards and those with serious mental illness in other units, due to the nature and severity of symptoms (Marshall et al., 2001).

3.1.3.2 Phenomena of interest

The inclusion criteria were intended to select any qualitative studies with a primary data focus on the process of shared decision making about treatment and care, or possible facilitators and/or barriers to shared decision making, for patients in secondary mental healthcare. Studies focused on engagement with or involvement in other activities, such as research, were excluded as these are not related to the aims of this review.

3.1.3.3 Design, evaluation, and research type

Qualitative studies that were published in English and which used individual interviews, focus groups or observations to explore the perspectives of shared decision making about treatment and care amongst patients in secondary mental healthcare were included in this review. In addition, research using mixed methods or involving data from participants other than the patients themselves was included only if the results had been reported separately.

All types of qualitative empirical studies were considered to meet criteria of this review, including grounded theory, phenomenology, ethnography, case studies, action research and thematic analysis. Abstracts and other grey literature such as conference papers were excluded as they did not provide enough detailed information to extract and analyse critically (Autorino et al., 2007).

3.1.4 Eligibility screening

Each article was initially screened, by title and abstract, by two reviewers separately (the author & Dr Laoise Renwick). The reviewers rated each article as either 'relevant', 'maybe' or 'irrelevant'. If there was any conflict, the paper was discussed and if agreement was not reached the paper was sent to a third reviewer (Professor Karina Lovell).

Full-text copies of potentially relevant articles were read by the author, using the inclusion and exclusion criteria. The third reviewer (Professor Karina Lovell) checked inclusion and exclusion reasons for the first eleven articles, to enhance rigour.

3.1.5 Data extraction

An extraction sheet was developed by the author after discussion with the supervisory team (see Appendix 1). In this study, study characteristics, demographics, methodology and outcomes were extracted from eligible studies.

3.1.6 Quality appraisal

To determine the quality of the included papers, the Consolidated Criteria for Reporting Qualitative Research (COREQ) proposed by Tong et al. (2007) was employed. The COREQ is presented in Part B of the extraction sheet (Appendix 1). The COREQ consists of a 32-item checklist and was designed for appraising qualitative studies across three domains, namely research team and reflexivity, study design, and data analysis and reporting. Detailed discussion of the quality of the included studies is presented in Section 3.2.3.

3.1.7 Synthesis of qualitative studies

Thematic synthesis was used in this review, based on the work of Thomas and Harden (2008), who proposed three steps:

- Free line-by-line coding of textual findings from primary studies;
- Organisation of these free codes into descriptive themes;
- Development of analytical themes.

3.1.7.1 Step one: Free line-by-line coding of textual findings from primary studies

The first step is to code texts line by line. The use of line-by-line coding aims to inductively create free codes to explain each sentence extracted from the findings of primary studies with a parallel structure (Thomas and Harden, 2008). All extracted data were coded by the author and a long list of codes, along with the extracted data, was created. An example of the studies included in this review is detailed below (Table 3.3).

	Table 3.3: A	n example of us	e of line-by-line	coding
--	--------------	-----------------	-------------------	--------

Data extraction	Code
"I also felt I could be more honest with	• Importance of support from other
GH because of the case manager's	professionals
support, and that perhaps GH would take	
my concerns more seriously in the	
<i>future</i> . " (Delman et al., 2015)	
"I think sometimes the doctors have too	Professionals have great power
much power, especially psychiatrists,	Potential coercion
they make decisions about where you're	
going to live. I mean, Dr M basically	
made a decision on my life. She decided	
that I was under her care, she decided	
what medication I was going to have, and	
she decided where I was going to work.	
Now, isn't that power?" (Laugharne et	
al., 2012)	

3.1.7.2 Step two: Organisation of free codes into descriptive themes

The purpose of the second step was to develop descriptive themes from included studies. Based on the similarities and differences between the themes, ten main descriptive themes were identified (see details in Table 3.4).

No	Descriptive themes
1	Lack of information about treatment and/or care
2	Lack of interaction with health professionals
3	Robotic care
4	Mental illness
5	Health professionals with paternalistic attitude
6	Health professionals are qualified to decide
8	Good relationship between patients and health professionals
9	Patient believing self to be an expert on his/her own mental illness
10	Sufficient information and resources

 Table 3.4: Summary of descriptive themes

3.1.7.3 Step three: Development of analytical themes

The final step in thematic synthesis aims to interpret the findings of the included studies (Thomas and Harden, 2008). This step often includes collapsing some themes and grouping them into new or existing themes. Also, in order to ensure the themes are well represented, it is important to re-check the data used in the included studies. The descriptive themes (see above) were restructured in this stage.

After re-examination of the themes, the sub-theme "robotic care" was combined with "lack of interaction with health professionals" and renamed "poor quality of interaction with health professionals", based on similarity. In addition, the theme "lack of information about treatment and/or care" was renamed "lack of professional knowledge and information". Furthermore, two sub-themes, "mental illness" and "health professionals are qualified to decide", were collapsed into "health professionals with paternalistic attitude". Based on similarity, three descriptive themes ("professionals with paternalistic attitude", "poor quality of interaction with health professionals, "lack of professional knowledge and information") were grouped into a single main theme: "barriers to shared decision making".

Conversely, in the included studies, patients mentioned not just barriers to shared decision making but also how some facilitators improved levels of patient involvement and shared decision making. A good relationship between patients and health professionals was renamed "supportive attitude to patient involvement". After reviewing the theme, "patient believing self to be an expert on his/her mental illness" was collapsed into the descriptive theme "supportive attitude to patient involvement". After this, "sufficient information and resources" was restructured into two different themes: "sufficient information exchange" and "other support resources".

The author grouped three descriptive themes, namely "supportive attitude to patient involvement", "sufficient information exchange", and "other support resources" into one main theme and named this main theme "facilitators of shared decision making". (For an overview of analytic themes, see Table 3.5).
Main themes	Sub-themes
Barriers to shared decision making	 Professionals with paternalistic attitude Poor quality of interaction with health professionals Lack of professional knowledge and information
Facilitators of shared decision making	 Supportive attitude to patient involvement Sufficient information exchange Other support resources

Table 3.5: Two primary analytical themes from the synthesis of included studies

3.2 Findings

3.2.1 Search results

The search identified 8,808 publications and dissertations (Figure 3.1). The title and abstract screening excluded 8,743 studies. Full text screening of the remaining 65 articles eliminated a further 52 studies. The remaining 13 studies were included in the review. A summary of this process and the reasons for exclusion are presented in Figure 3.1 and Table 3.6.

Reasons for exclusion	Number of studies	Number of studies
	excluded	retained (from 65 studies)
Sample/setting	14 studies	51
Phenomena of interest	4 studies	47
Design	20 studies	27
Evaluation	0 studies	27
Research type	14 studies	13
	Total: 52 excluded	Total: 13 retained

Figure 3.1: Flow of the review



3.2.2 Characteristics of the included studies

The key characteristics of each study are summarised in Table 3.7. The review included 13 studies. The included studies were predominantly conducted in the United States (n=8) or the United Kingdom (n=4). All included studies were published in a ten-year period between 2006 and 2015 (Table 3.7).

3.2.2.1 Methodological design

All included studies used a qualitative design. The studies used various data collection processes: six used in-depth interviews (Delman et al., 2015; Gioia et al., 2014; Goscha and Rapp, 2015; Laugharne et al., 2012; Wharne et al., 2012; Woltmann and Whitley, 2010), four conducted focus groups (Connor and Wilson, 2006; Jones et al., 2013; Tanenbaum, 2008; Velligan et al., 2016), two incorporated both focus groups and in-depth interviews (Dahlqvist et al., 2015; Farrelly et al., 2016), and one used observation (Matthias et al., 2012).

Five of the studies adopted grounded theory (Connor and Wilson, 2006; Dahlqvist et al., 2015; Farrelly et al., 2016; Goscha and Rapp, 2015; Jones et al., 2013). Velligan et al. (2016) employed the constant comparative methods approach to their analysis. The remaining seven studies used thematic analysis (Delman et al., 2015; Gioia et al., 2014; Laugharne et al., 2012; Matthias et al., 2012; Tanenbaum, 2008; Wharne et al., 2012; Woltmann and Whitley, 2010).

3.2.2.2 Characteristics of participants

In total, 397 participants were included in studies, of which 341 were patients, 48 were health professionals, and eight were carers. Health professionals' and carers' views were not included in this synthesis.

Sample sizes ranged from 3 to 80 participants. Of the included studies, 9 were patientonly samples (Connor and Wilson, 2006; Dahlqvist et al., 2015; Delman et al., 2015; Gioia et al., 2014; Jones et al., 2013; Laugharne et al., 2012; Tanenbaum, 2008; Wharne et al., 2012; Woltmann and Whitley, 2010), and the remaining four used mixed samples comprising patients and carers (n=1) (Velligan et al., 2016) or patients and professionals (n=3) (Farrelly et al., 2016; Goscha and Rapp, 2015; Matthias et al., 2012).

Although all included studies reported that the participants were recruited from secondary mental healthcare settings, only 10 studies reported participants' diagnoses. Three studies did not report discrete diagnoses but used "patients with mental illness" or "serious mental illness" to describe their samples (Connor and Wilson, 2006; Farrelly et

al., 2016; Jones et al., 2013). The majority of patient participants were recruited from outpatient clinics, community and other secondary mental healthcare (Table 3.7).

	Study det	ails			Sample Characteristics	Methodology		
No	Authors	year	Country	Patient number	Sample characteristics	Setting	Data collection	Analysis
1	Connor & Wilson	2006	UK	31	• people with serious mental illness, including schizophrenia, depression and bipolar disorder	User groups and community residential settings	Focus groups	Grounded theory methodology
2	Dahlqvist_ Jönsson	2015	Sweden	20	 Psychosis - 3, schizophrenia - 3, bipolar disorder - 4, depression - 4, and others - 6 Mean age =52 (range 26-64) 6 men and 14 women 	User-organised clubhouses and drop-in centres	Five focus groups and two individual interviews	Constructivist grounded theory methodology
3	Delman	2015	USA	24	 schizophrenia - 2, schizoaffective - 8, bipolar - 9, depression - 3, and personality disorder - 2 Mean age =24 (range = 19–30) 67% were female and 33% were male 	outpatient services	Semi-structured interviews	Thematic analysis
4	Farrelly et al	2015	UK	51	 Diagnosis: not stated Mean age=39 years old 52 % were female 	Generic and specialist community mental health teams in four English Mental Health Trusts	Focus groups and individual interviews	Grounded theory methodology
5	Gioia et al	2014	USA	20	 bipolar disorder - 10, schizophrenia - 6, and other (schizoaffective or depression) - 4 Mean age=52.9 (SD: 6.8) 18 men and 2 women 	Outpatient program	Semi-structured interviews	Thematic analysis (Template analysis)

 Table 3.7: Summary of the included studies on perspectives of shared decision making amongst patients

6	Goscha	2015	USA	12	 all clients with a severe and persistent chronic mental illness, including schizophrenia and other psychotic disorder Mean age =45 7 men and 5 women 	A community mental health centre	Semi-structured interviews	Grounded theory methodology
7	Jones et al	2013	USA	54	 Diagnosis: not stated Mean age = 48 (range 39-57) 17 men and 34 women 	Secondary mental health services	Focus groups	Grounded theory approach
8	Laugharne et al	2011	UK	22	 Schizophrenia/ schizoaffective - 6, bipolar disorder - 7, and psychosis - 7 Median Age = 51 (range 38- 62) in Cornwall Age range 21-40 in London 12 men and 10 women 	Secondary mental health services	in-depth interviews	Components of grounded theory and thematic analysis
9	Matthias, et al	2012	USA	40	 Schizophrenia/schizoaffective disorder-25, major depression -8, bipolar disorder -4, or other -3. Mean age = 43.5 (SD ±15.2) 19 men 21women 	A community mental health centre with a recovery-oriented focus	Direct observation	Thematic analysis
10	Tanenbaum	2008	USA	38	 people with serious mood, anxiety or thought disorder Age (range 20's to 60's) and most were in their 30's and 40's. 16 men and 22 women 	An urban clubhouse and two at a consumer- operated centre in a suburban/rural county	Four focus groups	Thematic analysis
11	Velligan et al	2016	USA	10	 Schizophrenia - 3, affective disorder -5, and anxiety disorder - 2 age: not stated 5 men and 5 women 	Outpatient Transitional Care Clinic	Two focus groups	Open coding framework and constant comparative method

12	Wharne, et al	2012	UK	3	 Psychosis - 3 aged from late 30s to early 50s three men Not stated Semi-structured interviews 	Thematic analysis
13	Woltmann & Whitley	2010	USA	16	 Schizophrenia - 8, bipolar disorder - 4, major depression - 2 and other - 2 Mean age = 45 (SD: 8.2; range 33-58) 9 men and 7 women Three sites of a large, urban community mental health agency Semi-structured interviews 	Thematic analyses

3.2.3 Quality of included studies

The COREQ quality appraisal tool was used to assess quality (see Part B, Appendix 1). There is no official scoring system for the tool. Therefore, the author used the number of items which had been well presented in the studies to indicate their level of quality. The possible range of numbers was from 0 to 32 items (Booth et al., 2014). Furthermore, there was no cut-off for the COREQ tool (Tong et al., 2007), but for the purposes of this review, three levels of quality were applied: low (0-10 items present), medium (11-21 items present), and high (22-32 items present). To help the reader to understand how these quality levels were arrived at, the author considered that if only one third of the items were explained then the quality level was low; more than one third but not two thirds represented medium quality; more than two thirds was high quality. Detailed discussion of the assessment outcomes of each study is presented below and in Table 3.8.

Among the 13 studies assessed, no study met all of COREQ's quality criteria. Quality scores ranged between 9 and 23. One study was assessed as high quality in this review (met 23 of the criteria): the one carried out by Farrelly et al. (2016). Most of the included studies (n=11) were recognised as medium quality. The most frequently occurring COREQ score (mode) was 17 out of 32, meaning that more than half of the criteria had been properly explained (median score was also 17 and the mean score was 15.62). One was low quality and only met 9 of the criteria.

The studies were recognised as medium quality and provided some details in the "study design" (15 criteria) and "analysis and findings" domains (9 criteria). However, compared with the high-quality study, these studies only met 50% of the criteria (12 out of 24 criteria) in these domains. Also, there was a significant lack of information in the "research team and reflexivity" domain. In some cases, none or only one of 8 criteria were reported.

The lowest score was 9 for the study conducted by Jones et al. (2013) and was characterised as low quality in the assessment. Similarly, a lack of detail in the "research team and reflexivity" domain was identified. Beyond this, discussion of "research design" and "analysis and findings" in this study was unclear and met less than 50% of the criteria. The inadequacy of reporting in these two domains made it impossible for the reader to judge the research process properly.

Table 3.8:	Quality	appraisal	outcomes
	<u> </u>		

	Included studies	Connor et al (2006)	Dahlqvist et al	Delman et al.	Farrelly et al.	Gioia et al. (2014)	Goscha et al (2015)	Jones et al. (2013)	Laugharn e et al.	Matthias et al.	Tanenbau m (2008)	Velligan et al.	Wharne et al.	Woltman n et al
the COREQ			(2015)	(2015)	(2016)				(2012)	(2012)		(2016)	(2012)	(2010)
Research team a	Research team and reflexivity													
1. Interviewer/fa	acilitator	×	✓	√	×	✓	×	×	×	×	✓	×	✓	×
2. Credentials		×	~	√	✓	×	×	×	×	~	×	√	×	×
3. Occupation		~	✓	×	✓	×	×	×	✓	1	×	×	✓	×
4. Gender		×	×	×	×	×	×	×	×	×	×	×	×	×
5. Experience an	nd training	×	~	×	×	×	×	×	×	×	×	×	✓	✓
6. Relationship	established	×	~	×	✓	×	×	×	×	×	×	×	1	×
7. Participant kr interviewer	nowledge of the	×	*	×	√	×	×	×	×	×	×	*	×	×
8. Interviewer c	haracteristics	×	×	×	✓	×	×	×	✓	×	×	×	×	×
Study design														
9. Methodologia Theory	cal orientation and	√	 ✓ 	×	√	-	-	√	×	×	×	×	•	×
10. Sampling		~	×	√	~	~	~	×	~	×	~	~	×	 ✓
11. Method of ap	proach	~	~	✓	✓	✓	×	×	~	×	~	√	×	×
12. Sample size		~	~	√	√	✓	1	✓	~	1	√	√	1	✓
13. Non-participa	ation	×	 ✓ 	√	×	 ✓ 	×	×	×	1	×	×	×	×
14. Setting of dat	a collection	√	×	√	✓	×	×	×	~	×	✓	√	×	×
15. Presence of n	on-participants	×	×	×	×	×	×	×	×	1	×	√	×	×
16. Description o	f sample	×	✓	√	✓	✓	1	✓	1	1	✓	√	✓	✓
17. Interview gui	de	✓	×	√	✓	✓	✓	x	✓	×	×	√	×	√

18. Repeat interviews	×	×	×	×	×	1	×	×	×	×	×	×	×
19. Audio/visual recording	1	✓	✓	√	✓	✓	×	✓	✓	√	√	×	✓
20. Field notes	1	4	1	√	1	×	×	×	×	×	√	×	×
21. Duration	√	1	~	√	✓	1	×	1	×	1	×	×	×
22. Data saturation	✓	×	~	√	×	×	×	✓	×	×	×	×	×
23. Transcripts returned	×	×	×	×	×	1	×	×	×	×	×	×	×
Analysis and findings													
24. Number of data coders	×	×	1	4	×	×	√	✓	1	1	~	×	✓
25. Description of the coding tree	~	×	×	×	×	×	×	×	×	×	×	×	×
26. Derivation of themes	1	1	~	√	✓	~	√	~	~	1	√	~	×
27. Software	×	×	✓	√	✓	1	×	×	1	×	√	×	√
28. Participant checking	×	1	×	√	×	~	×	×	×	×	×	×	×
29. Quotations presented	~	1	~	√	✓	~	√	~	~	1	√	~	✓
30. Data and findings consistent	✓	4	~	√	✓	4	√	1	1	4	√	✓	×
31. Clarity of major themes	✓	1	✓	√	✓	1	√	1	1	1	√	✓	~
32. Clarity of minor themes	~	√	~	×	~	×	1	~	×	×	1	~	√
Total	17/32	19/32	20/32	23/32	17/32	15/32	9/32	17/32	13/32	13/32	17/32	12/32	12/32
Outcome of the assessment	Medium	Medium	Medium	High	Medium	Medium	Low	Medium	Medium	Medium	Medium	Medium	Medium

3.2.4 Analytical themes

Key descriptive themes were grouped into two broad analytical themes, namely "barriers to shared decision making" and "facilitators of shared decision making". A summary of the findings of the included studies is presented in Table 3.9. The number of studies where each theme and sub-theme was identified as a barrier/facilitator influencing patient involvement is presented in Table 3.10.

Table 3.9: Summary of findings of the included studies on perspectives of shared decision making from patients

No	Authors (year)	Summary of findings/results
1	Connor & Wilson (2006)	 Barriers to shared decision making Disempowerment due to stigma of mental illness not only from society but also professionals Imbalanced power between patients and professionals Negative attitude of professionals Facilitators of shared decision making Reliable professional-patient relationship (good communication and sufficient time) Professionals having good supportive skills (empathy, respect, listening, common sense, humanity, sufficient time, reassurance, caring, understanding, patience, trust, compassion and practical organisational skills) Individual support for patients (personal life, medication, care, and occupational therapy) Peer support between patients Sufficient information for patients
2	Dahlqvist_Jönsson (2015)	 Sufficient monitation for patients Barriers to shared decision making Powerful role of professionals due to inequality in access to knowledge Insufficient information Untrusting professional-patient relationship Facilitators of shared decision making Patients feeling respected Flexible mental health care Alliance between patients and professionals (listening, accepting different opinions, understand needs and expectations of both sides, equal power, willingness to get involved in decision making) Active involvement in decision making Trust in patients' decision making abilities Other professionals providing advice without criticising Sufficient professional information from video or someone with similar experience Positive attitude of professionals Sufficient time Equal rights and equal responsibility
3	Delman (2015)	Barriers to shared decision making Health professionals with negative attitudes Lack of time for consultations

		Limited patient self-efficacy
		Facilitators of shared decision making
		Health professionals with positive attitudes
		• The psychiatrist's flexibility
		Other support resources
		• Self-confidence
4	Farrelly et al (2015)	Shared decision making is not widely implemented in mental healthcare
		Barrier to shared decision making
		Patients distrusting their health professionals
		 Powerful role of professionals due to professional knowledge
		Patients' low self-confidence in their abilities to make decisions
5	Gioia et al (2014)	Facilitators
		Support resources
		Individual information about illness
		Reliable relationship with professionals
6	Goscha (2015)	Barriers to shared decision making
		 Patients do not receive relevant information about their medication
		Facilitators of shared decision making
		Patients having a positive working relationship with professionals
		Patients having positive motivation for life
		 Patients having disclosed personal information and preferences
		Professional advisors (case managers)
7	Jones et al (2013)	Factors in shared decision making
		Patients experienced being controlled during any decision making
		Powerful roles of professionals
		 Peer support amongst patients helps improve self-determined motivation, self-confidence/self-efficacy, and behavioural education
		 Patients feel empowerment/self-confidence when making decisions by themselves
		• Patients have increasing understanding of triggers, warning signs, behaviour patterns, medication use and the role of
		professionals
		 Improvement of professional-patient relationship via WRAP (Wellness Recovery Action Plan, which help patients to make decisions)
8	Laugharne et al (2011)	Trust
		Facilitators

		Trusting therapeutic relationship
		• building trust by disclosing personal information
		o time
		o positive attitude and benaviour of professionals
		Barriers
		• undermining trust due to nature of illness
		 under mining if ust due to nature of miless coercive/negative experience of treatment
		Choice
		Facilitators
		• Enough knowledge of patients
		• Sufficient time with professionals
		Barriers
		Lack of information/knowledge about their treatment options
		Impaired decision making capability; serious illnesses
		Powerful roles of professionals
		Burden of decision making
		Negative experience of decisions they made previously
9	Matthias et al (2012)	The process of shared decision making required:
		Sharing opinions, concerns, and/or preferences
		Giving information about medication
		• Deliberation of a decision
		Mostly reaching an agreement
		What happened during decision making:
		Health professionals initiated most decisions
10		Decisions about follow-up appointments normally by agreement
10	Tanenbaum (2008)	Factors which influence decision making:
		• Doctor-patient relationship
		• Self-knowledge
		Personal meaning
		 Recovery-related values Need for information from multiple courses (professionale, corers, other recourses)
		 Received for information from multiple sources (professionals, carefs, other resources) Professionals' positive attitude toward sharing information
		A course of delivering professional knowledge
		Accuracy of derivering professional knowledge

11	Velligan et al (2016)	Barriers to shared decision making				
		Negative attitudes of health professionals toward patients				
		• Limited access to mental health system				
		Facilitators of shared decision making				
		Respect for therapeutic relationship				
		• Sufficient information				
		Easy access to services				
12 Wharne, et al (2012) Barriers to shared decision making						
		Being Excluded or withdrawing from decision making				
		Losing valued social positions				
		Being left unsupported				
		Not being consulted about treatment				
		Seeking isolation				
		Being in the wrong place at the wrong time				
		• Lives that are broken up by periods of detention in hospital, leading to problems with, e.g. relationships with others,				
		employment or somewhere to live.				
		Prone to wandering and random occurrences				
		Struggling to make experiences meaningful				
		 Not being able to share unusual beliefs and experiences 				
		Adapting to being different				
13	Woltmann & Whitley (2010)	Patients desire autonomous/shared decision making				
		Factors in shared decision making				
		Relationship and interaction with health professionals				

Themes	Barriers to shared decision making			Facilitators of shared decision making		
Sub-themes	Professionals with	Poor quality of	Lack of	Supportive attitude	Sufficient	Other support
	paternalistic attitude	interaction with	professional	to patient	information	resources
Studies		health professionals	knowledge/informa	involvement	exchange	
			tion			
Connor et al. (2006)	V	V	V	V	V	V
Dahlqvist et al. (2015)	V		V	V		V
Delman et al. (2015)	V	V	V	V	V	V
Farrelly et al. (2016)	V	V				
Gioia et al. (2014)			V		V	V
Goscha and Rapp (2015)		V	V	V	V	V
Jones et al. (2013)	V	V		V	V	V
Laugharne et al. (2012)	V		V	V	V	
Matthias et al. (2012)					V	
Tanenbaum (2008)	V			V	V	V
Velligan et al. (2016)	V	V	V	V	V	
Wharne et al. (2012)	V	V	V			
Woltmann et al. (2010)			V	V	V	V

Table 3.10: Overview of analytical themes

3.2.4.1 Theme 1: Barriers to shared decision making

This theme describes the factors patients perceived as barriers to involvement or inclusion in shared decision making in mental healthcare, consistently linked with "professionals with paternalistic attitude", "poor quality of interaction with health professionals", and "lack of professional knowledge and information".

3.2.4.1.1 Sub-theme: Health professionals with paternalistic attitude

When patients experienced paternalism from health professionals, they perceived that they would be unable to actively participate in decision making and tended to adopt a silent role. Although there was alternative evidence in some individual cases, the accounts indicated prevailing negative assumptions and paternalistic attitudes from health professionals about patient involvement in decision making. Patients often felt prohibited from speaking up in clinical encounters. For example:

"...It was just to do what they said, for they were always right, even if you felt that this can't be correct, that this is not right for me, they were right in all cases and that was not good at all." (Dahlqvist et al., 2015)

Similarly, even if patients desired or preferred a more active role in decision making, they perceived that their health professionals were unwilling to share decision making with them. This is illustrated in the example below:

"They (health professionals) are taught that they are the ones that provide the healthcare and so they see us agitators from their perspective." (Connor and Wilson, 2006)

The data showed that seeing health professionals as authority figures led to patients being reluctant to become involved in decision making. In this case, patients perceived that health professionals held a more powerful role in the decision-making process. This was highlighted by a patient in the following example:

"They're (health professionals) the authority, and I don't know anything – that's how they think." (Jones et al., 2013)

Such statements were perceived to demonstrate the health professionals' need to be the key decision makers. Patients felt health professionals considered themselves superior. Patients also stressed living in fear of further coercive treatment or care, and of decisions usually made by others. One patient described how health professionals acted when they

wanted to refuse treatment, and then how health professionals considered themselves authorities:

"Um, the very first time I was sectioned (sic), what really give (sic) you ECT against your will, and that's when I tried to run for the door."... "There was nothing nice or doctoring speciality or loving or caring – you know he thought he was he was a glorified police (sic) " (referring to a psychiatrist). (Wharne et al., 2012)

Furthermore, a few patients perceived existence of a power imbalance between health professionals and themselves. Patients felt powerless about the situation and that they were being lost in the decision-making process in all areas of their life. This is demonstrated in the following example:

"I think sometimes the doctors have too much power, especially psychiatrists, they make decisions about where you're going to live. I mean, Dr M basically made a decision about my life. She decided that I was under her care, she decided what medication I was going to have, and she decided where I was going to work. Now, isn't that power?" (Laugharne et al., 2012)

3.2.4.1.2 Sub-theme: Poor quality of interaction with health professionals

A number of studies highlighted that the lack of – or poor-quality – communication was a significant challenge for patients. Patients' descriptions of interactions with health professionals were generally negative. Barriers to communication and interaction between patients and health professionals could cause patients to become less able to be involved in decision making. These could involve the use of inappropriate language and responses in communication, or lack of time to communicate with health professionals during decision making.

Disrespectful language was often reported, and was perceived to reduce the level of patient involvement in decision making. Health professionals dominated decision making encounters. Patients felt that health professionals did not respect patients and provided examples of negative verbal or non-verbal behaviour, thereby creating barriers to shared decision making. There were more subtle forms which manifested themselves in professionals using challenging responses to patients, characterised by statements such as "*no, we can't do that*" or "*you should*".

Many patients commonly reported that health professionals unconsciously treated them like children, and it was also perceived by the patients that health professionals saw themselves as parents who should take the role of proxy to make decisions for patients (as for children). Patients explained this by the use of disrespectful responses to patient involvement from health professionals, for example:

"The need to be a good boy. Don't be a naughty boy. Get us to behave like children, manage us." (Connor and Wilson, 2006)

Similarly, from the patients' point of view, other people seem to think of people with a mental illness as being "child-like" and hence having impaired decisional capability, as illustrated below:

"There was nothing that stimulated me intellectually. And they automatically assume that if you have got a mental illness you are not intellectually capable." (Connor and Wilson, 2006)

Patients also felt that professionals used inappropriate language, and certain problematic behaviour, when interacting with patients. Not listening was widely reported by patients. One patient described a situation where he felt completely ignored by his health professionals:

"He wouldn't listen to anything. He didn't do anything. With Depakote, when I needed my blood test, he never got my blood test. He never rescheduled my appointments, he just didn't do ANYTHING for me." (Delman et al., 2015)

In some cases, health professionals were described as "rude", as illustrated below:

"I wanted a joint crisis plan 'cause I thought it might make a difference (...) with regard to how the psychiatrist would approach things if I got sick. 'Cause I've been sectioned so many times. But I remember, on the day that (the facilitator) came (the psychiatrist) was on the (computer), he was so rude (...) and he was on his (computer) most of the time when (the facilitator) was talking. He had his back turned." (Farrelly et al., 2016)

Health professionals' failure to respond to patients' questions were commonly reported by the patients. It was perceived that health professionals paid more attention to addressing their own concerns and just asking questions they thought were important to them. Of central concern to patients was being listened to, not just about their symptoms but their stories too. They felt that health professionals often paid attention only to the nature of their illness rather than their personal experience and preference, and this often resulted in failure to listen to patients as patients expected, for example:

"And they just dealt with symptoms of what was wrong with me: they never asked me about my father's death, they never asked me about the break up with my long-term girlfriend. There was never any reference made to that, not once, I was just pumped full of drugs." (Connor and Wilson, 2006)

Furthermore, some patients characterised health professionals as robot-like, fixated on standard procedure and unable to provide care at a human level. Health professionals failed to demonstrate an understanding of patients' suffering and were described as having "become desensitised" and treating patients "like cattle". For example:

"...didn't say hello, how are you, how are you feeling today, he just walked to the next person... and he did not speak to one patient, he didn't say, they were like cattle..." (Connor and Wilson, 2006)

In addition, patients particularly emphasised the lack of time they spent with health professionals. Some patients perceived that the limited time allocated for consultations with their health professionals was insufficient for patient involvement in decision making. This prevented patients from talking about any difficulties they might be experiencing, asking questions, or expressing their concerns or preferences. Accordingly, this influenced their willingness to engage in treatment or care, for example:

"Sitting for hours, waiting, and only seen for five minutes. As soon as I walk in, he's already writing (a prescription), not talking to me." (Velligan et al., 2016)

3.4.2.1.3 Sub-theme: Lack of professional knowledge and information

Patients repeatedly emphasised that knowledge gaps were a barrier to being able to participate in the decision-making process with their health professionals. For example, one patient commented that he/she saw his/her role as being very limited in making decisions about the treatment because he/she did not have the professional knowledge/information that the health professionals did.

"To be involved in decisions... well... of course, it's probably hard to participate in care when you have no education, and the others are sitting there with professional training and knowledge." (Dahlqvist et al., 2015) Beyond the nature of the knowledge gap between health professionals and patients, it was perceived that the patients did not receive sufficient information about their treatment and care. Patients commonly perceived that they were not given sufficient information and knowledge about their treatment. Patients said that they did not gain relevant information or knowledge about their treatment and care in medical encounters. From their perspective, the level of information about treatment plans provided by the healthcare professionals was insufficient. Therefore, they were not able to contribute more during the decision-making process. One patient explained how they accepted treatment but were not told the reasons for it:

"It feels like the ball is always in their court; they're making decisions and they're not explaining anything." (Velligan et al., 2016)

Furthermore, patients with mental illness judged their decisional capacity negatively because of their lack of professional knowledge. The sense of being negatively judged by others served to undermine their confidence in being involved in decision making. Patients frequently said that they were not qualified to express their views and then lost confidence. One patient explained:

"I don't make many decisions because I'm frightened of making the wrong one." (Laugharne et al., 2012)

Although there was a lack of receiving information and knowledge, the patients emphasised the desire for more information/knowledge. Patients stated that they particularly wanted information about medication, including the benefits and unwanted effects. This is illustrated in the quote below:

"...What I can sense when talking medicine / is / that you may only know what the idea is and how it should work, but then nothing is said about side effects." (Dahlqvist et al., 2015)

Less commonly, but importantly, the sense of having a right to be involved in decision making and ask for more information was not widespread. Without this, patients became accustomed to the traditional silent role in decision making. For example, one patient described that he/she was not aware of the application of shared decision making in healthcare. Therefore, he/she did not realise his/her right to take or refuse the medication given by the health professionals, so he/she was more likely to accept it submissively, as the following quote illustrates:

"I just didn't know I could choose which medications I could be on or that I could refuse at all." (Delman et al., 2015)

3.2.4.2 Theme 2: Facilitators of shared decision making

Although mostly it was barriers that patients reported, all of the studies found factors that facilitate shared decision making. Three major facilitators were found: supportive attitude to patient involvement, sufficient information exchange, and other support resources.

3.2.4.2.1 Sub-theme: Supportive attitude to patient involvement

Nine studies found that a supportive attitude to patient involvement was important in shared decision making. Patients believed that health professionals with supportive attitudes enabled them to ask questions, express their concerns or preferences, and share personal information. For example, one patient commented:

"In recent times, we have more or less the same ideas, but before that some of the ideas weren't the same. Because we weren't familiar with each other: they being a professional and me being a consumer, and learning to be more receptive to me being a consumer...we see more eye to eye on a lot of issues." (Woltmann and Whitley, 2010)

Patients emphasised positive/supportive attitudes from health professionals, such as kindness, empathy and friendliness. These helped health professionals and patients to build trusting relationships. Patients perceived that health professionals who displayed human qualities and treated them as independent people had been most helpful.

"They believed in my ability, and they did so I grew... The more you see and show that you can do, the more chances you get. So it has meant that I have grown and, yes, I feel a thousand times better now." (Dahlqvist et al., 2015)

Less commonly mentioned, but important, was how health professionals reacted when a disagreement emerged between them and the patients. Sometimes health professionals disagreed with what patients really wanted, such as reducing psychiatric medication, but still they both worked together to make a final decision. Even where health professionals were concerned about patients' decisions, they still tried to provide support to help patients work through their decisions. One patient shared a situation where he/she wanted to change medication and how the health professionals supported him/her in changing it:

"I wanted to go off meds. She said she didn't want me to do it but it was my decision and she'd help me do it in a way that would be most effective, and we can catch anything if I have an issue. She told me to call if I was having any delusions - to call the crisis line here; she wanted to make sure that nothing bad happens, or something, we can stop." (Delman et al., 2015)

3.2.4.2.2 Sub-theme: Sufficient information exchange

In order to reach a mutually agreed-upon decision, openly sharing opinions and/or concerns was recognised as an important facilitator by patients. From the patient views, both professional knowledge and information, and personal preference and concerns, were important in making a shared decision about treatment and care. Therefore, having their medical knowledge and information needs met in an appropriate way was an essential facilitator for many patients. However, if patients did not acknowledge their condition, and they did not understand their available options, they were not able to participate in decision making.

Unsurprisingly, several patients reported there was a need for, in particular, pharmacological information. The patients in the included studies most frequently paid attention to medical information and spoke of medication discussions, such as the side effects of medication, changes of medication, or reduced dosage. The following example illustrates this shared process:

"Consultants have looked at me and they have considered that I've got enough knowledge and experience to be able to have an input into what the medication is." (Laugharne et al., 2012)

Many patients believed information about treatment alternatives should be provided by health professionals, for example:

"Explain what it is and then give you an option between different medications" and "This one is for this, this one works really well with this, this one works really well with this." (Velligan et al., 2016)

Conversely, patients with their unique expertise believed they could offer more information about their treatment preferences and facilitate a process of sharing the information. The importance of patient knowledge about themselves and their own conditions was repeatedly pointed out by the patients. In particular, one patient stressed the importance of health professionals taking what patients say into account to ensure a more accurate clinical diagnosis. Some even mentioned that personal expertise and medical expertise were equally important in the decision-making process. Therefore, patients felt that their voice should be heard by health professionals, for example:

"The people who are trying to decide how to give the right kind of help need to hear from the people who need the help." (Gioia et al., 2014)

Similarly, some patients highlighted the importance of the professionals listening to patients rather than just adhering to theory or textbook facts about mental illness. They perceived that their first-hand accounts and research evidence were equally important in decision making. A patient who was diagnosed with depression asserted that health professionals should integrate what they heard from patients with what they learned from textbooks:

"Theory is good, it is good to understand theory but there has got to be a practical side too. You have got to get someone to say well I haven't read such and such a book on depression but I have suffered from depression and this is what I have gone though (sic) on a day to day basis." (Connor and Wilson, 2006)

From the patient views, one way that health professionals gained mutual understanding in decision making was to know the patient's personal details and social circumstances. Patients' personal preferences about treatment options, and their goals and beliefs could have an impact on what kind of decision they would like to make. One patient commented that it is important that health professionals understood them as a person and not just the mental illness:

"I've got a very good CPN... she knows a lot about my personal circumstances, she knows I've got two daughters and knows I want to be part of their lives... She's got a nice personal touch to her, she tells me about her family and things." (Laugharne et al., 2012)

3.2.4.2.3 Sub-theme: Other support resources

A prominent theme in people's experience of shared decision making in eight of the studies related to other supportive resources. The importance of multiple resources which could help them through decision making was identified by the patients in the included studies. They felt that it was very important that support resources were provided at any time they needed them. One patient mentioned:

"The resources have to be there. You could have the best heart in the world but the resources have to be there." (Connor and Wilson, 2006)

Consistent with the previous themes detailed above, some patients felt that the flow of communication could be difficult because of a knowledge gap or personal communication skills. The help offered by these supporting people was varied, and included professional skills and communication strategies, which could give patients more confidence to express themselves. It was also thought helpful if what was likely to be discussed with health professionals in the medical encounter was also discussed in advance by patients. Access to a person or tool that has professional knowledge or sufficient information about mental health decisions was needed before, during and after decision meetings with their health professionals. These support resources varied from people to tools, such as family involvement, other professionals, patients with similar mental illness, and decision aid tools.

The patients expressed the importance of feeling that a supportive person was on their side. A supportive person could help them to express their preferences and ask questions without feeling criticised. The encouragement of a support person in decision making might reduce negative experiences and enhance the level of involvement in decision making.

In terms of support from other professionals, the patients suggested that supportive professionals could provide their knowledge to offer guidance in making decisions, for example:

"You might have a support person or a counsellor or PSR (psychosocial rehabilitation) facilitators or even a psychiatrist. That person has the education level to help you determine what you need to do, also friends and family." (Jones et al., 2013)

Meanwhile, it was perceived that support was also to communication strategy. Some patients found they were more confident if they had support with communication skills before actually interacting with their health professionals. For example, one patient described a situation where one member of staff in a group home encouraged him/her to actively communicate with his/her health professional:

"The (group home staff member) encouraged me to write down what I wanted to say, or the questions I had. That way I didn't have to verbalize it. Having that paper in front of me, I was able to bring up medication problems right when we first sat down." (Delman et al., 2015)

Similarly, some patients pointed out that other patients who have similar conditions or experience could provide an insight into their mental health conditions. Advice from these people could help the patients focus on their points of concern and promote discussion with their health professionals. Two examples follow:

"I think it's good to hear what others are doing and what's working for them..." (Tanenbaum, 2008)

"It's better if someone who has had experience of it can describe it. Someone you feel understands and then you may feel oh well this might be something for me... or maybe you feel, no, this was not what I thought." (Dahlqvist et al., 2015)

Less commonly mentioned in studies but important to note were the views of family involvement in decision making. Some believed that active family involvement could help patients to gain more understanding of their conditions and provide someone to keep an eye on the warning signs of relapse. For example, one patient described how they gained more understanding of their medication plan and mental health conditions with their spouse's support:

"I had more of an understanding (when my spouse was involved) of my mental health, and my medication, and the effectiveness of both," declared one married veteran. (Gioia et al., 2014)

In addition, the importance of decision aids and interventions has been stressed in these data. A few patients felt that they were able to be involved in decision making because they used some decision aids or participated in shared decision making interventions, which could improve the implementation of shared decision making in mental healthcare. The intervention promoted participation by helping patients to gain sufficient knowledge/information, and asking questions during decision-making meetings. Therefore, some patients believed that decision support could potentially help them to be actively involved in decision making; for example:

"WRAP gave me the idea of taking my list of wellness tools to the psychiatrist's office and using it to discuss [things]... [It] made me bring up and talk about a lot of things that I wouldn't have otherwise." (Jones et al., 2013)

3.3 Strengths and limitations

To the author's knowledge, this review is the first qualitative systematic review to provide a thorough meta-synthesis of the experience of shared decision making and possible barriers and facilitators for patients of secondary mental healthcare. The reported findings should be carefully considered in the context of the strengths and limitations of the synthesis which was carried out.

First, the review incorporated the experiences of patients with mental illness from diverse backgrounds and circumstances. Also, the reported findings given are comprehensive, sourced from qualitative investigations undertaken in various regions of the world. However, all the studies were conducted in western countries, such as the US, the UK, Canada, and Sweden. Meanwhile, none of these studies was carried out in Asian countries, including Taiwan, the specific context of the author. Therefore, the implications cannot be translated to other non-western high- or low-income countries.

There is a lack of studies on decision making in Asian counties, particularly in Taiwan, suggesting that shared decision making has only recently been conceptualised and has not been widely introduced in practice, including in mental healthcare (Hou, 2015; Hsu et al., 2015; Yen and Hsu, 2007). To date, there are no relevant studies on decision making for patients with mental illness in Taiwan. This is a significant gap between western and Asian countries.

Second, only few published studies of perspectives on shared decision making concentrate the majority of their research on both patients and health professionals, despite evidence that in this context mental health patients' views can be lost. Therefore, it is possible for some studies to not reach saturation point to adequately address all potential issues that may have arisen from the patients' perspectives.

Third, some limitations of qualitative synthesis should be considered. Researchers have argued that qualitative data is not suitable for synthesis as every researcher could produce very different results from the same data, which leads to concerns about rigour in systematic reviews (Jensen and Allen, 1996). Despite this, in the interpretive paradigm, it is argued that reality is constructed by social phenomena created by humans, and exists in context-specific situations which are multiple rather than singular (Green and Thorogood, 2009; Guba and Lincoln, 1994). Synthesis of qualitative studies could help people to understand the essences of specific phenomena and advance knowledge (Sandelowski, 1993). Also, qualitative meta-synthesis relies on interpretation of narrative

inferences rather than statistical inferences (Jensen and Allen, 1996). Therefore, the validity of qualitative synthesis should be carefully considered when conducting reviews. To reach valid interpretations, this review strictly followed the recommended set of procedures to achieve credibility.

Finally, although all the studies were carefully considered by the author when deciding to include or exclude them, the quality of each study could vary. Overall, the quality of the studies was categorised as high, medium or low, and thus it is difficult to be confident of the overall findings as representative of patients' views. Furthermore, due to resource limitations, this review excluded non-English publications, which raises the possibility of missing relevant studies.

Hence, further studies could collect data directly from not only health professionals but also patients in other contexts. Moreover, it could address the methodological constraints of the studies, such as data analysis, in a rigorous way.

3.4 Summary

To summarise, this systematic review has produced important findings regarding patients' views about their experience of shared decision making, and possible barriers to and facilitators of shared decision making in mental healthcare. The findings of this review strongly supported the idea that patients in mental healthcare still played passive roles in the decision-making process. The findings of this review introduced two important themes: barriers to and facilitators of shared decision making.

There were three significant barriers to shared decision making in mental healthcare. The health professionals commonly adopted a paternalistic attitude to stop patients being actively involved in the decision-making process. Meanwhile, the poor quality of interaction with health professionals was also recognised as a major barrier to shared decision making, from the patients' accounts. Additionally, patients did not have the confidence to be actively involved due to their limited professional knowledge.

By contrast, some facilitators were also identified in the interviews. Health professionals with a supportive attitude could be the key to facilitating the implementation of shared decision making. Also, patients pointed out that sufficient information exchange could also be of benefit in implementing shared decision making. Moreover, some patients also realised the importance of other support resources, such as other health professionals and decision aids, which helped them to actively engage in decision making.

This chapter has provided an insight into the patient perspective of shared decision making in secondary mental healthcare, as described in the existing literature. However, there is a lack of direct evidence about the perspectives of either patients or mental health professionals about shared decision making in Asian countries, including Taiwan. Increased emphasis needs to be placed on how their views can change mental healthcare practice or services. In addition, the limitations of some methodologies, as described in Section 3.3, could be improved in some ways. Therefore, this chapter has established that, whilst the review highlighted some evidence that patients are being consulted in some contexts, there was a definite need to undertake studies of patients' and health professionals' views directly in this area in other cultural contexts.

Chapter Four: Working methods for patient and health professional interviews

The research design of this study was presented in Chapter Two, which described the underpinning methodology. This chapter will present the working methods for the two qualitative studies exploring patients' and health professionals' perspectives on how decision making was shared in secondary mental healthcare.

4.1 Participant selection

4.1.1 Patient interviews

The aim of using purposive sampling (described in Chapter Two) was to ensure the selection of information-rich participants for the most effective use of limited resources. The sampling strategy sought to recruit patients with a diagnosis of mental illness who were using halfway houses (for inclusion and exclusion details see Table 4.1).

Table 4.1: Patients – inclusion and exclusion criteria

Inclusion criteria	Exclusion criteria
 Adults aged 20 or over who have mental illness Using halfway houses Able to read and understand traditional Chinese or English, and speak Taiwanese, Mandarin, or English 	 Less than 20 years old Unwilling to participate; not able to communicate in Taiwanese, Mandarin, or English Refusing audio-recording

The reason patient participants were recruited from half-way houses was to involve patients with experience across numerous mental health secondary services, such as inpatient settings, outpatient departments, and other secondary mental health settings. Typically, patients in halfway houses have been referred from inpatient units and continue to attend out-patient services, so they are likely to have experience across the range of services provided in mental healthcare in Taiwan. In addition, according to the National Health Insurance system, patients who are referred for treatment in community rehabilitation centres and halfway houses typically have stable psychotic symptoms and long-term support needs, but they are considered to have the potential for rehabilitation to return to the community to live independently (Hsieh and Shiau, 2006). In other words, these patients now have a stable psychiatric condition but have broad experiences across

secondary mental healthcare. Therefore, patients in halfway house were selected for conducting this study.

Adults aged 20 or over were included because in Taiwan, research participants are legally required to be adults, i.e. over 20 years old (The Legislative Yuan of the Republic of China, 1929). This criterion also applied to the health professional group.

4.1.2 Health professional interviews

The sampling of health professionals aimed to capture a wide range of mental health settings with a range of clinical roles in secondary mental health service (see Table 4.2 for details). The rationale for including professionals with a minimum of one year's experience is that professionals need to have experience of the phenomenon of interest and trainees may not have direct experience.

Inclusion criteria	Exclusion criteria
 Registered mental health professionals including psychiatrists, mental health nurses, nurse practitioners, clinical psychologists, occupational therapists, case managers, and social workers Aged 20 or over Working in secondary mental healthcare, including community mental health centres, outpatient 	 Under 20 years old Less than one year's working experience Unwilling to participate Not able to communicate in Taiwanese, Mandarin, or English Refusing audio-recording
 departments, outpatient clinics, day centres, day hospitals, rehabilitation services, group homes, halfway houses, or other support services With at least one year's experience of working in mental healthcare Able to read and understand traditional Chinese or English, and speak Taiwanese, Mandarin or English 	

Table	4.2:	Health	professionals -	– inclusion an	d exclusion	criteria
IUNIC		IIcuitii	protessionals	merubion an	a cherabion	ci icci iu

4.2 Recruitment

Following confirmation of ethical approval from the University of Manchester (Ref: 2017-2009-3358) and Chang-Gung Medical Foundation Institutional Review Board

(Ref: 201700879B0) in Taiwan, the author visited the five settings (one outpatient department, one day care centre and three halfway houses) to explore recruitment for both studies. The department managers and mental health professionals were approached in their workplaces, either individually or at team meetings, in order to explain the purpose of the studies and data collection procedures, and to identify participants meeting the study criteria. There were individual meetings with managers in the various settings. The author also made ten presentations to health professionals in their workplaces to help them to gain more understanding of the purpose of the studies and the data collection procedures. After this, the managers and health professionals contacted three halfway houses to facilitate this study. Three halfway houses agreed to facilitate study and three health professionals in the halfway houses were willing to be the author's main contact person to facilitate data collection.

4.2.1 Patients

For the patient group, patients were identified through the managers and healthcare professionals in three halfway houses in Taiwan.

Two methods were used to recruit potential patient participants in the halfway houses:

- information packs (10/07/2017)
- posters (10/07/2017-01/09/2017)

The study information pack was distributed to each resident. The healthcare professionals were asked if there were any concerns about risks that may present to the patient or the author in conducting an individual interview before distributing packs, and no concerns were raised, thus no resident was excluded for reasons of safety. To promote recruitment, posters were displayed in the public areas of halfway houses.

Patients were given at least 48 hours to read the information sheet, and think about taking part in the research. If any patient was interested in taking part or wished to ask any further questions about the research, they could contact the author via phone or email. A contact number and email address were given in the invitation letter and the information sheet so that, if the potential patient participants required more information, they could contact the author directly. Also, in order to provide flexibility, potential patient participants could leave a message using the contact form to indicate they were interested (either giving it to their health professionals or leaving it in message boxes in the

reception of the halfway houses). The author regularly checked with the health professionals and the message box at least weekly.

4.2.2 Health professionals

The author used two stages to maximise recruitment and ensure variation in the types of healthcare professionals sampled.

- stage one:
 - posters in staff offices in the five selected settings (one outpatient department, one day care centre and three halfway houses) (10/07/2017-01/09/2017)
 - information pack allocated for the five selected settings (10/07/2017)
 - adverts on social media (10/07/2017-01/09/2017)
- stage two:
 - adverts on social media (01/11/2017-15/11/2017)

First, the author identified potential professional participants through the managers of one outpatient department, one day-care centre, and three halfway houses. During the initial stage of recruitment, research information had been presented by the author to managers of the settings through either one-to-one or group meetings. The author asked the managers of these five settings to send the information sheet, the non-named invitation letter, and the contact form to all registered mental health professionals there, to invite them to take part in the study. In addition, posters advertising the study were placed in staff offices of the five selected settings (one outpatient department, one daycare centre, and three halfway houses) to identify more mental health professionals. To maximise health professional recruitment the study was advertised via social media including Facebook, Twitter, and PTT Bulletin Board System (which is the largest terminal-based bulletin board system (BBS) based in Taiwan) from 10th July 2017 to 1st September 2017. The author's contact details (phone number and email address for work) were provided in the advert. Potential professional participants could contact the author via phone and email to gain further information. Information sheets were then provided via post or email to explain the details of the study. In this stage, the author explained further details of the study to potential professional participants via phone and email to clarify any questions they had. Then the health professionals were given at least 48 hours to consider whether they were willing to participate in the study. Professionals were then

asked to sign the consent form. In order to take part, they could contact the author via phone or email to arrange an interview.

Although this first stage was fruitful for recruiting health professionals (such as psychiatrists), no mental health nurses were recruited. To ensure the diversity of the health professional sample, a second stage of recruitment was implemented. The authors contacted other health professionals to spread the information about the study, and then re-posted the advert in some nurses' groups (professional nurses' groups on social media). One nurse colleague supported the study by spreading this information to relevant groups on social media.

4.3 Data collection – Qualitative interviews

The author conducted individual face-to-face, semi-structured interviews.

4.3.1 Topic guide

Topic guides for each of the patients and health professional groups (Appendix 2 and Appendix 3) were devised from the results of the qualitative systematic review and modified after discussion with supervisors. In order to become familiar with the topic guides and to ensure flow of the interview the author practiced with a supervisor and colleagues. Further minor modifications were made following this process.

4.3.1.1 Patient interviews

Patients were asked to discuss their experience of decision making in mental healthcare. Here, clarification was provided around the term "decision making": "making decisions about treatment and/or care in mental healthcare" because it was felt that patients may not be familiar with the concept of shared decision making. Patients were asked about their preferred level of involvement in treatment and care decisions and to identify facilitators or barriers influencing patient involvement in decision making.

The patient interview questions focused on the following issues:

- their experience of decision making in secondary mental healthcare
- their preference about decision making in secondary mental healthcare
- their expectation of decision making in secondary mental healthcare
- barriers to/facilitators of their involvement in decision making

4.3.1.2 Health professional interviews

Health professional interview questions focused on the following issues:

- their experience of decision making in secondary mental healthcare
- their attitude toward patient involvement in decision making
- their understanding of shared decision making
- barriers to/facilitators of patient involvement in decision making

4.3.2 Demographic data collection

To ensure the sample could be described, demographic data were gathered on both patients and health professionals using a brief questionnaire (Appendices 4 and 5).

4.4 Data analysis

All interviews were digitally recorded and transcribed verbatim by the author.

4.4.1 Qualitative data analysis software

Computer assisted qualitative data analysis software was employed to assist in the analysis of the data. Qualitative data analysis software was used because it supports several analytical styles (Fielding and Lee, 1991), helps portray data transparently and provides an audit of the data analysis process (Carcary, 2009; Welsh, 2002). Therefore, computer-based methods have been recognised as valuable means of improving the rigour of qualitative research (Alhojailan, 2012; Welsh, 2002).

NVivo software was used to manage and analyse the data in this study. NVivo (QSR International Pty Ltd. Version 11, 2015) is a qualitative data analysis software package consistent with thematic analysis. It can help analyse qualitative data at various stages: gathering all the evidence, subsequently organising it into codes and grouping it into similar themes (Alhojailan, 2012). Welsh (2002) also points out some features which make this software easy to use. For example, using this program, documents can be imported directly from a word processing package and coded. This helped the author to easily see the details of the data. Another strength is the model explorer tool in Nvivo, which is useful to map out diagrammatically how these themes relate to each other, in order to make sense of them (Welsh, 2002). Also, NVivo has been recognised as one of the most popular qualitative data management programs in the field of health (Zamawe, 2015). Therefore, in this study the information was entered in Microsoft Word and then transferred to NVivo.

4.4.2 Thematic analysis

During the data analysis process, thematic analysis as proposed by Braun and Clarke (2006) was applied. There are several steps that guided the author when interpreting the data:

- familiarisation with the data
- generating initial codes
- searching for themes
- reviewing themes
- defining and naming themes
- producing the report

4.4.2.1 Familiarisation with the data

All interviews were transcribed by the author herself. Following the completion of the transcriptions, the first three interviews in each group (patients and health professionals) were transcribed into English and read by the author and supervisory team to ensure they gathered in-depth information around each topic, and also to ensure completeness of the interviews. The first step was to gain familiarity with the data by listening to the recordings of the interviews and rapidly reading the transcripts of the interviews and field notes (Braun and Clarke, 2006). As part of the qualitative data analysis, all recordings were carefully listened to and transcribed in Chinese by the author using Microsoft Office Word 2010 software. Recordings were listened to repeatedly.

4.4.2.2 Generating initial codes

The next step after familiarisation is generating initial codes. This is an analytic process which involves a conceptual and semantic reading to generate the codes (Clarke and Braun, 2013) in NVivo. This second stage involves making concise notes about important features of the data related to the research. After reading all the transcripts of the interviews, the author coded the transcripts. Using NVivo software, extracts across transcripts that were related to each other were grouped together into codes (known as 'nodes' in the software). All the interviews were coded using this approach, and a final reading was undertaken to identify any possible missing codes. At this stage, long lists of codes were generated from both patient interviews and health professional interviews. Table 4.3 below provides an example of how initial codes were added to the data extraction.
Studies	Data extraction	Initial code
Patient	I told them: I don't want to take this	Patient cannot
interviews	medication But the staff told me I must	decide treatment or
	take it or they will take me to be	care
	hospitalised. So so I took it (09	• Verbal threat about
	Patient)	re-hospitalisation
		from professionals
Professional	I don't think it is a great idea to give them	• Felt a lot of
interviews	many treatment options. They feel panic	information would
	when they get this disease. If they are given	be overwhelming
	a lot of information at one time they	• Patients prefer a
	can't make a decision. You know the	passive role
	patients still ask the professionals to	Professionals have
	decide for them. They don't want to decide.	negative attitude to
	So isn't it pointless to provide options at	shared decision
	the beginning? (01 Psychiatrist)	making

 Table 4.3: Data extraction with initial codes applied

4.4.2.3 Developing themes

Following coding, a long list of potential themes and sub-themes was identified. (Appendix 6). Clarke and Braun (2013) describe this stage as being more like coding your initial codes to search for a pattern within your initial codes, or nodes. A theme is an interpretive concept which explains the characteristics of the data and identifies meaningful patterns within the data. Importantly, researchers should actively construct their themes rather than discover themes hidden in the data (Clarke and Braun, 2013). Further, the aim of this step is to understand the relationships between codes rather than identify an overarching theme structure.

Figure 4.1 presents an example of an initial thematic map for patient interviews. To ensure the themes were appropriate for analysis, all these themes were descriptive and were easy to allocate or assign at this stage.

Figure 4.1: Initial thematic map for patients



4.4.2.4 Reviewing themes

Following this process, the author checked that the themes represented the full data-set (Clarke and Braun, 2013). It is important to ensure the themes tell a compelling and conclusive story from the data (Clarke and Braun, 2013; Spencer et al., 2014). The author started to consider the essence of and relationships between the themes and discuss the emerging themes with the supervisory team. Although this is a time-consuming process it provides a greater level of interpretation across the data. This often involves collapsing or combining themes to explain the data comprehensively (Braun and Clarke, 2006).

By continued discussion with the supervisory team, the main themes and sub-themes were structured for both patient interviews and professional interviews.

Using NVivo, the author constantly reviewed the content of the nodes at this stage. Node classifications containing recognised characteristics from all sources were created. Memos were also used for documenting the author's questions, ideas, and thoughts during data analysis.

For the patient interviews, all potential themes were reviewed and two main themes were identified, namely "barriers to patient involvement in decision making" and benefits of shared decision making".

For the professional interviews, the author also reviewed all the themes based on the codes generated earlier, during step two and three, and identified two main themes,

namely "barriers to shared decision making" and "facilitators of shared decision making".

4.4.2.5 Defining and naming themes

This step involves conducting and writing up detailed analyses of individual themes, and researchers should ask themselves what story was told in any given theme, and how each theme represents a particular part of the whole story (Clarke and Braun, 2013). To ensure the themes were named appropriately, the author identified the nature of each theme and constructed an informative and convincing name for each. For example, at this stage, the theme "benefits of shared decision making" was renamed as "facilitators of shared decision making". All details of each developed theme will be presented in Chapters Five and Six. The interpretation of the result was agreed with the supervisory team.

4.4.2.6 Producing the report

In this stage it is essential to write the analytic narrative and data extracts all together, to explain the whole story in a coherent and persuasive way (Braun and Clarke, 2006). As the data analysis process reached an advanced level, a broader picture or visual interpretation of the data became necessary for producing reports and models. This also involved contextualising all the findings within the existing literature. These strategies were used at this stage to present the results from the data. A detailed discussion of the findings will be presented in subsequent chapters. At each stage of the data generation and analysis process, regular meetings were held with the supervisory team to review sampling strategies, transcripts, coding, memos and field-notes, and to identify categories and explain the relationships between codes.

4.5 Rigour

To ensure that the quality of this study was maintained throughout the research process, the principles of credibility, transferability, dependability, and confirmability were considered (Lincoln and Guba, 1985).

4.5.1 Credibility

Peer debriefing is a strategy where the author seeks scholarly guidance and support from professionals, such as supervisors, to improve credibility. The supervisory team, whose perspective of this study was more neutral, examined and provided feedback to support the development of this study and enhanced the quality and credibility of this thesis. Throughout the research process, regular meetings were held with the supervisory team

to discuss general methodology, working methods (such as interview topic guides and interview transcripts), and the final report.

4.5.2 Transferability

Transferability refers to the extent to which the results of qualitative studies can be transferred or generalised to different contexts or settings. A rich description of the research process and details of the participants should be provided to ensure transferability. In this thesis, detailed discussion of how the study was conducted are presented in Chapter Two (Methodology) and this chapter (Working methods). Data collection was conducted to substantiate the interpretation produced and the diversity of the population studied. Also, thick description of the participants will be presented in Chapters Five (Findings from patient interviews) and Six (Findings from professional interviews). Therefore, readers will be able to use the information to establish whether the findings of this study are applicable to their own settings or services.

4.5.3 Dependability

Dependability, regarded as reliability in qualitative studies, draws attention to the need for researchers to explain the changing context where the research was conducted (Korstjens and Moser, 2018; Lincoln and Guba, 1985). The criterion of replicability or repeatability, which forms part of a traditional quantitative view of reliability, could be challenging to meet in qualitative studies.

In this study, to ensure dependability, an audit trail was used and details of features of the study design, such as data collection and sampling, are presented within the thesis. Methodological and theoretical decisions were discussed in Chapter Two (Methodology). This chapter also provided all the details of the processes of data collection and data analysis. The research design and its application are presented in the thesis to ensure dependability.

4.5.4 Confirmability

Confirmability aims to ensure the results could be confirmed or corroborated by others to minimise potential misinterpretation. As in qualitative studies it is assumed that researchers bring their own perceptions to the studies, it was vital to ensure the objectivity of the whole study process. The strategies of an audit trail and a reflexive journal were used in this study. The audit trail was discussed above. In this section, the author therefore focuses on reflexivity.

4.5.4.1 Reflexivity

In order to enhance confirmability, a reflective journal was employed throughout data collection and analysis to help the author to assess incorrect assumptions, personal influence, and possible misinterpretations of the data. Reflexivity is a process of introspection for researchers into their subjectivity in the research process. In interpretivism, reflexivity, which is the appraisal of the influence of the author on the participants studied, is needed (Horsfall, 1995).

This is a dynamic process while the study is being conducted of recognising, examining, and understanding how researchers' social background and assumptions influence their research practice (Hesse-Biber, 2007). The principles of reflexivity are to ensure methodological cohesion, work inductively, be responsive as a researcher, recruit an appropriate and accurate sample, and attend to ethics (Morse et al., 2002). It will be discussed in detail in the following sections, including the positionality of the author, the use of field notes, and ethical approval.

4.5.4.1.1 The author

All aspects of the study, including participant recruitment (patients and health professionals), data collection (qualitative semi-structured interviews), and data analysis (thematic analysis and using the supervisory team for data analysis) were conducted by the author.

The author has long-term practice and interest in shared decision making, quality of life, evidence-based practice, long term mental ill health, and mental healthcare research, and has been involved in several research projects relating to patients in mental healthcare, such as a patient quality of life project and a stigma in mental healthcare project. The author has previously worked as a registered nurse in mental healthcare and as a lecturer in nursing colleges in Taiwan.

The author has spent a significant amount of time reading, understanding, and writing about decision making, so previous knowledge could have had a significant impact on the interpretation of the data. This background has its advantages but has some disadvantages the author was also aware of: for example, loyalty to mental health professional groups could have compromised objectivity. Due to the nature of qualitative studies, it is impossible to separate the author from the qualitative data collection (Patton, 2002). To minimise these potential problems, self-reflexivity and regular meetings with supervisors were used.

During the interviews, participants were aware that the author was a nurse in mental healthcare in Taiwan and a PhD student at the University of Manchester. However, these roles could have influenced patients' involvement or responses to the interviews in some ways. Some participants may have felt that it was difficult or unsafe to disclose their true feelings and behaviour about decision making in mental healthcare. This might have been exacerbated by a social desirability bias, where participants will often reply to questions on sensitive topics inaccurately to manage the impression they are giving, or for ego-defensive reasons (Fisher, 1993). In this case, participants are more likely to provide an answer which is socially acceptable rather than disclose his or her true perspective or behaviour. In order to mitigate this, the author therefore explained to the participants that the aim of study was to explore their perspectives, but not to judge them, and that there were no right or wrong answers to the interview questions.

At the same time, due to the author's circumstances and professional position, additional concerns might be raised about confidentiality. A full explanation of confidentiality and how this was maintained was provided to the participants prior to the interviews. Further details of how confidentiality was managed will be discussed in the autonomy section.

4.5.4.1.2 Field notes

The use of field notes is regarded as one of the core tools for qualitative studies (Tong et al., 2007). Field notes serve many functions, such as documenting valuable contextual data, constructing thick descriptions of the data, and encouraging the author to be reflective (Phillippi and Lauderdale, 2018). Most qualitative research methods suggest taking field notes to enhance rigour and trustworthiness (Creswell, 2007; Patton, 2002).

In this study, both descriptive information, including interview dates, settings, behaviours, and conversations which the author observed, and reflective information, such as personal thoughts, questions, ideas, and concerns during the interviews, were systematically taken down for each interview. Field notes were taken for all interviews to:

- encourage the author's reflection and identification
- document any social behaviour or environmental stimulus in which the author was interested
- provide valuable context to inform data analysis and facilitate the initial coding

Descriptive information in field notes (such as interview dates and settings) was used to provide thick description of the research process and details of the participants. Also, reflective information including the author's thoughts, ideas, questions, and concerns were reviewed by the author during data collection in order to maintain consistency in conducting this study. Meanwhile, field notes do not just foster self-reflection, but also are crucial for understanding and interpreting in a research study. Through the field notes, some ideas, initial codes and potential themes were identified, which fostered a more developed interpretation. During data analysis, a set of field notes helped the author recall the interviews and then distinguish relevant information and accurately code the data.

4.6 Ethical considerations

Through undertaking training in ethics and data collection prior to applying for ethical approval, the author felt comfortable with both the participants' rights during the study and the requirements of the health services. These courses included:

- Research Ethics Application: University & NHS
- Lone Worker Training for Field Researchers
- Data Protection
- Good Clinical Practice (number: 2017043)

Ethical considerations were taken into account and the application was submitted to:

- the University of Manchester Research Ethics Committee (ref: 2017-2009-3358) (see Appendix 7)
- the Chang-Gung Medical Foundation Institutional Review Board (ref: 201700879B0) (see Appendix 8) in Taiwan.

All the study procedures were approved by these bodies.

The procedures proposed by Beauchamp and Childress (2013) were applied to ensure that the four ethical principles of conducting health research were followed: autonomy, beneficence, non-maleficence, and justice. These are discussed below.

4.6.1 Autonomy

Potential participants were provided with an information sheet detailing the process of the study and potential benefits and risks of taking part, which was favourably reviewed by the relevant ethical committees. Both patients and health professionals had to provide informed written consent to participate in the study. Each participant's signature was obtained on an informed consent form. Details of informed consent are presented in Appendix 9 and Appendix 10. All participants were reminded by the author that their involvement was totally voluntary and not influenced by any other external pressure or discrimination.

The author ensured that participants fully understood the purposes and the process of this study by asking them to read an information sheet which the author provided prior to the interviews. They could decide if they were willing to take part or not without any pressure. Potential participants were given at least 48 hours to consider whether they wanted to participate or not. The whole process was confidential and the participants were fully informed.

Participants could withdraw from the study at any point during the interview without giving any reason, and up until the data was anonymised and analysed (approximately one month after the interview). If participants refused to take part in the study, there was no effect on their treatment and care/professional career and legal rights, and this was made clear in the study documentation.

A digital voice recorder was used during each interview, with the agreement of participants. Even if they either refused permission to use direct quotes in any report or declined to receive the summary of findings, they still could take part in this study. They could indicate via the consent form if they agreed to these two aspects.

4.6.1.1 Confidentiality

For both groups, the participants were informed that the discussion would be confidential unless they provided information that suggested suicide or risk of harm to themselves or others. Participants were informed that the author would inform their healthcare professionals or GPs if risk to self or others was identified. For the professional group, the participants were informed that confidentiality would only be broken if evidence of practice which was inappropriate or harmful to patients emerged during the interview. The author would in such circumstances inform their managers. Each participant was given a reference number and their data was coded with these IDs without linking to their names. After the interviews, written consents and any documents containing the participants' personal details were securely locked in a cabinet in Chang-Gung Medical Hospital. After data collection finished, these documents were transported securely by the author in her hand luggage to the University of Manchester, where they were locked

in a locked cabinet. The codes were linked to the relevant consent forms, which were stored separately.

During transcription, personal details of the participants were removed by the author. Audio-files were uploaded onto an encrypted server at the University of Manchester and then accessed only via a password protected computer and deleted from the digital recorder as soon as possible. All digital recording files on the encrypted server at the University of Manchester were deleted after transcription was completed. Information stored on the University of Manchester computer or the encrypted server at the University of Manchester is protected with passwords and the passwords are held by the author. Only the supervisory team (Professor Karina Lovell and Dr Laoise Renwick) in the UK and the principal investigator have access to identifiable data. Furthermore, when the study was completed, the results of the study were documented and a summary provided to all participants. The data generated is to be stored for 10 years after completion of the study. Following completion of the study, all participant contact information was destroyed. During the interview process, the privacy of the interview environment was managed carefully. Those who decided to proceed arranged a preferred date, time and place for a face-to face interview. For the patient group, the interviews were conducted in a private room located in one of the three halfway houses. For the health professional group, the interviews were conducted in a private room at their workplace or in the hospital as they preferred.

4.6.2 Beneficence and non-maleficence

The author clearly discussed the individual and practical benefits with the participants. In this study, although there was no obvious benefit from taking part, participants had an opportunity to express themselves. Furthermore, there was a possibility of patients' care being changed, or for health professionals to receive training based on the findings of this study. By publishing the results of this study, these might be used to improve the level of care for patients and provide health professionals with insight into how patient feel about the decision-making process. In order to facilitate this, the summary of findings of this study will be sent to the participants who indicated on their forms that they wished to receive it. During the ethics committee application, the potential risks and benefits of participating in this study were assessed by the author, the supervisory team, the external committee at the school, the University of Manchester Research Ethics Committee, and the Chang-Gung Medical Foundation Institutional Review Board.

4.6.2.1 Potential benefits of participation

As for benefits, the individual and practical benefits of taking part were discussed with the participants. Although there would be no direct benefit to research participants, understanding patients' and health professionals' experiences of decision making will help understand what kind of decision-making model is experienced by patients in mental healthcare. In addition, facilitators and barriers which influence this process might be identified from this study. Most importantly, this could provide an insight into the preference of patients with mental illness during decision making.

4.6.2.2 Potential risks of participation

In terms of potential risk, the two most significant risks of harm posed to the participants were breaching confidentiality and causing distress. The issue of confidentiality was discussed above. In this following section, the research will present how the risk of participant distress was managed.

• Distress management

Due to the potentially sensitive nature of the topics related to negative experiences of decision making, participants might have felt overwhelmed or become upset during interviews. During the interview, the author was careful and used words with no subjective judgment, and time was taken to end interviews well by asking participants about their experience of the interview, to check whether the interview had unsettled them in any way and to discuss any issues that had arisen. Some personal questions about the process of decision making in mental healthcare might have caused distress. A distress policy was written and was adhered to (Appendix 11).

4.6.3 Justice

To ensure justice, significant efforts were made, including equal opportunities for both patients and health professionals to participate in this study. Information packs were allocated for both patients and health professionals, and inclusion criteria were widely distributed for both groups (within the selected settings for both groups, and on social media for the professional group). In order to ensure that patients were not discriminated against, assessment of eligibility for participation should focus on the ability to understand and voluntarily participate without any external pressure. In addition, the length of interviews was kept to the minimum possible. Participants (patients and health professionals) were compensated for their time and effort when participating in the study with everyday supplies such as toiletries or a pre-paid gift card for no more than £10.

118

4.7 Summary

Overall, this chapter has discussed the working methods for studies based on interviews with patients and health professionals. These two studies aimed to explore their perspectives of how decision making was shared in mental healthcare. A qualitative method was used to generate data to provide a comprehensive and informative interpretation of the doctoral study. The qualitative studies incorporated thematic analysis and were underpinned by the philosophical position of interpretivism.

Chapter Five: Findings from patient interviews (Study Two)

This chapter will present the findings of the qualitative interview study which comprised in-depth semi-structured interviews with 20 patients in secondary mental healthcare (Study Two).

5.1 Characteristics of patient participants

Patients were purposively selected from three halfway houses between July 2017 and August 2017. A total of 92 information packs were distributed in the three halfway houses; 23 returned the contact form (20 contact forms through health professionals and three contact forms through the message boxes) indicating patients were interested in participating in the study. Of these, 20 patients agreed to be interviewed, and three felt they did not have enough available time to be interviewed and refused to take part. All interviews were held in interview rooms in the halfway houses. Interview duration ranged from 40 minutes to 79 minutes (average of 43 minutes).

The characteristics of the interviewed patients are presented in Table 5.1. The patients comprised 13 males and 7 females with an average age of 38 years. Most patients had been staying in halfway houses from a few months to 3 years (n=14). Five had been in halfway houses from 4 to 6 years and one for more than 10 years. Most (n=17) patients identified or reported that they had a diagnosis of schizophrenia, and three as having bipolar disorder, major depression, and schizoaffective disorder respectively.

Reference no.	Gender	Age	Self-Diagnosis	Duration of accommodation
1	Male	52	Schizophrenia	0 years 6 months
2	Female	55	Schizophrenia	1 year 1 month
3	Male	54	Schizophrenia	10 years 0 months
4	Female	39	Schizophrenia	3 years 0 months
5	Male	35	Schizophrenia	2 years 0 months
6	Male	46	Schizophrenia	1 years 9 months
7	Male	32	Schizophrenia	5 years 0 months
8	Female	57	Schizophrenia	4 year 0 months
9	Male	52	Schizophrenia	1 year 6 months
10	Male	52	Schizophrenia	6 years 1 month
11	Female	61	Bipolar disorder	3 years 0 months
12	Male	48	Schizophrenia	4 years 0 months
13	Male	32	Schizophrenia	2 years 0 months
14	Female	53	Schizophrenia	0 years 3 months
15	Female	28	Schizophrenia	1 year 0 months
16	Female	28	Major depression	0 years 1 month
17	Male	47	Schizophrenia	1 year 6 months
18	Male	39	Schizophrenia	1 year 2 months
19	Male	46	Schizophrenia	2 years 0 months
20	Male	35	Schizoaffective	4 years 0 months

Table 5.1: Characteristics of patient participants by reference number

5.2 Themes developed from the patient interviews

Details of the working methods and analysis of Study Two were given in Chapter Four. Data were managed in NVivo 11 (QSR International Pty Ltd., Australia) and analysed using Thematic Analysis encompassing six stages: familiarisation, generating initial codes, searching for themes, reviewing themes, defining and naming themes, and producing the report (Braun and Clarke, 2006). Using the principles of thematic analysis, two main themes were identified and several sub-themes were identified (Table 5.2).

- Theme 1 focused on barriers to shared decision making
- Theme 2 focused on facilitators of shared decision making

A discussion of these main themes and their sub-themes will be presented below. Example thematic frameworks can be viewed in Appendix 12 (Themes 1) and Appendix 13 (Theme 2).

Main themes	Sub-themes
1. Barriers to shared decision making	• Professional status of health
	professionals in submissive culture
	• Negative perception of making
	decisions
	• Limited time resource
2. Facilitators of shared decision making	Sufficient information exchange
	• Desire to make decisions

Table 5.2: Themes identified from patient interviews

5.2.1 Theme 1: Barriers to shared decision making

From the patients' accounts, shared decision making generally did not occur in mental health settings. Overwhelmingly, all patients described how the majority of decisions about treatment and care were made by health professionals and/or the families. A strong sense of powerlessness was widespread among patients, a feeling derived from the "professional status of health professionals in a submissive culture", "negative perception of making decisions", and "limited time resources" in their mental healthcare. These three issues will be explored in the sub-themes below.

5.2.1.1 Sub-theme: Professional status of health professionals in a submissive culture

One of the major causes of reluctance to be involved in decision making was the professional status of health professionals, leading to patients conforming to the socially sanctioned role of "patient". Fundamental beliefs in a submissive culture, such as that

"health professionals know better", and a fear of being labelled as a "difficult patient", strongly influenced how patients viewed the role of health professionals in the decisionmaking process. Meanwhile the power to decide that patients should be hospitalised was held by mental health professionals and therefore this reinforced the sense of power imbalance in the relationship.

Patients believed that both patients and health professionals in mental healthcare held different levels of power, based on the Taiwanese submissiveness culture. Patients viewed health professionals as authority figures and felt they should obey what mental health professionals said. Patients were relatively passive, did not assess their own conditions, and dependent on the health professionals' advice. This culture and obedience to authority are revealed in the following two examples:

"They are doctors. Doctors are qualified to control your treatment and care. Doctors are able to prescribe your medication. No one else can. They control... that's (making decisions) their responsibility." (08 F)

"There is no reason why... they should... They (professionals) are born to decide. They are professional. Yeah... that's why I should obey them." (04 F)

This not only referred to the individual roles of health professionals and patients, but was also related to how cultural beliefs influence patients when seeking medical help. The majority of the patients in the interviews shared a relatively similar pattern of treatment-seeking behaviours across mental health settings. The patients frequently perceived that there was a golden rule for how to interact with health professionals in Taiwan. The main role of patients was simply to provide medical symptoms to health professionals, and then let the health professionals decide for them. Patients perceived their role in relationships with their health professionals to be as "receivers" rather than "providers". The patients believed this – it is what they had been taught since they were children. This feeling therefore led to it becoming conventional wisdom to accept health professionals' advice. In this culture, it seemed to be unreasonable or unfamiliar for the patients to speak up or make their own decisions during the process. For example:

"It's just like seeing a doctor for a cold. I go to see a doctor for a cold and a doctor asks me my symptoms. And then does some examinations. And then you get some medication. It is a standard procedure. Like me. I told him (a doctor): 'I have cough, a runny nose, sore throat, and headache.' And then the doctor

gave me some medication to take. And then I took them... That's what happened to me. Never have I ever questioned this before. That's what I was told when I was a kid. I got used to this procedure. I think this is it. This is how the health system works." (05 M)

Furthermore, the sense that health professionals knew better was widely described in the patients' accounts. Throughout the interviews, more explanations referred to the fact that health professionals knew best due to their professional knowledge. Formal professional training and qualifications in medicine created a sense of correctness, authority, and superiority where "the doctor knows best". The view that health professionals all were experienced enough to recognise what was the best for patients could cause the latter to be reluctant to take an active role in decision making. The data showed that the feeling of deference to health professionals was considered socially appropriate, for example:

"As doctors, they have professional knowledge to make judgements. They are psychiatrists. Professional knowledge and professional staff. They went to medical school for seven years. They are experts on mental health. They definitely know how to cure patients and what to do." (16 F)

In these cases where the patients did not understand their disease or the available treatment/care options they cannot participate in decision making. Therefore, they tended to leave the decisions to the health professionals, who are "qualified". In secondary mental healthcare settings, this was mentioned by those patients who believed that a passive role was socially appropriate for them, for example:

"I don't know about medicine... don't understand. I just do what they say and take what they give me." (01 M).

Less commonly mentioned, but nevertheless prominent, was a fear of being labelled as a "difficult patient". There was considerable pressure on patients to conform to social roles and the social expectations of others, particularly of health professionals. For example, the health professionals gave positive feedback when the patients played their passive roles well but gave negative feedback when the patients played their roles badly, which normally meant including some behaviour the health professional did not agree with. Therefore, patients acted submissively during decision making to avoid displeasing their health professionals. For example, asking too many questions or refusing the health professionals' recommendations were not allowed, in the patients' view. "If they (health professionals) ask you to participate some activities (such as exercise), you should go and be well behaved. You should comply with it. Or if you refuse to take part or turn them down, they will shout at you... you know their faces... They don't like patients who ask a lot of questions or are not well behaved. I don't know how to describe it, but I should comply with them." (07 M)

Alongside this problem were the covert contracts that a few patients developed with their health professionals, where they felt coerced into adopting the role of a "good patient". This role was regarded as one of passivity and compliance. Patients described their sense of needing to be respectful of what health professionals suggested and expressed the negative implications of being a "difficult patient" if they did not. Many patients believed that they should not be involved in decision making, often due to fear of annoying the health professionals and the potential repercussions. The potential retribution prevented patients from actively engaging in decision making discussions. For example:

"You can't deny their professional status... Because they are doctors. They help us, but they don't require any input from you. If you refuse to take their advice... that means you are not respectful to them. You are too difficult to please." (20 M)

In addition, patients often perceived that patient hospitalisation and discharge schemes were all decided by the health professionals. Therefore, patients believed that health professionals were superior to them. The fear/threat was expressed by every patient, without probing, that unless they complied with the treatment and care decisions they would be re-hospitalised.

The threat of being sent to hospital, which resulted in a loss of liberty, indicated that patients felt controlled, rather than involved in decisions about treatment or care. This was strongly driven by a fear of mental hospitals: a sense that mental hospitals are effectively a restricted environment for people suffering from mental illness. In particular, the use of involuntary and coercive interventions in the hospitals was a major threat for the patients. They commonly reported that they were being physically restricted without their consent, leading to them adopting silent roles during decision making.

"In hospitals, we (patients) are nuts and they are health professionals. What can you do against them? If you don't follow their instructions, then they lock you up without asking you. Nothing you can say about it. They tie you up and even tell you: "anything you say is useless". "We are health professionals in a hospital" and "I can decide what to do to you (referring to coercive interventions)." (06 M)

Of central concern to the patients, the threat of greater use of involuntary interventions and coercive measures, such as physical restraint and enforced medication and injections, were widespread, and the patients often described these interventions and measures as traumatic and unnecessary.

"They felt I spoke loudly and got emotional. They wanted to tie me up. I said I was fine and no need to be tied up. The problem was they still thought I was not stable and did not take my medication regularly. But I did not want to.... I hated it (physical restraint). They just forced me." (20 M)

They explained how such measures and interventions were given in mental hospitals without their consent if judged by health professionals to be in their best interests. Most patients reported how such practices violated their sense of dignity.

"I am so scared (of physical restraint). So afraid of it! So I messed up. Like a dog, you know, you were tied by these belts. Really tight (around both hands and legs) and not able to go to the bathroom by myself... So devastated. I should listen to them." (18 M)

They also described having been hurt or injured by such coercive procedures. Therefore, the great fear/threat that patients lived with was that of being sent to the hospitals. For example:

"If you don't listen to the nurses, they tie you up. Even if you improve your behaviour, they still tie you up. They must tie you up for an hour. Some (nurses) even give you an injection. Even if you are yelling in the room, they won't listen to you. Like this. You must do what they (nurses) ask you to do." (10 M)

In addition, there were not just coercive measures or involuntary interventions, but also restrictions on patients' daily lives. From the patient views, hospitalisation caused disruption to their daily routine, such as social activities and working. Patients felt that health professionals had huge powers in that they decided if they were permitted to go outside, visit someone, have a meal with their friends or families, or go shopping. Patients

mentioned that the health professionals employed these "god-like" rules to force them to adhere to treatment regimens in the hospitals; for example:

"As the nurse said....if I lay on the bed in my room...they (nurses) would say: 'If you lie in like that, you will not sleep at night.' In the end, they (nurses) said: 'if you still lie on your bed, we will restrain you. Or not allow you go outside and go shopping'. Can't go outside or anywhere. So I don't really like going to hospitals" (15 F)

As part of the negative experience of being hospitalised, verbal threats of hospitalisation from health professionals were widely reported by patients. They reported that demanding language was used to force the patients to comply with instructions, such as *"you must"* and *"you should"*. They expressed such threats as frightening and perceived them as way of gaining power and control over their lives, for example:

"I told them: I don't want to take this medication... But the staff told me I must take it... or they will take me to be hospitalised. So... so... I took it... and after this, I told my doctor what happened to me... You know what?... My doctor just kept silent... Like nothing had happened... He said: 'Adjust... it is just fine... to adjust.'" (09 M)

Patients described many of the characteristic statements used to them, including "take your medication or go to hospital", "if you are well-behaved, you won't need to go to hospital", "listen to me, and nothing bad will happen to you again", and "you have to take medication". Patients complied and tolerated such threats but were clear that they were an effective "weapon" used by health professionals to force patients to comply with instructions.

Patients acknowledged that they were powerless, in that it was impossible to discharge themselves from services, particularly acute wards. It was perceived that there was no exit for them to escape. Patients said they were locked in wards and told they could not leave even if they wanted to. Some patients compared being detained to imprisonment and even described themselves as "prisoners" locked up in the hospitals, as illustrated by the example below:

"Because... because... this belongs... this is same as (being a) prisoner'...because they are locked up as well. We have to accept assessment. This is all decided by health professionals... same as prisoners. Many prisoners are locked up as a big group. I certainly felt uncomfortable. But what can I do about it?" (07 M)

More interestingly, in some cases, the patients described a feeling of culpability when they did not fulfil the health professionals' or families' expectations, behaved unacceptably, or were forced to comply with the professionals' instructions more than usual. As a result, the patients tended to submissively follow the rules to earn "privileges". One patient described it as follows:

"If you are well-behaved, you can go outside and you can be discharged from here soon. Just follow the instructions. This is good for you." (08 F)

Some family members also treated the patients the same way to force them to comply with instructions. For example, one patient described how his brother asked him to take medication regularly and told him what would happen if he refused to do so. He explained:

"My brother said: 'If you are taken there (the hospital)... we will lock you up for more than a year.'... No doubt... You can't say no... In the hospital, your family has to agree or you can't be discharged from the hospital. You know my friend has been locked up for a year... He is not allowed to discharge himself from the hospital. Can you imagine that? I told a nurse: 'I don't want to stay here'... But I got no reply... I had been locked up for one year... You know my friend... he is still locked up... Not like here (the halfway house)... you have more freedom." (18 M)

5.2.1.2 Sub-theme: Negative perception of making decisions

In patients' accounts, there was a concern about their decisional capability due to their mental illness. A variety of negative perceptions of their decisional ability led patients to have no confidence in expressing their perspectives or asking to be involved in decision making. It was generally more acceptable for the patients to keep silent in the decision-making process. Some patients believed that it was appropriate that mental health professionals did not actively engage with them, due to their mental conditions.

These negative perceptions originated from various sources, including the patients themselves, health professionals, and the public. Surprisingly, the strongest sense of impaired decisional ability was reported by the patients themselves. In this regard, a

number of patients reported that they were afraid of making unwise or irrational decisions. For example:

"I am ill (mental illness). How could I make decisions? I feel great if they decide for me." (12 M)

From the patient views, the patients with a first or recurrent psychotic episode tended to have experienced being out of control of their actions and movements, and being a risk to the public and themselves. Owing to these negative experiences in the past, the patients found it was difficult to believe that they were still capable of making any decisions about their treatment and care. In particular, patients who experienced intolerable symptoms (delusions or auditory hallucination) often felt very afraid that they were losing control again. Therefore, patients believed they were not qualified to make any decision because of having mental illness. For example:

"I was losing control. I don't know what disease I have. I could not sleep at night. Every day there were a lot of voices (auditory hallucination). I was sick. Not clear... There were a variety of treatments. If you are sick, this (treatment option) should be decided by doctors." (10 M)

There were further examples in the data of patients who "lost control" and deemed themselves incapable of making reasoned medical decisions.

"Yeah... if I don't take medication, I get out of control. Because... I have mental illness, I lose control. I couldn't make my own decisions. How can I make my own decisions? You must take medication. That's it. Because you are hospitalised... you lose control. You are not able to make decisions." (07 M)

Furthermore, patients also felt that the public's negative judgement aggravated the situation. One patient gave an example of a murderer, who the television news implied was suffering from mental illness. It appeared that the awareness of this trend shaped patients' attitudes towards themselves and reinforced a sense that patients were not qualified or not able to be involved in decision making.

"I think any patient with mental illness is not able to make their own decisions. Some of them may lose control and then kill someone. For example, XXX, he lost control and killed someone. Can someone like me make their own decisions?" (03 M) In addition, patients felt that health professionals also negatively judged them and viewed them as "patients without the ability to make decisions". The sense that the patients who were admitted to "mental health hospital" were viewed negatively also influenced the level of patient involvement in decision making. One patient described how he tried to ask more questions about his condition when he was first admitted, but the health professionals refused to answer. He perceived that the health professionals believed that he (the patient) was now suffering from mental illness and had been sent to the hospital, so he was no longer capable of making his own decisions. Therefore, the health professional refused to provide any information to him (the patient). The patient used the term "psychiatric hospital" to imply his decisional capability was impaired in the other people's view:

"I went to XXX hospital. I wanted to ask some questions about my condition. But they (health professionals) told me this is a psychiatric hospital and I am a person with mental illness. I had a headache and auditory hallucination. Someone (auditory hallucination) was talking to me. And then I was admitted (to the hospital). 32 years old. I wonder if anyone still wants to discuss anything with me." (06 M)

Although most of patients paid attention to decisions about treatment and care, such as medication and recovery plans, a few patients also mentioned how these judgements negatively influenced decisions about their daily life. For example, one patient gave an impassioned explanation about how she could not decide to live with her own son. Her health professionals perceived that she was suffering with her mental health and made risky decisions. She reported the feelings of powerlessness as follows:

"For example, they aren't willing for me to take my baby along. This is a clear rule. I don't think I need to stay in the halfway house. I am able to find a job and live on my own. Why do I need to stay here? They think I am ill, that I am not able to take care of myself or my baby and decide. So they decide everything for me." (04 F)

5.2.1.3 Sub-theme: Limited time resources

Most patients felt that health professionals, even if they agreed to shared decision making, were under too much time pressure and/or had limited resources available to give patients as much information or support in making decisions as they wanted. Patients felt that health professionals were required to see a lot of patients every day. The

ideas that "health professionals are busy", "there are a lot of patients waiting" and they "do not have sufficient time" were frequently reported in the data. Because they were not given adequate time to ask questions or respond, the patients often felt unable to be involved in the decision-making process. Also, this feeling put pressure on the patients to shorten their time with the health professionals. One patient described it as follows:

"I don't know. Every appointment with him (the health professional) finished very quickly. Because there are always other patients who are waiting. He needed to see a lot of patients in one day. So we did not talk too long. I hoped we could talk for a longer time. You know." (04 M)

Similarly, feelings of a lack of available time, along with the number of patient contacts, led patients to hesitate to ask more questions or express their expectations. This limited time could result in less detailed explanations to patients about treatment and care. Even if the patients were deeply concerned about their condition, there was not enough time for them to be actively involved in decision making. For example:

"If some professionals are able to give me some information that will be great. How to use the medication and what side effects I might notice. I want to know... but, probably, the doctors don't have much time. Like me, I am going to see my doctor tomorrow. I am number 130. I bet the doctor won't have time for me." (05 M)

Due to limited time being available, the patients noticed the health professionals tried to make more efficient use of what time there was. This caused some problems in health professional-patient interaction across the mental healthcare settings, such as no privacy, no proper attention, and inefficient communication in medical consultations.

Patients described their appointments with health professionals as having a lack of privacy during their consultations. It was relatively common practice in Taiwan to see a number of patients together in one appointment. Owing to the limited available time, the health professionals tended to see as many patients as they could in as short a time as possible. In some situations, health professionals asked more than one patient into the room at once. For example, one patient discussed his/her case with a health professional while other people (other patients) were in the same room to wait for their appointments. Even if the patients would have liked to discuss something with the professionals, it was difficult for them to express deep feelings in front of others:

"Three were in the same room. How could you say something sensitive to the doctor? How could you discuss anything with the doctor? How could I react to this? Three of us... say something.... The doctor gave us medication. That's all. And then I got medication. The end." (09 M)

A few patients believed that the performance of the mental healthcare professionals needed to be improved in some ways. It was perceived that the health professionals did not pay proper attention in the consultation. It seemed that the patients were not the primary focus of attention and a large of amount of the health professionals' time was spent interacting with a "computer" rather than "patients". Health professionals interacted with their patients for only a small part of the time; for example:

"He (the health professional) *did not say much. He just typed into his computer. How can I express anything to him?"* (18 M)

In addition, there was no opportunity to have direct conversation with health professionals, particularly psychiatrists, which was also pointed out by a few patients. An unsatisfactory amount of communication often caused the patients to be frustrated and limited their active involvement in decision making. A patient described the situation when he refused to take medication and intended to discuss it with his psychiatrist, and how the health professionals responded to him, as follows:

"We didn't really discuss it directly... He didn't come. But my nurse told him... He replied and the nurse told me what he said. He said: "Take it or leave it"." (03 M)

5.2.2 Theme 2: Facilitators of shared decision making

In the accounts, patients placed more emphasis on barriers to shared decision making as opposed to facilitators. Yet it was clear from the data that patients also saw examples of how shared decision making could be facilitated. Although the levels of involvement in decision making desired varied, many patients said that they wanted a more active role in the decision-making process. Almost every patient desired more information, but not everyone wanted to make final decisions. This clear desire motivated them to consider the possibility of their involvement in decision making and the potential of implementing shared decision making.

Active involvement in decision making was conceptualised by patients as a process whereby they could be engaged in their treatment and care, making it more likely to empower their decisional control. To this end, the two notions of shared decision making which were consistent with these principles are explored in the following two subthemes. These are: sufficient information exchange and desire to make decisions.

5.2.2.1 Sub-theme: Sufficient information exchange

Patients felt adequate information exchange was essential for them to gain a comprehensive understanding of what they should expect to happen next, and what conditions they might experience. Some patients directly mentioned that patient involvement could bring some benefits for treatment and care. Moreover, their knowledge about their bodies could also improve health professionals' understanding of their health conditions. Their unique expertise was recognised as important. The patients believed that long-term illness made them good doctors, and they were experts on their own diseases and bodies. They perceived that it was worth listening to people who actually required this help. For example, one patient explained that he had suffered from mental illness for a long time, which meant that prolonged illness had made him an expert on his condition:

"Long illness makes the patient a good doctor." (01 M)

Furthermore, active patient involvement could facilitate mutual understanding between health professionals and patients. In particular, from patients' points of view, they emphasised more the importance of how their voice could be heard by health professionals and others. This active involvement gave the patients the unique opportunity to express their views on their medical condition and treatment. One patient said:

"Because doctors should discuss with us to see how this medication works for us. They (doctors) prescribed medications to us. They should show respect for my choices. To see if I am cured, they should observe how I react to the medications." (19 M)

Similarly, patients believed that providing information on treatment and care could help alleviate concerns and reduce the uncertainty of the patients. The patients would perceive that their expectations and concerns about their treatment and care for their mental illness

133

had been considered in the process. The patients would take part as information-givers and not just passive recipients of medical information; for example:

"Based on this (active involvement), I could be cured quickly. I could understand if I am ok or not. I could explain my feelings about medication to them. They don't need to spend much time conducting experiments on me." (03 M)

In the process of exchanging information, it was not just that health professionals could gain the most effective understanding of patients' conditions and their expectations, but also that patients could have an insight into their mental illness and recovery plan. By making sure that patients understood their recovery goals, their conditions, symptoms, and the choice of treatment, patient perceived that they could be engaged in their prevention plan. One patient explained:

"If...(I can be actively involved in decision making).... Because I can understand my symptoms and conditions. I could understand how to prevent a relapse, to understand better if my condition is worse or better." (05 M)

Numerous patients recognised the desirability of gaining more information about their treatment/care. Unsurprisingly, most of the patients frequently paid attention to information/knowledge about medication, such as its effects and side-effects, changes of medication, or dosages. For example:

"I want to ... I want to know the effects of the medication. In fact, I really want to ask the doctors." (19 M)

Patients rarely mentioned their desire to be given information about relevant treatment options rather than being presented with a single treatment plan decided by the health professionals. For example, one patient stated:

"I think I do... I do want to know about other treatment options. For example, treatment one, treatment two, and treatment three. This would be really great. I don't know why the health professionals don't implement that. I don't know how they think." (08 F)

5.2.2.2 Sub-theme: Desire to make decisions

Although almost all patients desired more information about their treatment and care, they had different views of the right to make final decisions. One of the key notions of shared decision making, as described by a few patients, was making their own decisions. The desired levels of involvement in decision making varied from making their own decisions to tending to leave it to health professionals to make decisions about their treatment and care. Some patients directly mentioned they would prefer to make their own decisions about treatment and care, for example:

"I certainly do (want to make my own decisions). *But no way... The only thing I can do is live here* (in the halfway house)". (14 F)

More than sharing information on treatment and care, the patients would like to have opportunities to deliberate with health professionals about their expectations, based on their preferences and personal values. The view that patients and health professionals should jointly engage in the decision-making process and come to a negotiated agreement about which treatment is chosen and implemented was widely identified by the patients. Here are two examples:

"I want to discuss with my doctors that I don't want to take medication... yeah. Cause I took medication for many years. I don't think I need to take this... I think I will be fine even if I don't take any medication." (04 F)

"I think... if... medication... I want to discuss with him (the professional) whether I could adjust my medication. Now what he gave me... I can't get out my bed in the morning, I feel drowsy. I can't get out of my bed. I would like to tell him this is not right for me. I would like to change it." (12 M)

From some patient perspectives, patients should have the right to speak up and make their own decisions because the decisions on treatment or care would have consequences for their everyday lives. The patients believed they were more comfortable making these kinds of decisions and taking responsibility. For example, some patients directly mentioned they would prefer to make their own decisions on treatment and care;

"He (professional) never discussed it with me (treatment options). If possible, we could discuss it... and then I could decide myself. If you can decide, you feel better in your mind. This is all decided by me! No matter if it results in good or bad. I feel more comfortable about this. It wasn't decided by anyone else but me. Although I am a little slow on the draw and have a bad memory. But I believe I am able to make my own decisions." (06 M)

Another patient who suffered medication side effects explained why it would have been better for him to make his own decision in the first place:

"It is better they tell me (more about treatment options). It is better to have choices. You could find a treatment that is more perfectly suitable for you. That would be really great... After I took medication, I didn't know about its side effects. I gained weight - from 70 kg to 90 kg. I didn't know about it at the beginning. I hope I have other options." (05 M)

In the interviews, the patients also recognised that deliberation involves two-way communication between patients and health professionals. The patients should provide their views on treatment and care. Meanwhile, the professionals also were experts providing medical facts and experience. Therefore, from patient perspectives, suggestions about treatment and care offered by professionals could be equally important to patients to make more considered decisions about their treatment and care. One patient mentioned:

"I also would like to hear some professional advice from the professionals. I think if you got many professional suggestions, you could compare other treatments from different angles. You could hear different solutions." (05 M)

Some patients recognised the importance of professional recommendations in making informed decisions. Even though the patients would like to make their own decisions, they believed they still needed the advice of health professionals. One patient explained this in the following quotation:

"It will be great if I have medication options. But I would love to hear some advice from my health professionals. And then I can choose the best one for myself." (17 M)

In their accounts, patients focused more on the decisions about their treatment and care. However, a few patients also reported the importance of making their own decisions about their daily routine. For example, one patient explained her preference for making decisions, and that she desired to make not only medical decisions but also daily life decisions. She said:

I hope I can decide where I want to live and…if I am stable. The doctor can reduce the dosage of my medication. (07 M)

Patients had different preferences about the level of their involvement in decision making. Although some said they wanted to be involved in the decision-making process, in fact some often preferred the health professionals to make the final decision for them. Despite some patients not wishing to make their own final decisions, most still would like to have more information about their treatment and care. For example:

"Actually, I just want to understand what my medication is for... I don't need to know what choices I have... For example, what treatment I take... particularly what medication is for... Effects or side-effects. That's all I want." (02 F)

"Just information about my medication. Just let me know what medication they give me and why. What the medication is. Other things... that's fine. I don't need to decide." (16 F)

5.3 Strengths and limitations

To the author's knowledge, this is the first study exploring patients' perspectives of decision making in mental health in Taiwan. This study has provided an insight into how patients view the decision-making process and what they need. The importance of patients' suggestions has been identified by previous research and by the findings of this study. It is evident that patients should be involved in developing any new healthcare delivery and care approaches, such as shared decision making. Therefore, the findings of this study provide a unique opportunity for health professionals and policy makers to explore how to improve levels of patient involvement in decision making and the quality of healthcare.

There are limitations to this study. The findings should be treated cautiously because of the small number of respondents (n=20). Although qualitative studies are not intended to be generalisable, their strength depends on transferability: whether the results are transferable to other settings or contexts (Korstjens and Moser, 2018). However, the sample in this study comprised patients only from halfway houses and therefore were not transferable to other mental health settings. The largest number of patients were diagnosed with schizophrenia (n=17) and thus this limits the findings to this group. In addition, patients in the halfway houses could have more complex long-term mental health conditions and support needs in terms of shared decision making than other patients in secondary mental health settings. Yet, these patients also have rich experience of making decisions across different settings (inpatient and outpatient departments).

Moreover, some studies also found that even amongst patients with different forms of mental illness, most of them wanted a more collaborative role in decision making than they usually experienced (De Las Cuevas et al., 2014; Hill and Laugharne, 2006). Meanwhile, there is a significant knowledge gap about patient perspectives of decision making in Taiwan. Exploring the desires of patients with different psychiatric diagnoses or in various mental healthcare settings might be valuable, and would be imporaant to explore in future studies.

5.4 Summary

Overall, two main themes were identified: barriers to and facilitators of shared decision making. Evidence from this study suggested that patients in mental healthcare in Taiwan were not involved in the decision-making process, and paternalistic and coercive decision making was commonly reported. Three main barriers to shared decision making were identified, which included the idea that the powerful status of health professionals led patients to believe that the health professionals were more qualified to be the decision makers than the patients themselves. Also, patients had negative perceptions of their ability to make right decisions and therefore did not have the confidence to be involved in the decision-making process. Furthermore, a lack of sufficient and quality time with health professionals prevented patients expressing their deeper feelings, and therefore had a negative impact on patient involvement in decision making.

Despite these barriers, the patients in this study still desired a more active role in the decision-making process. These desires drove patients to engage more in the decision-making process and therefore facilitated shared decision making. Currently, opinions from the patients in Taiwan were divided about what level of more active involvement in decision making they desire. Generally, all patients looked forward to more information about treatment and care, such as the effects and side effects of medications. Interestingly, not everyone would like to make final decisions about their treatment and care. Some directly expressed that they desired not just to gain information but also to deliberate with mental health professionals and then to make a final decision about their treatment and care. However, a few patients would like to avoid making their own decisions and chose to leave the final decisions to the health professionals alone. Although there were differences in desired levels of patient involvement, all patients expected a more active role than now.

The next chapter will provide a detailed discussion of the interviews conducted with mental health professionals.

Chapter Six: Findings from health professional interviews (Study Three)

This chapter will present the findings of the qualitative study which consisted of in-depth semi-structured interviews with 24 health professionals in secondary mental healthcare in Taiwan.

6.1 Characteristics of professional participants

Professionals were purposively selected from the five chosen settings between July 2017 and November 2017. A two-stage recruitment strategy was implemented as detailed in the working methods chapter (Chapter Four):

- **stage one**: of the 95 packs distributed, 23 returned consent to contact forms; of these 20 agreed to interviews. (Three refused to participate due not having time to undertake an interview.)
- **stage two**: of the 20 packs distributed, ten returned consent to contact forms; of these six agreed to interview. The four who did not consent did not have time to take part in an interview. During stage one, professionals were identified through the managers of one mental health outpatient department, one day-care centre, and three half-way houses. 20 study information packs were distributed, reflecting the number of qualified health professionals in the settings. Furthermore, recruitment adverts were posted in health professional groups on social media. 75 information packs were distributed, reflecting the number of potential professional participants who responded. Therefore, 95 information packs in total were distributed in stage one. Following this, 23 potential professional participants returned the contact form, but three refused to participate in an interview due to the long duration of the interviews. Therefore, 20 professional participants agreed to interview at this stage.

In stage two, due to the initial lack of response from nurses, a second stage of recruitment was carried out. Adverts were posted in particular nurses' groups on social media. 20 study information packs were sent out when potential professional participants contacted the author and then 10 returned the contact forms. However, four refused to take part in interviews for two reasons, including no mutually suitable time and long duration of interviews. In the end, there were six professionals who agreed to interviews in the second stage.

26 interviews were conducted with health professionals. However, as two professionals withdrew their consent following the interview, only 24 interviews were retained for data

analysis. Interview duration ranged from 38 minutes to 1 hour and 32 minutes (average of 55 minutes). The characteristics of the interviewed professionals are presented in Table 6.1. Professionals comprised 13 males and 11 females, aged between 28 and 54 (average of 37.8 years); there were 13 psychiatrists, 6 mental health nurses, 3 occupational therapists, one psychologist, and one social worker, and their durations of clinical experience ranged from 1 year and 8 months to 28 years (average of 12 years and 11 months).

Reference no.	Gender	Age	Professional role	Working experience
1	Male	40	Psychiatrist	11 years 0 months
2	Female	35	Psychiatrist	3 years 2 months
3	Male	33	Psychiatrist	4 years 1 month
4	Male	39	Psychiatrist	7 years 3 months
5	Male	43	Psychiatrist	16 years 1 month
6	Male	54	Psychiatrist	28 years 0 months
7	Male	44	Psychiatrist	16 years 8 months
8	Male	38	Psychiatrist	12 years 0 months
9	Male	42	Psychiatrist	16 years 1 month
10	Male	35	Psychiatrist	3 years 4 months
11	Female	41	Occupational Therapist	15 years 0 months
12	Female	45	Occupational Therapist	20 years 1 month
13	Male	40	Psychiatrist	10 years 0 months
14	Male	41	Psychiatrist	11 years 2 months
15	Female	49	Psychologist	15 years 0 months
16	Male	28	Occupational Therapist	3 years 2 months
17	Female	43	Social worker	7 years 0 months
18	Male	39	Psychiatrist	3 years 9 months
19	Female	34	Registered Nurse	13 years 0 months
20	Female	36	Registered Nurse	12 years 7 months
21	Female	39	Registered Nurse	17 years 4 months
22	Female	31	Registered Nurse	4 years 5 months
23	Female	40	Registered Nurse	14 years 0 months
24	Female	37	Registered Nurse	1 year 8 months

Table 6.1: Characteristics of professional participants by reference number

6.2 Themes from the health professional interviews

The data were managed in NVivo 11 (QSR International Pty Ltd., Australia) and analysed using Thematic Analysis encompassing six stages: familiarisation, generating initial codes, searching for themes, reviewing themes, defining and naming themes, and producing the report (Braun and Clarke, 2006). The analysis process was described in full in the working methods chapter (Chapter Four). In brief, a long-list of codes was gathered in the initial stage of data analysis. Based on the similarities and differences, the codes were combined or collapsed into themes. Example thematic frameworks can be viewed in Appendix 14 and Appendix 15. Two main themes and sub-themes were identified from the data (see Figure 6.1 and Table 6.2).

- Theme 1: barriers to shared decision making provides a discussion about barriers negatively affecting the implementation of shared decision making, and is divided into four sub-themes.
- Theme 2: facilitators of shared decision making focuses on to what extent health professionals are aware of patient autonomy in mental healthcare and understand the benefits of implementing shared decision making, and its potential positive impacts on patient outcomes. It is divided into two sub-themes.

Table 6.2: Th	emes identified	from health	professional	interviews
---------------	-----------------	-------------	--------------	------------

Main themes	Sub-themes
1. Barriers to shared decision	Powerful status of health professionals
making	and families
	• Patients with impaired decisional
	ability due to mental illness
	Health professionals lack
	understanding of shared decision
	making
	Insufficient time
2. Facilitators of shared decision	Awareness of patients' right to
making	autonomy
	• Understanding of potential benefits of shared decision making

6.2.1 Theme 1: Barriers to shared decision making

Overall the data showed that professionals failed to practice or embrace the concept of shared decision making. The majority of professionals thought there were significant challenges to implementing shared decision making. Key barriers were identified from the professionals' statements, and included "the powerful status of health professionals

and families", "patients with impaired decisional ability due to mental illness", "health professionals lacking understanding of shared decision making", and "insufficient time". Further discussion of each theme will be presented in the following sections.

6.2.1.1 Sub-theme: Powerful status of health professionals and families

This sub-theme revealed that professionals often had a powerful status in their relationships with patients, and indicated that they and family members were considered superior to patients. From the health professionals' perspective, it was socially acceptable that health professionals and family members took a proxy role and decided for patients in mental healthcare. This idea had a negative impact on active patient involvement and the implementation of shared decision making. In the following sections, the powerful status of health professionals and family members will be presented.

6.2.1.1.1 Professional-patient relationship

All professionals held strong beliefs that health professionals should have more power than patients and perceived that such power was justified as it was their duty to make the best decision for patients. The professionals articulated that there were different social roles for patients and professionals. They considered themselves to be "providers" who should provide treatment and care for patients, and that patients should be "receivers" of their decisions. The professionals asserted that the traditional allocation of responsibility meant decision-making mainly belonged to the health professionals, particularly psychiatrists. This belief strongly drove how health professionals viewed their role in decision making. Two professionals explained:

"We are more like 'a provider' to provide resources and to solve problems." (06 M, Psychiatrist)

"I don't know, this is kind of a tradition of healthcare? They (patients) are more like a receiver rather than a provider. They don't speak up for themselves." (11 F, Occupational Therapist)

Some professionals explained this situation by citing the cultural perspective that "health professionals know better" or "doctors know better". A fundamental virtue in Taiwanese culture was respect for one's parents, elders, and ancestors, and this extended to some professions, such as teachers, doctors, and lawyers. One professional explained this idea and linked it to how the culture meant that they were reluctant to actively involve patients in decision making. For example:

"I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this... You should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide)". (14 M, Psychiatrist)

Similarly, there was broad agreement that the best approach to making decisions about treatment and care was based on professional knowledge. It was perceived that health professionals were justified in making treatment and care decisions for patients as they had superior levels of knowledge. Professionals described how their beliefs about the superiority or authority of health professionals had been reinforced since childhood, i.e. that doctors' and other health professionals' orders and advice should be adhered too. One professional reported:

"This is traditional eastern culture. It is kind of authoritative. A doctor is more...(qualified). When people were children, they were told to comply with doctors. We should completely obey them and do what they ask. That's what people have been taught since they were children." (12 F, Occupational Therapist)

In health professionals' accounts, knowledge was believed to require professional training and education programmes to prepare health professionals for a continuous process of learning how to make clinical decisions, which patients did not have. Because of biomedical and practical expertise, professional training, and experience, health professionals were in the best position to assess trade-offs between different treatment options and to make the best decisions for patients. For example:

"I think [shared decision making] could be difficult in mental healthcare. It could be challenging. For example, we (health professionals) assess [patients' conditions] according to the standard assessment. After professional assessment, there is the best (treatment) option and then I make a (treatment) plan. This is what we do." (16 M, Occupational Therapist)

Furthermore, some health professionals also proposed that they would be well versed in the newest and most valid clinical thinking. Health professionals would not only know the best treatment option available, but also they would consistently apply the updated knowledge when selecting treatments for the patients. For example:
"Because psychiatrists have practical expertise in mental healthcare. There are psychiatrists for mental healthcare and physicians for general medicine. You don't ask physicians for information about general surgery. And they have a lot experience of taking care of patients every day. They are very familiar with this. They know how to deal with this and make adequate medication decisions." (19 F, Registered Nurse)

The professionals discussed clinical guidelines which allowed them to correctly decide about appropriate care for specific clinical circumstances. This is illustrated in the quote below:

"For example, I would tell them that if you get mental illness, you need to take medication. We would let them know what their medication is, but what kinds of medication they need to take is decided by you (professionals) based on the clinical guidelines." (03 M, Psychiatrist)

Interestingly, some professionals emphasised the legitimacy of their right to prescribe medication for patients. Based on this legal right, they felt empowered to be the decision maker. One professional reported:

"I believe the right to prescribe medication belongs to doctors. Although this is doctors' right, we all hope the patients take their medication regularly." (10 M, Psychiatrist)

The importance of the duty of prescribing medication and control of the hospital discharge scheme were reported to reinforce the power of health professionals. The right to hospitalise and discharge patients was mainly held by the health professionals and this again reinforced their power to exert control over patients. Health professionals felt that patients themselves had a similar perception of health professionals' powerful roles. In order to avoid staying in hospital longer, patients tended to ingratiate themselves with health professionals by following their instructions. This led to health professionals clearly feeling their powerful status of being health professionals. Furthermore, this sense of enhanced knowledge legitimated health professionals' use of control and coercion. For example,

"Because patients have an expectation: I need to be discharged from the hospital first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctor has asked them to do. And then they could leave hospital." (14 M, Psychiatrist)

An interesting issue reported by a few health professionals was that by offering options to a patient or embracing the principles of shared decision making might make them look indecisive to patients. They felt that by embracing shared decision making this would show 'weaknesses and professional incompetence' to the patient. Hence this might go some ways to explain why health professionals perceived a paternalistic or authoritative attitude to be correct and why they found it difficult to embrace the concept of shared decision making, this may give rise to confusion or a sense of insecurity. The perception that "the professionals are not sure what they are doing" was identified from the interviews. One professional mentioned concern about disclosing uncertainty about treatment, which would adversely affect the relationship:

"If you (professionals) provide more decision options... no matter how you explain it... they would think you are not 100% sure about this... You tell them too much and this means you're not doing your job, you know. This could bring unnecessary trouble (medical malpractice) for yourself." (02 F, Psychiatrist)

Professionals also described the hierarchy within the health profession, with doctors at the top of the pyramid. Many of the health professionals discussed this authoritative/paternalistic view as coming between psychiatrists and patients, and that they themselves (i.e. other health professionals) followed the psychiatrists' instructions as they were top of the hierarchy and therefore had the most power and control. Health professionals themselves perceived that it was the psychiatrist who should make the decisions. The perception that there was no room for an inter-professional healthcare team was commonly reported by professionals (except psychiatrists). One mental health nurse described this as follows:

"They (psychiatrists) don't offer options. They just... explain. Nurses just explain what choices the psychiatrists made, such as dose or frequency. Actually these medication treatments were mainly explained by the doctors rather than us (nurses). But so far, we've never seen any doctors offering options to patients. It is all decided by the doctors." (22 F, Registered Nurse) The roles of doctors and other professionals, such as nurses, were inherently different, which was reflected in their education and responsibilities. These differences had led to power stereotypes in each profession and influenced how they recognised their role in decision making. Interestingly, even other professionals believed medication was only managed by "doctors."

"This is a doctor's job. Doctors are responsible for deciding what medication they should take. Because patients don't have any medical background. They haven't studied in a specialist medical educational system. Even me (nurse), I have some basic knowledge but I am still not a doctor to know which medication will be effective for patients." (24 F, Registered Nurse)

6.2.1.1.2 Family-patient relationship

From health professionals' perspectives, the family structure exerted significant influence on the level of patient involvement in medical decisions. Although it was clear that health professionals were seen as – and felt justified as acting as – authority figures, this also extended to some degree to family members. Therefore, the negative influence of the powerful status of family members was also reported in health professionals' accounts.

Health professionals discussed the traditional patriarchal family structure in Chinese society and its impact on patients' decision-making authority regarding mental health treatment options. The professionals reported that there was normally one family member who held the most entitlement to make all decisions in family. In traditional families, decision making was mainly vested in the father or older family members. These family members commonly played authoritative roles in decision making in healthcare. A complex interplay of cultural factors related to family could deprive patients of all their rights and thus reduced their levels of involvement in decision making in mental healthcare. For example:

"This is the power structure in a patient's family. If the family power type is an authoritative one, the father takes more power. And then he (a patient) can only listen to his father. This is possible. In this situation, no matter what he does, or tries to do against it, or how he tries to communicate (with his father), it is useless. Eventually, he can only listen to the decision maker in his family and follow his instructions." (23 F, Registered Nurse)

There were factors which drove this idea in the culture, including family responsibility for patients' behaviour and the need for family support. It was perceived that these family members should prevent the patient with mental illness from misbehaving. Also, family members were not able to simply abandon their responsibilities for the patients' behaviour. The strong sense of moral responsibility for patients with mental illness reinforced the power of family members. It was perceived that "whoever takes the responsibility makes the decisions." One specific example was given by a nurse, who explained that family members of one patient needed to respond in public to the murder committed by the patient.

"The family is too closed. Right, you should read the current news. XX (the killer on TV news and indicated as a patient with mental illness) is in his 20s. He is a legal adult in law. In the West, if someone commits a murder like that, his parents do not need to show up or do anything else. But in Chinese society, his parents needed to kneel in front of the public. At the beginning, they (the parents) did not do anything. They were attacked in the newspaper or community. It appeared that they have responsibility for this (behaviour). This (patient involvement) is really challenging." (21 F, Registered Nurse)

In particular, health professionals considered there was an obvious need for patients with mental illness to gain family support. They pointed out the importance of family involvement in decision making in order to be able to count on family support. There was widespread concern that if family members were not the main decision makers, the patients would lose all aspects of family support. Therefore, the health professionals believed it was important to take the family's voice into account during decision making. In most situations, from the health professionals' perspective, the family's voice was much more important than patient's. One example:

"In Taiwan, we (professionals) need to take a family as a unit. Without family intervention, in fact it is difficult for you to do anything (about patients). For example, we (decide) to transfer a patient to a halfway house. But the family does not support this... don't agree with this. They will take a patient out within one or two weeks." (09 M, Psychiatrist)

Similarly, health professionals were fearful of being criticised by family members, which might also explain their reluctance to engage in shared decision making. It was perceived that any conflict could bring a lack of family support, which could be detrimental to

patient recovery. One professional explained that there were consequences, such as poor treatment or care adherence, if families did not support patients. For example:

"Family members will question you in the end. Or they will criticise you: 'Why did you let the patient decide alone?' In order to protect ourselves (professionals), we need family permission. You know even if patients feel they do not need to stay here (in mental healthcare), but their family disagree, I think we wouldn't take the patient's opinion to discharge him. This must be agreed by family members, or the services or nurses won't be able to take their (patients') opinion." (22 F, Registered Nurse)

A few health professionals also noted that for patients to survive in the community their financial needs must be met. From their observation, family members provided financial support to patients. One professional explained that therefore, if there was any disagreement between patients and family members, the preference of the family members was more vital than the patients' as family members were the ones who paid the fees. This situation therefore enhanced the power status of family members; for example:

"It (patient involvement) is difficult in mental healthcare. First, this disease is a long-term condition. Maybe 40 or 50 years. Sometimes even 50 or 60 years. For example, some family members don't agree to patients being discharged from the hospital. If we (professionals) insist on doing so based on patients' wishes, their family members may say: 'How can they live without financial support from us? They don't have a home.' What can we do about this? Let them live on streets? You see what I mean." (21 F, Registered Nurse)

6.2.1.2 Sub-theme: Patients with impaired decisional ability due to mental illness

A clear sub-theme that was generated from the data was that health professionals fundamentally believed that patients had impaired decisional ability due to their mental illness. Health professionals were concerned about this impaired decisional capacity and therefore recognised it as a major barrier to shared decision making. As health professionals saw it, patients normally made unwise or irrational decisions which worsen their outcomes, such as refusing to take medication.

"Let patients decide?... If you ask me to do something I believe could worsen his/her outcome, I will definitely refuse. Some patients with mental illness don't have the ability to make decisions." (08 M, Psychiatrist)

Frequently professionals recognised cognitive deficits as a general problem for all patients with mental illness, but particularly patients with schizophrenia. From the professionals' perspective, cognitive impairments were critically associated with functional disability, particularly decisional functioning, which was a core characteristic of the illness. Cognitive impairment was reported frequently by patients in practice and therefore brought concerns to the health professionals' minds: *"whether patients are able to make the "right" decision*". One professional mentioned:

"For example, patients with schizophrenia, because their cognitive function and understanding facility are really poor, it is really hard...(to involve them)." (14 M, Psychiatrist)

Due to their concerns about cognitive deficits, the health professionals felt that patient involvement could have serious consequences. It appeared that involving patients in these medical decisions could be problematic if their preferred course of treatment contradicted a professional recommendation. For example:

"Because they normally ask to stop major medication. For example, one depressed patient doesn't want to take anti-depressive medication but asks for sleeping pills. Basically, I think the request is against the treatment guidelines and will lead to a depressive episode. I would turn them down." (09 M, Psychiatrist)

The health professionals believed that the more severe the symptoms an individual experienced the more decisional capacity was impaired. Many interviewees mentioned that shared decision making in acutely ill patients was nearly impossible. This might be even more complicated to implement as fears that the patients had limited ability were justified. One example:

"One of my patients, he has delusions and auditory hallucinations. In this situation, his opinion cannot be involved." (05 M, Psychiatrist)

These challenges focused on the patients' capacity for decision making, which could be adversely affected by psychosis. The concern was that decisions were made without taking adequate account of possible factors such as their ability to cooperate with a plan, the outcomes and risks associated with the decisions, or their social circumstances. One health professional described his experience of patient involvement and how the patient made a harmful decision due to psychiatric symptoms:

"Say one schizophrenia patient breaks his leg, and he refuses to accept treatment. And then the professionals follow his wishes as we need to respect his wishes? Don't treat him and let it be out of shape? In this case, he has delusions about surgery. Due to this illness, he cannot make this decision." (07 M, Psychiatrist)

By contrast, a few professionals considered that patients in stable remission were easier to involve in the process. They believed that those patients deemed "mentally stable" were more likely to be competent to choose from a variety of care plans or treatment options. Therefore, even if shared decision making was carried out, in practice the feasibility might depend on the severity of psychiatric symptoms or the phase of the illness. One professional addressed how different levels of psychiatric symptoms influenced patient involvement in decision making.

"For our patients, this (shared decision making) helps them to make autonomous decisions and (professionals) to respect their autonomous decisions. I think this idea is quite nice. For people in a stable condition, this is suitable. But acutely ill patients may refuse to take treatment. You still need to respect the wishes of this kind of patient? This is not suitable for them." (06 M, Psychiatrist)

The health professionals indicated that as a patient was mentally unwell then this clearly meant that they had no or limited decision making capacity. Poor insight was recognised by health professionals to be a standard indicator of patients' lack of awareness of their conditions. In this situation, health professionals felt that shared decision making might be diminished or absent entirely. From the professional views, these patients did not believe that they are suffering from mental illness and easily refused to take the recommended treatment and care. The health professionals described the patients as "having lack of insight" or "having poor insight", for example:

"On the other hand, some patients don't accept their mental illness. They don't understand their mental illness. It would probably be difficult to involve them in the decision-making process." (01 M, Psychiatrist) Consistent with above, lack of insight was believed by health professionals to lead to a high rate of medication non-adherence for patients in mental healthcare. In the interviews they often correlated lack of insight with poor medication adherence. They considered that there were various negative consequences of non-adherence for patients, including symptom exacerbation, worse functional performance, and re-hospitalisation, which damages patient outcomes. This resulted in a reduced desire on the part of professionals to engage patients in the decision-making process. One professional directly linked these factors and said:

"(Why patients are not involved)... because they have poor insight. (Patients think): "I don't have a disease. Why do I need to take medication". They (the patients) are discharged from hospitals but then don't get followed up in the outpatients department. And then they have an acute episode and call an ambulance. Because they experience repeat onsets and this brings really bad outcomes for them." (19 F, Registered Nurse)

6.2.1.3 Sub-theme: Health professionals lack understanding of shared decision making

Another key barrier to health professionals' willingness to engage in shared decision making was their lack of knowledge of shared decision making. Although there were various levels of understanding of shared decision making, most had low levels of understanding. More than half of the professionals had heard of shared decision making but were unable to correctly describe its principles and implementation. Some professionals had heard of the term but had forgotten the interpretation. For example:

"I can't remember... It was kind of a long time ago when I heard this term (shared decision making). It might be four or five years ago. I also read some related books or articles... I remember I read this term somewhere. But you ask me to explain this right now. I don't think I am able to explain it clearly." (05 M, Psychiatrist)

However, some professionals had not heard of shared decision making. One professional showed his lack of information and knowledge of shared decision making by repeatedly asking the author to define it:

"So... I think I don't know much more about the shared decision making tool. Or could you explain it to me in one minute. I don't really have the relevant information. I can only imagine what this would look like." (04 M, Psychiatrist)

A few professionals were able to understand and interpret the term, for example:

"I feel shared decision making is a way to meet patient needs. Based on their wishes, I believe it is kind of application of shared decision making. You (health professionals) provide some treatment options. And then patients to decide which one it is they prefer. Of course you have to explain the pros and cons of the options. (03 M, Psychiatrist)

From the professional views, the importance of understanding of all aspects of shared decision making was a key step for implementing it in healthcare. Although casual awareness did not guarantee familiarity with shared decision making and, critically, the ability to apply it correctly in practice, this was definitely the first step towards implementing shared decision making by health professionals. Conversely, if there was a lack of knowledge, it could be challenging to facilitate or implement it. One professional said that they must gain the relevant knowledge about shared decision making as the first step.

"I don't know what SDM is... I have to have some knowledge first. And then I can decide if I need this kind of training or not. If this model... could bring some benefits to patients and to me, I may want to understand. But first I should know what it is." (01 M, Psychiatrist)

Commonly, health professionals recognised training programmes were useful for collecting and organising information about shared decision making contexts. Although some professionals did not understand shared decision making well, they had still noticed that a few government activities had been developed to support patient involvement in decision making. Despite this, insufficient materials or training to put shared decision making into practice were widely reported by the health professionals. Lack of learning resources limited their desire to adopt shared decision making; for example:

"There might be some courses (about shared decision making), but I don't get the information. Perhaps there are some, but they're not really advocated... It is difficult for us (to adopt it)." (13 M, Psychiatrist) In addition, during the interviews, interestingly, some professionals agreed there was a lack of understanding of shared decision making but reported no motivation to learn more about it. Some health professionals seemed to believe that shared decision making was occurring in mental healthcare in Taiwan. In the minds of some healthcare professionals, it appeared that shared decision making had already been integrated into patient care, so there was no need to know more about it. One participant described how he thought about shared decision making, which he believed had been implemented already.

"In mental healthcare, I think we have already done this for such things as discharge planning. I do believe we (health professionals) are doing this. We just don't use data based evidence to do so." (12 F, Occupational therapist)

6.2.1.4 Sub-theme: Insufficient time

A further barrier which related to health professionals was insufficient time to put shared decision making into practice for patients. Commonly, health professionals believed that a clinical practice was a very busy place and there was a variety of knowledge which needed to be transferred. Given the time-poor environments of many healthcare settings, this was a valid concern. The working conditions for healthcare professionals were becoming increasingly stressful.

Although health professionals desired a more manageable workload, they felt the National Health Insurance made this impossible. Across the interviews, professionals, particularly doctors, told the interviewer that they had up to 100 patient contacts per day (and sometimes more than 100). They felt the number of patients they saw was excessive, with just 5-10 minutes for each appointment in some settings. It was perceived that this normally resulted in a lack of time in their working day to offer high quality communication with their patients. Consequently, health professionals perceived that time limitations were key factors in not implementing shared decision making.

"If you are working in an outpatient department, it could be difficult due to our National Health Insurance. Like me, you can only spend three hours finishing morning or afternoon appointments. You need to see 30 or 40 patients in one morning or afternoon session. That means you only can spend 5 minutes for per patient. 5 minutes! You know in this situation, we (professionals) don't have any opportunity to do it (shared decision making). So it is more efficient if we decide for them (patients)." (08 M, Psychiatrist) To address this in more detail, within the National Health Insurance, some even mentioned the strongest influence on professional income was the number of patient visits, for which health professionals were paid on a piecework basis. This meant the high earners needed to see a large number of patients every day. Some hospitals set up "sales targets" for the health professionals and expected to increase the "sales figures". As a higher level of service would need more money from the National Health Insurance, health professionals in higher-level practices would be required to see more patients.

"For example, the hospitals expect you have good sales figures. So you are not allowed to only see 3 or 4 patients. Or just 10 patients. No quantity means no profit, which is related to your income. Some professionals are even criticised by the hospitals. Maybe some new doctors do not have many patients. They may spend 40 minutes or 1 hour seeing a patient. But this is difficult. (04 M, Psychiatrist)

Due to high workloads the professionals felt there was no time for training or to implement shared decision making. For example:

"I think this (shared decision making) will only bring a burden and extra workload to mental healthcare. Because you have to have comprehensive knowledge about mental health resources. I have a lot of experience of many different settings, so I know. But others don't. If you want to implement it, they would have to undertake a lot of training. We are already too busy." (06 M, Psychiatrist)

Furthermore, the time-consuming shared decision making also added to the heavy workload of health professionals. Although most of them had little understanding of shared decision making, they recognised that it could be very time-consuming. Those professionals placed primary emphasis on providing sufficient information for patients, which should be the first step in shared decision making. In the professionals' view, the most time-consuming course regarding shared decision making was to enable patients to understand possible treatment options. One psychiatrist explained:

"Because it (shared decision making) needs a long time to explain. To be honest, this will take a long time to implement. Also there is a lot of work I need to do in the outpatient department. This is time-consuming." (09 M, Psychiatrist) Similarly, a few professionals believed that shared decision making might require the involvement of multiple internal professionals, which could be a big barrier in a busy health service context. One professional explained they therefore needed to set up group meetings which involved other mental health professionals such as nurses, psychologists, and social workers. There was a strong and widespread sense that shared decision making would inevitably take more time, which the health professionals did not have, and would stop health professionals implementing it. For example:

"I believe it (shared decision making) would be difficult to implement in practice. I suppose... Say you want to apply it in every case, getting different professions to work as a team. All the professionals come to discuss everything with the patients. I think time management would be really difficult. For example, doctors could not come as they have a lot of meetings, outpatient appointments and so on." (11 F, Occupational Therapist)

One interesting issue referred to the inappropriateness of adopting shared decision making in mental healthcare from the health professionals' perspectives. It was more likely that health professionals had concerns that more shared decision making across the mental healthcare continuum increased the demand for unnecessary interventions. This led some professionals to perceive that their workload would be significantly increased by implementing shared decision making and they would not be able to use their time in an efficient way. One professional explained:

"I think the medical environment in Taiwan is relatively special. So sometimes you want to introduce some new intervention into our society and this needs time to reshape the intervention. And you should be sure this is beneficial for health professionals. Because I think this is a mutual thing. This should be beneficial for both (patients and professionals) and then you adopt it. But if this is only beneficial for patients it is not necessary to implement it. It will bring a huge workload for health professionals." (10 M, Psychiatrist)

6.2.2 Theme 2: Facilitators of shared decision making

The interviews showed that shared decision making was not the norm in mental healthcare in Taiwan. In their accounts, the health professionals paid far more attention to the barriers to shared decision making than to possible facilitators: not every professional seemed to have the desire or the necessary understanding to improve or implement shared decision making.

Despite this, more than half of the professionals were aware that the implementation of shared decision making could be facilitated in some ways. Professionals noted that the lack of patient involvement in decision making might be problematic, and patient involvement might be positive for the health outcomes of some patients with mental illness. This awareness motivated health professionals to reflect on how decisions are shared in the current health system and to consider the possibility of shared decision making in mental healthcare. Therefore, two sub-themes were identified in the data: "awareness of patients' right to autonomy" and "understanding of the potential benefits of shared decision making".

6.2.2.1 Sub-theme: Awareness of patients' right to autonomy

The idea of patient autonomy was valued by some of the professionals. It was perceived that the principle of patient autonomy was usually associated with allowing or enabling patients to make their own decisions about which treatment or care interventions they will or will not receive.

Some professionals recognised the importance of patient autonomy and believed that patient involvement in decision making was needed in mental healthcare. With this awareness, health professionals were more likely to facilitate the implementation of shared decision making. There were two notions driving this motivation, including human rights over one's own body and uncertainty about treatment and care.

A few professionals reported that active patient involvement was a basic human right, and a key concern in biomedical ethics. With this perception, the professionals believed that patients were able to control their own bodies and lives rather than live under the control of others (health professionals or family members). The sense that patients should be offered options and allowed to make informed decisions about treatment or care interventions was crucial. Two examples:

"Because you own your body. You (patients) can decide how you want to use your body in the next decades. This body is mine. You should understand what will happen in the next decades if you decide to take this intervention or treatment. I believe this is the right thing to do." (12 F, Occupational Therapist)

"I hope they (patients) can be involved in decision making, because they should have the right to decide what treatment they want to receive." (17 F, Social Worker) In order to protect patient autonomy, one professional felt that health professionals should discourage inappropriate paternalism and protect patients from undesired intervention. The idea that patients should be enabled to make informed decisions rather than be forced was explained in the following example:

"It is not right to violate their human rights. It is just not right. Because they don't meet the criteria for compulsory admission. They have the right (to refuse to be admitted). You cannot do this." (18 M, Psychiatrist)

It was notable that some showed empathy with patients and experienced concern for them. One professional put herself in the 'patient role' to understand and address their patients' suffering. It appeared that empathy supported patients' autonomy and it also allowed the health professionals to self-reflect about patients participating in decision making. For example:

"If I was a patient (with mental illness), I suppose I wouldn't like someone else deciding for me. I would think I had been deprived of my liberty and dignity. Besides I could only follow the instructions of other people, including doctors or family members. It is just (better) decided by me. If I was a patient, I would think like that." (23 F, Registered Nurse)

Another driver of patient autonomy was uncertainty about health treatment and care. The idea there was uncertain or unknown evidence about the risks and benefit of treatment options and no clear standards of care was commonly reported by the health professionals to support the idea of shared decision making. Apparently, there was no exact right course or best medical choice for patients in practice from the professionals' perspectives. Although health professionals could easily access up-to-date knowledge about medication, a psychiatrist said that there is still no "perfect answer" in treatment options and patient preference could be a key in this. One specific example of the clinical uncertainty of medication was given by another health professional. He said:

"For example, he felt bad about Seroquel (Quetiapine). And he wants to change to Risperidone (Risperdal). That's fine. Just change it. Because to be honest with you - this is my personal view - even now the newest research cannot predict which medication is the most effective for patients. So why not (implement shared decision making)?" (09 M, Psychiatrist) Similarly, some therefore perceived that there was no perfect treatment and therefore the duty of health professionals was to communicate this uncertainty to patients rather just guiding them in clinical decisions. This led health professionals to see the potential of shared decision making in mental healthcare. Actively involving patients in the decision-making process was seen as more sensible in this situation. For example,

"There is no 100% correct answer for medicine. This is probably why it (shared decision making) might be implemented. Doctors might have better knowledge. But not all decisions are suitable for everyone. There might have been some (medical) mistakes, so a new policy (shared decision making) came out." (10 M, Psychiatrist)

6.2.2.2 Sub-theme: Understanding of potential benefits of shared decision making

Despite the findings described above, i.e. that most professionals did not involve patients in decision making; some professionals felt that there were potential benefits of shared decision making, including improving responsibility, motivation, and understanding of patients' conditions.

Some professionals described that active involvement in shared decision making could help patients with mental illness to understand their responsibility regarding important recovery goals in healthcare. They felt that encouraging patients to be involved in treatment decisions would allow patients to take responsibility for their treatment and care. They felt that greater patient control over their care and treatment would lead patients to feel empowered and relate positively to taking responsibility. Some professionals believed this effort could help to improve health outcomes, such as medication adherence. One professional described this as below:

"If today we have a child, he asks for something. And then we discuss with him and then decide together. It is better, isn't it? OR would he like to let the parents decide everything for him? I believe this is the same with the patients. Now we hope patients will decide to take their medication regularly and this is decided by us (health professionals) and the patients together. In this situation, responsibility plays a very important role. If you do it that way, the patients will take responsibility, you know? (05 M, Psychiatrist)

Conversely, as a health professional explained, sometimes health professionals purposely ignored the fact that patients were able to make decisions about their medical care,

particularly hospitalised patients or those who were seriously ill. This could bring a sense of who the main decision maker was, and thus who took all the responsibility. Both health professionals and patients wondered if they were capable of achieving any recovery goals or had no confidence in it. One professional mentioned what happened if health professionals tended to decide everything for patients.

"If I ignore their ability to make decisions about treatment and care, they could think: 'These (decisions) are all decided by doctors or other professionals'. It was like they didn't need to take any responsibility... Even if he is a psychosis patient, he could still decide. These things should be not decided by doctors... I should have noticed this (that the decisions were all made by professionals) but I didn't. It's as though the patient sees me as some sort of barrier to stop him recovering. I just see him as a child. And then this means he never ever takes responsibility." (14 M, Psychiatrist)

The importance of motivation was also pointed out by the professionals as having the potential to improve the health outcomes of patients with mental illness. Along with more engagement in decision making, the theme of patient motivation was identified as being frequently reported in the interviews. From the professionals' point of view, motivated patients could have more desire and energy to be continually committed to a treatment plan and their conditions were more inclined to improve. This was also described as an important determinant of rehabilitation outcome for mental healthcare. In more detail, motivation directly contributed to patient outcomes in terms of their demeanour and their compliance with a long-term rehabilitation plan. Two instances follow; one refers to accommodation and another is about medication treatment:

"I believe patients all want to have options, because they don't just go the place (possible accommodation) for one or two days. Sometimes it is more likely to be about their future - the rest of their life. If they can decide on their own and they like it, they could stay longer." (06 M, Psychiatrist)

"You let him decide about his medication. He has more motivation. He is more willing to take the medication." (03 M, Psychiatrist)

Furthermore, some of the health professionals had found that it was meaningless to force patients to follow instructions. If patients experienced compulsion, it appeared they had

no motivation to take more positive action in terms of their treatment or care. One occupational therapist reported:

"If he was forced or pushed, he didn't like (the intervention). We (health professionals) didn't really want this. For example, for occupational training, this is invalid and not effective. Patients without any motivation were forced. This is meaningless. They didn't want to stay." (11 F, Occupational Therapist)

A few professionals explained that there was an opportunity for health professionals to gain insight into exactly what happened to patients when they actively engaged in decision making. In this process, patients were supported in articulating their understanding of their medical conditions and also what they hoped treatment would achieve. Some professionals reported more generally that it could help them to gain a more precise understanding of the patients' conditions. Therefore, health professionals felt more confident in supporting patients in treatment or care plans. One professional mentioned:

"If patients have a full discussion with me, I feel more confident in the treatment. For example, the patient discussed all the details about how he took Lithium before. And then we decided he would take Lithium next. I did not really have to worry about side effects. I could have more information about the patient. And this helps me to have more confidence. And therefore in the decision-making process, I think patient involvement is positive." (02 F, Psychiatrist)

6.3 Strengths and limitations

The findings of this study have provided insight into how mental health professionals perceive shared decision making in Taiwan. To the author's knowledge, this is the first study that has explored health professional perceptions on shared decision making in Taiwan. Therefore, this study could have the potential to support and build further knowledge to provide in-depth descriptions of specific shared decision making interventions and relevant training, as well as the facilitators of and barriers to implementing shared decision making in healthcare, particularly in mental healthcare.

However, a number of limitations of this study are identified: for example, the study involved only healthcare professionals in Taiwan. There could be some differences in healthcare across different countries and this may influence how health professionals perceive the decision-making process in practice. A further limitation is that there is a strong possibility of sample selection and recruitment biases as it may be that the professionals who participate in studies are more interested or enthusiastic about shared decision making and therefore not representative of health professionals in general. In particular, the number of mental health nurses recruited was lower than expected. Data on who refused to participate and who withdrew from the study was not collected, thus not allowing comparison between the sample and the general population of mental health professionals meeting the including criteria.

Also, this study only collected perspectives about healthcare professionals' attitudes and self-reported behaviours; objective observation of healthcare professionals actually interacting with patients was not used in this study. Thus, it may be difficult to draw a firm conclusion about whether the situations described were a true reflection of reality. In order to fully explore the dynamics of decision making, observation research would be useful to examine the decision-making process in mental healthcare in real time. In spite of these limitations, the findings have provided a unique opportunity for health professionals to offer some insight towards re-evaluating shared decision making in mental healthcare.

6.4 Summary

This chapter has provided the findings of how health professionals perceive shared decision making in mental healthcare. The findings indicated that paternalism was still widespread in mental healthcare and the absence of patient involvement in decision making was acceptable to the mental health professionals. Two main themes were generated from the mental health professionals' accounts: barriers to shared decision making and facilitators of shared decision making.

Although health professionals' opinions varied based on their experiences of decision making in mental healthcare, most health professionals felt there were various difficulties related to implementing shared decision making. Numerous concerns about patients taking a more active role in decision making were identified from the data. The health professionals perceived that decision making was their duty as health professionals, and that they were more qualified to make decisions about treatment and care. Interestingly, the importance of family involvement was also pointed out by the health professionals. Due to the moral responsibility to take care of patients with mental illness, families held higher status than patients. Therefore, it was perceived that the family should be the proxy for patients in decision making. Also, concerns about patients' impaired decisional ability were recognised to have negative impacts on active patient involvement in the decision-making process. Additionally, health professionals had limited knowledge about shared decision making and therefore they were unable to implement it. Finally, the negative perception of limited time resources also stopped the implementation of shared decision making.

Conversely, some health professionals realised the importance of patient involvement in decision making and believed that shared decision making could bring benefits to patient outcome and recovery goals. These could motivate health professionals to implement shared decision making in mental healthcare. Some health professionals believed patient involvement contributes to autonomy and is a basic right for patients. As a matter of ethical principle, the health professionals felt that every patient should have rights over their own bodies and make decisions about their bodies. Shared decision making was seen as a way to make this possible. Additionally, it was highlighted by health professionals that shared decision making could help patients to engage in their treatment and care and therefore have potential benefits for patient outcomes.

Chapter Seven: Synthesis

This chapter will provide details of the process and outcome of the synthesis of research related to patients' and health professionals' perspectives of shared decision making. It aims to synthesise barriers to and facilitators of shared decision making from the three studies in this thesis, to extract evidence-based mechanisms through which the process of decision making may be enhanced, and to identify possible solutions to overcome these barriers.

7.1 Synthesis methods

7.1.1 Evidence resources

There were three sources of data from which evidence was synthesised:

- a systematic review of qualitative studies of patient perspectives of shared decision making in secondary mental healthcare (detailed in Chapter Three)
- qualitative interviews with patients to explore their perspectives of how decision making is shared in secondary mental healthcare in Taiwan (detailed in Chapter Five)
- qualitative interviews with health professionals to explore their perspectives of shared decision making in secondary mental healthcare in Taiwan (detailed in Chapter Six).

7.1.2 Synthesis process

Firstly, the author reviewed all transcriptions of both health professional interviews and patient interviews to become familiar with the data again. Following this, a summary table of background information and the findings from each evidence source was created (Appendix 16). All findings of each study were extracted in the summary table and reread to identify all potential mechanisms through which patients' and health professionals' attitudes and actions could improve patient involvement in decision making. All three evidence sources were equally weighted by the author and all potential improvement mechanisms found were taken into account. The mechanisms initially identified are presented in Table 7.1.

Next, the author re-read all the evidence and the table of extracted improvement mechanisms. The visualisation of these data helped the author to gain an overview of the findings and to review the applicability of the improvement mechanisms. Initial synthesis outcomes and solutions identified from the data were collapsed or combined into a final list of improvement mechanisms. Following completion of data extraction, a narrative

summary of each improvement mechanism was produced, which included the barriers, facilitators and potential solutions (Table 7.2).

	Qualitative review	Patient interviews	Professional interviews	Synthesis	Possible Solutions
Misuse of restrictive/involuntary interventions to force patients to follow instructions	Barriers Professionals paternalistic attitude towards decision- making Facilitators The importance of professionals adopting a supportive attitude	Barriers Powerful status of health professionals and paternalist relationship between health professionals and patients Clear evidence of misuse of compulsory and involuntary interventions Seeing restrictive/involuntary interventions as a consequence of not following instructions of health professionals Verbal threats of being hospitalised from health professionals and families	Barriers Belief that health professionals are more qualified to be the main decision-makers was widely reported Failure to use involuntary interventions as the last resort Facilitators Sense of patient autonomy enhances patient involvement	Both health professionals and patients confirm that health professional hold paternalistic attitude toward patients in decision making Patients consistently reported health professionals used restrictive/involuntary interventions to force them to comply with instructions Patients and health professionals realised the importance of positive attitude of patients and health professionals	 Approaches which are supportive of shared decision making should be applied in mental healthcare. Training could include Emphasis on promoting patients' rights to autonomy Emphasis on methods to promote active patient involvement in decision-making Opportunities to reflect on attitudes and behaviour not consistent with sharing decisions Providing a working definition of restrictive/involuntary interventions could provide greater clarity for health professionals around when it is acceptable to use these interventions, in order to avoid their misuse as punishments/sanctions. Health professionals and patients should be given opportunity to discuss causes of power imbalance in decision making and influence of each parties.
Health professionals and patients do not	Barriers	Barriers	Barriers	Both patients and health professionals confirm that they have	Training for incorporating evidence individualised to patients' unique

Table 7.1 Summary of initial synthesis outcomes and possible solutions

promote patient decisional capability	Patients with a limited professional knowledge levels	Widespread reporting of having no confidence to make decisions about their treatment and care due to their mental illness Negative perception of decisional capability due to their mental illness Clear evidence of sense of "health professional knows better"	Belief that patients with impaired decisional capability were not able to make decisions Information asymmetry between health professionals and patients was identify by mental health professionals	no confidence in the decisional capability of patients. Patients and health professionals identify patients with limited professional knowledge, which could prevent them being involved.	circumstances and presented in a format they can understand. Optimising means of delivering professional information, such as decision aids, flyers and websites developed by health professionals.
Limited knowledge of shared decision making-Professionals	N/A	N/A	Barriers Limited knowledge of implementation of shared decision making from health professionals. Facilitators The importance of understanding of benefit of shared decision making had been identified	Health professionals identified limited knowledge about shared decision making as a barrier to implementation of it.	Increased understanding of potential benefits of patient involvement and shared decision making. Training in the knowledge and principles of shared decision making
Health professionals are unable to provide sufficient time	Barriers Poor quality of communication due to	Barriers Clear evidence that health professionals provided	Barriers Widespread reporting of heavy workloads which	Both health professional and patients confirm that environment in mental	Optimising work flow for and demands on health professionals

limited time was reported by patients.	limited time with patients: no privacy, no attention, and not enough time.	left no room for health professionals to have time to involve patients in decision making. Due to the NHI system, health professionals need to see a lot patients every day	healthcare is time- poor.	 simplify health records and working systems in-depth discussion about the pressure from health insurers to keep patient visits short and the increasing number of patients Support tools and resources shared decision making tools
---	--	--	------------------------------	--

Table 7.2 Synthesis summary	of improvement mechanisms
-----------------------------	---------------------------

Improvement mechanism	Barriers	Facilitators	Possible solutions			
Capability/skill improvement						
Increasing patient decisional capability	Both patients and health professional confirm that they have no confidence in the decisional capability of patients (patient interviews and health professional interviews) Patients and health professionals say that patients' limited professional knowledge could prevent them being involved (systematic review, patient interviews, and health professional interviews)	Clear evidence of sufficient information exchange, including professional knowledge and personal preference facilitate patient involvement (systematic review and patient interviews)	 Step one: understanding patients' information need Training for health professionals reflecting that they are essential sources of health advice that patients access Step two: Offering information in the way patients can understand Optimising means of delivering professional information, such as decision aids, flyers and websites developed by health professionals. Training for incorporating evidence individualised to patients' unique circumstances and presented in a format they can understand. Step three: offering opportunity for patients to reflect on their 			
Improving professional knowledge levels of shared decision making	Health professionals identified limited knowledge about shared decision making as a barrier to implementing it (health professional interviews)	The importance of understanding the benefit of shared decision making was identified (health professional interviews)	Increased understanding of potential benefits of patient involvement in shared decision making			

			Training about the principles of shared
			decision making:
			definition
			evidence
			implementation
			effectiveness/outcome
			Increased flexibility for training in
			shared decision making through
			case studies
			 workshops
			• seminars
			online courses
Attitude/motivation modifica	tion and improvement		
Attitude modification of	Both health professionals and patients	Patients and health professionals realised	Approaches which are supportive of
health professionals about	confirm that health professionals hold	the importance of positive attitude of	shared decision making should be
patient involvement	paternalistic attitudes toward patients in	patients and health professionals	applied in mental healthcare.
	decision making (systematic review,	(systematic review, patient interviews,	Training should encourage health
	patient interviews, and health	and health professional interviews)	professionals to reflect on their
	professional interviews)		paternalist attitude and behaviour
	Patients consistently reported health		Awareness of patients' right to
	professionals used restrictive/involuntary		autonomy should be stressed in mental
	interventions to force them to comply with		health education program or
	instructions (patient interviews)		professional training
			I raining should help health
			professionals to understand now to
			decision making
			Health professionals and patients
			should be given the opportunity to
			discuss the causes of power imbalance
			uiscuss me causes of power mibalance

			in decision making and the influence of
Attitudinal modification of use of involuntary intervention	Clear evidence of misuse of compulsory and involuntary interventions (patient interview) Seeing restrictive/involuntary interventions as a consequence of not following the instructions of health professionals (patient interviews and health professional interviews) Verbal threats of being hospitalised from health professionals and families (patient interviews)	N/A	each party The purposes and working definition of applying restrictive/involuntary interventions should be clearly identified as being to help patients to regain control rather than consequences or punishment for patients. Mental health professionals should understand the negative impacts of restrictive/involuntary interventions on patients Respectful communication strategies
			during the interventions
Limited time resources impro	ovement		
Using limited time efficiently	Both health professionals and patients confirm that the environment in mental healthcare is time-poor (patient interview and health professional interviews)	Clear evidence that sufficient information exchange improves patient involvement levels (systematic review and patient interviews)	 Optimising work flow of and demands on health professionals simplify health records and working systems in-depth discussion about the pressure from health insurers to keep patient visits short and the increasing number of patients Support tools and resources shared decision making tools

7.2 Synthesis outcomes

Three improvement mechanisms were identified across the three evidence sources (Table 7.2). Based on these, implementation of shared decision making could be improved through: capability/skill improvement, attitude/motivation modification and improvement, and limited time resources improvement. The following sections will present an in-depth discussion of the synthesis outcomes and possible solutions for overcoming barriers to the improvement mechanisms.

7.2.1 Capability/skill improvement

7.2.1.1 Increasing patient decisional capability

Synthesis outcome

It was evident in both patient and professional interviews that there was little confidence in patients' decisional capacity. Most participants (both health professionals and patients) felt that patients with mental illness suffered from impaired decisional capacity and difficulty in absorbing knowledge about their treatment and care in order to make rational decisions.

Health professionals were concerned that patients with mental illness, who generally lacked insight and were suffering from psychotic symptoms, would make unreasonable or unwise decisions which might worsen their health outcomes or jeopardise their recovery goals, such as refusing to take medication and thus causing relapse. It was assumed by health professionals that patients did not have the capability to take an active role in decision making. Therefore, health professionals, whose reason was not impaired, took responsibility to decide for patients with mental illness.

Patients who were interviewed and the patients' views in the systematic review expressed that they had no confidence to participate in the decision-making process due to limited knowledge levels and long-term mental health conditions, such as hallucination and delusion. Therefore, there was clear evidence that patients had negative perceptions of their capability to understand the information provided by others and underestimated their knowledge gain relative to that of health professionals. Thus, it was perceived that the decisions should be left to the experts who own the knowledge.

Possible solutions

The synthesised data indicated that efforts to promote mutual communication and sufficient information exchange during decision making may enhance the implementation of shared decision making. Patients could gain more understanding of their treatment and care and therefore enhance their confidence to participate in decision making (Clarke et al., 2016; Joseph-Williams et al., 2014a). There are three steps to improve patient knowledge levels. Step one: understanding patients' information needs; step two: offering information in a way patients can understand; step three: offering an opportunity for patients to reflect on their understanding of the information (Clarke et al., 2016). Detailed discussions of each step will be presented in the following sections.

Step one: Understanding patient information needs

It was evident that the patients often reported that they desired more information than they received. While health professional participants felt that they spent a substantial proportion of time providing information and education to patients, the patient data contradicted this as they described not receiving information, or receiving unclear information and knowledge about their illness or treatment.

Because of the perception among patients that seeking more information about their health was critical, and that the information should be tailored to their specific clinical conditions, it may be essential to promote an ethos consistent with the exploration of patient information needs. Understanding what information patients really need could therefore provide greater clarity for health professionals about what information and knowledge should be delivered in order to increase patients' limited knowledge levels and meet their information needs.

Step two: Offering information in a way patients can understand

Sharing information appropriately was recognised as essential to providing effective care for patients in both the patient interviews and the systematic review. Although the health professionals agreed with the importance of patient education, most patients – by their accounts – had difficulties in understanding the information given to them. Specifically, both health professionals and patients had some concerns about patients' capability to absorb professional knowledge. Therefore, it seems essential to find an appropriate approach to delivering medical information to improve patient knowledge levels.

It was evident that patients' perception of the ways health professionals deliver information and knowledge was negative. The information was often partly understood, misunderstood, or misinterpreted. It could be important to look at knowledge delivery through the patients' eyes. Assessment of patients' capacity to understand specific healthcare information could be important for health professionals in deciding how to provide the information. A large amount of knowledge needs to be covered in order to present integrated clinical information in diverse health settings, and the use of 'bite size chunks' or a 'lay' written leaflet or website would be appropriate options to ensure patients receive understandable, accurate and evidence-based information. Patient education is effective when patients receive accurate, timely, and unambiguous information from health professionals in a way they are able to understand (Marcus, 2014) and then patients are enabled to participate responsibly in decision making.

Step three: Offering opportunities for patients to reflect on their understanding of the information received

The patient data was replete with evidence of health professionals' inability to provide opportunities for patients and health professionals to communicate fully in medical encounters. Examples were given of professionals persistently failing to ask or to understand what information patients needed and whether patients had gained sufficient knowledge to take part in decision making. Furthermore, after being provided with a little knowledge or information about their treatment and care, patients felt that health professionals then attempted to ignore their further requirements or questions. This was particularly true for patients who had not been professionally trained, and only partially understood – and quickly forgot – the information they had received. As they pay more attention to processing the information they are given about a new diagnosis, a new treatment, or a new medication, patients may need an opportunity to review all the details with their health professionals and then clarify where they require further information from them.

In addition, although the health professionals believed they had spent time during the encounters verbally providing medical education, the patients' perception was that health professionals, particularly psychiatrists, did not spend sufficient time with them. In the interviews, both health professionals and patients agreed that there was a large amount of medical information which needed time to be thoroughly explored and understood. As the patients described it, the quality of the medical encounters is often seen as time-poor:

174

patients rarely had sufficient time to formulate their questions in the outpatient appointments. For these reasons, consideration must be given to how to improve the time-poor environment, which is also discussed in Section 7.2.3, and then the effectiveness of the educational process might be significantly enhanced.

7.2.1.2 Improving professionals' knowledge of shared decision making <u>Synthesis outcome</u>

The health professionals recognised that a potential barrier to implementing shared decision making was their limited knowledge about it. Although health professionals' knowledge levels about shared decision making ranged from 'never heard of it' to 'fully understand it', most did not have a clear understanding of shared decision making. Even for those health professionals who felt that there were benefits to shared decision making, their overall knowledge, conceptualisation and practical implementation were limited.

Factors identified by the health professionals as contributing to this included a lack of learning materials or training, no motivation to understand, and limited time resources. Although the government had made some efforts to promote the implementation of shared decision making, limited learning materials and training for health professionals was recognised as a major barrier. Furthermore, because of limited knowledge about it, a few health professionals felt that shared decision making had already been implemented in mental healthcare, or was not suitable for patients with mental illness. Therefore, they were unable to be motivated to learn about the new idea.

Possible solutions

As described earlier, the health professionals strongly indicated that even if shared decision making could bring benefit to their practice, as a first step they needed to gain knowledge about shared decision making. Some barriers should be taken into account in developing the education programmes and training for health professionals.

The importance of learning materials and training was identified by all the health professionals. As some health professionals mistrusted shared decision making as a norm in mental healthcare, they felt there was no need to learn more about it. However, a few essentials that the health professionals mentioned in the interviews should be covered in this training, including what shared decision making is, how to implement it and what its benefits are. These could help health professionals to clarify the real situation of making decisions in mental healthcare. Meanwhile, because of health professionals' busy

programmes, it may be important to consider the flexibility of training for mental health professionals. In order to enhance accessibility of this training, some other forms of training should be also considered, such as online courses.

In addition, it may be worth considering that a few health professionals indicated that decision making would be a process of communication between health professionals and patients. Knowledge of effective communication is important when exploring what type of relationship with clinicians could maximise patient involvement in their own health and care, and how this might be reflected in medical education and training. This training and these learning materials would support health professionals in adopting shared decision making and information giving. Some communication skills training may be needed to make shared decision making possible in mental healthcare.

7.2.2 Attitude/motivation modification and improvement

7.2.2.1 Modification of health professionals' paternalistic attitude to patient involvement

<u>Synthesis outcome</u>

Across the evidence sources, both health professionals and patients confirmed a widespread paternalistic attitude to patients amongst health professionals. It was widely practised because health professionals were usually expected to make the best decision for the patient. Power imbalance between health professionals and patients was significant in the decision-making process and has existed for a long time in all healthcare fields (Joseph-Williams et al., 2014a).

There was good evidence that these attitudinal barriers were delaying progress in implementing shared decision making. There were natural duties and roles for health professionals and patients; health professionals should be the ones to make decisions and patients should submissively accept a silent role during the process. It was more rational that decisions should be left to health professionals, who were more qualified to decide. Therefore, the dominance of health professionals and the passivity of patients were commonly adopted roles in healthcare.

Another potential barrier that exacerbated the power imbalance between health professionals and patients was that the power to order hospitalisation and other involuntary and restraint interventions was held only by the health professionals. Patients, therefore, were afraid of speaking up and felt there would be consequences if they refused to follow instructions. These barriers will be discussed in depth in Section 7.2.2.2.

Interestingly, the importance of family involvement was also recognised in both patient and health professional interviews. Both groups also believed that elderly, male family members had a dominant role in decision making, usually more than patients. Meanwhile, as families had a moral responsibility for patients, the family held much more power in mental healthcare. Thus, many participants argued that the dominance and decision-making authority of families in society in general was reflected in mental healthcare.

Possible solutions

Unsurprisingly, having sufficient knowledge and information is key for patient involvement in decision making. There was a natural knowledge gap between health professionals and patients. Therefore, improvement of patient knowledge levels could be helpful to improve the power imbalance in the health professional-patient relationship. Some possible solutions for the knowledge gap have been discussed in the previous sections.

However, knowledge alone may be insufficient to change this trend of paternalistic relationships between health professionals and patients. Attitudinal barriers from health professionals were widely reported in the studies as preventing patients being involved in decision making. There was a persistent belief amongst the professionals that they were more qualified to make decisions, and their failure to involve patients in decision making, such as refusing to explain the rationale for treatment and intervention, not responding to patients' requirements, or when they expressed their preference about treatment and care, could be a barrier to shared decision making: not only could it cause increased patient passivity, it could also erode health professional-patient relationships.

For these reasons, the importance of supportive approaches in decision making should be stressed in professional (medical and nursing) education and training. Meanwhile, implementation of shared decision making training for health professionals should consider focusing on communication skills training in decision making, including sharing information, deliberation and making final decisions (The Health Foundation, 2012b). Training options to meet these needs are explored by discussion; a series of meetings would need to be held in order to thoroughly explore their incorporation in implementing shared decision making (The Health Foundation, 2012a).

In order to reduce the dominance of close family and other relatives, respecting patient autonomy could be important for health professionals to consider. Understanding patients' cultural and family backgrounds could also be a key to developing proper communication strategies with patients and family members. Also, careful observation of family interaction is critical to ensure patients make decisions based on their personal preference and desire rather than coercion by the family.

7.2.2.2 Attitudinal modification regarding use of involuntary intervention <u>Synthesis outcome</u>

The misuse of involuntary interventions, such as restraint, seclusion, and injected medication, was evident in both the patient and the health professional studies. All patients in the studies had experienced being hospitalised or other involuntary interventions. They often felt powerless and hopeless during these experiences and some of them saw the experience as traumatic. Therefore, patients tended to submissively accept requests from health professionals and act in the way they thought health professionals would like.

Although both groups believed that health professionals' power was reinforced in this way, patients strongly perceived this power as a threat in the decision-making process. Across all the patients, during hospitalisation and involuntary intervention their voice was often ignored by the health professionals. In order to keep patients passive in decision making, both health professionals and patients described that health professionals tended to use involuntary intervention as a threat to force patients to follow their instructions. In some examples, patients described how health professionals used involuntary intervention as "punishment" if they did not listen. Intense fear and feelings of being threatened were consistently widespread in the patients' accounts.

Possible solutions

The purpose of involuntary interventions and hospitalisation should be clearly stressed as a way to help people to regain control rather than as a tool for forcing them to follow instructions. The use of these interventions should only be adopted when other preventive strategies, such as de-escalation techniques, have failed (NICE, 2017). Owing to the perception among health professionals that coercion is an effective way to control patients, an ethos consistent with the development of a definition may be needed, to provide an alternative way of offering less coercive methods and routines on which to depend to manage patients in the various settings. Therefore, a clear working definition of these interventions and criteria for hospitalisation should be developed for health professionals in practice. With these clear definitions, health professionals could clarify the proper situations in which to apply these interventions and how to use them (Horsburgh, 2004). Furthermore, promotion of rapid deployment of other possible interventions and the minimisation of coercion could also be helpful to ensure equality of patient and health professional roles.

Even if, in some situations, involuntary interventions or hospitalisation are inevitable, health professionals should use more respectful communication strategies during intervention. It may be useful for health professionals to explore patients' views on the situation and discuss their needs and their concerns, offering sufficient time for patients to respond and have the opportunity to speak up. Seeing patients as the central focus of care is key to communicating with patients. Some training in communication skills may help health professionals to provide higher quality care.

7.2.3 Limited time resources improvement

7.2.3.1 Using limited time efficiently

Synthesis outcome

The issue that health professionals had limited time with patients was recognised to prevent effective communication between health professionals and patients. The patient data strongly indicated the poor quality of medical encounters with mental health professionals. For example, some patients complained that there was no privacy in the clinical room at all, as the health professional asked more than two patients to wait in the same room, in order to see as many patients as possible. Similarly, both health professional and patient groups confirmed that patients normally had only five to ten minutes with their health professionals in each medical encounter. The issue of limited time resources was identified by both parties as impeding shared decision making.

Furthermore, the causes of professionals failing to provide enough time for patients were discussed in both studies. Patients paid attention only to the numbers of patients that health professionals needed to see each day, which they used to explain why the health professionals were always busy, whereas health professionals provided more insight into the reasons why they needed to see a large number of patients in one day. The health

professionals said that the National Health Insurance system required them to increase their "sales figures" in the practice. The numbers of patients they saw therefore had significant influence on the hospitals' and clinics' revenues from the National Health Insurance, as well as their personal incomes. Accordingly, they must complete patient visits efficiently and quickly.

Another barrier to shared decision making was concern about its time-consuming nature. Although health professionals did not have a full understanding of shared decision making, the data showed a widespread belief that it was time-consuming. Concern that implementing shared decision making would bring increased workloads was a potential barrier for health professionals. For these reasons, they were reluctant to expedite shared decision making.

Possible solutions

The importance of how to use limited time could be a key to improving and facilitating shared decision making in mental healthcare. However, in the data, the health professionals often struggled with high workloads. Therefore, optimising work flow for and demands on health professionals may be a key to help health professionals to use their limited time efficiently. For example, complicated medical records and working systems could be made simpler and more flexible. Thus, increasing the availability of health professionals for patients may improve the quality of medical encounters.

It also worth clarifying the belief that shared decision making is time consuming. Although some studies believe that shared decision making needs time to perform in practice (Ford et al., 2002; Légaré et al., 2008), there is also evidence that implementation of shared decision making does not need extra time (Stacey et al., 2014). Therefore, it may be important to gain more evidence to decide if shared decision making requires more time in practice.

While it could be challenging, some system-level efforts could be needed to improve the often time-poor environment. The significant stressor that health professionals need to see a large numbers of patients each day needs to be improved. Therefore, an in-depth discussion is needed within the health system about the pressure from health insurers to keep patient visits short in order to increase the number of patients seen. Striking a balance between demand and workloads could be a key to dealing with this challenge.
Therefore, the authorities should carefully consider the negative impacts of this pressure and review the work flow for health professionals.

In addition, due to the concern about the time-consuming nature of shared decision making, it may be worth introducing some support tools and resources for shared decision making. For instance, information in a variety of formats, such as flyers and booklets, could be used to help patients to gain the knowledge to participate in decision making (Drug and Therapeutics Bulletin, 2013). However, some patients might still prefer to receive material about their specific circumstances that health professionals recommend (Clarke et al., 2016). Thus, the process of knowledge delivery could be carried out using alternative means such as online services or the telephone. Additionally, information from support groups or internet-based discussion groups could be also helpful, to enable patients to learn from others with similar medical conditions. These resources would give patients more flexible ways to exchange information and increase knowledge levels about treatment and care.

7.3 Summary

This chapter has presented a synthesis of one qualitative systematic review and two qualitative interview studies (patient interviews and health professional interviews). The findings of the synthesis have provided a detailed discussion on the decision-making process in secondary mental healthcare in Taiwan, and how health professionals and patients felt about the process. Three main improvement mechanisms were identified: capability/skill improvement, attitude/motivation modification improvement, and limited time resources improvement. First, the knowledge levels of both health professionals and patients should be improved to ensure they are able to undertake shared decision making. Second, attitudinal modification regarding paternalism could also be a key to facilitate shared decision making. It could be important for health professionals to critically review and reflect on their part in the decision-making process and then provide proper support to enable patients with mental illness to actively participate in decision making. Finally, the importance of improving the time-poor environment was identified by both groups. Although some systemic efforts need to be made, optimising work flows and introducing support tools may also help health professionals use their limited time more efficiently. With these three improvement mechanisms, the implementation of shared decision making could progress in mental healthcare. The next chapter will discuss the findings of the synthesis, its implications, and the direction of further studies.

Chapter Eight: Discussion

This chapter will provide a detailed discussion of this thesis. The first section of this chapter will present a summary of the main findings in relation to the aims and objectives of the thesis. The second section will provide a discussion of the strengths and weaknesses of the synthesis. The third section will review the findings and compares them with the current literature. Following this, the fourth section will discuss possible further studies in this area. Finally, the conclusion will be presented.

8.1 Summary of key findings

8.1.1 Qualitative systematic review (Study One, Chapter Three, pg. 65)

Aim: To examine and synthesise perspectives of shared decision making amongst patients in secondary mental healthcare.

Methods: Using a systematic search strategy, the meta-synthesis identified qualitative studies exploring patient perspectives of shared decision making in secondary mental healthcare. The review followed the principle of thematic synthesis, as outlined by Thomas and Harden (2008).

Findings: 13 studies were included in the synthesis. Patients in mental healthcare played passive roles in the decision-making process. There were two main themes: "barriers to shared decision making" and "facilitators of shared decision making".

Barriers

- professionals with a paternalistic attitude
- poor quality of interaction with health professionals
- patients lacking professional information and knowledge

Mental health professionals' paternalistic attitudes prevented patients being actively involved in the decision-making process. Another barrier to shared decision making recognised by patients was the poor quality of interaction with health professionals Furthermore, patients were concerned about their limited professional knowledge preventing them from participating in decision making.

Facilitators

- supportive attitude to patient involvement
- sufficient information exchange
- other support resources

A supportive attitude from mental health professionals could facilitate the implementation of shared decision making. Sufficient information exchange was also reported by the patients as being a significant benefit to implementing shared decision making. Furthermore, some patients also felt that support resources, including decision aids, could help them to actively engage in decision making.

8.1.2 Patient interviews (Study Two, Chapter Five, pg. 120)

Aim: To explore patient perspectives of shared decision making in secondary mental healthcare in Taiwan.

Methods: Semi-structured qualitative interviews were used. Interviews were conducted with 20 patients sampled purposively and recruited from three halfway houses in Taiwan. The interviews were analysed using thematic analysis (Braun and Clarke, 2006).

Findings: 20 patients were included in this study. Patients in mental healthcare in Taiwan were not involved in the decision-making process. Two main themes were identified: barriers to and facilitators of shared decision making.

Barriers

- the professional status of health professionals in a submissive culture
- a negative perception of making decisions
- limited time resources

As the health professionals had power and status, patients perceived that decision making should be left to those who were more qualified, i.e. the health professionals. Patients had negative perceptions of their ability to make decisions and therefore did not have the confidence to be involved in the decision-making process. Furthermore, a lack of sufficient and quality time with health professionals, particularly psychiatrists, prevented patients expressing their deeper feelings, and therefore had a negative impact on patient involvement in decision making.

Facilitators

- sufficient information exchange
- desire to make decisions

Patient participants wanted to gain sufficient information to improve their level of involvement in the process. One interesting finding regarding the need to make final decisions was that some patients preferred to make the decisions themselves, but some

tended to leave final decisions to the health professionals alone. Although not everyone attempted to make their own final decisions, all patients reported a clear need for more active patient involvement in decision making.

8.1.3 Health professional interviews (Study Three, Chapter Six, pg. 140)

Aim: To explore health professionals' perspectives of shared decision making in secondary mental healthcare in Taiwan.

Methods: Semi-structured qualitative interviews were employed. The semi-structured interviews were analysed using thematic analysis (Braun and Clarke, 2006).

Findings: 24 mental health professionals, purposively sampled from secondary mental healthcare in Taiwan, were individually interviewed. Overall, paternalism was still widespread in mental healthcare and the absence of patient involvement in decision making was common, according to the mental health professionals. Two main themes were generated: "barriers to shared decision making" and "facilitators of shared decision making".

Barriers

- powerful status of health professionals and families
- patients with impaired decisional ability due to mental illness
- health professionals lack understanding of shared decision making
- insufficient time

From the health professionals' perspective, decision making about treatment and care was a natural responsibility of being a health professional. Interestingly, the powerful status of not just health professionals but also families negatively influenced active patient involvement. Additionally, patients' impaired decisional ability also brought some concerns to health professionals. Meanwhile, there was a significant lack of knowledge about shared decision making amongst health professionals, leading to them being unable to implement it. Another barrier was that health professionals perceived that they struggled to spend sufficient time with patients, and therefore felt that shared decision making would be impossible to implement.

Facilitators

- awareness of patients' right to autonomy
- understanding of potential benefits of shared decision making

Although health professionals focused more on the barriers, some were aware of significant potential benefits of patient involvement in decision making, and were more motivated to facilitate shared decision making in practice. It was perceived that active patient involvement is an approach to ensure patient autonomy. Furthermore, the health professionals also discussed the importance of shared decisions in enabling patients to engage more in their treatment and care, which would then bring potential benefits to patient outcomes. Therefore, they were more willing to adopt shared decision making in practice.

8.1.4 Synthesis of the studies (Chapter Seven, pg. 164)

Aim: To synthesise barriers to/facilitators of shared decision making from the three studies in this the thesis, to extract evidence-based mechanisms through which the process of decision making may be enhanced, and to identify possible solutions to overcome these barriers.

Methods: A synthesis of the findings from the qualitative systematic review, patient interviews, and health professional interviews was presented in Chapter Seven.

Findings: Although patients desired shared decision making, it has not yet been implemented in mental healthcare in Taiwan. Three types of barriers were identified: capability/skills barriers, attitude/motivation barriers, and limited time resources. Three improvement mechanisms were identified:

Capability/skill improvement

- increasing patients' decisional capability
- improving professionals' knowledge of shared decision making

Concerns that patients with mental illness might have impaired decisional capacity were widespread in the synthesis. Therefore, it was perceived by all the groups that patients had difficulty absorbing knowledge about their treatment and care in order to make rational decisions. To overcome this, increasing professional knowledge levels could be a key to increasing patients' decisional capability so that they may participate in decision making. Patients could gain professional knowledge from their health professionals and other resources such as literature and the internet, and then could build enough confidence to be actively involved.

Another barrier to shared decision making was that health professionals had a significant lack of understanding of shared decision making. Although their knowledge levels

varied, most did not have a clear understanding of the subject and felt it could be difficult to implement. Before they were able to implement shared decision making, the first step was that they must have knowledge about it. Thus, the importance of improving professionals' level of knowledge of shared decision making was also identified in the study.

Attitude/motivation modification and improvement

- modification of health professionals' paternalistic attitude to patient involvement
- attitudinal modification regarding use of involuntary interventions

It was evident that a widespread paternalistic attitude to patients existed amongst health professionals. Therefore, modifying health professionals' attitudinal barriers to patient involvement could be a key to facilitating shared decision making in mental healthcare. Importantly, the power imbalance between health professionals and patients could also be improved by continued reflection on paternalism in the patient-health professional relationship. Health professionals should, therefore, understand how to take a supportive approach to encourage patients to explore their preferences, and then form an alliance with patients during decision making. Furthermore, the purpose and working definition of involuntary interventions should clearly emphasise helping patients to regain control, rather than being a punishment imposed by health professionals.

Limited time resources improvement

• using limited time efficiently

Health professionals and patients recognised that the issue of health professionals having limited time prevented effective communication between health professionals and patients. Therefore, using limited time resources efficiently may facilitate the application of shared decision making. The work flow and demands on health professionals may need optimising to make sure health professionals have sufficient time with patients. Also, some tools and resources to support shared decision making, such as decision aids, could potentially be used to reduce the time it takes to exchange information, and then shared decision making could be used more efficiently in a limited time.

8.2 Strengths and limitations

The strengths and limitations of each study (one systematic review and two qualitative studies) were presented in depth earlier in this thesis. Only a brief summary will be

therefore presented in the following sections. Following this, discussion on strengths and limitations for the final synthesis will also be provided.

8.2.1 Qualitative systematic review (Study One)

The systematic review (Chapter Three) provided an in-depth discussion directly from patients' perspectives of shared decision making. A key limitation of the review was that it only included studies from western countries, which made it difficult to adapt the findings to other cultural contexts. In addition, there were the limitations of qualitative synthesis to be considered. As this synthesis only focused on qualitative evidence, objectivity should be carefully taken into account.

8.2.2 Patient interviews (Study Two)

This was the first study directly exploring patient perspectives of decision making in mental healthcare in Taiwan (Chapter Five). Therefore, the study provided a unique opportunity for patients to speak up about their views of this process and for health professionals and policy makers to explore how to improve levels of patient involvement. However, there were limitations to consider: although generalisability is not the main purpose of qualitative studies, the small population should be taken into account. Furthermore, the patients in this study were only recruited from the halfway houses. This could lead to the conclusion that the findings may not be transferable to other mental health settings. Although the patient participants were recruited only from halfway houses, they had long-term and rich experience of mental healthcare.

8.2.3 Health professional interviews (Study Three)

In terms of health professional interviews (Chapter Six), this study was the first to provide an insight into how mental health professionals perceive shared decision making in Taiwan. However, there were limitations which included the potential threat of sample selection and recruitment biases. It may be that professionals were more engaged in the topic, shared decision making, and therefore not representative of health professionals in general. One major factor in this was that the number of mental health nurses recruited was lower than expected. Also, two professional participants asked to be withdrawn from the analysis. Although they did not provide reasons for withdrawing, sensitive topics about decision making for patients with mental illness could have negatively influenced how the health professionals viewed their own behaviour in practice, leading them to refuse to continue to participate. Therefore, this could influence how well the study

represents the general population from which the professionals were drawn (Sica, 2006; Smith and Noble, 2014). Furthermore, the data were only collected from self-reported behaviours, which may be difficult to be totally objective about (Rosenman et al., 2011). Therefore, it might be difficult to draw any firm conclusions.

8.2.4 Synthesis

All details of the strengths and limitations of each study have been presented in the previous chapters. This section now focuses on the overall synthesis of this doctoral study. Its main strength is that it has addressed a gap in knowledge relating to patients' perspectives of shared decision making, by both examining the existing literature and obtaining new evidence obtained directly from patients and health professionals in secondary mental healthcare in Taiwan. There have been no published studies exploring decision making in mental healthcare in Asia generally or in Taiwan in particular. Therefore, this study could provide an initial insight into shared decision making in a different context, rather than only a western one.

Furthermore, the study captured the phenomena of the decision-making process comprehensively. It was not just limited to understanding how decision making is shared in mental healthcare, and barriers to and facilitators of it, but also explored improvement mechanisms to overcome the barriers to shared decision making. Meanwhile, both health professionals' and patients' voices were taken into account at the same time, in order to provide a more representative account and a more in-depth discussion of shared decision making through a different lens than other qualitative research in this area.

However, there are some limitations which should be considered while examining the findings of this doctoral study. Although this synthesis did not aim to achieve quantitative generalisability, only the findings from three data resources, including one qualitative systematic review and two qualitative studies (patient and health professional groups), were included in the synthesis. However, the data was recognised by the author, with the agreement of the supervisory team, to have indeed achieved saturation. Therefore, the quality of the data would be able to provide an in-depth insight into the phenomena in which the author was interested. Furthermore, although there are various approaches to qualitative syntheses, this overall synthesis only focuses on improvement mechanisms. In order to pay attention not just to themes which were generated from the data but also to possible improvement, and to develop strategies for overcoming these barriers to

shared decision making, the main emphasis was recognised before the synthesis was conducted.

8.3 Current literature

There has been increasing scholarly attention paid to shared decision making in mental healthcare since the author commenced this doctoral study. This section will therefore provide a discussion of the findings of this study and compare them with the most recent literature.

The participants' (both patients' and health professionals') accounts in this study presented a widespread view that there has been a failure to actively involve patients in decision making, and it is evident in recent literature that patients are still commonly excluded from the decision-making process (Care Quality Commission, 2017). Equally common is the belief that patients with mental illness should not be involved in decision making, and widespread use of potential coercive interventions was widely reported by both health professionals and patients in this study; this, too, is echoed in the most recent literature (Saya et al., 2019; Shepherd et al., 2014). These practices are inconsistent with patients' legal right to be involved in their treatment and care, which has in recent years been enhanced and added to in law by lawmakers world-wide, including in Taiwan (Liao et al., 2017). The practices also violate ethical principles of health practice, such as autonomy, which encompasses showing respect for the patients' choices, and often refers to the protection of a patient's ability to make his or her own decisions (Beauchamp and Childress, 2013). Despite some advances, this remains an area of concern (Slade, 2017).

It is worth noting that some policies have been created and efforts made to improve shared decision making in healthcare internationally (Slade, 2017). Furthermore, some countries have been integrating shared decision making into their standard principles in mental healthcare to ensure patients have the right to make their own informed decisions (Hem et al., 2016). Despite this, one recent cross-national mixed-method comparative case study in the UK found that patients in mental healthcare were not really involved in the decision making process (Coffey et al., 2019) and one systematic review also supported the suggestion that shared decision making has failed in mental healthcare (Bee et al., 2015). It is clear that patient passivity is still common in the decision-making process around the world (Care Quality Commission, 2017) and specifically in Taiwan (Liao et al., 2017), and the use of potential coercion is still widespread in mental healthcare (Care Quality Commission, 2019).

In Taiwan, some relevant laws, such as the Patient Self-Determination Act, have been enacted to ensure patients' right to make decisions (Liao et al., 2017). Although a nationwide shared decision making programme has been in place since 2016, shared decision making is still seen as a new concept (Liao et al., 2017). Consistent with the findings of this study, however, there is still a significant lack of understanding of shared decision making and it is still not applied in practice in Taiwan (Liao et al., 2017). Therefore, it is clear that implementation of shared decision making still needs further work to make it possible in mental healthcare, particularly in Taiwan.

From patients' views in this study, there was a strong perception that patients had a clear desire to be involved in decision making, which was also frequently reported in the literature (Chewning et al., 2012; Kiesler and Auerbach, 2006). One systematic review aimed to explore the preference-match strategy in physician-service user communication. This review included 69 studies and revealed that patients expected more active roles in decision making than they experienced (Kiesler and Auerbach, 2006). One more recent Cochrane systematic review, including 115 studies, also concluded that most patients desired shared decision making in medical encounters (Chewning et al., 2012). These findings are consistent with this study showing patients are largely positive about patient involvement.

However, the finding of this study showed some divergence in patients' desire for active involvement in some ways. Similarly, some previous studies have suggested that patient preferences about involvement in decision making could vary (Levinson et al., 2005). Although the patients studied were clear in their desire for more active involvement in decision making, their desired involvement levels varied from more information to making their own decisions in this study. Despite not every patient wanting to make their own decisions, all patient participants still desired more information about their treatment and care. These findings converge with current literature: many would prefer not to remain passive, and to receive more information about their health conditions (Bowling and Ebrahim, 2001; Dahlqvist et al., 2015; James and Quirk, 2017). According to NICE, some patients may not want to take an active role during decision making, but they should always have an opportunity to decide what level of involvement they want (NICE, 2019).

Substantial evidence suggests that patients in mental healthcare are motivated to be actively involved, both in this study and in the literature (Bowling and Ebrahim, 2001; Dahlqvist et al., 2015; Fisher et al., 2016; James and Quirk, 2017; Woltmann and

Whitley, 2010), but this study found that numerous barriers are created through insufficient information exchange and a lack of opportunities for deliberation. Significantly, a variety of challenges to embedding shared decision making in routine services have been highlighted in this study and in the literature (Bee et al., 2015; Joseph-Williams et al., 2014a; Joseph-Williams et al., 2014b; Joseph-Williams et al., 2017; Slade, 2017; Stovell et al., 2016), including in Taiwan (Liao et al., 2017). Other research has found that because of these barriers, integrating shared decision making into routine practice has been surprisingly impeded (Joseph-Williams et al., 2017). Comparison and consideration of the findings from these patient and health professional interviews expose some barriers to shared decision making where greater attention may be required.

The importance of providing sufficient information for patients has been identified in the literature (Faller et al., 2016; Joseph-Williams et al., 2014b), as in this study. It could be difficult for patients to participate in decision making without any knowledge about their illness, health conditions and available options (Joseph-Williams et al., 2014a). The findings of this study and in the literature point out the importance to the decision-making process of patients and health professionals having sufficient knowledge (Légaré et al., 2018; Stiggelbout et al., 2015). As education initiatives that enhance health professionals' ability to integrate shared decision making into their practice are significant, a variety of training programmes have identified to support health professionals in this process (Diouf et al., 2016; Légaré et al., 2018). However, Joseph-Williams et al. (2014b) also stated that although providing information is important to shared decision making, it is not enough to facilitate it in practice and they believed more actions needed to be taken to deal with attitudinal barriers.

It is evident that attitudinal barriers to shared decision making are more significant than other barriers in this study. This study showed that health professionals reported that they ignored patients' desire for active involvement, and a similar conclusion was drawn in the literature (Dahlqvist et al., 2015). The issue of attitudinal change in health professionals has been pointed out in recent years (Joseph-Williams et al., 2017). This study's negative findings are consistent with the existing literature, which is replete with examples of professionals adopting paternalistic attitudes to patient involvement in decision making, particularly in mental healthcare (Hem et al., 2016; Quirk et al., 2012). Health professionals' attitudes were identified as a significant barrier, with the implication that patients should not be involved in decision making (Bae, 2017). To explain this, health professionals believed they should take responsibility as the main

decision makers, as they act in the patients' best interest based on their biomedical ethical codes (Slade, 2017).

This sense of power imbalance between health professionals and patients was widespread across health settings around the world (Joseph-Williams et al., 2014a; Ljungberg et al., 2016). Also, Taiwan has a long tradition of paternalism in the therapeutic relationship (Tai and Tsai, 2003), as is seen in many other countries (Hu et al., 2014; Murgic et al., 2015). This could also explain why patients feel they cannot participate in shared decision making, rather than not wanting to (Joseph-Williams et al., 2014a; Ljungberg et al., 2016). Furthermore, the use of involuntary interventions exacerbates the power imbalance between health professionals and patients (Ljungberg et al., 2016). In mental healthcare, there is a rising trend in the use of coercion and involuntary interventions (Sashidharan and Saraceno, 2017). Patients with mental illness who were subjected to coercive psychiatric interventions had negative experiences and views of the experience, such as trauma/retraumatisation, distress, fear, feeling ignored, and being dehumanised, both in this study and literature (Cusack et al., 2018). These experiences of being restrained in hospitals create trauma and fear regarding future treatment and care (Brophy et al., 2016). In most patient accounts, the overwhelming majority of patients had experienced the use of coercion in mental healthcare, such as taking medication, physical restraint and rapid tranquilisation in hospitals. In patient accounts, the power to use coercive interventions is generally held by health professionals rather than patients. Therefore, patients and health professionals believed the powerful status of health professionals. Patients with mental illness are therefore constrained from expressing their perspectives openly and honestly to health professionals because of fear of the "consequences" (National Mental Health Consumer and Carer Forum, 2009). This sense of health professionals having the more powerful role could explain why shared decision making is severely lacking in mental healthcare.

Meanwhile, both this study and the literature found that the sense that patients with mental illness often suffer from a lack of insight into their mental health conditions leads health professionals to believe they might refuse to accept any treatment, which could create a conflict with the biomedical ethical principles and worsen patient health outcomes. (Shepherd et al., 2014; Slade, 2017). Concerns about patients having impaired decisional ability were also found in other studies, and are highlighted in the literature as barriers to involvement (Scholl and Barr, 2017; Slade, 2017). The consistency of this finding across our patient and health professional interviews can be seen in a number of

ways. For example, health professionals believed that patients commonly refused to take medications. Impaired decisional capability of patients with mental illness has been frequently argued in this study and in the literature (Seale et al., 2006; Shepherd et al., 2014). This may be explained by concerns that patients would make risky and irresponsible decisions leading to worse health outcomes (Shepherd et al., 2014). Therefore, these concerns led to health professionals not attempting to involve patients with mental illness. This may also explain why the patients in Taiwan also undervalue their ability to understand professional information about treatment and care, and their lack of confidence in asking questions. However, only if patients acknowledge that their unique expertise on their own bodies is valuable in the decision making process and they are capable of understanding information will they be able to actively enage in decision making (Joseph-Williams et al., 2014a).

In addition, as described in this study, lack of knowledge of shared decision making among health professionals was a major barrier to implementing it. Only a few health professionals articulated clear values and understanding reflecting components of shared decision making and the potential benefits of shared decision making which have been promoted to improve healthcare in Taiwan. From this study, there is a significant knowledge gap amongst health professionals in terms of how to practise shared decision making. Lack of awareness of and familiarity with the intervention could negatively affect health professionals' knowledge about implementing it (Cabana et al., 1999). A prerequisite for implementing shared decision making is that mental healthcare professionals are not just willing but also have the ability to involve patients in decisions (Grim et al., 2016).

One more concern, about limited time resources, is also widespread among health professionals in many current studies (Joseph-Williams et al., 2017; Liao et al., 2017; Terry and Coffey, 2019). Health professionals perceive that they struggle to provide additional time for patients in order to implement shared decision making (Liao et al., 2017). As found in this study, patient participants also consider health professionals' limited time as a significant barrier to patient involvement in decision making. However, a number of decision aids have been developed to overcome the challenges of limited time resources (Légaré et al., 2018). These helpful tools could overcome this barrier effectively.

8.4 Clinical implications

Attention should be paid to these suggestions for facilitating the implementation of shared decision making in mental healthcare. Although patients in mental healthcare reported a clear desire for a more active role during decision making, shared decision making is not actually the norm in mental healthcare. The findings of this study could be an important resource for patients to have this unique opportunity to explain their perspectives on shared decisions making to the public. Also, health professionals have specific insights into their interactions with patients and what it is important for them to explain to the public.

Furthermore, to develop and implement shared decision making in practice, various barriers and facilitators which were identified by this doctoral study should also be taken into account. First, as earlier described, some training is required for both health professionals and patients to increase their knowledge levels to enable them to participate. Second, attitudinal barriers evidently influence health professionals' and patients' beliefs about patient involvement. Therefore, the need for modification of these negative perceptions has been pointed out in the study. Finally, the issues of a time-poor environment in practice were also identified and need to be improved.

In addition, the significance of family involvement in decision making was identified in the study. The complicated structure of the patient-family relationship and implicit moral responsibility could be a key to successfully implementing shared decision making in mental healthcare. It is worth mentioning that health professionals should consider how to handle disclosure and understand the family dynamic process to improve patient involvement.

Meanwhile, training for health professionals to facilitate shared decision making is also essential to make this possible in practice. However, there are few education programmes on shared decision making available for health professionals to improve their knowledge and develop their skills. While patients indicated in the interviews that they needed more understandable knowledge delivery, health professionals admitted they have limited knowledge and skills to provide it. Therefore, the findings of this study could be used to develop medical education programmes for health professionals to improve these.

8.5 Further research

The findings of this synthesis serve as relevant evidence of what is available and what patients desire from mental healthcare, and how health professionals view patient

involvement in decision making. However, the importance of family involvement is also recognised in the study. Given the limitations of this study, it would be worth strengthening the evidence through similar qualitative research reaching out to other groups, such as family caregivers.

Furthermore, the study suggested three improvement mechanisms to overcome patient passivity and barriers to shared decision making. However, these potential improvements and levels of involvement were only expressed in qualitative evidence in this synthesis. Further studies may need more objective and quantitative measurements of the effectiveness of these improvements, such as randomised controlled trials, to justify these suggestions.

Meanwhile, the effectiveness of these shared decision making interventions still remains unclear in a wide range of health settings (Légaré et al., 2018). In order to embed shared decision making into routine practice in mental healthcare, it may be important to establish an understanding of the effectiveness of these interventions through a future study. Also, as there is a lack of intervention studies in Taiwan, it may be difficult to determine which of these interventions will work in the different cultural context, or how.

8.6 Conclusion

The study aimed to explore the decision-making process in secondary mental healthcare in Taiwan from the perspective of patients and health professionals. It consists of three studies: 1) a systematic review of qualitative studies of patients' perspectives of shared decision making in secondary mental healthcare; 2) qualitative interviews with patients in secondary mental healthcare in Taiwan; 3) qualitative interviews with health professionals in secondary mental healthcare in Taiwan.

The exploration resulted in the identification of patient passivity and health professional dominance, and other barriers to and facilitators of shared decision making. Despite the limitations of this study, it would be valuable to consider the findings to facilitate evidence-based implementation of shared decision making.

References

- Adams, J. R. & Drake, R. E. (2006). 'Shared decision-making and evidence-based practice', *Community Mental Health Journal*, 42(1), pp. 87-105.
- Al-Busaidi, Z. Q. (2008). 'Qualitative research and its uses in health care', *Sultan Qaboos University Medical Journal*, 8(1), pp. 11-19.
- Alhojailan, M. I. (2012). 'Thematic analysis: a critical review of its process and evaluation', *West East Journal of Social Sciences*, 1(1), pp. 39-47.
- Aliyu, A., Umar Bello, M., Kasim, R. & Martin, D. (2014). 'Positivist and non-positivist paradigm in social science research: conflicting paradigms or perfect partners?', *Journal of Management and Sustainability*, 4(3), pp. 79-95.
- Anney, V. N. (2014). 'Ensuring the quality of the findings of qualitative research: looking at trustworthiness criteria', *Journal of Emerging Trends in Educational Research* and Policy Studies, 5(2), pp. 272-281.
- Aslam, S. & Emmanuel, P. (2010). 'Formulating a researchable question: a critical step for facilitating good clinical research', *Indian Journal of Sexually Transmitted Diseases*, 31(1), pp. 47-50.
- Auerbach, S. M. (2000). 'Should patients have control over their own health care?: Empirical evidence and research issues', *Annals of Behavioral Medicine*, 22(3), pp. 246-259.
- Autorino, R., Quarto, G., Di Lorenzo, G., De Sio, M. & Damiano, R. (2007). 'Are abstracts presented at the EAU meeting followed by publication in peer-reviewed journals? A critical analysis', *European Urology*, 51(3), pp. 833-840.
- Ayres, L. (2012). 'Semi-structured Interview', in Given, L. M. (ed.) The SAGE encyclopedia of qualitative research methods. Thousand Oaks: SAGE Publications.
- Bae, J. M. (2017). Shared decision making: relevant concepts and facilitating strategies. *Epidemiology and Health* [Online], 39. Available at: https://www.eepih.org/journal/view.php?doi=10.4178/epih.e2017048 (Accessed: 22 July 2018).
- Balint, M. (1957). *The doctor, his patient and the illness*. 2nd edn. London: Pitman Medical.
- Barusch, A., Gringeri, C. & George, M. (2011). 'Rigor in qualitative social work research: a review of strategies used in published articles', *Social Work Research*, 35(1), pp. 11-19.

- Bearman, M. & Dawson, P. (2013). 'Qualitative synthesis and systematic review in health professions education', *Medical Education*, 47(3), pp. 252-260.
- Beauchamp, T. L. (1984). *Medical ethics: The moral responsibilities of physicians*. New Jersey: Prentice-Hall.
- Beauchamp, T. L. & Childress, J. F. (2013). Principles of biomedical ethics. 7th edn. New York: Oxford University Press.
- Bee, P., Price, O., Baker, J. & Lovell, K. (2015). 'Systematic synthesis of barriers and facilitators to service user-led care planning', *The British Journal of Psychiatry*, 207(2), pp. 104-114.
- Beitinger, R., Kissling, W. & Hamann, J. (2014). 'Trends and perspectives of shared decision-making in schizophrenia and related disorders', *Current Opinion in Psychiatry*, 27(3), pp. 222-229.
- Bilderbeck, A. C., Saunders, K. E., Price, J. & Goodwin, G. M. (2014). 'Psychiatric assessment of mood instability: qualitative study of patient experience', The British Journal of Psychiatry, 204(3), pp. 234-239.
- Booth, A. (2016). 'Searching for qualitative research for inclusion in systematic reviews: a structured methodological review', *Systematic Reviews*, 5(74), pp. 1-23.
- Booth, A., Hannes, K., Harden, A., Noyes, J., Harris, J. & Tong, A. (2014). 'COREQ (Consolidated Criteria for Reporting Qualitative Studies)', in Moher, D., Altman, D. G., Schulz, K. F., Simera, I. & Wager, E. (eds.) *Guidelines for reporting health research: a user's manual*. Oxford: John Wiley & Sons, pp. 214-226.
- Booth, A., Papaioannou, D. & Sutton, A. (2012). *Systematic approaches to a successful literature review*. London: SAGE Publications.
- Bowen, G. A. (2009). 'Supporting a grounded theory with an audit trail: an illustration', *International Journal of Social Research Methodology*, 12(4), pp. 305-316.
- Bowling, A. & Ebrahim, S. (2001). 'Measuring patients' preferences for treatment and perceptions of risk', *Quality in Health Care*, 10(Suppl 1), pp. i2-i8.
- Braun, V. & Clarke, V. (2006). 'Using thematic analysis in psychology', *Qualitative Research in Psychology*, 3(2), pp. 77-101.
- Brett, S. (2018). 'Generalizability in qualitative research: misunderstandings, opportunities and recommendations for the sport and exercise sciences,' *Qualitative Research in Sport, Exercise and Health*, 10(1), pp. 137-149.
- Brophy, L. M., Roper, C. E., Hamilton, B. E., Tellez, J. J. & McSherry, B. M. (2016). 'Consumers and carer perspectives on poor practice and the use of seclusion and

restraint in mental health settings: results from Australian focus groups', International Journal of Mental Health Systems, 10(6), pp. 1-10.

- Bunn, M. H., O'Connor, A. M., Tansey, M. S., Jones, B. D. W. & Stinson, L. E. (1997).
 'Characteristics of clients with schizophrenia who express certainty or uncertainty about continuing treatment with depot neuroleptic medication', *Archives of Psychiatric Nursing*, 11(5), pp. 238-248.
- Bureau of the National Health Insurance (2015). *The monthly report of the National Health Insurance*. Taipei: National Health Insurance Administration in Taiwan.
- Cabana, M. D., Rand, C. S., Powe, N. R., Wu, A. W., Wilson, M. H., Abboud, P.-A. C.
 & Rubin, H. R. (1999). 'Why don't physicians follow clinical practice guidelines? A framework for improvement', *JAMA*, 282(15), pp. 1458-1465.
- Caelli, K., Ray, L. & Mill, J. (2003). 'Clear as mud': toward greater clarity in generic qualitative research. Article 1. *International Journal of Qualitative Methods* [Online], 2(2). Available at: https://sites.ualberta.ca/~iiqm/backissues/2_2/pdf/caellietal.pdf (Accessed: 29 October 2018).
- Capers, M. (2010). Sahred decision-making in mental health care: practice, research, and future directions: Center for Mental Health Services, Substance Abuse and Mental Health Service Administration. Available at: https://store.samhsa.gov/product/Shared-Decision-Making-in-Mental-Health-Care/sma09-4371 (Accessed: 20 August 2017).
- Carcary, M. (2009). 'The research audit trial-enhancing trustworthiness in qualitative inquiry', *Electronic Journal of Business Research Methods*, 7(1), pp. 11-24.
- Care Quality Commission (2017). 2017 Community mental health survey: statistical relase, Care Quality Commission. Available at: https://www.cqc.org.uk/publications/surveys/community-mental-health-survey-2017 (Accessed: 20 July 2018).
- Care Quality Commission (2019). *Monitoring the Mental Health Act in 2017/18*, Care Quality Commission. Available at: https://www.cqc.org.uk/sites/default/files/20190320_mhareport1718_report.pdf (Accessed: 25 August 2019).
- Carpenter, W. T. & Conley, R. R. (1999). 'Sense and nonsense: an essay on schizophrenia research ethics', *Schizophrenia Research*, 35(3), pp. 219-225.
- Carpenter, W. T., Gold, J. M., Lahti, C., Queern, C. A., Conley, R. R., Bartko, J. J., Kovnick, J. & Appelbaum, P. S. (2000). 'Decisional capacity for informed

consent in schizophrenia research', Archives of General Psychiatry, 57(6), pp. 533-538.

- Cavanagh, S. (1997). 'Content analysis: concepts, methods and applications', *Nursing Research*, 4(3), pp. 5-16.
- Chang, R., Lin, S. & Aron, D. C. (2012). 'A pay-for-performance program in Taiwan improved care for some diabetes patients, but doctors may have excluded sicker ones', *Health Affairs*, 31(1), pp. 93-102.
- Charavel, M., Bremond, A., Moumjid-Ferdjaoui, N., Mignotte, H. & Carrere, M. O. (2001). 'Shared decision-making in question', *Psycho-Oncology*, 10(2), pp. 93-102.
- Charles, C., Gafni, A. & Whelan, T. (1997). 'Shared decision-making in the medical encounter: what does it mean? (or it takes at least two to tango)', *Social Science* & *Medicine*, 44(5), pp. 681-692.
- Charles, C., Gafni, A. & Whelan, T. (1999). 'Decision-making in the physician-patient encounter: revisiting the shared treatment decision-making model', *Social Science & Medicine*, 49(5), pp. 651-661.
- Cheng, T. M. (2015). 'Reflections on the 20th anniversary of Taiwan's single-payer National Health Insurance System', *Health Affairs*, 34(3), pp. 502-510.
- Chewning, B., Bylund, C. L., Shah, B., Arora, N. K., Gueguen, J. A. & Makoul, G. (2012). 'Patient preferences for shared decisions: a systematic review', *Patient Education and Counseling*, 86(1), pp. 9-18.
- Chou, F. H. (2015). 'Addressing issues of Taiwan psychiatry in the early 21st century', *Taiwanese Journal of Psychiatry* 29(3), pp. 36-38.
- Chuang, M., Cheng, J., Huang, H., Ke, S. & Chu, T., Yuan, E. (1995). *Review of psychiatric service systems*. Taipei: Research Development and Evaluation Commission.
- Cipriani, A. & Geddes, J. (2003). 'Comparison of systematic and narrative reviews: the example of the atypical antipsychotics', *Epidemiologia and Psichiatria Sociale*, 12(3), pp. 146-153.
- Clark, A. M. (1998). 'The qualitative-quantitative debate: moving from positivism and confrontation to post-positivism and reconciliation', *Journal of Advanced Nursing*, 27(6), pp. 1242-1249.
- Clarke, E., Puschner, B., Jordan, H., Williams, P., Konrad, J., Kawohl, W., Bar, A., Rossler, W., Del Vecchio, V., Sampogna, G., Nagy, M., Suveges, A., Krogsgaard Bording, M. & Slade, M. (2015). 'Empowerment and satisfaction in a

multinational study of routine clinical practice', *Acta Psychiatrica Scandinavica*, 131(5), pp. 369-378.

- Clarke, M. & Oxman, A. D. (2003). *Cochrane reviewer's handbook 4.2.0*. Oxford: The Cochrane Collaboration.
- Clarke, M. A., Moore, J. L., Steege, L. M., Koopman, R. J., Belden, J. L., Canfield, S. M., Meadows, S. E., Elliott, S. G. & Kim, M. S. (2016). 'Health information needs, sources, and barriers of primary care patients to achieve patient-centered care: a literature review', 22(4), pp. 992-1016.
- Clarke, V. & Braun, V. (2013). 'Teaching thematic analysis: overcoming challenges and developing strategies for effective learning', *The Psychologist*, 26(2), pp. 120-123.
- Coffey, M., Hannigan, B., Barlow, S., Cartwright, M., Cohen, R., Faulkner, A., Jones, A. & Simpson, A. (2019). 'Recovery-focused mental health care planning and coordination in acute inpatient mental health settings: a cross national comparative mixed methods study', *BMC Psychiatry*, 19(1), p. 115.
- Cohen, L., Manion, L. & Morrison, K. (2007). *Research methods in education*. 6th edn. London: Routledge.
- Connor, S. L. & Wilson, R. (2006). 'It's important that they learn from us for mental health to progress', *Journal of Mental Health*, 15(4), pp. 461-474.
- Cook, D. J., Mulrow, C. D. & Haynes, R. B. (1997). 'Systematic reviews: synthesis of best evidence for clinical decisions', *Annals of Internal Medicine*, 126(5), pp. 376-380.
- Cooke, A., Smith, D. & Booth, A. (2012). 'Beyond PICO: the SPIDER tool for qualitative evidence synthesis', *Qualitative Health Research*, 22(10), pp. 1435-1443.
- Cooper, S. & Endacott, R. (2007). 'Generic qualitative research: a design for qualitative research in emergency care?', *Emergency Medicine Journal*, 24(12), pp. 816-819.
- Coulter, A. (2010). 'Do patients want a choice and does it work?', *The BMJ*, 341, pp. 973-975.
- Coulter, A. & Collins, A. (2011). Making shared decision making a reality: no decision about me, without me, London: King's Fund. Available at: https://www.kingsfund.org.uk/sites/default/files/Making-shared-decisionmaking-a-reality-paper-Angela-Coulter-Alf-Collins-July-2011_0.pdf (Accessed: 26 May 2016).

- Coulter, A., Edwards, A., Elwyn, G. & Thomson, R. (2011). 'Implementing shared decision making in the UK', *Zeitschrift für Evidenz, Fortbildung und Qualität im Gesundheitswesen*, 105(4), pp. 300-304.
- Coulter, A. & Jenkinson, C. (2005). 'European patients' views on the responsiveness of health systems and healthcare providers', *European Journal of Public Health*, 15(4), pp. 355-360.
- Creswell, J. W. (2007). *Qualtative inquiry and research design: choosing among five approaches*. 2nd edn. London: SAGE Publications.
- Cribb, A. & Entwistle, V. A. (2011). 'Shared decision making: trade-offs between narrower and broader conceptions', *Health Expectations*, 14(2), pp. 210-219.
- Crotty, M. (1998). *The foundations of social research: Meaning and perspective in the research process*. London: SAGE Publications.
- Cusack, P., Cusack, F. P., McAndrew, S., McKeown, M. & Duxbury, J. (2018). 'An integrative review exploring the physical and psychological harm inherent in using restraint in mental health inpatient settings', *International Journal of Mental Health Nursing*, 27(3), pp. 1162-1176.
- Dahlqvist, J. P., Schön, U. K., Rosenberg, D., Sandlund, M. & Svedberg, P. (2015).
 'Service users' experiences of participation in decision making in mental health services', *Journal of Psychiatric and Mental Health Nursing*, 22(9), pp. 688-697.
- De las Cuevas, C., Peñate, W. & de Rivera, L. (2014). 'Psychiatric patients' preferences and experiences in clinical decision-making: examining concordance and correlates of patients' preferences', *Patient Education and Counseling*, 96(2), pp. 222-228.
- Deegan, P. E. (2007). 'The lived experience of using psychiatric medication in the recovery process and a shared decision-making program to support it', *Psychiatric Rehabilitation Journal*, 31(1), pp. 62-69.
- Del Piccolo, L. & Goss, C. (2012). 'People-centred care: New research needs and methods in doctor-patient communication. Challenges in mental health', *Epidemiology and Psychiatric Sciences*, 21(2), pp. 145-149.
- Delbanco, T., Berwick, D. M., Boufford, J. I., Edgman-Levitan, Ollenschläger, G., Plamping, D. & Rockefeller, R. G. (2001). 'Healthcare in a land called PeoplePower: nothing about me without me', *Health Expectations*, 4(3), pp. 144-150.
- Delman, J., Clark, J. A., Eisen, S. V. & Parker, V. A. (2015). 'Facilitators and barriers to the active participation of clients with serious mental illnesses in medication

decision making: the perceptions of young adult clients', *The Journal of Behavioral Health Services & Research*, 42(2), pp. 238-253.

- Denzin, N. & Lincoln, Y. (2003a). *Collecting and interpreting qualitative materials*. 2nd edn. London: SAGE Publications.
- Denzin, N. & Lincoln, Y. (2003b). *Strategies of qualitative inquiry*. 2nd edn. London: SAGE Publications.
- Denzin, N. & Lincoln, Y. (2005). *The SAGE handbook of qualitative research*. 3rd edn. London: SAGE Publications.
- Denzin, N. K. (1989). The research act: a theoretical introduction to sociological methods . 3rd edn. Chicago: Aldine.
- Denzin, N. K. & Lincoln, Y. S. (1998). *The landscape of qualitative research: theories and issues*. London: SAGE Publications.
- Dey, I. (1993). *Qualitative data analysis: a user-friendly guide for social scientists*. London: Routledge.
- Diouf, N. T., Menear, M., Robitaille, H., Painchaud Guérard, G. & Légaré, F. (2016).
 'Training health professionals in shared decision making: update of an international environmental scan', *Patient Education and Counseling*, 99(11), pp. 1753-1758.
- Dixon-Woods, M., Agarwal, S., Jones, D., Young, B. & Sutton, A. (2005). 'Synthesising qualitative and quantitative evidence: a review of possible methods', *Journal of Health Services Research & Policy*, 10(1), pp. 45-53.
- Dixon-Woods, M., Booth, A. & Sutton, A. J. (2007). 'Synthesizing qualitative research: a review of published reports', *Qualitative Research*, 7(3), pp. 375-422.
- Dixon-Woods, M., Cavers, D., Agarwal, S., Annandale, E., Arthur, A., Harvey, J., Hsu, R., Katbamna, S., Olsen, R., Smith, L., Riley, R. & Sutton, A. J. (2006). Conducting a critical interpretive synthesis of the literature on access to healthcare by vulnerable groups. *BMC Medical Research Methodology* [Online], 6(35). Available at: http://dx.doi.org/10.1186/1471-2288-6-35 (Accessed: 25 May 2016).
- Doyal, L. (2002). 'Good clinical practice and informed consent are inseparable', *Heart* (*British Cardiac Society*), 87(2), pp. 103-106.
- Drake, R. E. & Deegan, P. E. (2009). 'Shared decision making is an ethical imperative', *Psychiatric Services*, 60(8), p. 1007.
- Drug and Therapeutics Bulletin (2013). An introduction to patient decision aids. TheBMJ[Online],346(f4147).Availableat:

https://www.bmj.com/content/bmj/347/bmj.f4147.full.pdf (Accessed: 16 June 2016).

- Duncan, E., Best, C. & Hagen, S. (2010). 'Shared decision making interventions for people with mental health conditions', *Cochrane Database of Systematic Reviews*, (1), p. Cd007297.
- Earthy, S. & Cronin, A. (2008). 'Narrative analysis', in Gilbert, N. (ed.) *Researching social life*. 3rd edn. London: SAGE Publications, pp. 420–439.
- Edwards, A. & Elwyn, G. (2009). 'Shared decision-making in health care: achieving evidence-based patient care', in Edwards, A. & Elwyn, G. (eds.) *Shared decisionmaking in health care.: Vol. 1.* 2nd edn. Oxford: Oxford University Press, pp. 3-10.
- Elam, G. & Fenton, K. A. (2003). 'Researching sensitive issues and ethnicity: lessons from sexual health', *Ethnicity and Health*, 8(1), pp. 15-27.
- Ellis, P. (2013). *Understanding research for nursing students*. 2nd edn. London: SAGE Publications.
- Ellwood, P. M. (1988). 'Shattuck lecture--outcomes management. A technology of patient experience', *The New England Journal of Medicine*, 318(23), pp. 1549-1556.
- Elo, S. & Kyngas, H. (2008). 'The qualitative content analysis process', *Journal of Advanced Nursing*, 62(1), pp. 107-115.
- Elwyn, G., Edwards, A., Kinnersley, P. & Grol, R. (2000). 'Shared decision making and the concept of equipoise: The competences of involving patients in healthcare choices', *The British Journal of General Practice*, 50(460), pp. 892-899.
- Elwyn, G., Frosch, D., Thomson, R., Joseph-Williams, N., Lloyd, A., Kinnersley, P., Cording, E., Tomson, D., Dodd, C., Rollnick, S., Edwards, A. & Barry, M. (2012). 'Shared decision making: a model for clinical practice', *Journal of General Internal Medicine*, 27(10), pp. 1361-1367.
- Elwyn, G., Laitner, S., Coulter, A., Walker, E., Watson, P. & Thomson, R. (2010). 'Implementing shared decision making in the NHS', *The BMJ*, 341(2010), pp. 971-975.
- Elwyn, G., O'Connor, A., Stacey, D., Volk, R., Edwards, A., Coulter, A., Thomson, R., Barratt, A., Barry, M., Bernstein, S., Butow, P., Clarke, A., Entwistle, V., Feldman-Stewart, D., Holmes-Rovner, M., Llewellyn-Thomas, H., Moumjid, N., Mulley, A., Ruland, C., Sepucha, K., Sykes, A. & Whelan, T. (2006). Developing a quality criteria framework for patient decision aids: online international Delphi

consensus process. *The BMJ* [Online], 333(417). Available at: <u>https://www.bmj.com/content/bmj/333/7565/417.full.pdf</u> (Accessed: 12 July 2016).

- Elwyn, G., Tilburt, J. & Montori, V. (2013a). 'The ethical imperative for shared decisionmaking', *European Journal for Person Centered Healthcare*, 1(1), pp. 129-131.
- Elwyn, G., Tsulukidze, M., Edwards, A., Legare, F. & Newcombe, R. (2013b). 'Using a 'talk' model of shared decision making to propose an observation-based measure: observer OPTION 5 Item', *Patient Education and Counseling*, 93(2), pp. 265-271.
- Emanuel, E. J. & Emanuel, L. L. (1992). 'Four models of the physician-patient relationship', *The Journal of the American Medical Association*, 267(16), pp. 2221-2226.
- Emanuel, E. J., Wendler, D. & Grady, C. (2000). 'What makes clinical research ethical?', *The Journal of the American Medical Association*, 283(20), pp. 2701-2711.
- Enosh, G. & Buchbinder, E. (2005). 'The interactive construction of narrative styles in sensitive interviews: the case of domestic violence research', *Qualitative Inquiry*, 11(4), pp. 588-617.
- Entwistle, V. (2009). 'Patient involvement in decision-making: the importance of a broad conceptualization', in Edwards, A. & Elwyn, G. (eds.) *Shared decision-making in health care*. 2nd edn. Oxford: Oxford Univesity Press, pp. 17-22.
- Etikan, I., Musa, S. A. & Alkassim, R. S. (2016). 'Comparison of convenience sampling and purposive sampling', *American Journal of Theoretical and Applied Statistics*, 5(1), pp. 1-4.
- Faden, R. R. & Beauchamp, T. L. (1986). A history and theory of informed consent. New York: Oxford University Press.
- Faller, H., Koch, U., Brähler, E., Härter, M., Keller, M., Schulz, H., Wegscheider, K., Weis, J., Boehncke, A., Hund, B., Reuter, K., Richard, M., Sehner, S., Szalai, C., Wittchen, H. U. & Mehnert, A. (2016). 'Satisfaction with information and unmet information needs in men and women with cancer', *Journal of Cancer Survivorship*, 10(1), pp. 62-70.
- Farrelly, S., Lester, H., Rose, D., Birchwood, M., Marshall, M., Waheed, W., Henderson, R. C., Szmukler, G. & Thornicroft, G. (2016). 'Barriers to shared decision making in mental health care: qualitative study of the Joint Crisis Plan for psychosis', *Health Expectations*, 19(2), pp. 448-458.

- Fielding, N. G. & Lee, R. M. (1991). *Using computers in qualitative research*. London: SAGE Publications.
- Fisher, A., Manicavasagar, V., Kiln, F. & Juraskova, I. (2016). 'Communication and decision-making in mental health: a systematic review focusing on bipolar disorder', *Patient Education and Counseling*, 99(7), pp. 1106-1120.
- Fisher, R. J. (1993). 'Social desirability bias and the validity of indirect questioning', *Journal of Consumer Research*, 20(2), pp. 303-315.
- Fitzpatrick, R. & Boulton, M. (1994). 'Qualitative methods for assessing health care', *Quality in Health Care*, 3(2), pp. 107-113.
- Flick, U. (2007). Designing qualitative research. London: SAGE Publications.
- Flick, U. (2009). An Introduction to qualitative research. 5th edn. London: Sage Publications.
- Floyd, J. & Fowler, J. (2004). Structured interview. *In:* Lewis-Beck, M. S., Bryman, A.
 & Futing Liao, T. (eds.) *The SAGE encyclopedia of social science research methods*. Thousand Oaks: SAGE Publications.
- Fontana, A. (2004). 'Unstructured Interview', in *The SAGE encyclopedia of social science research methods* [Online]. Version. Available at: <u>http://methods.sagepub.com/reference/the-sage-encyclopedia-of-social-science-research-methods</u>.
- Ford, S., Schofield, T. & Hope, T. (2002). 'Barriers to the evidence-based patient choice (EBPC) consultation', *Patient Education and Counseling*, 47(2), pp. 179-185.
- Friedrichs, A., Spies, M., Harter, M. & Buchholz, A. (2016). Patient preferences and shared decision making in the treatment of substance use disorders: a systematic review of the literature. *PLoS One* [Online], 11(1). Available at: https://journals.plos.org/plosone/article?id=10.1371/journal.pone.0145817 (Accessed: 16 July 2017).
- Frosch, D. L., May, S. G., Rendle, K. A., Tietbohl, C. & Elwyn, G. (2012). 'Authoritarian physicians and patients' fear of being labeled 'difficult' among key obstacles to shared decision making', *Health Affairs*, 31(5), pp. 1030-1038.
- Fu, T., Lee, C., Gunnell, D., Lee, W. & Cheng, A. (2013). 'Changing trends in the prevalence of common mental disorders in Taiwan: a 20-year repeated crosssectional survey', *The Lancet*, 381(9862), pp. 235-241.
- Fukui, S., Matthias, M. S. & Salyers, M. P. (2015). 'Core domains of shared decisionmaking during psychiatric visits: scientific and preference-based discussions',

Administration and Policy in Mental Health and Mental Health Services Research, 42(1), pp. 40-46.

- Garcia, J., Bricker, L., Henderson, J., Martin, M. A., Mugford, M., Nielson, J. & Roberts, T. (2002). 'Women's views of pregnancy ultrasound: a systematic review', *Birth*, 29(4), pp. 225-250.
- Gerrish, K., Lacey, A. & Cormack, D. (2010). *The research process in nursing*. Chichester: Blackwell.
- Gilbar, R. & Miola, J. (2015). 'One size fits all? On patient autonomy, informed consent and the impact of culture', *Medical Law Review*, 23(3), pp. 375-399.
- Gillon, R. (1994). 'Medical ethics: four principles plus attention to scope', *The BMJ*, 309(6948), pp. 184-188.
- Gioia, D., Autrey, S., Drapalski, A. L., Glynn, S., Cohen, A. N. & Dixon, L. B. (2014).
 'Veterans' views of a shared decision-making process: a qualitative substudy of REORDER', *American Journal of Psychiatric Rehabilitation*, 17(4), pp. 348-364.
- Given, L. M. (2016). *100 Questions (and answers) about qualitative research*. Thousand Oaks: SAGE Publications.
- Glaser, B. G. & Strauss, A. (1967). *The discovery of grounded theory: strategies for qualitative research*. Chicago: Aldine.
- Gopalakrishnan, S. & Ganeshkumar, P. (2013). 'Systematic reviews and meta-analysis: understanding the best evidence in primary healthcare', *Journal of Family Medicine and Primary Care*, 2(1), pp. 9-14.
- Goscha, R. & Rapp, C. (2015). 'Exploring the experiences of client involvement in medication decisions using a shared decision making model: results of a qualitative study', *Community Mental Health Journal*, 51(3), pp. 267-274.
- Goss, C., Moretti, F., Mazzi, M. A., Del Piccolo, L., Rimondini, M. & Zimmermann, C. (2008). 'Involving patients in decisions during psychiatric consultations', *The British Journal of Psychiatry*, 193(5), pp. 416-421.
- Grant, M. J. & Booth, A. (2009). 'A typology of reviews: an analysis of 14 review types and associated methodologies', *Health Information & Libraries Journal*, 26(2), pp. 91-108.
- Greaney, A. M., Sheehy, A., Heffernan, C., Murphy, J., Mhaolrunaigh, S. N., Heffernan,E. & Brown, G. (2012). 'Research ethics application: a guide for the novice researcher', *British Journal of Nursing*, 21(1), pp. 38-43.

- Green, B. N., Johnson, C. D. & Adams, A. (2006). 'Writing narrative literature reviews for peer-reviewed journals: secrets of the trade', *Journal of Chiropractic Medicine*, 5(3), pp. 101-117.
- Green, J. & Thorogood, N. (2009). *Qualitative methods for health research*. 2nd edn. London: SAGE Publications.
- Greenhalgh, T. (2010). *How to read a paper: the basics of evidence-based medicine*. 4th edn. Oxford: Wiley-Blackwell.
- Grim, K., Rosenberg, D., Svedberg, P. & Schön, U.-K. (2016). 'Shared decision-making in mental health care—A user perspective on decisional needs in communitybased services', *International Journal of Qualitative Studies on Health and Wellbeing*, 11(1), p. 30563.
- Guba, E. G. (1990). The aradigm dialog. Newbury Park: SAGE Publications.
- Guba, E. G. & Lincoln, Y. (1994). 'Competing paradigms in qualitative research', in Denzin, N. & Lincoln, Y. (eds.) *Handbook of qualitative research*. Thousand Oaks: SAGE Publications, pp. 105-117.
- Gulmezoglu, A. M., Chandler, J., Shepperd, S. & Pantoja, T. (2013). Reviews of qualitative evidence: a new milestone for cochrane. *Cochrane Database of Systematic Reviews* [Online], 8(11). Available at: https://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.ED000073/full (Accessed: 8 December 2016).
- Hamann, J., Cohen, R., Leucht, S., Busch, R. & Kissling, W. (2007). 'Shared decision making and long-term outcome in schizophrenia treatment', *Journal of Clinical Psychiatry*, 68(7), pp. 992-997.
- Hamann, J., Langer, B., Winkler, V., Busch, R., Cohen, R., Leucht, S. & Kissling, W. (2006). 'Shared decision making for in-patients with schizophrenia', *Acta Psychiatrica Scandinavica*, 114(4), pp. 265-273.
- Hamann, J., Mendel, R., Meier, A., Asani, F., Pausch, E., Leucht, S. & Kissling, W. (2011). "How to speak to your psychiatrist": shared decision-making training for inpatients with schizophrenia', *Psychiatric Services*, 62(10), pp. 1218-1221.
- Happell, B., Manias, E. & Roper, C. (2004). 'Wanting to be heard: mental health consumers' experiences of information about medication', *International Journal* of Mental Health Nursing, 13(4), pp. 242-248.
- Harter, M., van der Weijden, T. & Elwyn, G. (2011). 'Policy and practice developments in the implementation of shared decision making: an international perspective',

Zeitschrift für Evidenz, Fortbildung und Qualität im Gesundheitswesen, 105(4), pp. 229-233.

- Hauser, K., Koerfer, A., Kuhr, K., Albus, C., Herzig, S. & Matthes, J. (2015). 'Outcomerelevant effects of shared decision making: a systematic review', *Deutsches Ärzteblatt International*, 112(40), pp. 665-671.
- Hayes, D., Edbrooke-Childs, J., Town, R., Wolpert, M. & Midgley, N. (2018). 'Barriers and facilitators to shared decision making in child and youth mental health: clinician perspectives using the Theoretical Domains Framework', *European Child & Adolescent Psychiatry*, 28(5), pp. 655-666.
- Health and Social Care Information Centre (2015). *Quality and outcomes framework, prevalence, achievements and exceptions report.* Leeds: The Government Statistical Service.
- Health Promotion Administration (2014). *The annual statistics of district public health centres* Taipei: Ministry of Health and Welfare in Taiwan.
- Hem, M. H., Gjerberg, E., Husum, T. L. & Pedersen, R. (2016). 'Ethical challenges when using coercion in mental healthcare: a systematic literature review', *Nursing Ethics*, 25(1), pp. 92-110.
- Heron, J. & Reason, P. (1997). 'A participatory inquiry paradigm', *Qualitative Inquiry*, 3(3), pp. 274-294.
- Hesse-Biber, S. N. (2007). *Handbook of feminist research: theory and praxis*. Thousand Oaks: SAGE Publications.
- Heyvaert, M., Hannes, K. & Onghena, P. (2016). Using mixed methods research synthesis for literature reviews. Thousand Oaks: SAGE Publications.
- Higgins, J. P. T. & Green, S. (2011). Cochrane Handbook for Systematic Reviews of Interventions [Online]. The Cochrane Collaboration. Available at: http://handbook.cochrane.org.
- Hill, S. A. & Laugharne, R. (2006). 'Decision making and information seeking preferences among psychiatric patients', *Journal of Mental Health*, 15(1), pp. 75-84.
- Hoffmann, T. C., Légaré, F., Simmons, M. B., McNamara, K., McCaffery, K., Trevena,
 L. J., Hudson, B., Glasziou, P. P. & Del Mar, C. B. (2014). 'Shared decision making: what do clinicians need to know and why should they bother?', *Medical Journal of Australia*, 201(1), pp. 35-39.
- Holloway, I. & Todres, L. (2007). 'Grounded theory', in Gerrish, K. & Lacey, A. (eds.) *The research process in nursing*. 5th edn. Oxford: Blackwell, pp. 192-207.

- Holloway, I. & Wheeler, S. (2010). *Qualitative research for nurses*. 3rd edn. Chichester: Wiley-Blackwell.
- Horsburgh, D. (2004). 'How, and when, can I restrain a patient?', *Postgraduate Medical Journal*, 80(939), pp. 7-12.
- Horsfall, J. M. (1995). 'Madness in our methods: nursing research, scientific epistemology', *Nursing Inquiry*, 2(1), pp. 2-9.
- Hou, W. H. (2015). 'Patient centered evidence-based healthcare: shared decision making', *Journal of Healthcare Quality*, 9(5), pp. 4-9.
- Hsieh, C. J. & Shiau, S. J. (2006). 'The current and prospective issues of community services for psychiatric rehabilitation in Taiwan', *The Journal of Psychiatric Mental Health Nursing*, 1(2), pp. 41-50.
- Hsieh, H. F. & Shannon, S. E. (2005). 'Three approaches to qualitative content analysis', *Qualitative Health Research*, 15(9), pp. 1277-1288.
- Hsu, W. P., Chang, R. Y., Lu, M. C., Chou, M. C. & Hsiao, P. C. (2015). 'Shared decision making in clinical practice', *Cheng Ching Medical Journal*, 11(5), pp. 24-29.
- Hu, L., Yin, X., Bao, X. & Nie, J. B. (2014). 'Chinese physicians' attitudes toward and understanding of medical professionalism: results of a national survey', *Journal* of Clinical Ethics, 25(2), pp. 135-147.
- Jacox, A., Suppe, F., Campbell, J. & Stashinko, E. (1999). 'Diversity in philosophical approaches', in Hinshaw, A. S., Feetham, S. L. & Shaver, J. L. F. (eds.) *Handbook* of clinical nursing research. London: SAGE Publications.
- James, K. & Quirk, A. (2017). 'The rationale for shared decision making in mental health care: a systematic review of academic discourse', *Mental Health Review Journal*, 22(3), pp. 152-165.
- Jensen, L. A. & Allen, M. N. (1996). 'Meta-synthesis of qualitative findings', *Qualitative Health Research*, 6(4), pp. 553-560.
- Jones, N., Corrigan, P. W., James, D., Parker, J. & Larson, N. (2013). 'Peer support, selfdetermination, and treatment engagement: A qualitative investigation', *Psychiatric Rehabilitation Journal*, 36(3), pp. 209-214.
- Joosten, E. A. G., de Jong, C. A. J., de Weert-van Oene, G. H., Sensky, T. & van der Staak, C. P. F. (2009). 'Shared decision-making reduces drug use and psychiatric severity in substance-dependent patients', *Psychotherapy and Psychosomatics*, 78(4), pp. 245-253.
- Joosten, E. A. G., DeFuentes-Merillas, L., de Weert, G. H., Sensky, T., van der Staak, C.P. F. & de Jong, C. A. J. (2008). 'Systematic review of the effects of shared

decision-making on patient satisfaction, treatment adherence and health status', *Psychotherapy and Psychosomatics*, 77(4), pp. 219-226.

- Joseph-Williams, N., Edwards, A. & Elwyn, G. (2014a). Power imbalance prevents shared decision making. *The BMJ* [Online], 348(g3178). Available at: <u>https://www.bmj.com/content/bmj/348/bmj.g3178.full.pdf</u> (Accessed: 15 July 2017).
- Joseph-Williams, N., Elwyn, G. & Edwards, A. (2014b). 'Knowledge is not power for patients: a systematic review and thematic synthesis of patient-reported barriers and facilitators to shared decision making', *Patient Education and Counseling*, 94(3), pp. 291-309.
- Joseph-Williams, N., Lloyd, A., Edwards, A., Stobbart, L., Tomson, D., Macphail, S., Dodd, C., Brain, K., Elwyn, G. & Thomson, R. (2017). Implementing shared decision making in the NHS: lessons from the MAGIC programme. *The BMJ* [Online], 357(j1744). Available at: https://www.bmj.com/content/bmj/357/bmj.j1744.full.pdf (Accessed: 14 April 2018).
- Kadam, P. & Bhalerao, S. (2010). 'Sample size calculation', *International Journal of Ayurveda Research*, 1(1), pp. 55-57.
- Kahlke, R. M. (2014). 'Generic qualitative approaches: pitfalls and benefits of methodological mixology', *International Journal of Qualitative Methods*, 13(1), pp. 37-52.
- Kessler, R. C., Berglund, P. A., Bruce, M. L., Koch, J. R., Laska, E. M., Leaf, P. J., Manderscheid, R. W., Rosenheck, R. A., Walters, E. E. & Wang, P. S. (2001).
 'The prevalence and correlates of untreated serious mental illness', *Health Services Research*, 36(6 Pt 1), pp. 987-1007.
- Kiesler, D. J. & Auerbach, S. M. (2006). 'Optimal matches of patient preferences for information, decision-making and interpersonal behavior: evidence, models and interventions', *Patient Education and Counseling*, 61(3), pp. 319-341.
- King, N. (2004). 'Using templates in the thematic analysis of text', in Cassell, C. & Symon, G. (eds.) *Essential guide to qualitative methods in organizational research* London: SAGE Publications, pp. 257–270.
- Kon, A. A. (2010). 'The shared decision-making continuum', *The Journal of the American Medical Association*, 304(8), pp. 903-904.

- Korstjens, I. & Moser, A. (2018). 'Series: Practical guidance to qualitative research. Part
 4: trustworthiness and publishing', *European Journal of General Practice*, 24(1),
 pp. 120-124.
- Koster, J. (2014). PubMed PubReMiner version 1.14. Amsterdam: Department of Oncogenomics, AMC. Available at: https://hgserver2.amc.nl/cgibin/miner/miner2.cgi (Accessed: 14 May 2019).
- Krefting, L. (1991). 'Rigor in qualitative research: the assessment of trustworthiness', *The American Journal of Occupational Therapy*, 45(3), pp. 214-222.
- Kuhn, T. (1970). *The structure of scientific revolutions*. Chicago: University of Chicago Press.
- Kvale, S. & Brinkmann, S. (2009). *Interview: learing the crafts of qualitative research interviewing*. 2nd edn. London: SAGE Publications.
- Lan, J. Y. (2017). 'Achieving and sustaining universal health coverage: fiscal reform of the National Health Insurance in Taiwan', *Applied Health Economics and Health Policy*, 15(6), pp. 717-731.
- Laugharne, R., Priebe, S., McCabe, R., Garland, N. & Clifford, D. (2012). 'Trust, choice and power in mental health care: experiences of patients with psychosis', *International Journal of Social Psychiatry*, 58(5), pp. 496-504.
- Légaré, F., Adekpedjou, R., Stacey, D., Turcotte, S., Kryworuchko, J., Graham, I. D., Lyddiatt, A., Politi, M. C., Thomson, R., Elwyn, G. & Donner-Banzhoff, N. (2018). Interventions for increasing the use of shared decision making by healthcare professionals. *Cochrane Database of Systematic Reviews* [Online], (7). Available at: https://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.CD006732.pub4/

full (Accessed: 3 July 19).

- Légaré, F., Ratté, S., Gravel, K. & Graham, I. D. (2008). 'Barriers and facilitators to implementing shared decision-making in clinical practice: update of a systematic review of health professionals' perceptions', *Patient Education and Counseling*, 73(3), pp. 526-535.
- Légaré, F. & Stacey, D. (2009). 'Shared decision-making: the implications for health care teams and practice', in Edwards, A. & Elwyn, G. (eds.) *Shared decision-making in health care*. 2 edn. Oxford: Oxford University Press, pp. 23-28.
- Légaré, F. & Witteman, H. O. (2013). 'Shared decision making: examining key elements and barriers to adoption into routine clinical practice', *Health Affairs*, 32(2), pp. 276-284.

- Levinson, W., Kao, A., Kuby, A. & Thisted, R. A. (2005). 'Not all patients want to participate in decision making: a national study of public preferences', *Journal of General Internal Medicine*, 20(6), pp. 531-535.
- Liao, H. H., Liang, H. W., Chen, H. C., Chang, C. I., Wang, P. C. & Shih, C. L. (2017).
 'Shared decision making in Taiwan', *Zeitschrift für Evidenz, Fortbildung und Qualität im Gesundheitswesen*, 123-124(2017), pp. 95-98.
- Lincoln, Y. S. & Guba, E. G. (1985). *Naturalistic inquiry*. Newburry Park: SAGE Publications.
- Ljungberg, A., Denhov, A. & Topor, A. (2016). 'Non-helpful relationships with professionals-a literature review of the perspective of persons with severe mental illness', *Journal of Mental Health*, 25(3), pp. 267-277.
- LoBiondo-Wood, G. & Haber, J. (2013). Nursing research: methods and critical appraisal for evidence-based practice. 8 th edn. Missouri: Mosby.
- Loh, A., Simon, D., Wills, C. E., Kriston, L., Niebling, W. & Härter, M. (2007). 'The effects of a shared decision-making intervention in primary care of depression: a cluster-randomized controlled trial', *Patient Education and Counseling*, 67(3), pp. 324-332.
- Mack, L. (2010). 'The philosophical underpinnings of educational research', *Polyglossia*, 19, pp. 5-11.
- Mack, N., Woodsong, C., MacQueen, K., Guest, G. & Namey, E. (2005). *Qualitative research methods: a data collector's field guide*. Research Triangle Park: Family Health International.
- Major, C. & Savin-Baden, M. (2010). An introduction to qualitative research synthesis: managing the information explosion in social science research. New York: Routledge.
- Makoul, G. & Clayman, M. L. (2006). 'An integrative model of shared decision making in medical encounters', *Patient Education and Counseling*, 60(3), pp. 301-312.
- Malm, U., Ivarsson, B., Allebeck, P. & Falloon, I. R. (2003). 'Integrated care in schizophrenia: A 2-year randomized controlled study of two community-based treatment programs', *Acta Psychiatrica Scandinavica*, 107(6), pp. 415-423.
- Marcus, C. (2014). 'Strategies for improving the quality of verbal patient and family education: a review of the literature and creation of the EDUCATE model', *Health Psychology and Behavioral Medicine*, 2(1), pp. 482-495.
- Marshall, M., Crowther, R., Almaraz-Serrano, A., Creed, F., Sledge, W., Kluiter, H., Roberts, C., Hill, E., Wiersma, D., Bond, G. R., Huxley, P. & Tyrer, P. (2001).

'Systematic reviews of the effectiveness of day care for people with severe mental disorders: (1) acute day hospital versus admission; (2) vocational rehabilitation; (3) day hospital versus outpatient care', *Health Technology Assessment*, 5(21), pp. 1-75.

- Marshall, M. N. (1996). 'Sampling for qualitative research', *Family Practice*, 13(6), pp. 522-525.
- Martínez-Mesa, J., González-Chica, D. A., Duquia, R. P., Bonamigo, R. R. & Bastos, J.
 L. (2016). 'Sampling: how to select participants in my research study?', *Anais* Brasileiros de Dermatologia, 91(3), pp. 326-330.
- Mason, M. (2010). 'Sample size and saturation in PhD studies: using qualitative interviews', *Forum: Qualitative Social Research*, 11(3), p. Art. 8.
- Matthias, M. S., Salyers, M. P., Rollins, A. L. & Frankel, R. M. (2012). 'Decision making in recovery-oriented mental health care', *Psychiatric Rehabilitation Journal*, 35(4), pp. 305-314.
- Maxwell, J. A. (2009). 'Designing a qualitative study', in Bickman, L. & Rog, D., J. (eds.)
 The SAGE handbook of applied social research methods. 2nd ed. Thousand Oaks:
 SAGE Publications, pp. 214-253.
- Mays, N. & Pope, C. (2000). 'Qualitative research in health care: assessing quality in qualitative research', *The BMJ*, 320(7226), pp. 50-52.
- McInnes, R. J. & Chambers, J. A. (2008). 'Supporting breastfeeding mothers: qualitative synthesis', *Journal of Advanced Nursing*, 62(4), pp. 407-427.
- Mead, N. & Bower, P. (2000). 'Patient-centredness: a conceptual framework and review of the empirical literature', *Social Science & Medicine*, 51(7), pp. 1087-1110.
- Merriam, S. B. (1998). *Qualitative research and case study applications in education*. San Francisco: Jossey-Bass.
- Meyer, J. (2000). 'Using qualitative methods in health related action research', *The BMJ*, 320(7228), pp. 178-181.
- Mhaskar, R., Emmanuel, P., Mishra, S., Patel, S., Naik, E. & Kumar, A. (2009). 'Critical appraisal skills are essential to informed decision-making', *Indian Journal of Sexually Transmitted Diseases*, 30(2), pp. 112-119.
- Miles, B. (2010). 'Discourse analysis', in Salkind, N. J. (ed.) Encyclopedia of research design. Thousand Oaks: SAGE Publications, pp. 368-370
- Miller, J. & Classner, B. (2011). 'The "inside" and the "outside": finding realities in interview', in Sliverman, D. (ed.) *Qualitative research*. 3rd edn. London: SAGE Publications, pp. 131-148.

- Morse, J. M., Barrett, M., Mayan, M., Olson, K. & Spiers, J. (2002). 'Verification strategies for establishing reliability and validity in qualitative research', *International Journal of Qualitative Methods*, 1(2), pp. 13-22.
- Morton, R. L., Tong, A., Howard, K., Snelling, P. & Webster, A. C. (2010). The views of patients and carers in treatment decision making for chronic kidney disease: Systematic review and thematic synthesis of qualitative studies. *The BMJ* [Online], 340(c112). Available at: http://www.bmj.com/content/bmj/340/bmj.c112.full.pdf (Accessed: 22 May 2016).
- Munhall, P. L. (1993). "Unknowing': toward another pattern of knowing in nursing', *Nursing Outlook*, 41(3), pp. 125-128.
- Munson, R. (2004). *Intervention and reflection: basic issues In medical ethics*. 7th edn. Victoria: Thomson Wadsworth.
- Munyaradzi, M. (2012). 'Critical reflections on the principle of beneficence in biomedicine', *The Pan African Medical Journal*, 11(29), pp. 1-5.
- Murgic, L., Hébert, P. C., Sovic, S. & Pavlekovic, G. (2015). 'Paternalism and autonomy:
 Views of patients and providers in a transitional (post-communist) country', *BMC Medical Ethics*, 16(65), pp. 1-9.
- Murphy, E., Dingwall, R., Greatbatch, D., Parker, S. & Watson, P. (1998). 'Qualitative research methods in health technology assessment: a review of the literature', *Health Technology Assessment*, 2(16), pp. iii-ix, 1-274.
- National Health Insurance Administration in Taiwan (2015). The current situation of mental health services and resourses in Taiwan. Taipei: Executive Yuan. Available at: www.mohw.gov.tw/cht/DOS/DisplayStatisticFile.aspx?d=50230 (Accessed: 30 May 2016).
- National Health Insurance Administration in Taiwan (2016a). *Introdction to National Health Insurance in Taiwan*. Taipei: Executive Yuan. Available at: http://www.nhi.gov.tw/index2015.aspx (Accessed: 10 May 2016).
- National Health Insurance Administration in Taiwan (2016b). *The statistical report of IC cards of severe illness on March of 2015*. Taipei: Executive Yuan. Available at:

http://www.nhi.gov.tw/webdata/webdata.aspx?menu=17&menu_id=661&webd ata_id=805&WD_ID=689 (Accessed: 12 May 2016).

National Health Insurance Administration in Taiwan (2016c). The statistics of numbersofhealthservices.Taipei:ExecutiveYuan.Availableat:

https://dep.mohw.gov.tw/DOS/cp-1735-3244-113.html (Accessed: 10 May 2016).

- National Institute for Health and Care Excellence (2017). *Violent and aggressive behaviours in people with mental health problems*, London: NICE. Available at: <u>https://www.nice.org.uk/guidance/qs154/resources/violent-and-aggressive-</u> <u>behaviours-in-people-with-mental-health-problems-pdf-75545539974853</u>.
- National Institute for Health and Care Excellence (2019). *Shared decision making*, London: NICE. Available at <u>https://www.nice.org.uk/advice/ktt23/chapter/Key-points</u>.
- National Mental Health Consumer and Carer Forum (2009). Ending seclusion and restraint in Australian mental health services., Deakin West: National Mental Health Consumer and Carer Forum.
- Nelson, G., Lord, J. & Ochocka, J. (2001). Shifting the paradigm in community mental
- health: toward rmpowerment and community. Buffalo: University of Toronto Press.
- Neuman, L. W. (2011). 'The meanings of methodology', in Neuman, L. W. (ed.) Social research methods: qualitative and quantitative approaches. 3rd edn. Harlow: Pearson, pp. 90-122.
- Ng, C. J., Lee, Y. K., Lee, P. Y. & Abdullah, K. L. (2013). 'Health innovations in patient decision support: Bridging the gaps and challenges', *The Australasian Medical Journal*, 6(2), pp. 95-99.
- Noblit, G. W. & Hare, R. D. (1988). *Meta-ethnography: synthesizing qualitative studies*. London: SAGE Publications.
- Nowell, L. S., Norris, J. M., White, D. E. & Moules, N. J. (2017). 'Thematic analysis: striving to meet the trustworthiness criteria', *International Journal of Qualitative Methods*, 16(1), pp. 1-13.
- Noyes, J. & Lewin, S. (2011). 'Chapter 5: Extracting qualitative evidence', in Supplementary guidance for inclusion of qualitative research in cochrane systematic reviews of interventions [Online]. Version. Available at: http://cqrmg.cochrane.org/supplemental-handbook-guidance.
- O'Connor, A. (2001). 'Using patient decision aids to promote evidence-based decision making', *Evidence Based Medicine*, 6(4), pp. 100-102.
- Orb, A., Eisenhauer, L. & Wynaden, D. (2001). 'Ethics in qualitative research', *Journal* of Nursing Scholarship, 33(1), pp. 93-96.
- Paavilainen, E., Lepisto, S. & Flinck, A. (2014). 'Ethical issues in family violence research in healthcare settings', *Nursing Ethics*, 21(1), pp. 43-52.

- Paré, G. & Kitsiou, S. (2017). 'Methods for literature review', in Lau, F. & Kuziemsky,
 C. (eds.) *Handbook of eHealth evaluation: An evidence-based approach*.
 Victoria, BC: University of Victoria Library Press, pp. 151-173.
- Patel, S. R., Bakken, S. & Ruland, C. (2008). 'Recent advances in shared decision making for mental health', *Current Opinion in Psychiatry*, 21(6), pp. 606-612.
- Patton, M. Q. (1999). 'Enhancing the quality and credibility of qualitative analysis', *Health Services Research*, 34(5), pp. 1189-1208.
- Patton, M. Q. (2002). *Qualitative research & evaluation methods*. 3rd edn. Thousand Oaks: SAGE Publications.
- Percy, W. H., Kostere, K. & Kostere, S. (2015). 'Generic qualitative research in psychology', 20(2), pp. 76-85.
- Petticrew, M. (2001). 'Systematic reviews from astronomy to zoology: myths and misconceptions', *The BMJ*, 322(7278), pp. 98-101.
- Petticrew, M. & Roberts, H. (2005). *Systematic reviews in the social sciences: a practical guide*. Malden: Wiley-Blackwell.
- Phillippi, J. & Lauderdale, J. (2018). 'A guide to field notes for qualitative research: context and conversation', *Qualitative Health Research*, 28(3), pp. 381-388.
- Polit, D. F. & Beck, C. T. (2014). *Essentials of nursing research: appraising evidence for nursing practice*. 4th edn. Philadelphia: Lippincott Williams and Wilkins.
- Polit, D. F. & Tatano Beck, C. (2004). *Nursing research: principles and methods*. 7th edn. Philadelphia: Lippincott Williams & Wilkins.
- Pope, C., van Royen, P. & Baker, R. (2002). 'Qualitative methods in research on healthcare quality', *Quality & Safety in Health Care*, 11(2), pp. 148-152.
- President's Commission for the Study of Ethical Problems in Medicine and Behavioral Research (1982). *Making health care decision: a report on the ethical and legal implications of informed consent in the patient-practitioner relationship.* Washington: Government Printing Office.
- Pring, R. (2000). Philosophy of educational research. London: Continuum.
- Quirk, A., Chaplin, R., Lelliott, P. & Seale, C. (2012). 'How pressure is applied in shared decisions about antipsychotic medication: a conversation analytic study of psychiatric outpatient consultations', *Sociology of Health & Illness*, 34(1), pp. 95-113.
- Rahman, M. (2016). 'The advantages and disadvantages of using qualitative and quantitative approaches and methods in language "testing and assessment"
research: a literature review', *Journal of Education and Learning*, 6(1), pp. 102-112.

- Reeves, S., Kuper, A. & Hodges, B. D. (2008). 'Qualitative research methodologies: ethnography', *The BMJ*, 7(337), p. a1020.
- Richardson, W. S., Wilson, M. C., Nishikawa, J. & Hayward, R. S. (1995). 'The wellbuilt clinical question: a key to evidence-based decisions', *ACP Journal Club*, 123(3), pp. A12-A13.
- Ring, N., Ritchie, K., Mandava, L. & Jepson, R. (2011). A guide to synthesising qualitative research for researchers undertaking health technology assessments and systematic reviews, Glasgow: Quality Improvement Scotland (NHS QIS). Available http://www.healthcareimprovementscotland.org/programmes/clinical_cost_eff ectiveness/programme_resources/synth_qualitative_research.aspx (Accessed: 27 March 2016).
- Rosenman, R., Tennekoon, V. & Hill, L. G. (2011). 'Measuring bias in self-reported data', International Journal of Behavioural & Healthcare Research, 2(4), pp. 320-332.
- Rowley, J. & Slack, F. (2004). 'Conducting a literature review', *Management Research News*, 27(6), pp. 31-39.
- Salzburg Global Seminar (2011). Salzburg statement on shared decision making. *The BMJ* [Online], 342(d1745). Available at: <u>https://www.bmj.com/content/bmj/342/bmj.d1745.full.pdf</u> (Accessed: 12 May 2016).
- Sandelowski, M. (1993). 'Rigor or rigor mortis: the problem of rigor in qualitative research revisited', *Advances in Nursing Science*, 16(2), pp. 1-8.
- Sandelowski, M. (1994). 'We are the stories we tell: narrative knowing in nursing practice', *Journal of Holistic Nursing*, 12(1), pp. 23-33.
- Sashidharan, S. P. & Saraceno, B. (2017). 'Is psychiatry becoming more coercive?', *The BMJ*, 357(j2904), pp. 1-2.
- Saya, A., Brugnoli, C., Piazzi, G., Liberato, D., Di Ciaccia, G., Niolu, C. & Siracusano, A. (2019). 'Criteria, procedures, and future prospects of involuntary treatment in psychiatry around the world: a narrative review', *Frontiers in Psychiatry*, 10(271), pp. 1-22.
- Scholl, I. & Barr, P. J. (2017). 'Incorporating shared decision making in mental health care requires translating knowledge from implementation science', World Psychiatry, 16(2), pp. 160-161.

- Scotland, J. (2012). 'Exploring the philosophical underpinnings of research: relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms', *English Language Teaching*, 5(9), pp. 9-16.
- Seale, C., Chaplin, R., Lelliott, P. & Quirk, A. (2006). 'Sharing decisions in consultations involving anti-psychotic medication: a qualitative study of psychiatrists' experiences', *Social Science & Medicine*, 62(11), pp. 2861-2873.
- Seers, K. (2012). 'What is a qualitative synthesis?', *Evidence-Based Nursing*, 15(4), p. 101.
- Seers, K. (2015). 'Qualitative systematic reviews: their importance for our understanding of research relevant to pain', *British Journal of Pain*, 9(1), pp. 36-40.
- Shenton, A. K. (2004). 'Strategies for ensuring trustworthiness in qualitative research projects', *Education for Information*, 22(2), pp. 63-75.
- Shepherd, A., Shorthouse, O. & Gask, L. (2014). 'Consultant psychiatrists' experiences of and attitudes towards shared decision making in antipsychotic prescribing, a qualitative study', *BMC Psychiatry*, 14(127), pp. 1-10.
- Sheridan, S. L., Harris, R. P. & Woolf, S. H. (2004). 'Shared decision making about screening and chemoprevention', *American Journal of Preventive Medicine*, 26(1), pp. 56-66.
- Shih, C., Su, Y., Liao, C. & Lin, J. (2010). Patterns of medical pluralism among adults: results from the 2001 national health interview survey in Taiwan. *BMC Health Services Research* [Online], 10(191). Available at: http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2910697/ (Accessed: 2 May 2016).
- Shu, B., Lung, F., Lu, Y., Chase, G. A. & Pan, P. (2001). 'Care of patients with chronic mental illness: comparison of home and half-way house care', *International Journal of Social Psychiatry*, 47(2), pp. 52-62.
- Slade, M. (2017). 'Implementing shared decision making in routine mental health care', World Psychiatry, 16(2), pp. 146-153.
- Smith, J. & Noble, H. (2014). 'Bias in research', *Evidence Based Nursing*, 17(4), pp. 100-101.
- Spencer, L., Ritchie, J., O'Connor, W., Morrell, G. & Ormston, R. (2014). 'Analysis: practices, principles and processes', in Ritchie, J., Lewis, J., McNaughton Nicholls, C. & Ormston, R. (eds.) *Qualitative research practice: a guide for*

social science students and researchers. 2nd edn. London: SAGE Publications, pp. 199-218.

- Stacey, D., Légaré, F., Col, N. F., Bennett, C. L., Barry, M. J., Eden, K. B., Holmes-Rovner, M., Llewellyn-Thomas, H., Lyddiatt, A., Thomson, R., Trevena, L. & Wu, J. H. (2014). 'Decision aids for people facing health treatment or screening decisions', *Cochrane Database of Systematic Reviews*, 28(1), p. CD001431.
- Stacey, D., Legare, F., Pouliot, S., Kryworuchko, J. & Dunn, S. (2010). 'Shared decision making models to inform an interprofessional perspective on decision making: a theory analysis', *Patient Education and Counseling*, 80(2), pp. 164-172.
- Steadman, H. J., Osher, F. C., Robbins, P. C., Case, B. & Samuels, S. (2009). 'Prevalence of serious mental illness among jail inmates', *Psychiatric Services*, 60(6), pp. 761-765.
- Stern, C., Jordan, Z. & McArthur, A. (2014). 'Developing the review question and inclusion criteria', *The American Journal of Nursing*, 114(4), pp. 53-56.
- Stiggelbout, A. M., Pieterse, A. H. & De Haes, J. C. J. M. (2015). 'Shared decision making: concepts, evidence, and practice', *Patient Education and Counseling*, 98(10), pp. 1172-1179.
- Stovell, D., Morrison, A. P., Panayiotou, M. & Hutton, P. (2016). 'Shared treatment decision-making and empowerment-related outcomes in psychosis: systematic review and meta-analysis', *The British Journal of Psychiatry*, 209(1), pp. 23-28.
- Stroup, S., Appelbaum, P., Swartz, M., Patel, M., Davis, S., Jeste, D., Kim, S., Keefe, R., Manschreck, T., McEvoy, J. & Lieberman, J. (2005). 'Decision-making capacity for research participation among individuals in the CATIE schizophrenia trial', *Schizophrenia Research*, 80(1), pp. 1-8.
- Substance Abuse and Mental Health Service Administraction (2010). Shared decisionmaking in mental health care: practice, research, and further directions. Rockville: U.S. Department of Health & Human Services.
- Sutton, J. & Austin, Z. (2015). 'Qualitative research: data collection, analysis, and management', *The Canadian Journal of Hospital Pharmacy*, 68(3), pp. 226-231.
- Sica, G. T. (2006). 'Bias in research studies', *Radiology*, 238(3), pp. 780-789.
- Tacconelli, E. (2010). Systematic reviews: CRD's guidance for undertaking reviews in health care, York: Centre for Reviews and Dissemination, University of York. Available at: https://www.york.ac.uk/media/crd/Systematic_Reviews.pdf (Accessed: 15 May 2016).

- Tanenbaum, S. J. (2008). 'Consumer perspectives on information and other inputs to decision-making: implications for evidence-based practice', *Community Mental Health Journal*, 44(5), pp. 331-335.
- Tang, W. H. (1997). 'The political process of mental health law in Taiwan: the politycentered perspective', *Chinese Journal of Mental Health*, 10(1), pp. 1-27.
- Teddlie, C. & Tashakkori, A. (2009). Foundations of mixed methods research: Integrating quantitative and qualitative approaches in the social and behavioral sciences. London: SAGE Publications.
- Teddlie, C. & Yu, F. (2007). 'Mixed methods sampling: A typology with examples', *Journal of Mixed Methods Research*, 1(1), pp. 77-100.
- Terry, J. & Coffey, M. (2019). 'Too busy to talk: examining service user involvement in nursing work', *Issues in Mental Health Nursing*, 40(11), pp. 957-965.
- The Dispute of Committee in the National Health Insurance (2008). *The dispute of extended hospitalization at acute psychiatric admission*. Taipei: Ministry of Health and Welfare in Taiwan. Available at: https://service.mohw.gov.tw (Accessed: 25 April 2016).
- The Health Foundation (2012a). *Evidence scan: quality improvement training for healthcare professionals*. London: The Health Foundation. Available at: https://www.health.org.uk/publications/quality-improvement-training-forhealthcare-professionals.
- The Health Foundation (2012b). Evidence: helping people share decisions. London: The
Health Foundation. Available at:
https://www.health.org.uk/sites/default/files/HelpingPeopleShareDecisionMaki
ng.pdf (Accessed: 25 May 2016).
- The Civic Code.
- The Taiwanese Mental Health Law.
- Thomas, J. & Harden, A. (2008). Methods for the thematic synthesis of qualitative research in systematic reviews. *BMC Medical Research Methodology* [Online], 8(45).
 Available at: https://bmcmedresmethodol.biomedcentral.com/articles/10.1186/1471-2288-8-

45 (Accessed: 25 May 2016).

- Thomson, R. (2009). 'Evidence based implementation of complex interventions', *The BMJ*, 339(b3124), pp. 355-356.
- Thomson, S. (2011). 'Sample size and grounded theory', *The Journal of Administration* & *Governance*, 5(1), pp. 45-52.

- Thorne, S. (2000). 'Data analysis in qualitative research', *Evidence Based Nursing*, 3(3), pp. 68-70.
- Tobin, G. A. & Begley, C. M. (2004). 'Methodological rigour within a qualitative framework', *Journal of Advanced Nursing*, 48(4), pp. 388-396.
- Tong, A., Sainsbury, P. & Craig, J. (2007). 'Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups', *International Journal for Quality in Health Care*, 19(6), pp. 349-357.
- Towle, A., Godolphin, W., Grams, G. & Lamarre, A. (2006). 'Putting informed and shared decision making into practice', *Health Expectations*, 9(4), pp. 321-332.
- Travers, M. (2008). *Qualitative research through case studies*. London: SAGE Publications.
- Tuli, F. (2010). 'The basis of distinction between qualitative and quantitative research in social science', *Ethiopian Journal of Education and Sciences*, 6(1), pp. 97-108.
- United Nations (2017). *World economic situation and prospects 2017*. New York: United Nations Publication (Accessed: 10 April 2018).
- Urquhart, C. (2013). *Grounded theory for qualitative research: a practical guide*. Thousand Oaks: SAGE Publications.
- Vaismoradi, M., Turunen, H. & Bondas, T. (2013). 'Content analysis and thematic analysis: implications for conducting a qualitative descriptive study', *Nursing & Health Sciences*, 15(3), pp. 398-405.
- van Manen, M. (2007). 'Phenomenology of practice', *Phenomenology & Practice*, 1(1), pp. 11-30.
- Velligan, D. I., Roberts, D. L., Sierra, C., Fredrick, M. M. & Roach, M. J. (2016). 'What patients with severe mental illness transitioning from hospital to community have to say about care and shared decision-making', *Issues in Mental Health Nursing*, 37(6), pp. 400-405.
- Volk, R. J., Llewellyn-Thomas, H., Stacey, D. & Elwyn, G. (2013). 'Ten years of the International Patient Decision Aid Standards Collaboration: evolution of the core dimensions for assessing the quality of patient decision aids', *BMC Medical Informatics and Decision Making*, 13(Suppl 2), p. S1.
- Wallendorf, M. & Belk, R. W. (1989). 'Assessing trustworthiness in naturalistic consumer research', *Association for Consumer Research*, pp. 69-84.
- Weaver, K. & Olson, J. K. (2006). 'Understanding paradigms used for nursing research', *Journal of Advanced Nursing*, 53(4), pp. 459-469.

- Welsh, E. (2002). 'Dealing with data: using NVivo in the qualitative data analysis process', Forum Qualitative Sozialforschung/Forum: Qualitative Social Research, 3(2), p. Art. 26.
- Wetherell, M., Taylor, S. & Yates, S. J. (2001). *Discourse as data: a guide for analysis*. London: SAGE Publications.
- Wharne, S., Langdridge, D. & Motzkau, J. (2012). 'Decision-making in mental healthcare: a phenomenological investigation of service user perspectives', *The Humanistic Psychologist*, 40(2), pp. 153-165.
- White, A. & Schmidt, K. (2005). 'Systematic literature reviews', *Complementary Therapies in Medicine*, 13(1), pp. 54-60.
- WHO Regional Office for Europe (2018). Outpatient contacts per person per year. Copenhagen: WHO Regional Office for Europe. Available at: <u>https://gateway.euro.who.int/en/indicators/hfa_543-6300-outpatient-contacts-per-person-per-year/</u> (Accessed: 21 June 2019).
- Woltmann, E. M. & Whitley, R. (2010). 'Shared decision making in public mental health care: perspectives from consumers living with severe mental illness', *Psychiatric Rehabilitation Journal*, 34(1), pp. 29-36.
- Wong, D. F. & Xuesong, H. (2011). 'Schizophrenia literacy among Chinese in Shanghai, China: a comparison with Chinese-speaking Australians in Melbourne and Chinese in Hong Kong', *Australian & New Zealand Journal of Psychiatry*, 45(7), pp. 524-531.
- Wu, H. Y. & Cheng, A. T. (2017). 'A history of mental healthcare in Taiwan', in Minas,
 H. & Lewis, M. (eds.) *Mental health in Asia and the Pacific: historical and cultural perspectives*. Boston: Springer, pp. 107-121.
- Wu, T., Majeed, A. & Kuo, K. N. (2010). 'An overview of the healthcare system in Taiwan', *London Journal of Primary Care*, 3(2), pp. 115-119.
- Yen, J. C. & Hsu, M. H. (2007). 'The evolution of doctor-patient rapport from the perspective of informed consent and co-decision-making', *Taipei City Medical Journal*, 4(8), pp. 643-649.
- Zamawe, F. C. (2015). 'The implication of using NVivo software in qualitative data analysis: evidence-based reflections', *Malawi Medical Journal*, 27(1), pp. 13-15.
- Zhuang, X., Wong, D. F., Cheng, C. & Pan, S. (2017). 'Mental health literacy, stigma and perception of causation of mental illness among Chinese people in Taiwan', *International Journal of Social Psychiatry*, 63(6), pp. 498-507.

Appendix 1: Extraction sheet for qualitative systematic review

To be extracted: information refers to perspectives of shared decision making from health users

Part A: Demographic information

General information

Title:

Year:

Name of authors:

Nation of data collection:

Aim:

Methodology

Study design:

Theoretical framework for the study

Data collection

Data analysis

Participant selection

Target population:

Number of participants:

Setting/context:

Sampling:

Recruitment method:

Inclusion criteria/Exclusion criteria:

Characteristics of study population

Gender:

Age:

Diagnosis:

Part B: Consolidated criteria for reporting qualitative studies (COREQ) Checklist

Don	nain 1 Research team and reflexivity		
	Characteristics		Comments
1.	Interviewer/facilitator	□Yes	
	Which author/s conducted the interview or focus group?	□No	
2.	Credentials	□Yes	
	What were the author's credentials? E.g. PhD, MD	□No	
3	Occupation	ΠYes	
5.	What was their occupation at the time of the study?	$\square N_0$	
4	Gender	\square Yes	
••	Was the author male or female?	\Box No	
5.	Experience and training	□Yes	
	What experience or training did the author have?	□No	
	Relationship with participants		
6	Relationship established	ΠYes	
0.	Was a relationship established prior to study	\Box No	
	commencement?		
7.	Participant knowledge of the interviewer	□Yes	
	What did the participants know about the author? e.g.	□No	
	personal goals, reasons for doing the research		
8.	Interviewer characteristics	□Yes	
	What characteristics were reported about the	□No	
	interviewer/facilitator? e.g. Bias, assumptions, reasons		
	and interests in the research topic		
Don	nain 2: study design		
	Theoretical framework		
9.	Methodological orientation and Theory	□Yes	
	What methodological orientation was stated to underpin	□No	
	the study? e.g. grounded theory, discourse analysis,		
	ethnography, phenomenology, content analysis		
	Participant selection		
10.	Sampling	□Yes	
- • •	How were participants selected? e.g. purposive.		
	convenience, consecutive, snowball		
11.	Method of approach	□Yes	
	How were participants approached? e.g. face-to-face,	□No	
	telephone, mail, email		
12.	Sample size	□Yes	
	How many participants were in the study?	□No	
13.	Non-participation	□Yes	
	How many people refused to participate or dropped out?	□No	
	Reasons?		
	Setting		
14.	Setting of data collection	□Yes	
	Where was the data collected? e.g. home, clinic,	□No	
	workplace		
15.	Presence of non-participants	□Yes	
	Was anyone else present besides the participants and	□No	
	researchers?		

16.	Description of sample	□Yes	
	What are the important characteristics of the sample? e.g.	□No	
	demographic data, date		
	Data collection		
17.	Interview guide	□Yes	
	Were questions, prompts, guides provided by the authors?	□No	
	Was it pilot tested?		
18.	Repeat interviews	□Yes	
	Were repeat interviews carried out? If yes, how many?	□No	
19.	Audio/visual recording	□Yes	
	Did the research use audio or visual recording to collect	□No	
	the data?		
20.	Field notes	□Yes	
	Were field notes made during and/or after the interview or	□No	
	focus group?		
21.	Duration	□Yes	
	What was the duration of the interviews or focus group?	□No	
22.	Data saturation	□Yes	
	Was data saturation discussed?	□No	
23.	Transcripts returned	□Yes	
	Were transcripts returned to participants for comment	□No	
	and/or correction?		
Don	nain 3: analysis and findings		
	Data analysis		
24.	Number of data coders	□Yes	
	How many data coders coded the data?		
25.	Description of the coding tree	□Yes	
	Did authors provide a description of the coding tree?		
26.	Derivation of themes	∐Yes	
	Were themes identified in advance or derived from the	LNO	
07			
27.	Software		
20	What software, if applicable, was used to manage the data?		
20.	Participant checking Did participants provide feedback on the findings?	\square I es	
	Reporting		
20	Orestetiene mesente l		
29.	Ware neutricipant quotations presented to illustrate the		
	the second device the second d		
	\mathbf{T}		
	themes / findings? Was each quotation identified? e.g. participant number		
30	Was each quotation identified? e.g. participant number		
30.	Was each quotation identified? e.g. participant number Data and findings consistent Was there consistency between the data presented and the	□Yes □No	
30.	Was each quotation identified? e.g. participant number Data and findings consistent Was there consistency between the data presented and the findings?	□Yes □No	
30.	Was each quotation identified? e.g. participant number Data and findings consistent Was there consistency between the data presented and the findings? Clarity of major themes	□Yes □No	
30. 31.	Was each quotation identified? e.g. participant number Data and findings consistent Was there consistency between the data presented and the findings? Clarity of major themes Were major themes clearly presented in the findings?	□Yes □No □Yes □No	
30. 31.	Was each quotation identified? e.g. participant number Data and findings consistent Was there consistency between the data presented and the findings? Clarity of major themes Were major themes clearly presented in the findings? Clarity of minor themes	□Yes □No □Yes □No □Yes	
30. 31. 32.	Was each quotation identified? e.g. participant number Data and findings consistent Was there consistency between the data presented and the findings? Clarity of major themes Were major themes clearly presented in the findings? Clarity of minor themes Is there a description of diverse cases or discussion of	□Yes □No □Yes □No □Yes □No	

Part C: Finding from the paper

Appendix 2: Topic guide for patients Detailed interview topic guide

Health services user perspectives of decision making in secondary mental health services in Taiwan

Introduction:

Go through the information sheet to introduce the details of the study, including purpose, method, process of the study, confidentiality, and the right to withdraw from the study at any time during the interview without giving any reason and up until the data is anonymised and analysed (approximately one month after the interview).

Ensure informed written consent is obtained.

Even if they refuse the use of direct quotes in any report or decline to receive the summary of findings of this study, they still can take part in this study.

The role of the author

Background information

Warm-up question

Could you please tell me your name?

Could you please tell me what shared decision making means to you?

Experience of the process of decision making

Can you tell me about your experience of how decisions are made in mental healthcare?

Prompts:

You take medication everyday could you please tell me how and who made this decision?

Could you please describe how you decide to move in the halfway house?

What exactly happened when decisions were being made about your treatment or care?

What was it like for you?

How did you feel about this? Why?

Do you receive the information you needed during decision making?

What was this?

How did you receive this?

Was anything missing?

How satisfied were you with the health professional during this process?

Probe (attitudes, behaviours and outcomes)

Preference

What level of involvement do service users want in decisions about their treatment and care?

Prompts:

When decisions arise in relation to your medication or your care plan, would you like to be involved in these decisions? Why?

What are potential benefits? What are the potential difficulties?

Are there any decisions you would not want to be involved in? (Probe specific examples)

Barriers to and facilitators of shared decision making

What barriers are there to involving patients in decision making in mental healthcare? (probe individual and cultural barriers)

What would help you be involved in decision making?

Prompts:

What are the challenges you encounter when faced with making decisions about your mental healthcare?

What specific things might help you be involved in decision making in mental healthcare?

How do you think you might influence the process of making decisions?

What role do mental health professionals currently have in decision making about treatment or care?

Any support? What kind of support?

What are mental health professionals/providers' attitudes towards involving patients in decision making in mental health care? Do you think this could have an impact on how involved patients are in the decision making process?

Is there anything else you think of which may be helpful or challenging when decision are made in relation to your treatment or care?

Is there anything you would like to add or share with me about your perspectives?

Additional probes will be used during interviews to encourage participants to explore, such as:

Can you say something more about....

Then what happened?

Do you have further examples of this?

Can you give me more detailed descriptions of what happened?

You said... What do you mean by that?

Conclusion

Is there anything you would like to add?

Thank you very much for participating.

Appendix 3: Topic guide for health professionals **Detailed interview topic guide**

Mental health professionals' perspectives of decision making in secondary mental health services in Taiwan

Introduction:

Go through the information sheet to introduce the details of the study, including purpose, method, process, confidentiality, and the right to withdraw from the study at any time during the interview without giving any reason and up until the data is anonymised and analysed (approximately one month after the interview).

Ensuring written informed consent is obtained.

Even if they refuse the use of direct quotes in any report or decline to receive the summary of findings of this study, they still can take part in this study.

Clarify the role of the author

Background information

Warm-up question

Could you please tell me about your role in secondary mental health services?

Attitude toward shared decision making

How do you feel about shared decision making in mental healthcare?

Prompts:

Do you have any experience of involving patients with mental illnesses in decision making? What happened? How do you feel about this?

When do you think it is acceptable to involve patients with mental illness in decision making on their treatment or care? Why?

Has involving patients in decision making been a negative or positive experience? Why?

Should patients and their carers be involved in making decisions on their treatment and care? Why/why not?

Advantages/disadvantages

Supports for professionals

What support would professionals need to undertake shared decision making successfully within mental health services?

Prompts:

What knowledge would you need to enable you to involve patients with mental illness in decision making?

What resources might be helpful during shared decision making in mental healthcare?

What kind of training might be helpful for staff to improve patient involvement in the decision making process?

Barriers to and facilitators of implementing shared decision making

Prompts:

What makes it most difficult for you to involve patients in making decisions about their treatment and care? Any there any other difficulties? (Probe individual and cultural factors).

What is your role in the decision making process? How do you think you might influence the process of making decisions? Why?

How does patient attitudes/behaviours influence your willingness and ability to involve them in decision making about their care/treatment?

What factors may have a negative effect on shared decision making in mental healthcare?

What factors may have a positive effect on shared decision making in mental healthcare?

Is there anything that could make it better or worse? Could you give examples?

Is there anything you would like to add or share with me about your perspectives?

Additional probes will be used during interviews to encourage participants to explore, such as:

Can you say something more about...

Then what happened?

Do you have further examples of this?

Can you give me more detailed descriptions of what happened?

You said... What do you mean by that?

Conclusion

Is there anything you would like to add?

Thank you very much for participating.

Appendix 4: Background of mental health patients



The University of Manchester

Background of mental health service users

Date:

Reference number:

Date of birth: _____

Gender:

 \Box Male \Box Female

Psychiatric diagnosis:

Onset age of mental illness: _____ years old

Duration of stay in the mental health service: _____ month(s) _____ year(s)

Appendix 5: Background of mental health professionals



The University of Manchester

Background of mental health professionals

Date:

Reference number:

Date of birth:

Gender:

 \Box Male \Box Female

Occupation

□ Psychiatrists □Mental health nurses □Nurse practitioners

Clinical Psychologists DOccupational Therapists

□Case Managers □Social Workers

What type of mental health service you work at: _____

Duration of working in mental health service: _____ month(s) _____ year(s)

Appendix 6: Initial themes

1. Patient Interviews

Passive role for patient

Main themes	Sub-themes
Professionals decide/Family decides	Decision about treatment and care
(power imbalance)	Decision about daily life
Lack of information	Professional knowledge about treatment
	Sense of patient involvement

Barriers to patient involvement

Main themes	Sub-themes	
Power of professional/Consequence of Staying in hospital for a long tim		
resisting professionals or families/Fear of	Changing of medication treatment	
being hospitalised	Involuntary treatment	
	Relapsing	
	Negative experience of making	
	own decision	
Social role of being patients in Taiwanese	wanese Well behaviour	
culture	Family pressure	
Mental illness/doubtful ability to make	Psychiatric symptoms	
right decisions	Loss control	
	Not able to make right decision	
A lack of communication between	Limited time resources	
professionals and patients	No privacy	

Patient desires different level of patient involvement

Main themes	Sub-themes
More information	Treatment and care
	Side effects of medication
More treatment options	Medication options
	Other treatment options
More discussion	Patient preference
	Professional recommendation
	Reasons for suggesting the treatment
	option
	Further plans
Make own decision	make own decision based on own
	preference

2. Professional Interviews

Main themes	Sub-themes
Concern about impaired decisional	Serious mental illness
ability	Psychiatric symptoms
•	Medication adherence
	Lack of insight
	Impaired cognitive function
	Lack of professional knowledge
Taiwanese submissive culture	Lack of motivation for involvement
	Powerful role of professional/discomfort
	of power exchange
	Family's negative attitude to patient
	involvement
A lack of information about shared	Never heard about shared decision
decision making	making
	Heard about shared decision making but
	does not fully understand all aspects of
	shared decision making
	No relevant learning resource
Mental health system problems	Staff shortage
	Limited time resources

Passive motivation to implement shared decision making

Lack of patient involvement in mental healthcare

Main themes	Sub-themes	
Decided by others	Professional decides treatment	
	Family decides daily life	
Limited information	No information	
	Wrong information	
Limited options	No options	
	Accepted options	

Moral sense of patient involvement

Main themes	Sub-themes
Basic human right	Respect to patient
	Well informed
	Autonomy
Uncertainty of treatment	Risk of treatment
	There is no 100% right answer
Empathy	Role play (if I were a patient)

Benefit to shared decision making/active patient involvement

Main themes	Sub-themes
Mutual understanding	Patient preference
	Patient condition

	Professional recommendation	
The therapeutic alliance	collaborative relationship between	
	patient and professionals	
	taking Responsibility	
	Setting recovery goal together	
Motivation	Engagement in treatment	

Appendix 7: Ethical approval from the University of Manchester



Ref 2017-2009-3356

26/06/2017

Dur Ms Chia-Yi Lin, , Prof Karina Lovell, Dr Laoise Renwick

Study Title: The decision-making process in secondary mental healthcare in Taiwan

University Research Ethics Committee 3

Iwrite to thank you for submitting the final version of your documents for your project to the Committee on 21/06/2017 20:49. I ampleased to confirm a favourable ethical opinion for the above research on the basis described in the application form and supporting documentation as submitted and approved by the Committee.

Please see below for a table of the tife, version numbers and dates of all the final approved documents for your project:

Document Type	File Name	Date	Version
Advertisement	Appendix 13 contact form	27/04/2017	1.1
Distress Protocol/Debrief Sheet	01062017-V2-distress policy	01/06/2017	2
Lone Worker Polic y/Procedure	01062017-V2-Appendix 25-lone worker assessment	01/06/2017	2
Lone Worker Polic y/Procedure	01062017-lone worker	01/06/2017	2
Additional does	01062017-V2-Appendix 25-lone worker assessment	01/06/2017	2
Questionnaire	Appendix 7&8 Background questionnaire-	01/06/2017	1
Additional does	The letter from the committee	01/06/2017	1
Topic Guide	08062017-Appendix 1&2-topic guide	08/06/2017	2
Additional does	12062017-V2-Social media recruitment text	12/06/2017	2
Additional does	a letter from a gatekeeper	12/06/2017	1
Additional does	12062017-V2-response letter	12/06/2017	2
Advertisement	21062017-V3-Appendix 9-12 letter and advert	21/06/2017	3
Participant Information Sheet	21062017-V4-Appendix 3&4 infomation sheet-2	21/06/2017	4
Consent Form	21062017-V4-Appendix 5&6 consent forms-2	21/06/2017	4
Additional does	21062017-Traditional Chinese version (Appendix 14-24)	21/06/2017	3
Additional does	21062017-response letter-Chiuyi, Lin	21/06/2017	1

This approval is effective for a period of five years however please note that it is only valid for the specifications of the research projectas outlined in the approved documentation set. If the project continues beyond the 5 year period or if you wish to propose any changes to the methodology or any other specific swithin the project, an application to seek an amendment must be submitted for review. Failure to do so could invalidate the insurance and constitute research misconduct.

You are reminded that, in accordance with University policy, any data carrying personal identifiers must be encrypted when not held on a secure university computer or kept secure ly as a hard copy in a location which is accessible only to those involved with the research.

Reporting Requirements:

You are required to report to us the following:

- 1. Amendments 2. Breaches and adverse events
- 3. Notification of progress/end of the study

Fee dback

It is our aim to provide a finely and efficient service that ensure stransparent, professional and proportionate ethical review of research with consistent outcomes, which is supported by clear, accessible guidance and training for applicants and committees. In order to assist us with our aim, we would be grateful if you would give your view of the service that you have received from us by completing a UREC Feedback Form. Instructions for completing this can be found in your approval email.

Page 1 of 2

Research Governance, Ethics and Integrity
2 nd Floor Christic Building
The University of Manchester
Oxford Road
Manchester
M13.9PL
Tel: 0161 275 2206 2674
Email: <u>concerch_obicaijon_anchester_acuk</u>

10 z We wish you every success with the research.

Yours sincerely,

Adran Jours.

Mr Adrian Jarvis

Secretary to University Research Ethics Committee 3

Page 2 of 2

長庚醫療財團法人人體試驗倫理委員會

臨床試驗同意證明書

地 址:105台北市我化北路199號 真:03-3494549 佳 聯絡人及電話: 萘宜君(03)3196200#3716 電子郵件信稿: yijiun@cgnh.org.tw 試驗名稱: 台灣精神康復之家病人之決策過程 本院業號: 201700879B0 試驗期間: 2017年7月10日~2018年2月28日 本次核准執行期間:2017年7月10日-2018年2月28日 主持人: 精神科護理師組 辜爾卉 內科NP1-N3(含)以下 共同主持人: 林秋谊 執行機構: 嘉義長庚醫院 核准之計畫書版本: 20170629 Version2 核准之同意書版本: 受訪者同意書: 20170629 Version3 核准之招募文宣版本: 20170606 Version1 通過日期:2017年7月4日 ※本研究將於2018年2月28日到期,請於試驗到期後三個月內繳交結案報告至本 會審查,若需要展延試驗期間者,請於到期前兩個月提出變更案,並繳交期中報 告,本會未核准之試驗期間,本研究不得繼續執行。

長庚醫療財團法人 人體試驗倫理委員會謝燦堂主席



中華民國 106 年 07 月 04 日

Chang Gung Medical Foundation

Institutional Review Board

199, TUNG HWA NORTH ROAD, TAIPEI, TAIWAN, 10507 REPUBLIC OF CHINA Tel: (03) 3196200 Fax: (03) 3494549

Date 2017/07/04

Protocol Title: The Decision-making Process for Service Users in Secondary Mental Health Services in Taiwan. IRB No.: 201700879B0 Principal Investigator(s): KU, YU-HUI Co-Investigator(s): Lin, Chiu-Yi Duration of Approval: From 2017/07/10 TO 2018/02/28 Approved Protocol: 20170629 Version2 Approved Informed Consent: 20170629 Version3 Advertisement: 20170606 Version1 Date of Approval: 2017/07/04

, was approved by the Institutional Review Board (the "IRB") of Chang Gung Medical Foundation on 2017/07/04. The IRB is organized and operates according to Good Clinical Practice and the applicable laws and regulations.

Sincerely Yours,

OT. I. Hstel

Tsang-Tang Hsieh,MD Chairman Institutional Review Board Chang Gung Medical Foundation

【主持人須知】

- 一、 實施人體研究計畫前,應擬定研究計畫,經人體試驗倫理委員會審查通過,始得為 之。另醫療法所稱之人體試驗案及應用人體生物資料庫檢體進行之案件,尚需經衛 生福利部核准,方可進行。人體試驗倫理委員會或主管機關命令中止/终止試驗案件 時,不得繼續執行。
- 二、 廠商贊助試驗於執行前,應協助試驗執行醫院完成臨床試驗合約書簽署,並提供本 會影本乙份。合約書簽署流程及注意事項應依本院之,廠商贊助計畫管理作業準則 」辦理。展延試驗期間者,經人體試驗倫理委員會同意後,應進行合約變更。
- 三、 試驗進行前,主持人應確實核對試驗計畫書、受試者同意書等之正確版本,以及人 體試驗倫理委員會與衛生福利部核准之試驗進行期間;醫療法所稱之人體試驗案 ,應於本院人體試驗倫理委員會與衛生福利部皆已核准,方可進行;並以人體試驗 倫理委員會核准之同意臨床試驗證明這日為試驗截止日。
- 四、應完全熟悉試驗藥品/醫療材料、醫療技術在試驗計畫書、最新主持人手冊及其他由 試驗委託者提供的相關資訊中描述的使用方法。
- 五、應明瞭並遵守「醫療法」、「人體試驗管理辦法」、「人體研究法」、「人體生物 資料庫管理條例」、「藥品優良臨床試驗準則」、「醫療器材優良臨床試驗準則」 等相關法規,以及本院「人體試驗作業管理辦法」、「人體生物資料庫管理辦法」、「研究材料外送作業準則」等之規定,善盡保護受試者之責任,並配合主管機關 、人體試驗倫理委員會及執行機構的查核。
- 六、 應確保所有協助臨床試驗的相關人員,對試驗計畫書及試驗藥品/醫療材料、醫療技 術有充分的了解,以及他們在臨床試驗中相關的責任和工作,並定期召開會議討論
- 七、 試驗主持人應負責所有臨床試驗相關的醫療決定。
- 八、 在受試者參加試驗與後續追蹤期間,試驗主持人應對受試者任何與試驗相關的不良 反應,提供充分的醫療照護,包括重要實驗室檢查值等。當試驗主持人察覺試驗期 間受試者有疾病需要醫療照護時,必須告知受試者。
- 九、 主持人須遵照「嚴重藥物不良反應通報辦法」及「藥品優良臨床試驗準則」、「人 體試驗管理辦法」之規定,於時效內向相關單位完成通報。發生非預期且相關之嚴 重藥品不良反應,試驗主持人應立即通知人體試驗倫理委員會,如為衛生福利部列 管案件,並需確認試驗委託者已通報主管機關;如為主持人自行發起之案件,主持 人需通報主管機關。但若試驗計畫書或其他文件明確排除者,不在此限。如為已上 市藥品試驗案發生藥品不良反應(ADR),應依規定至院內網頁之安全通報作業系統 完成通報。
- 十、 試驗主持人應依本院「人體試驗作業管理辦法」規定繳交期中報告及結案報告,未 繳者不得再申請新案,案件審議如有必要時須配合於會議列席報告。
- 十一、 受試者同意書使用原則須遵照「藥品優良臨床試驗準則」之規定,若具有重要性之 新資訊可能影響受試者之同意時,應修訂受試者同意書及提供受試者之任何其他書 面資料,並應立即告知受試者、法定代理人或有同意權之人,並重新取得書面同意 。修訂後之受試者同意書及提供受試者之任何其他書面資料,應先得到人體試驗倫 理委員會之核准;經主管機關核准進行之臨床試驗,並應得到主管機關之核准。
- 十二、 經人體試驗倫理委員會核准之計畫內容,凡於核准期間內任一變更(包含試驗計劃書、主持人手冊、受試者同意書、問卷、量表、執行院區、計畫主持人、試驗委託者等),應於執行前送變更案審查,於接獲書面核准函後,始可為之。另計劃主持人因故自行暫停、終止或結案時,應以書面通知本會。
- 十三、 試驗研究團隊執行試驗時發生未遵照審查通過之計畫書、主管機關所訂立之法令規 章或本會規定之情事,應遵照本院「人體試驗作業管理辦法」之規定,於時效內完 成通報。
- 十四、 人體試驗倫理委員會同意之試驗期間到期,但需要展延試驗期間者,應於有效期限 到期前2個月提出申請。
- 十五、 醫療法所稱人體試驗範圍之案件, 使入性檢查或治療,以及使用上市藥品之案件需 至『HIS系統-研究計畫-研究計畫申請-臨床試驗執行管控作業。登錄受試者資料。
- 十六、 試驗主持人於門診(住院)時發現受試者發生與試驗相關的不良反應,需轉診時,試 驗主持人應主動向受試者及家屬解釋,並協助轉診(會診),且於病歷(會診單)上詳 細記載本試驗目的、副作用、目前受試者病情及處置,並口頭告知轉診主治醫師。
- 十七、 若為多國多中心之案件,計畫主持人須確認第一個Kit之正確性。

十八、 應遵照「人體試驗管理辦法」、「醫療機構及醫事人員發布醫學新知或研究報告倫 理守則」及本院「研究成果發表於報掌媒體作業準則」規定,醫療法所稱人體試驗 之研究成果須經衛生福利部審核通過後,試驗主持人始得發表。

十九、 因试验计畫主持人過失造成醫院或他人受損害時,須由計畫主持人負法律責任。

- 二十、 免除受試者同意後仍會適時提供受試者試驗相關訊息。
- 二十一、 當受試者在納入研究後成為受刑人,主持人得知後應以非預期問題通報本會及試驗 委託者。
- 二十二、 其餘未盡事宜,請參照本院「人體試驗作業管理辦法」辦理。

CHANG GUNG MEDICAL FOUNDATION 199 TUNG HWA NORTH ROAD TAIPEI, TAIWAN, 10507 REPUBLIC OF CHINA TEL : (03)3196200 FAX : (03)3494549

Institutional Review Board A

MEMBER	GENDER	PRIMARY SPECIAL TY	AFFILIATION OF INSTITUTION
主席:谢煤堂 Chairman: Tsang-Tang Hsieh	Male 男性	Obstetrics/Gynecology 婦產科專科	CGMF (Taipei) 台北長庚紀念醫院
陳順勝	Male	Neurology	CGMF (Kaohsiung)
Shun-Sheng Chen	男性	神經內科專科	高雄長庚紀念醫院
1박 想과 Morgan Fu	用性	心臟內科專科	高雄長庚紀念醫院
沈一娟	Female	HepatoGastroentero logy	CGMF (Lin-Kou)
I-Shyan Sheen	女性	買腸肝膽科專科	林口長庚紀念醫院
王正炮	Male	Hematology/Oncology	CGMF (Kee-Lung)
Cheng-Hsu Wang	男性	腫瘤科專科	基隆長庚紀.念髓院
郭 亮鉾	Male	General Surgery	CGMF (Jiayi)
Liang-Mou Kuo	男性	一般外科專科	嘉義長庚紀.念醫院
蘇建安	Male	Psychiatry	CGMF (Jiayi)
Jian-An Su	男性	精神科專科	嘉義長庚紀念醫院
厚.維洲 Chi-Chou Liao	Male 男性	Pharmacy 藥事	Institute for Biotechnology and Medicine Industry 社團法人國家生技醫療產業策進會
游 静宜	Female	Social Work	CGMF (Lin-Kou)
Ching-Yi Yu	女性	≹f ⊤	林口長康紀念醫院
黄怡婷	Female	Statistics	National Taipei University
Yi-Ting Hwang	女性	統計	台北大學
黃 英 寛	Female	Law	None
Ying-Ni Huang	女性	法律	fiit
鄭 欽明	Male	Business Management	Children's Hearing Foundation
Kenny Cheng	男性	企業管理	雅文兒童聽語文教基金會
曾旭民	Male	Psychology	Chang Gung University
Hsu-Min Tseng	男性	心理學	長庚大學
張 統翰	Male	Orthopaedic	CGMF (Lin-Kou)
Yuhan Chang	男性	骨科	林口長庚紀念醫院
林 欣柔	Female	Law	Chang Gung University
Shin-Rou Lin	女性	法律	長庚大學

Chairman :

OT. I. Hstah

Tsang-Tang Hsieh, M.D.

Appendix 9: Consent form (patient version)



The University of Manchester

Consent Form (service user version)

(Date: June 2017, Version: 4)

Study Title: The Decision making Process for Service users in Secondary Mental Health Services in Taiwan.

Name of Researcher: Chiu-Yi, LinPlease initial all boxes				
1	I confirm that I have read the attached information sheet dated (version 4) on the above project and have had the opportunity to information and ask questions, and had these answered satisfact	June, 2017 to consider the torily.		
2	I understand that my participation in the study is voluntary and withdraw at any time during the interview and up until the data and analysed (approximately one month after the interview) wi reason and without my medical care or legal rights being affect be made clear in study documentation.	that I am free to is anonymised thout giving any ted and this will		
3	I understand that data collected during the study may be looked individuals from the University of Manchester, from regulatory from the Chang-Gung Medical Foundation Institutional Review Taiwan, where it is relevant to my taking part in this research. I individuals having access to this information.	l at by authorities or v Board in I agree to these		
4	I agree to the use of digital recording			
5	I agree to the possible use of anonymised quotations.			
6	I agree to take part in the above study.			
7	I would like to be informed of the findings of this research and name and email / address being added to a mailing list that will for the purposes of disseminating the findings of this research.	agree to my l be used solely		
Name	e of participant: Name of researcher:			

Date:	Date:
Signature:	Signature:

Appendix 10: Consent form (health professional version)



The University of Manchester

Consent Form (health professional version)

(Date: June 2017, Version: 4)

Study Title: The Decision making Process for Service users in Secondary Mental Health Services in Taiwan.

Name of Researcher: Chiu-Yi, LinPlease initial all boxes					
1	I confirm that I have read the attached on the above project and have had the and ask questions, and had these answ	information dated June, opportunity to consider t vered satisfactorily.	2017 (version 4) the information		
2	I understand that my participation in the withdraw at any time during the interv until the data is anonymised and analy interview) without giving a reason and rights being affected and this will be n	he study is voluntary and view without giving any r vsed (approximately one r d without my professiona nade clear in study docur	that I am free to reason and up month after the l career or legal mentation.		
3	I understand that data collected during the study may be looked at by individuals from the University of Manchester, from regulatory authorities or from the Chang-Gung Medical Foundation Institutional Review Board in Taiwan, where it is relevant to my taking part in this research. I agree to these individuals having access to this information				
4	I agree to the use of digital recording, quotations.	with possible use of anot	nymised		
5	I agree to the possible use of anonymised quotations.				
6	I agree to take part in the above study.				
7	I would like to be informed of the find name and email / address being added for the purposes of disseminating the	lings of this research and to a mailing list that will findings of this research.	agree to my l be used solely		
Name	of participant:	Name of researcher:			
Date:		Date:			
Signature: Signature:					

Appendix 11: Distress policy

If this happened, there were steps to manage distress:

The author would ask whether she/he needs to stop the interview immediately or needs a short break.

If service users had any questions, the author would address them immediately. The author would provide some support to the participants and reiterate that they have the right to refuse to answer some question if they did not want to.

The author would ask them if they were willing to continue the interview and assess whether the participant was still able to continue the interview based on the author's professional knowledge (the author is a mental health nurse who is trained and experienced at identifying and managing distress).

If not, the interview would stop immediately, and there were sources of support which would be provided (see 6-9 below). The author would ask the participants if they would like their data to be removed from the study.

If the participant continued the interview, there were some further steps that would be followed during and after the interview:

During the interviews

The author would continue to observe the participants' non-verbal and verbal cues.

Participants would be asked to share their experiences, as far as they were comfortable with doing so.

At the end of the interview, the participants would be given time to express their emotions about the interview and reflect on the experience of being part of this project.

Some support resources would be provided:

For the service user group, relevant support resources would be provided to participants, such as the consultant's phone number or/and appointments with their GPs or mental healthcare providers:

For the health professional group, the authors would provide the consultant's phone number and ask participants to contact their close family or friends for continued emotional support. Contact details for the author were provided to each participant in both groups.

After the interview

The author would inform mental healthcare providers or GPs if necessary (including suicide or risk of harm to themselves or others).

If they were concerned about the information they have provided, participants could contact the author to withdraw at any time during the interview and up until the data was anonymised and analysed (approximately one month after the interview), and their data would be destroyed upon request.

Appendix 12: Example of thematic frameworks for the patient interviews (Theme 1)

Theme 1: Barriers to shared decision making

Themes	Professional status of health professionals in a	Negative perception of making decisions	Limited time resources
	submissive culture		
Number	20/20	9/20	10/20
of			
reports			
1	I believed him. I have been seeing him (the doctor) for a	N/A	N/A
	long time. I have never changed my doctor.		
	I don't know about medicine don't understand. I just		
2	<i>He said he is able to cure my auditory hallucinations</i>	Ν/Δ	I felt it was not great. The most important thing was the
-	only if I follow the instructions It is not really had for		three patients' conditions I wanted to discuss my
	me I don't know how other people feel about this		condition with the doctor but it was very difficult.
	but it is fine for me.		
			But you know, he (the doctor) had to see more than
	Anyway he told me not to ask too many questions I		one hundred patients in one day. He did not have much
	am afraid to ask you know like my roommate(to		time for me.
	be hospitalised)		
3	I do believe psychiatrists are professional. We should	I think any patient with mental illness is not able to	We didn't really discuss it directly He didn't come.
	respect to them This is why our behaviour should be	make their own decisions. Some of them may lose	But my nurse told him He replied and the nurse told
	controlled by professionals.	control and then kill someone. For example, XXX, he	me what he said. He said: Take it or leave it.
	Constitues I don't no lle const to take and is stice. Dut	make their own decisions?	
	sometimes I aon I really want to take medication. But	make men own accisions.	
	medications just go to hospital I don't think I need it		
	but I still take it.		
4	There is no reason why they should They	For example, they aren't willing for me to take my	I don't know. Every appointment with him (the health
	(professionals) are born to decide. They are	baby along. This is a clear rule. I don't think I need	professional) finished very quickly. Because there is
	professional. Yeah that's why I should obey them.	to stay in the halfway house. I am able to find a job	always another patient who is waiting. He needed to
		and live on my own. Why do I need to stay here?	see a lot of patients in one day. So we did not talk too
		They think I am ill; that I am not able to take care of	long. I hope we could talk for longer. You know.

		myself or my baby and decide. So they decide	
		everything for me.	
5	It's just like seeing a doctor for a cold. I go to see a doctor for a cold and a doctor asks me my symptoms. And then does some examinations. And then you get some medication. It is a standard procedure. Like me. I told him (a doctor): 'I have cough, a runny nose, sore throat, and headache.' And then the doctor gave me some medication to take. And then I took them That's what happened to me. Never have I ever questioned this before. That's what I was told when I was a kid. I got used to this procedure. I think this is it. This is how the health system works.		If some professionals are able to give me some information, that will be great. How to use the medication and what side effects I might notice. I want to know but, probably, the doctors don't have much time. Like me, I am going to see my doctor tomorrow. I am number 130. I bet the doctor won't have time for me.
6	I have not discussed it with him (the health professional) yet. ButI was trying to discuss it with him before, he was just staring at me. He looked not happy about it. So I am not sure I should say something about it. Because you say more and you take more medication or you get sent to hospital. You ask more questions and this is what happens next.	I went to XXX hospital. I wanted to ask some questions about my conditions. But they (health professionals) told me this is a psychiatric hospital. I am a person with mental illness. I had a headache and auditory hallucination. Someone was taking to me. And then I was admitted. 32 years old. I wonder if anyone still wants to discuss anything with me.	I would love to know (the treatment options) but the doctor won't tell us this. They are too busy. My doctor needs to see at least 38 patients a day. Don't have time to discuss with every patient for this.
7	The right of making decisions belongs to them (health professionals). The only thing we should do is to tell them your condition. Because because this belongs this is same as (being a) 'prisoner'because they are locked up as well. We have to accept assessment. This is all decided by health professionals same with prisoners. Many prisoners are locked up as a big group. I certainly felt uncomfortable. But what can I do about it?	Yeah if I don't take medication, I get out of control. Because I have mental illness, I lose control. I couldn't make my own decisions. How can I make my own decisions? You must take medication. That's it. Because you are hospitalised you lose control. You are not able to make decisions.	N/A
8	They are doctors. Doctors are qualified to control your treatment and care. Doctors are able to prescribe your	N/A	I feel it was very quick (the appointment). It is not allowed to prolong the time. If you delay all patients are delayed.

	medication. No one else can. They control that's		
	(making decisions) their responsibility.		
	They are authority. You just a patient. This is the rule		
	you must follow the instructions.		
	If you are well-behaved you can so outside and you		
	can be discharged from here soon. Just follow the		
	instructions. This is good for you		
	This is why I don't want to take medication but I still		
	do. XXX hospital you know. Yeah, here (the halfway		
	house) is more open. But the hospital is locked. Living		
	here has more freedom but if you had been sent to the		
	hospitalyou know it is differentjust follow the		
	instructions and listen to them (health professionals).		
9	Yes, this is a doctor's duty. He (the doctor) sent me to	N/A	Doctors nowadays need to see a lot of patients in one
	the psychiatric hospital. I cannot say anything. I told		day. Sometimes, you know, more than 100 patients.
	him I don't want to. But it is their job to decide.		Perhaps they don't really remember who they met
			today.
	If you don't listen to the nurses, they tie you up. Even if		
	you improve your behaviour they still tie you up. They		Three were in the same room. How could you say
	must tie vou up for an hour. Some (nurses) even give		something sensitive to the doctor? How could you
	you an injection Even if you are velling in the room		discuss anything with the doctor? How could I react to
	they won't listen to you Tike this You must do what		this? Three of us say something The doctor gave
	they (nurses) ask you to do		us medication That's all And then I got medication
	incy (nurses) usk you to uo.		The and
			The chu.
	I told them: I don't want to take this medication But		
	the staff told me I must take it or they will take me to		
	be hospitalised. So so I took it and after this, I		
	told my doctor what happened to me You know		
	what? My doctor just kept silent Like nothing had		
	happened He said: 'Adjust it is just fine to adjust		
10	If you didn't listen to the nurses, they tied you up. Even	I was losing control. I don't know what disease I	N/A
	you improved your behaviours, they still tied you up.	have. I could not sleep at night. Every day there were	

	The second discussion from the second from the	a let a familie a familie and the allowing attemption). Let an attemption	
	They must the you up for an nour. Someone (nurse) even	a lot of voices (auaitory nativicination). I was sick.	
	gave you an injection. Like this. You must do what they	Not clear There were a variety of treatments. If you	
	(nurses) ask you to do.	are sick, this (treatment option) should be decided by	
		doctors.	
11	I must respect doctors. Or why do we go to see them?		N/A
	They give us medication and we take it. They are		
	professional What they said should be right.		
12	They are doctors! They are doctors. We are patients.	I am ill (mental illness). How could I to make	
	They asked how are you lately? They ask some	decisions? I feel great they decide for me	
	questions And then they preservibe some medications	accisions. I jeer great they accue for the.	
	for you. They are destors. Or you think you can deside?		
	You do side? And they follow? No sume They are		
	Tou aeciae? And they jollow? No way They are		
	professional, but you are not.		
	The manager (in the halfway house) said if I don't take		
	medications, I will be sent to the hospital. This is what		
	they say.		
13	They are doctors. They are trying to help you rather	<i>My aunt said we are psychiatric patients. Just do</i>	N/A
	than hurting you. You should believe this. This is what	what they (health professionals) say. No arguing with	
	they do	anyone No choice	
	There is no other way. They (headleh profession als) and		
	There is no other way. They (neutin projessionals) are		
	the most powerful in the ward. There is no way to		
	discuss with them. Just do what they ask and don't make		
	any trouble.		
14	This is their job (health professionals). They are helping	N/A	Four patients once in the same appointment. There
	me. They are taking care of me.		were many patient waiting outside. We (the four
			patients) were seeing the same doctor. We took turns to
			talk with our doctor. Only little time for each of us.
15	Because he is my doctor. They can decide for me. They	N/A	N/A
	can fix it for me.		
	The doctor told me: if you don't take medication you		
	will be taken to hospital. I falt really upset really		
	will be luxen to hospital. I jell really upset really		
	upsel. And then I went to see the doctor and I said: I		
	will take the medication and cooperate with you. But you please don't give me too strong a drug. I don't want to be like a scarecrow feeling stiff in my whole body. As the nurse saidif I lay on the bed in my roomthey (nurses) would say: 'If you lie in like that, you will not sleep at night.' In the end, they (nurses) said: 'if you still lie on your bed, we will restrain you. Or not allow you go outside and go shopping' Can't go outside or		
----	---	---	---
16	anywhere. So I don't really like going to hospitals. As doctors, they have professional knowledge to make judgements. They are psychiatrists. Professional knowledge and professional staff. They went to medical school for seven years. They are experts on mental health. They definitely know how to cure patients and	N/A	Sometime you waited for so long and did not want to talk a lot. I just want to leave there as soon as possible. I was no. 91.
	what to do. If you keep askingthat means you are not stable. I don't want to go to the hospital. This is the rule here. The only thing I can do is to accept it.		
17	 Based on their profession (health professionals), they know what is suitable for me. They are mental health professionals. They know patients' needs. I told them my symptoms. (I) just follow my mum's instruction. Take medication regularly. No hallucinations. No self-taking. Or if you 	I just follow the health professionals' instructions. They told me I am permanently disabled because of mental illness. This means I am really serious. So I just follow what they decide and take medication.	N/A
18	My brother said: 'If you are taken there (the hospital) we will lock you up for more than a year.' No doubt You can't say no In the hospital, your family has to agree or you can't be discharged from the hospital. You know my friend has been locked up for a year He is not allowed to discharge himself from the hospital. Can you imagine that? I told a nurse: 'I don't want to stay	I never read these prescription. Doctors read. I don't know what it says. Because it is too difficult for me. I haven't liked to read since I was a little. Life passes. My brain is not clear (because of mental illness). You know (the patient used the finger to point himself head)	He (the health professional) did not say much. He just typed in his computer. How can I express anything with him?

	here' But I got no reply I had been locked un for		
	one year You know my friend he is still locked up		
	Not like here (the halfway house) you have more		
	freedom		
	jreedon.		
	One staff member was really aggressivehe made all the patients so afraid of him nervous. I am not going to hospital no wayI was terrified of him He said: "why you are theredon't know what to do? Hurry up!" He kept saying kept saying: "Why do you do nothing but just lay down?" Very, very aggressive. Kept swearing. If you talk back you will be pinned to the ground and the nurses will give you a shot. They are really aggressive.		
19	I want to ask but I am afraid to ask. I think patients should have right to understand their conditions. Sometime you should ask. But maybe you should not. This might cause an argument with your health professionals. This is not necessary. It is difficult to say no to them (health professionals). They are the boss. If they send you to hospital, you must stay there.	N/A	N/A
20	You can't deny their professional status Because they are doctors. They help us, but they don't require any input from you. If you refuse to take their advice that means you are not respectful to them. You are too difficult to please. If you were not well-behaved, you were "tied up"	N/A	N/A
	Yeah, some doctors did this		

Themes	Sufficient Information exchange	Desire to make decisions
Number of reports	17/20	16/20
1	Long illness makes the patient a good doctor.	N/A
2	To my mind, I do want the doctor explain it to me. But he didn't have enough time.	Actually, I just want to understand what my medication is for I don't need to know what choices I have For example, what treatment I take particularly what medication is for Effects or side-effects. That's all I want.
	The medications you take don't suit you. They [professionals] can't understand your symptoms very quickly. Right? If they know a lot about you you will get better in a fortnight.	
3	Based on this (active involvement), I could be cured quickly. I could understand if I am ok or not. I could explain my feelings about medication to them. They don't need to spend much time conducting experiments on me.	N/A
	um in fact I would like to take part Because I am back to normal. Of course I want to know what happened to me For example, I want know why I need to use each psychiatric medication.	
4	I want to discuss with my doctors that I don't want to take medication yeah. 'Cause I took medication for many years. I don't think I need to take this I think I will be fine even if I don't take any medication.	I want to discuss with my doctors that I don't want to take medication yeah. 'Cause I took medication for many years. I don't think I need to take this I think I will be fine even if I don't take any medication.
5	<i>If(I can be actively involved in decision making) Because I can understand my symptoms and conditions. I could understand how to prevent a relapse, to understand better if my condition is worse or better.</i>	I hope I have other treatment choices. I can find which one is the best for me. This is good for me I don't want this side effect (gaining weight). I want to be skinny. It is better for my health. Look healthy. This is also good for job

Appendix 13: Example of thematic frameworks for the patient interviews (Theme 2)

		interview you know. So I nope the doctor gives me a list of treatment options
		und I can make my own decision.
		It is better they tell me (more about treatment options). It is better to have choices. You could find a treatment that is more perfectly suitable for you. That would be really great After I took medication, I didn't know about its side effects. I gained weight - from 70 kg to 90 kg. I didn't know about it at the beginning. I hope I have other options. I also would like to hear some professional advice from the professionals. I
		think if you got many professional suggestions, you could compare other treatments from different angles. You could hear different solutions.
6	And I want to know more about the medication I take now. Because I'm having injections right now. I wonder if there is another option but the same type of the medication. I'm kind of allergic to this medication. I want them to understand my condition. They (the health professionals) can understand me and give me the right medication have donor and constraints of the more confectable.	He (professional) never discussed it with me (treatment options). If possible, we could discuss it and then I could decide myself. If you can decide, you feel better in your mind. This is all decided by me! No matter if it results in good or bad. I feel more comfortable about this. It wasn't decided by anyone else but me. Although I am a little slow on the draw and have a bad memory. But I believe I am able to make my own decisions.
7	I hope to understand the reason why I need to stay in the halfway house and why I need to take this medication. This is what I hope. I want to know. But they (health professionals) never discuss it with me.	I hope I can decide where I want to live andif I am stable. The doctor can reduce the dosage of my medication.
8	I think I do I do want to know about other treatment options. For example, treatment one, treatment two, and treatment three. This would be really great. I don't know why the health professionals don't implement that. I don't know how they think	I want to make my own decisions too.
9		Of course I want to (make decisions). I want to move out (the halfway house). But they say this is not allowed.
10	I want to ask them about my medication treatment but I am afraid of them (health professionals).	I don't want to make medication decisions. I don't know much about medication. Just two pills. Should be fine.

11	I don't know, I just took what they gave to me. I am not sure. Perhaps I want know more. I am not sure.	N/A
12	I think if medication I want to discuss with him (the professional) whether I could adjust my medication. Now what he gave me I can't get out my bed in the morning, I feel drowsy. I can't get out of my bed. I would like to tell him this is not right for me. I would like to change it.	I think if medication I want to discuss with him (the professional) whether I could adjust my medication. Now what he gave me I can't get out my bed in the morning, I feel drowsy. I can't get out of my bed. I would like to tell him this is not right for me. I would like to change it.
13	I never thought about this. I just took medication you know. Or I should ask them (the health professionals). Are you sure I can ask them? I shouldright I want to ask them what this medication for. I should ask them next time.	N/A
14	Of course, I love to know more about my treatment. But I am not sure I can	I certainly do (want to make my own decisions). But no way The only thing I can do is live here (in the halfway house).
15	I want to discuss with the health professionals. I want to tell them I feel the medication is fine for me. But I feel I want to eat more food. I wonder if there is another way to strike a balance.	I want to decide, you know, about my medication. I want to manage my medication.
16	I want know what medication they gave to me. Why they chose this and why I need to adjust the medication this time.	Just information about my medication. Just let me know what medication they give me and why. What the medication is. Other things that's fine. I don't need to decide.
	It is right they are professional. But some medication is not just suitable for me. Last time, I wanted them to prescribe FM2 (Flunitrazepam), but they refused. And gave me another medication. I could not sleep that night. This is not helpful at all.	
17	<i>I hope I will be told about other choices (about medication treatment). They will tell me the advantages and disadvantages of treatment.</i>	It will be great if I have medication options. But I would love to hear some advice from my health professionals. And then I can choose the best one for myself.

18	N/A	Doctors I would like to know what I can't eat or drink. I want them to tell me what medication I can take. Some information about how to take medications.
19	Because doctors should discuss with us to see how this medication works for us. They (doctors) prescribed medications to us. They should respect to my choices. To see if I am cured, they should observe how I react to the medication. I want to I want to know the effects of the medication. In fact, I really want to ask the doctors.	I want to I want to know the effects of the medication. In fact, I really want to ask the doctors.
20	N/A	Because I don't know what medication they are prescribing for me. I don't know if this medication is right for me. I want to ask them what medication they give to me. What the effects or side effects of this medication are. I want to know this.

Appendix 14: Example of thematic frameworks for the health professional interviews (Theme 1)

Themes	Powerful status of health professionals and families	Patients with impaired decisional ability due to mental illness	Health professionals have lack of understanding of shared decision making	Insufficient time
Number of reports	23/24	24/24	24/24	15/24
1	In addition, in Taiwan, I think this is the culture If you get sick you go to see a doctor and you get medicine. All they need to do is just go to see a doctor. And a doctor tells them what to do.	On the other hand, some patients don't accept their mental illness. They don't understand their mental illness. It would probably be difficult to involve them in the decision making process.	I don't know what SDM is I have to have some knowledge first. And then I can decide if I need this kind of training or not. If this model could bring some benefits to patients and to me, I may want to understand. But first I should know what it is.	For example, some professionals are really busy. They might just tell patients to take the medication they give them, without any discussion. We just decide for them.
2	If you (professionals) provide more decision options no matter how you explain it they would think you are not 100% sure about this You tell them too much and this means you're not doing your job, you know. This could bring unnecessary trouble (medical malpractice) for yourself. Patients in Taiwan don't have this kind of thought about patient involvement in medical decisions or relevant information about this. For example, some of my friends or family, if doctors offered treatment options for them, they would probably feel odd, you know. Because we are in Taiwan, we (health professionals) should respect their families. They (patients) will be taken	Because patients don't accept their mental illness. I am afraid they can't comply with medication if they know much about their treatment. This also causes conflict between patients and their families. He/she (patient) may feel he/she doesn't have mental illness, complains about side effects and eventually refuse to take any medication.	As far as I know, but I have only had a look at two PPTs (Power- point slideshows). It seems that they include professionals and patients. And they discuss the case together. And then they decide together? But I am not sure.	I feel the big problem is we (health professionals) don't have much time to discuss with patients in this medical environment. Maybe patients also feel they don't have time to have an opportunity to ask more questions.

Theme 1: Barriers to shared decision making

	care of by their family. I think if we don't respect the families during making decision, patients will lose the support of their families. This would make their healthcare difficult. So I think I need to involve the family to make decisions for patients.			
3	For example, I would tell them that if you get mental illness, you need to take medication. We would let them know what their medication is, but what kinds of medication they need to take is decided by you (professionals) based on the clinical guidelines. We have the professional knowledge to judge (treatment options). This is what doctors should do.	Because some cases insisted on making bad choices for themselves. So we have a responsibility to talk with them Some have really poor judgement you know We discussed this (in some lectures). The lecturers didn't have any answers for that. From my perspective, those patients lose their ability to make a judgement. Their will can't be seen as will, you know? This is important. He has poor judgement. He is a seriously ill patient and out of touch with reality. You talk about sharing decision making with this kind of patient. That's strange. Patients with psychosis or patients without any insight into their mental illness, of course, are less involved (in decision making). I think it (patient involvement) is difficult for them (patients with psychosis) because they have low motivation. So they don't ask questions. And all this because they have poor cognitive function and negative symptoms. Patients with avolition or patients who don't care about their	"I feel shared decision making is a way to meet patient needs. Based on their wishes, I believe it is kind of application of shared decision making. You (health professionals) provide some treatment options. And then patients to decide which one it is they prefer. Of course you have to explain the pros and cons of the options. But I wonder if this is suitable for patients with mental illness. For example, today, if a patient intended to commit suicide, it would be impossible for me to agree with his/her decision. This is odd. You know, if shared decision making is implemented in mental healthcare.	You (health professionals) are not able to explain all the information for every patient. Because there are 118 beds in the acute wards and almost 100 patients in the day care centre.

		disease I think there is no way to involve patients like that. I believe the level of insight is really important to determine if patients are able to be involved. If you don't know you have mental illness, how can you be involved?		
4	The role of doctors is naturally to have responsibility (to decide). We would guide him or take a dominant role. Because we know well his disease and support resources. This is an unavoidable responsibility.	Another example is his symptoms. Sometimes he is really depressed; often he could hesitate or wouldn't be able to decide. In this case, we (professionals) would make most of the decisions. Because you ought to be more dominant, to guide him. I suppose this is a better way.	So I think I don't know much more about the shared decision making tool. Or could you explain it to me in one minute. I don't really have the relevant information. I can only imagine what this would look like.	For example, the hospitals expect you have good sales figures. So you are not allowed to only see 3 or 4 patients. Or just 10 patients. No quantity means no profit, which is related to your income. Some professionals are even criticised by the hospitals. Maybe some new doctors do not have many patients. They may spend 40 minutes or 1 hour seeing a patient. But this is difficult.
5	From my perspective of being a health professional, we should choose the best treatment for patients. Help them to get better and use some ways to help them, to let them stay (in hospitals). This is what we should do. During decision making, you will find that sometimes, particularly for some young adults, the right of making the decision belongs to their significant family member rather than themselves. For example, their father.	One of my patients, he has delusions and auditory hallucinations. In this situation, his opinion cannot be involved. They are not stable. We can't follow their instructions and let them discharge themselves from the hospitals, right? For example, patients with poor medication adherence cannot take responsibility for their decisions. This could influence how much they can be involved. For example, one of my patients had delusions about how Satan is coming to our world. If he died he could save all people in the world. So he refused to	I can't remember It was kind of a long time ago when I heard this term (shared decision making). It might be four or five years ago. I also read some related books or articles I remember I read this term somewhere. But you ask me to explain this right now. I don't think I am able to explain it clearly.	N/A

		take any food. In this case, he was not able to take responsibility for his life. So we should take responsibility to help him, to give him some involuntary intervention.		
6	We are more like 'a provider' to provide resources and to solve problems.	For our patients, this (shared decision making) helps them to make autonomous decisions and (professionals) to respect their autonomous decisions. I think this idea is quite nice. For people in a stable condition, this is suitable. But acutely ill patients may refuse to take treatment. You still need to respect the wishes of this kind of patient? This is not suitable for them. If patients say they don't have mental illness – why do they need to take medication – "these medications are poison" – you are psychiatrist, should you respect his/her decisions? If they are in an acute episode, they have delusions, they have wrong ideas about their mental condition. It would be wrong if I still respected these decisions. It could hurt them.	I know shared decision making. Our institution also wants to implement shared decision making. We hope patients will have autonomy. I tell them (patients) the pros and cons of each option. Eventually they decide. I know now it is widely acceptable in end-of-life care. But if this is implemented in mental healthcare, it will only increase our workload.	I think this (shared decision making) will only bring a burden and extra workload to mental healthcare. Because you have to have comprehensive knowledge about mental health resources. I have a lot of experience of many different settings, so I know. But others don't. If you want to implement it, they would have to undertake a lot of training. We are already too busy.
7	Because this is a pattern in Taiwan, how you see the health professionals. We need to comply with this principle. So you normally take medication provided by the health professionals. We had been taught when we are kids. You get used to this. In term of types of medication, this decided by me (health professional)	Say one schizophrenia patient breaks his leg, and he refuses to accept treatment. And then the professionals follow his wishes as we need to respect his wishes? Don't treat him and let it be out of shape? In this case, he has delusions about surgery. Due to this illness, he cannot make this decision. Another possibility is that patients with serious conditions or psychosis don't	I seems to have heard this (shared decision making) before. But I don't really understand what it means, exactly. I suppose that we (health professionals) should sit down with patients and discuss all medical decisions. And then we should respect his/her decision. And also explain more details of the treatment? I wonder if informed	It is difficult for you (health professionals) to provide a variety of treatment options for patients like a vending machine. In this case, in the NHI, the effectiveness of your outpatient clinic is very poor.

	based on your condition and my profession. This because I could choose the treatment that is in the best interest of the patients.	 have any sense of reality. Or they don't have insight. Or even worse, they might have some risk of violence or suicidal. They are not able to take part in medical decision making. We (health professionals) are worried that they are not able to accept treatment, so we don't let them be involved. Because they have no insight, they refuse to take medication. 	consent is a kind of shared decision making. Is it? I would like to know. I don't know the definition of shared decision making.	
8	Many patients get used to decision making being the doctors' responsibility. Or they would wonder why they need to see a doctor and spend money on an outpatient appointment. Not every patient wants to make decisions. They believe you are doctors. You are professionals, you decide.	Let patients to decideIf you ask me to do something I believe could worsen his/her outcome, I will definitely refuse it. Some patients with mental illness don't have the ability to make decisions.	I have heard this term (shared decision making) before, but I don't really understand it. Maybe this should involve patients, doctors, and other professionals? For example, some patients suffer from both mental and psychical problems. Not just psychiatrists but also physicians discuss together? I am not sure.	If you are working in an outpatient department, it could be difficult due to our National Health Insurance. Like me, you can only spend three hours finishing morning or afternoon appointments. You need to see 30 or 40 patients in one morning or afternoon session. That means you only can spend 5 minutes for per patient. 5 minutes! You know in this situation, we (professionals) don't have any opportunity to do it (shared decision making). So it is more efficient if we decide for them (patients).
9	In Taiwan, we (professionals) need to take a family as a unit. Without family intervention, in fact it is difficult for you to do anything (about patients). For example, we (decide) to transfer a patient to a halfway house. But the family does not support this don't agree with this. They will take a patient out within one or two weeks.	Because they normally ask to stop major medication. For example, one depressed patient doesn't want to take anti-depressive medication but asks for sleeping pills. Basically, I think the request is against the treatment guidelines and will lead to a depressive episode. I would turn them down.	If my memory is correct, I seem to have heard this term (shared decision making) in some workshops or from one of my colleagues. I think what you need to do is: first we (professionals) need to know what benefit it (shared decision making) brings for health professionals. And then I might like to know, or have some interest in learning about it. If there are	Because (shared decision making) needs a long time to explain. To be honest, this will take a long time to implement. Also there is a lot of work I need to do in the outpatient department. This is time-consuming.

			hanafits (to professionals) I will	
			have more motivation	
10	I believe the right of prescribing medication belongs to doctors. Although this is doctors' right, we all hope the patients take their medication regularly.	But for some patients with serious mental illness or with long-term conditions, decisions should be made by us or their family. They (patients) follow our decisions. For example, some patients with schizophrenia and bipolar disorder, because they need treatment. These kinds of patients need long-term treatment. Normally they are not able to make their own decisions. Or they don't have the ability to research information or gain correct knowledge about medication.	I have heard this term (shared decision making) recently. As I understand, it was as if there are many treatment options. Health professionals could provide their understanding of different options. And then patients make the final decisions. I think the medical environment in Taiwan is relatively special. So sometimes you want to introduce some new intervention into our society and this needs time to reshape the intervention. And you should be sure this is beneficial for health professionals. Because I think this is a mutual thing. This should be beneficial for both (patients and professionals) and then you adopt it. But if this is only beneficial for patients it is not necessary to implement it. It will bring a huge workload for health professionals.	 From a psychiatrist's point of view, if there are many patients in one day, it could be difficult if I spend too much time on one patient – particularly for patients with minor conditions. Because we (health professionals) don't have enough time. But it (shared decision making) needs to take a lot of time. And maybe it isn't really very effective. Overall, it is difficult to implement it for all patients. I think the medical environment in Taiwan is relatively special. So sometimes you want to introduce some new intervention into our society and this needs time to reshape the intervention. And you should be sure this is beneficial for health professionals. Because I think this is a mutual thing. This should be beneficial for both (patients and professionals) and then you adopt it. But if this is only beneficial for health professionals.
11	I think they (patients) can comply with our orders well. It is a doctor's role to be an authority figure. I don't know, this is kind of a tradition of healthcare? They (patients) are more like a receiver rather than a provider. They don't speak up for themselves.	I believe this (patient involvement) should depend on the severity of their mental illness. I mean sometimes patients have been influenced by their psychosis symptoms. Then it would be unreasonable to accept their decisions.	I've never heard of this (shared decision making). I know about some ideas of patient involvement. But I never heard this term.	I believe it (shared decision making) would be difficult to implement in practice. I suppose Say you want to apply it in every case, getting different professions to work as a team. All the professionals come to discuss everything with the patients. I think time management would be really difficult. For example, doctors could not come as they

	This is our healthcare system. Like my parents: they said, "I believe what doctors say, but I don't believe what our daughter (the health professional) says. This is kind of our tradition. Everyone think doctors are very intelligent. For example, the patients and family in the wards took the right position when they see the doctors. So you say don't doctors have authority status?			have a lot of meetings, outpatient appointments and so on.
12	This is traditional eastern culture. It is kind of authoritative. A doctor is more(qualified). When people were children, they were told to comply with doctors. We should completely obey them and do what they ask. That's what people have been taught since they were children. We (health professionals) can provide some information. But I think it is up to their family to decide. The family takes over you know. We are not the ones who need to take care of them (patients) forever. But their family do.	I believe their decisional capability and motivation are impacted by mental illness. They are suffering from negative symptoms. So they don't engage in decision making and have no motivation.	I know about this (shared decision making). My hospital wants to implement this. I believe that shared decision making means that we (health professionals) provide some treatment decisions. For example, option A, option B, and option C. and then some evidence for each option. For example, breast cancer. Survivability of each surgery option. Patients could make their own decisions based on this evidence. In mental healthcare, I think we have already done this for such things as discharge planning. I do believe we (health professionals) are doing this. Just we don't use data based evidence to do so.	I think we (health professionals) now facing a shortage of human resources in practice. There are a lot of patients but only few professional staff. It is a difficult time right now.
13	No (we don't provide treatment options), because we (professionals) know best.	Some patients have conditions such as psychosis, schizophrenia, or bipolar disorder. Their decisional function is not good enough.	There might be some courses (about shared decision making), but I don't get the information. Perhaps there are some, but they're not	N/A

 Not Not (Providing treatment options) we are the ones who know which one is the best. They don't inderstand. Some people with bipolar or schizophrenia, or depression. They is to to take medication so they can control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a person with schizophrenia, because and make it dirty. In fact, we feel wor do may be developed are spopsed to listen to the elderly and authorities. I think people get used to this You should listen to people with professional ability. I guess they reprietive get used to putting doctors or nurses into position (to decide). Because patients have an expectation: I need to be discharged from the hospital. first. I feel, in this situation, they will follow the orders of doctors. And they would do yhout fue or ductors. And they would do yhout they think keotors think or want. Based on this, they would do hey they the the ottors. And they would do his, they would do his, they will his for patients. 					
 we are the ones who know which one is Some people with bipolar or state best. They don't understand. If they have strong authority. They believe patient should follow the instructions. Until now, some patients sit believe this what doctors should the is what doctors should do they are powerful, and we (patients) should listen to it. Beside, taking care of patients is family's responsibility: I think patients can go home but family don't. I think we need to respect family. And funity sample, if there is a person with every single night. Or he might make a lot of rubbish in the public area and make it dirty. In fact, we feel scan and make it dirty. In fact, we feel to respect family. And funity sample, if there is a person with schizophrenia l'invig the cascing to future to the inspinal. Decause this is future, they are nove like a los of noise or yell in his hould make decision for them (patients). think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to the resymple, futients by thus do to puting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide). Because patients have an expectation: I need to be discharged from the hospital, I will actively engage from the hospital, first. I fiel, in this situation, they will follow the orders of doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position to doctors or nurses into positions to the order of the will not take and promote the will be a will be wil		No! No! (Providing treatment options)		really advocated It is difficult for	
14 I the best. They don't understand. Like some older doctors, such as Dr. XXX, they have strong authority. They believe patient should follow the instructions. Unit now, some weath still believe this is what doctors should do. They are powerful, and we ty atients) should listen to it. schizophrenia, or depression. They refuse, they are more like a loose control their symptoms. But if they refuse, they are more like a loose difficulty in the community. For easy bowerful, and we ty atients) should listen to it. schizophrenia, or depression. They refuse, they are more like a loose difficulty in the community. For easy bowerful, and we easy bowerful, and we ty atients). I think patients can go home but family dort. I think we need to respect family. And family should make decision for them ty atients). iterad some policy makers want to nother building refuse. For example, patients with schizophrenia, because their cognitive mode in the obe building, they are in the position (to decide). N/A 14 I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people gets weat furthis they are in the position (to decide). I heard some policy makers want to make indires. Som, I will not compromise with this, fi and a betieve he will hospitalised again soon, I will not compromise with this, fi are do to disclores and marses think they are in the position (to decide). N/A Because patients have an expectation: I need to be discharged from the hospital. Fin: the docisor and marse thory thank or want. Based on this, they would by wolf guere out what doctors whink or want. Based oot this, they would by wolf guere out what doctors		we are the ones who know which one is	Some people with bipolar or	us (to adopt it).	
Ide some older doctors, such as Dr. XXX, they have storng authority. They believe patient should follow the instructions. Until now, some patients still believe this is what doctors should do. They are powerful, and we (patients) should listen to it. have to take medication so they can control their symptoms. But if they refuse, they are more like a loose camom. This kind of person will cause camom. This kind of person will cause this of person will solitizely in the community. For example, if there is a person with schizophrenia loing downstairs, he might make a lot of noise or yell in his home every single night. Or he might you hould make decision for them (patients), Idea (absect came cacept their). And family should make decision for them (patients), N/A 14 I think in Chinese society, people are supposed to listen to the elderly and nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide). For example, patients with schizophrenia, because their cognitive fraction and understanding facility ar really poor. It is really hard (To involve them, If he decides not to take medication and I believe he will hospitalised again son, I will not compromise with this, if the vare in the position (to decide). N/A Because patients have an expectation: follow the orders of doctors. Add they would try to figure out what doctors think or want. Based on this, they would by wold the orders of doctors. Add there would ave of dustration, they will yold ow hat they think the doctor For example, patients is the medication and I believe he will hospitalised again son, I will not compromise with his, if the causot to		the best. They don't understand.	schizophrenia, or depression. They		
 XXX, they have strong authority. They believe patients should follow the instructions. Until now, some patients still believe this to what doctors should did. They are powerful, and we (patients) should listen to it. Beside, taking care of patients is cango to more tike a portion with schizophrenia living downstairs, he more every single night. Or he might make a lot of noise or yell in his home every single night. Or he might make a lot of noise or yell in his home every single night. Or he might make a lot of noise or yell in his home every single night. Or he might make a lot of noise or yell in his home every single night. Or he might make a lot of noise or yell in his home every single night. Or he might make a lot of noise or yell in his home every single night. Or he might make a lot of noise or yell in his home every single night. Or he might make a lot of rubbish in the public area and make it dirty. In fact, we feel scared. One day he may start a fire and to noise or yell in his home every single night. Or he might make a lot of rubbish in the public area and make it dirty. In fact, we feel scared. One day he may start a fire and to noise or yells in the value get used to their decisions (to refuse to take medication)? I think in Chinese society, people are supposed to listen to the elderly and and live inderstanding factors on nurses think profeosional ablity. I guess they (patients) get used to putting doctors or nurses think profeosional ablity. I guess they will hor orders of doctors and nurses think for the opple will have point to take medication first. If yeel, in this situation, they will follow the orders of doctors. And they would my to figure our what doctors that we medication they will follow the orders of doctors. And therefore be sent to the hospital for the hospital first. If yeel, in this situation, they will would will be best or the hospital for the hospital for the would my to figure our what doctors of the hospital for the hospital for the wo		Like some older doctors, such as Dr.	have to take medication so they can		
 believe patient should follow the instructions. Until now, some patients still believe this is what doctors should do. They are powerful, and we (patients) should listen to it. Beside, taking care of patients is family's responsibility. I think patients can go home but family don't. I think we need to response (family, and family should make decision for them (patients). I think in Chinese society, people are to take medication? I think in Chinese society, people are to take medication? I think in Chinese society, people are to take medication? I think in Chinese society, people are to take medication? I the constraint of the constra		XXX, they have strong authority. They	control their symptoms. But if they		
instructions. Until now, some patients still believe this is what doctors should do. They are powerful, and we (patients) should listen to it. camon. This kind of person will cause difficulty in the community. For example, if there is a person with schizophrenia living downstairs, he might make a lot of noise or yell in his home every single night. Or he might put a lot of or itability. In this home every single night. Or he might put a lot of or itability. In this home every single night. Or he might put a lot of or itability. In the public area and make if alot of noise or yell in his home every single night. Or he might put a lot of or itability. In the community. For example, eit there, is a person with schizophrenia living downstairs, he mad make if alot of noise or yell in his home every single night. Or he might put a lot of ruits they we feel scared. One day he may start a fire and burn down the whole building. How can we accept their decisions to refixee to take medication? I heard some policy makers want to implement this (shared decision making). As far as I know, put simplement this (shared decision making). As far as I know, put simplement this (shared decision making). As far as I know, put simplement this (shared decision making). As far as I know, put simplement this (shared decision making). As far as I know, put simplement this (shared decision making). As far as I know, put simplement this (shared decision make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all soon, I will not compromise with this, fit he says directly that he will not take any medication after being discharged him. I mean he has to take medication. He cannot to refuse it. He will herefore be sent to the hospital professionals to understand patient preference and patients. And hen they combine that with evidence and find the best option for patients. N/A <th></th> <th>believe patient should follow the</th> <th>refuse, they are more like a loose</th> <th></th> <th></th>		believe patient should follow the	refuse, they are more like a loose		
still believe this is what doctors should do. They are powerful, and we (patients) should listen to it.difficulty in the community. For example, if there is a person with schizophrenia living downstains, he might make a lot of noise or yell in his home every single night. Or he might year of or the best of rubbish in the public area and make it diry. In fact, we feel scared. One day he may start a fire and burn down the whole building. How can we accept their decisions (to refuse to take medication)?N/A14I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this You should listen to people with (patients) ye used to puting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).For example, patients with schizophrenia, because their cognitiv aud dow the whole building facility are really poor. It is really hard (To involve them).I heard some policy makers want to implement this (shared decision making). As far as I knov, put simply, most of the time doctors make almost every decision for and I believe he will hospitalised again and I believe he will hospitalised again first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, hey would by the try think the doctors think or want. Based on this, hey would by the try think the doctors think or want. Based on this, hey would by that the doctors think or want. Based on this, hey would by that the doctors think or want. Based on this, heydifficulty in the combine that with ether fore be sents to take medication. He camot to refuse it. He will therefore be sents to take medication. He camo		instructions. Until now, some patients	cannon. This kind of person will cause		
do. They are powerful, and we (patients) should lisen to it. example, if there is a person with schizophrenia living downstairs, he might make at lot of noise or yell in his home every single night. Or he might make at lot of noise or yell in his home every single night. Or he might make at lot of noise or yell in his home every single night. Or he might make at lot of noise were yingle night. Or he might make at lot of noise were yingle night. Or he might make at lot of noise were yingle night. Or he might make at lot of noise were yingle night. Or he might make at lot of noise were yingle night. Or he might make at lot of noise were yingle night. The work of building. How can we accept their decisions (to refuse to take medication)? I heard some policy makers want to implement this (shared decision for make almost every decision for patients. You should listen to people with they are in the position (to decide). NA Because patients have an expectation: I meed to be discharged for the hospital first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would try to figure out what doctors there first. He event would by the orders of doctors. And they would try to figure out what doctors further benet the doctors further the hospital . Will actively engage the. I mean the tas to take medication. He cannot to refuse it. He will more first it. He will hor take and make it. He will hor the hey combine that with evidence and find the best option for patients. NA		still believe this is what doctors should	difficulty in the community. For		
 (patients) should listen to it. Beside, taking care of patients is family's responsibility. I think patients can go home but family does and make it dirty. In fact, we feel or sepect family. And family should make decision for them (patients). I think in Chinese society, people are substrained burn down the whole building. How can we accept their decisions (to refuse to take medication)? I think in Chinese society, people are substrained burn down the whole building. How can we accept their decisions (to refuse to take medication)? I think in Chinese society, people are substrained burn down the whole building. How can we accept their decisions (to refuse to take medication)? I think in Chinese society, people are substrained burn down the whole building. How can we accept their decisions (to refuse to take medication)? I think in Chinese society, people are substrained burn down the whole building. How can we accept their decisions (to refuse to take medication)? I think in Chinese society, people are substrained burn down the whole building. How can we accept their decisions (to refuse to take medication)? I think in Chinese society, people are substrained burn down the whole building. How can we accept their decision and understanding facility are really poor. It is really hard (To invove them). I he decides not to take medication and understanding facility are really poor. It is really hard (To invove them). I he decides not to take medication argument. Because this is all decided by doctors, it is more likely that the doctors for mathers of doctors. And they weil doctweils of the hospital. I will actively engage from the hospital. I weil any medication affer being discharged from the hospital. I weil any medication affer being discharged rom the hospital. I weil any medication affer being discharged rom the hospital. I weill actively engage from the hospital.		do. They are powerful, and we	example, if there is a person with		
 It might make a lot of noise or yell in his home every single night. Or he might must a lot of rubistic harming is responsibility. I think patients is family is responsibility. I think patients and make it dirty. In fact, we feel scared. One day he may start a fare and bur down the whole building. How can we accept their decisions (to refuse to take medication)? It hink in Chinese society, people are used to this You should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide). Because patients have an expectation: I need to be discharged from the hospital first. I feel, in this situation, they will follow the orders of doctors. And they will follow the orders of doctors. And they will pollow the orders of doctors. And they would dury to figure out what doctors think the doctor would do what they think the doctors with the doctors would do what they think the doctors what they think the doctors would do what they th		(patients) should listen to it.	schizophrenia living downstairs, he		
Beside, taking care of patients is family's responsibility. I think patients can go home but family don't. I think patients can go home but family don't. I think patients we need to respect family. And family should make decision for them (patients). home every single night. Or he might put a lot of rubbish in the public area and make i dirty. In fact, we feel scared. One day he may start a fire and burn down the whole building. How can we accept their decisions (to refuse to take medication)? I heard some policy makers want to implement this (shared decision making). As far as I know, put schicophrenia, because their cognitive function and understanding facility are really poor. It is really hard (To involve them). N/A 14 I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this You should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide). I heard some policy makers want to involve them). N/A Because patients have an expectation: I need to be discharged from the hospital first. I feel, in this situation, they will follow the orders of doctors. And they would do what they divink the doctor think or want. Based on this, they num medication and in the basi to take medication. therefore be sent to the hospital, I will actively engage thin. I mean he has to take medication. think for wout. Based on this, they num medication therefore be sent to the hospital soon.		(1	might make a lot of noise or yell in his		
family's responsibility. I think patients can go home but family don't. I think we need to respect family. And family should make decision for them (patients).put a lot of rubbish in the public area and make it dirty. In fact, we feel scared. One day he may start a fire and burn down the whole building. How can we accept their decisions (to refuse to take medication)?NA14I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this You should listen to people with professional ability. I guess they (patients) get used to putting doctors on nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).For example, patients with schizophrenia, because their cognitive involve them).I heard some policy makers want to implement this (shared decision making). As far as I know, put simplement this (shared decision for patients. But when the outcome of murses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).N/ABecause patients have an expectation: I need to be discharged from the hospital first. I fiel, in this situation, they will would dury to figure out what doctors think or want. Based on this, hey would try to figure out what doctors think for doctors would what mey think the doctorI heard some policy makers want to immake in the solital for the policition and understanding facility are rareally hard (To involve them).NA		Beside, taking care of patients is	home every single night. Or he might		
Just of the part of the pa		family's responsibility. I think patients	put a lot of rubbish in the public area		
We need to respect family. And family should make decision for them (patients).Scared. One day he may stort a fire and burn down the whole building. How can we accept their decisions (to refuse to take medication)?N/A14I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this You should listen to people with professional ability. I guess they (patients) get used to putting doctors or Therefore, doctors and nurses think they are in the position (to decide).For example, patients with schizophrenia, because their cognitive involve them).I heard some policy makers want to implement this (shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, in is more likely that the will not take any medication after being discharged from the hospital, I will actively engage him. I mean he has to take medication from the hospital, I will actively engage him. I mean he has to take medication from the hospital, I will actively engage him. I mean he has to take medication professional try to figure out what doctors think or want. Based on this, they would try to figure out what doctorsI mean he has to take medication any medication after being discharged from the hospital, I will actively engage him. I mean he has to take medication, therefore be sent to the hospital soon.N/AI be and to want. Based on this, they would do what they think the doctorI mean he has to take medication, any medication after being discharged from the hospital soon.I mean he has to take medication any medication after being di		can go home but family don't I think	and make it dirty. In fact, we feel		
should make decision for them (patients).burn down the whole building. How can we accept their decisions (to refuse to take medication)?I heard some policy makers want to implement this (shared decision making). As far as I know, put simply, most of the time doctors making). As far as I know, put simply, most of the time doctors make almost every decision for patients bur the outcome of the decides not to take medication arrses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).N/ABecause patients have an expectation: I need to be discharged from the hospital, follow the orders of doctors. And they would do what they think the doctorI heard some policy makers want to implement this (shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decides not to take medication and I believe he will hospitalide again soon, I will not compromise with this. If he says directly that he will not take any medication after being discharged from the hospital, I will actively engage him. I mean he has to take medication. He cannot to refuse it. He will therefore be sent to the hospital soon.N/A		we need to respect family. And family	scared. One day he may start a fire and		
Image: constraint of particular structureCan we accept their decisions (to refuse to take medication)?14I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this You should listen to people with professional ability. I guess they (patients) get used to puting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).For example, patients with solution and understanding facility are in the position (to decide).I heard some policy makers want to involve them).N/ABecause patients have an expectation: I need to be discharged from the hospital, first. I feel, in this situation, they will follow the orders of doctors. And they would do what they think the doctorI will not compromise with the solution of the refore be sent to the hospital soon.I will actively engage him. I mean he has to take medication. the cannot to refuse it. He will therefore be sent to the hospital soon.making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.		should make decision for them	burn down the whole building. How		
14I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this J vou should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).For example, patients with schizophrenia, because their cognitive function and understanding facility are involve them).I heard some policy makers want to implement this (shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all soon, I will not compromise with this. If he says directly that he will not take any medication after being discharged first. I feel, in this situation, they will follow the orders of doctors. And they would do what they think the doctorsIn. I mean he has to take medication. therefore be sent to the hospital soon.I heard some policy makers want to implement this (shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that the doctors should take all the professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.		(patients)	can we accept their decisions (to refuse		
14I think in Chinese society, people are supposed to listen to the elderly and authorities. I think people get used to this You should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).For example, patients with schizophrenia, because their cognitive function and understanding facility are really poor. It is really hard (To involve them).I heard some policy makers want to implement this (shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that they ult not take any medication after being discharged first. I feel, in this situation, they will follow the orders of doctors. And they would dry to figure out what doctors think or want. Based on this, they would do what they think the doctorII heard some policy makers want to implement this (shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that the doctors should take all the responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And therefore be sent to the hospital soon.N/A		(partenis).	to take medication)?		
supposed to listen to the elderly and authorities. I think people get used to this You should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).schizophrenia, because their cognitive function and understanding facility are really poor. It is really hard (To involve them).implement this (shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that the doctors should take all the responsibility. Shared decision making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that the doctors should take all the responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.	14	I think in Chinese society, people are	For example, patients with	I heard some policy makers want to	N/A
authorities. I think people get used to this You should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).function and understanding facility are really poor. It is really hard (To involve them).making). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decides not to take medication and I believe he will hospitalised again soon, I will not compromise with this. If he says directly that he will not take any medication after being discharged from the hospital, I will actively engage from the hospital, I will actively engage him. I mean he has to take medication. think or want. Based on this, they would do what they think the doctormaking). As far as I know, put simply, most of the time doctors make almost every decision for patients. But when the outcome of the decides not to take medication argument. Because his is all decided by doctors, it is more likely that the doctors should take all the responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.		supposed to listen to the elderly and	schizophrenia, because their cognitive	implement this (shared decision	
this You should listen to people with professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power.really poor. It is really hard (To involve them).simply, most of the time doctors make almost every decision for patients. But when the outcome of the decides not to take medication and I believe he will hospitalised again soon, I will not compromise with this. If he says directly that he will not take any medication after being discharged first. I feel, in this situation, they will first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctorreally poor. It is really hard (To involve them).simply, most of the time doctors make almost every decision for patients. But when the outcome of the decided by doctors, it is more likely that the doctors should take all the responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And the nethy combine that with evidence and find the best option for patients.		authorities. I think people get used to	function and understanding facility are	making). As far as I know, put	
professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power.involve them).make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that the doctor should take all the responsibility. Shared decision making aims to help health professionals to understand patient professionals to understand patient professional ability. I guess they (patients) get used to putting doctors or nurses into positions of power.involve them).If he decides not to take medication and I believe he will hospitalised again soon, I will not compromise with this. If he says directly that he will not take any medication after being discharged from the hospital, I will actively engage him. I mean he has to take medication. He cannot to refuse it. He will therefore be sent to the hospital soon.make almost every decision for patients. But when the outcome of the decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that the doctor should take all the responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.		this You should listen to people with	really poor. It is really hard (To	simply, most of the time doctors	
Image: product of provide the product of product of provide the product of provide the product of product of provide the product of		professional ability. I guess they	involve them).	make almost every decision for	
nurses into positions of power. Therefore, doctors and nurses think they are in the position (to decide).If he decides not to take medication and I believe he will hospitalised again soon, I will not compromise with this. If he says directly that he will not takethe decision is not good, there is an argument. Because this is all decided by doctors, it is more likely that the doctor should take all the responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients,		(patients) get used to putting doctors or	,	patients. But when the outcome of	
Therefore, doctors and nurses think they are in the position (to decide).and I believe he will hospitalised again soon, I will not compromise with this. If he says directly that he will not take any medication after being discharged first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctorand I believe he will hospitalised again and I believe he will hospitalised again he will not compromise with this. If he says directly that he will not take any medication after being discharged from the hospital, I will actively engage him. I mean he has to take medication. He cannot to refuse it. He will therefore be sent to the hospital soon.argument. Because this is all decided by doctors, it is more likely that the doctors should take all the responsibility. Shared decision making aims to help health preference and expectations. And then they combine that with evidence and find the best option for patients.		nurses into positions of power.	If he decides not to take medication	the decision is not good, there is an	
they are in the position (to decide).soon, I will not compromise with this. If he says directly that he will not take any medication after being discharged from the hospital first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctorsoon, I will not compromise with this. If he says directly that he will not take any medication after being discharged from the hospital, I will actively engage him. I mean he has to take medication. He cannot to refuse it. He will therefore be sent to the hospital soon.decided by doctors, it is more likely that the doctors should take all the responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.		Therefore, doctors and nurses think	and I believe he will hospitalised again	argument. Because this is all	
Image: Construction of the second s		they are in the position (to decide).	soon, I will not compromise with this. If	decided by doctors, it is more likely	
Because patients have an expectation: I need to be discharged from the hospital first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctorany medication after being discharged from the hospital, I will actively engage from the hospital, I will actively engage him. I mean he has to take medication. He cannot to refuse it. He will therefore be sent to the hospital soon.responsibility. Shared decision making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.			he says directly that he will not take	that the doctors should take all the	
need to be discharged from the hospital first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctorfrom the hospital, I will actively engage him. I mean he has to take medication. He cannot to refuse it. He will therefore be sent to the hospital soon.making aims to help health professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.		Because patients have an expectation: I	any medication after being discharged	responsibility. Shared decision	
first. I feel, in this situation, they will follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctorhim. I mean he has to take medication. He cannot to refuse it. He will therefore be sent to the hospital soon.professionals to understand patient preference and expectations. And then they combine that with evidence and find the best option for patients.		need to be discharged from the hospital	from the hospital, I will actively engage	making aims to help health	
follow the orders of doctors. And they would try to figure out what doctors think or want. Based on this, they would do what they think the doctorHe cannot to refuse it. He will therefore be sent to the hospital soon.preference and expectations. And then they combine that with evidence and find the best option for patients.		first. I feel, in this situation, they will	him. I mean he has to take medication.	professionals to understand patient	
would try to figure out what doctors think or want. Based on this, they would do what they think the doctortherefore be sent to the hospital soon.then they combine that with evidence and find the best option for patients.		follow the orders of doctors. And they	He cannot to refuse it. He will	preference and expectations. And	
think or want. Based on this, they would do what they think the doctor think or want. Based on this, they would do what they think the doctor think or want. Based on this, they would do what they think the doctor		would try to figure out what doctors	therefore be sent to the hospital soon.	then they combine that with	
would do what they think the doctor		think or want. Based on this, they	,	evidence and find the best option	
		would do what they think the doctor		for patients.	

	has asked them to do. And then they could leave hospital.			
15	N/A	It is different. Let me think. Because it is really hard to talk with or involve patients with psychosis.	Yes, I heard about this (shared decision making) recently. I think shared decision making is making an informed decision. For example there are treatment A and treatment B. health professionals discuss the advantages and disadvantages of these treatments with patients. And then support the patients to make decisions.	N/A
16	I think (shared decision making) could be difficult in mental healthcare. It could be challenging. For example, we (health professionals) assess (patients' conditions) according to the standard assessment. After professional assessment, there is the best (treatment) option and then I make a (treatment) plan. This is what we do.	I think this (the level of patient involvement) depends on cognitive function. For example, some patients in chronic wards – they are not able to have a discussion with you.	Shared decision making? There seem be some workshops. But I personally don't know about this.	N/A
17	If you have been to hospitals, you should know doctors are the most powerful in hospitals. They are the boss.	Some patients are not able to go into the group you know. They keep talking about their delusions. How can you interact with them?	Shared decision makingI think I saw this in an article. But I don't know any details of it.	It (shared decision making) is difficult to take it further in practice because of time limitations. There are many tasks which I need to finish. For example, groups for patients, groups for sexual victims, outpatient appointments, and so on.
18	This is natural there is a knowledge gap. We are professionals (to decide).	When they need hospitalisation, we need to have a conversation with the family and patients. This is an art Because patients are ill. They don't have an insight into their mental illness. How can we discuss treatment with patients in this condition?	I have heard of shared decision making. I think it could have been difficult before we had the Mental Health Act. Although I don't really understand it, I would say the aim of it is that we (health professionals and patients) all discuss everything together and then make decisions.	N/A

19	Because psychiatrists have practical expertise in mental healthcare. There are psychiatrists for mental healthcare and physicians for general medicine. You don't ask physicians for information about general surgery. And they have a lot experience of taking care of patients every day. They are very familiar with this. They know how to deal with this and make adequate medication decisions. The decision about whether patients are able to be discharged from hospital is down to their family. Because their family are the people who will take care of them after a period of hospitalisation	"(Why patients are not involved) because they have poor insight. (Patients think): "I don't have a disease. Why do I need to take medication". They (the patients) are discharged from hospitals but then don't get followed up in the outpatients department. And then they have an acute episode and call an ambulance. Because they experience repeat onsets and this brings really bad outcomes for them.	I seem to have heard of shared decision making before. It involves nurses, doctors, and patients? I cannot remember. But as I understand this literally, all parties discuss it together? I am not sure, really. To be honest, I am not sure this is really helpful. Even if it could bring some benefits, I still think it is difficult, you know. How to do it and I believe there is a gap with reality. And there are many challenges.	<i>Time. Because health professionals are busy. Their families have jobs. It is difficult to arrange time for all the people.</i>
20	Why aren't patients with mental illness able to take part (in decision making)? Because they don't have professional knowledge. Unless we (health professionals) explain to them what this medication for they don't know. The doctor would not normally agree that a patient can be discharged from the service. There is no reason. They (patients) still have family. Their family will be worried about their safety. For example, if they (a patient) discharge themselves from the services on their own; they are hit by a car on their way home. And then the family will criticise us (health professionals) for letting them discharge themselves.	They (patients) may say they don't have psychiatric symptoms and why do they need to take this medication. They are not able to make right decisions.	I've never heard of this (shared decision making) before.	N/A

21	The family is too closed. Right, you should read the current news. XX (the killer on TV news and indicated as a patient with mental illness) is in his 20s. He is a legal adult in law. In the West, if someone commits a murder like that, his parents do not need to show up or do anything else. But in Chinese society, his parents needed to kneel in front of the public. At the beginning, they (the parents) did not do anything. They were attacked in the newspaper or community. It appeared that they have responsibility for this (behaviour). This (patient involvement) is really challenging. It (patient involvement) is difficult in mental healthcare. First, this disease is a long-term condition. Maybe 40 or 50 years. Sometimes even 50 or 60 years. For example, some family members don't agree to patients being discharged from the hospital. If we (professionals) insist on doing so based on patients' wishes, their family members may say: how can they live without financial support from us? They don't have a home." What we can do about this? Let them live on streets? You see what I mean	I think we label patients with mental illness as not able to make their own decisions. This may be influenced by the news or movies. We believe they are crazy and are not able to judge. Even I have been influenced by these reports. I believe it (shared decision making) is difficult. In particular, our patients don't like to take medication and have some bizarre behaviour. For example, one of our cases was suffering from a hernia and a doctor suggested he had surgery. But he said no. We tried to communicate with his family. But the patient still refused. We encouraged the family to talk with him. His sister-in-law came bybecause his brother was ill. So only his sister-in- law. His sister-in-law talked to him and suggested he had surgery. The doctor thought it was better for him to have surgery as soon as possible. But the patient still said no. So surgery was delayed. The patient thought he was not ill at all. [He thought]: "No hernia. Why do I need to have surgery? Yeah I haven't got a hernia. Why do I need to?" And then the doctor explained it to him. He said to us: "The doctor is	Yeah, I heard about it (shared decision making). It is quite popular recently. I know the XXX emergency department is now working on it. So that's why I heard about this. It seems to involve patients and their family and doctors provide some information? And then decide together? Is that correct?	Shared decision making takes a lot of time which we (health professionals) don't have. It is challenging.
22	For now doctors don't provide and	lying to me. I don't have a hernia."	Someone mentioned this (shared	N/A
22	treatment options for patients. This is all decided by doctors. This is	if you are a patient with psychosis and disorganised behaviour, you are not able to make a good decision. We or	Someone mentioned this (shared decision making) recently. In mental healthcare, patient, family,	N/A

	 professional. The right of prescribing belongs to the doctors. I believe this is related to Taiwan's traditions. Everything is decided by the family. Family members will question you in the end. Or they will criticise you: "Why did you let the patient decide alone?" In order to protect ourselves (professionals), we need family permission. You know even if patients feel they do not need to stay here (in mental health services), but their family disagree, I think we wouldn't take the patient's opinion to discharge him. This must be agreed by family members, or the services or nurses won't be able to take their (patients') opinion. 	the family should decide. This is the pattern. There are rare families that will take patients with mental illness seriously. This is what I think. I believe they (patients) have been labelled as incompetent.	psychiatrist, nurse, social worker, occupation therapist, and psychologist work together when making decisions. When there is a problem, they all work together to sort it out.	
	They (psychiatrists) don't offer options. They just explain. Nurses just explain what choices the psychiatrists made, such as dose or frequency. Actually these medication treatments were mainly explained by the doctors rather than us (nurses). But so far, we've never seen any doctors offering options to patients. It is all decided by the doctors.			
23	This is how health professionals foster their profession. For example, only doctors can prescribe medication. This is how people were educated. We have seen this a lot. You can get any Chinese	I wonder if patients in a psychotic condition really make the right choice. Or if it is decided by his/her symptoms.	I've heard of this (shared decision making) before. But I don't really know about the content of it. There are some workshops in the	Normally it (shared decision making) does not happen in mental healthcare. It takes too much time in the outpatient department. But doctors only have a little time. For example, one doctor needs to see 100 patients in one

	 medication. You can make a tea with any Chinese herb. But in healthcare, this is the rule. Only doctors have the right to prescribe. This is their power structure in a patient's family. If the family power type is an authoritative one, the father takes more power. And then he (a patient) can only listen to his father. This is possible. In this situation, no matter what he does, or tries to do against it, or how he tries to 		hospitals. But I am too busy to take the class. So I am not clear about it.	morning, within 3.5 hours. You want to have a discussion with them in only a few minutes? It is impossible. If we do (implement shared decision making), doctors may need to spend one or two hours seeing one patient.
24	communicate (with his father), it is useless. Eventually, he can only listen to the decision maker in his family and follow his instructions.	I believe that patients with mental	Share decision making sorry I	N/A
27	responsible for deciding what medication they should take. Because patients don't have any medical background. They haven't studied in a specialist medical educational system. Even me (nurse), I have some basic knowledge but I am still not a doctor to know which medication will be effective for patients.	illness have been seen as being not able to judge stuff correctly. From my perspective, I may feel they are not able to make the right decision. But I believe they are of relative low status.	have never heard of this before.	

Theme 2: F	acilitators of shared decision making	
Themes	Awareness of patients' right to autonomy	Understanding of the potential benefits of shared decision making.
Number of reports	14/24	16/24
1	N/A	N/A
2	N/A	If patients have a full discussion with me, I feel more confident in the treatment. For example, the patient discussed all the details about how he took Lithium before. And then we decided he would take Lithium next. I did not really have to worry about side effects. I could have more information about the patient. And this helps me to have more confidence. And therefore in the decision making process, I think patient involvement is positive.
3	N/A	You let him decide about his medication. He has more motivation. He is more willing to take the medication.
4	N/A	N/A
5	If today we have a child, he asks for something. And then we discuss it with him and then decide together. It is better, isn't it? OR would he like to let the parents decide everything for him? I believe it is the same with the patients. At least he has right to decide who should decide it.	If today we have a child, he asks for something. And then we discuss with him and then decide together. It is better, isn't it? OR would he like to let the parents decide everything for him? I believe this is the same with the patients. Now we hope patients will decide to take their medication regularly and this is decided by us (health professionals) and the patients together. In this situation, responsibility plays a very important role. If you do it that way, the patients will take responsibility, you know.
6	For our patients, this (shared decision making) helps them to make autonomous decisions and (professionals) to respect their autonomous decisions. I think this idea is quite nice.	I believe patients all want to have options, because they don't just go the place (possible accommodation) for one or two days. Sometimes it is more likely to be about their future - the rest of their life. If they can decide on their own and they like it, they could stay longer.

Appendix 15: Example of thematic frameworks for the health professional interviews (Theme 2)

7	We should implement patient-centred care and discuss their case with them. It depends on them. They have the right to know what treatment they are taking now.	Because the future of mental illness is becoming a long-term condition for patients. They should improve their insight. If they are involved, they are more willing to fight for this mental illness.
8	(Why do you think patients should take part in decision making?)Because this medication will be taken by patients. They can ask questions about it.	
9	For example, he felt bad about Seroquel (Quetiapine). And he wants to change to Risperidone (Risperdal). That's fine. Just change it. Because to be honest with you - this is my personal view - even now the newest research cannot predict which medication is the most effective for patients. So why not (implement shared decision making)?	I think this (shared decision making) could improve the therapeutic relationship and the quality of care. I'm guessing. I don't really know.
10	There is no 100% correct answer for medicine. This is probably why it (shared decision making) might be implemented. Doctors might have better knowledge. But not all decisions are suitable for everyone. There might have been some (medical) mistakes, so a new policy (shared decision making) came out.	Basically, I personally believe that this (patient active involvement) is better. Because they discuss it with us and they have some ideas about their treatment and care. They can engage in the process. This is not only about health professionals.
11	N/A	If he was forced or pushed, he didn't like (the intervention). We (health professionals) didn't really want this. For example, for occupational training, this is invalid and not effective. Patients without any motivation were forced. This is meaningless. They didn't want to stay.
12	Because you own your body. You (patients) can decide how you want to use your body in the next decades. This body is mine. You should understand what will happen in the next decades if you decide to take this intervention or treatment. I believe this is the right thing to do.	In fact, I have been a patient (with a long-term condition) since I was little. I actually have a thought. For me, because I had serious surgery in my mouth, I don't want to talk with others close up, because I easily spit on them. So when I recalled that time. I thought: if I hadn't accepted surgery, I wouldn't have needed to experience this side effect. But there was no choice. No one told me. I thought you (the health professional) were supposed to tell me all the options, or you should tell me what will happen next. It is a better way. For patients here it should be the same (12 Occupational Therapist)
13	N/A	N/A
14	For example one of my patients who was suffering with nasopharyngeal carcinoma was recommended by one doctor to take surgery and then radiotherapy and chemo therapy. My patient complied with the doctor. Based on the evidence base, the doctor didn't do anything wrong, but he	If I ignore their ability to make decisions about treatment and care, they could think: 'These (decisions) are all decided by doctors or other professionals'. It was like they didn't need to take any responsibility Even if he is a psychosis patient, he could still decide. These things should

	didn't pay attention to how the patient think about this. The truth is, my patient was worried about side effects of this treatment. He had been diagnosed with bladder cancer a long time before. He had been through this treatment once. He didn't want to undergo all these processes again. He has the right to say something about his treatment. I think this is important to think about.	be not decided by doctors I should have noticed this (that the decisions were all made by professionals) but I didn't. It's as though the patient sees me as some sort of barrier to stop him recovering. I just see him as a child. And then this means he never ever takes responsibility.
15	Let me think this (patient involvement) may be a great idea. I've never thought about this. In this way, I can show respect to their rights (to decide).	Maybe this (shared decision making) is right. They can decide which therapy they need. I can respect all their decisions. There is no argument between us (the health professional and patients). We could work together.
16	N/A	Patient involvement also influences their motivation. For example, if they are involved and have some discussion with us For example, they choose one occupation therapy. For example he/she decides to help sell happy meals on the wards (one type of occupation therapy in the setting). This is decided by him/her. When he/she faces some challenge, they are more likely have the motivation to continue. This helps them to reach the goal.
17	I hope they (patients) can be involved in decision making, because they should have the right to decide what treatment they want to receive.	But this is their life, long life. We need to let them know how to take responsibility for their own life. If their decisions are decided by others, they refuse to take any responsibility. They cannot engage in their treatment. It is impossible for them to recover.
18	"It is not right to violate their human rights. It is just not right. Because they don't meet the criteria for compulsory admission. They have the right (to refuse to be admitted). You cannot do this.	N/A
19	N/A	To be honest, I am not sure this is really helpful. Even if it could bring some benefits, I still think it is difficult, you know. How to do it. And I believe there is a gap with the reality. And there are many challenges.
20	N/A	N/A
21	To be honest, I believe that if they (patients) are stable, you know, they should have right to say they want to be discharged from the hospital or something (treatment).	N/A
22		N/A

23	If I was a patient (with mental illness), I suppose I wouldn't like someone	I believe their (patients') views should be involved. I mean, if I'm setting a	
	else deciding for me. I would think I had had been deprived of my liberty	goal with them Not just "We (health professionals) think this is best for	
	and dignity. Besides I could only follow the instructions of other people,	you (patients)". "We (health professionals) are on your (patients') side.	
	including doctors or family members. It is just (better) decided by me. If	We are working together. You (patients) will have more motivation to	
	I was a patient, I would think like that.	reach the goal."	
24	I believe they have rights, because they are independent people. They	If we have a recovery plan which he/she doesn't have any motivation for,	
	have a right to decide (treatment or care).	for example, you deliver a plan which he doesn't agree with, he doesn't	
		want to do it. It is useless. But if he has the motivation, I believe he can	
		implement his plan successfully.	

Appendix 16: Extraction sheet for synthesis

To be extracted: information refers to perspectives of shared decision making in secondary mental healthcare (one systematic review and two qualitative studies)

Part A: Demographic information

General information

Title:

Year:

Aim:

Methodology

Study design:

Data collection:

Data analysis:

Participant selection

Target population:

Number of participants:

Setting/context:

Sampling:

Recruitment method:

Inclusion criteria/Exclusion criteria:

Characteristics of study population

Gender:

Age:

Diagnosis/profession:

Part B: Findings from the studies